

In Vitro and In Vivo neuronal differentiation capacity of human adult bone marrow-derived mesenchymal stem cells

Author: Khoo, Melissa Li Meng

Publication Date: 2009

DOI: https://doi.org/10.26190/unsworks/4010

License:

https://creativecommons.org/licenses/by-nc-nd/3.0/au/ Link to license to see what you are allowed to do with this resource.

Downloaded from http://hdl.handle.net/1959.4/44841 in https:// unsworks.unsw.edu.au on 2024-04-30

IN VITRO AND IN VIVO NEURONAL DIFFERENTIATION CAPACITY OF HUMAN ADULT BONE MARROW-DERIVED MESENCHYMAL STEM CELLS

Melissa Li Meng Khoo

BAdvSc(Hons)

Submitted in fulfilment of the requirements for

the award of Doctor of Philosophy

Faculty of Medicine

The University of New South Wales

2009

PLEASE TYPE THE UNIVERSITY Thesis/E	Y OF NEW SOUTH WALES Dissertation Sheet
Surname or Family name: KHOO	
First name: Melissa	Other name/s: Li Meng
Abbreviation for degree as given in the University calendar: PhD	
School: St Vincent's Clinical School	Faculty: Medicine
Title: In Vitro and In Vivo Neuronal Differentiation Capacity of Huma	an Adult Bone Marrow-Derived Mesenchymal Stem Cells

Abstract 350 words maximum: (PLEASE TYPE)

Discovery of the ability of mesenchymal stem cells (MSCs) to differentiate into cells of non-mesodermal tissues, particularly neuronal cells, have raised the possibility of utilising MSCs in regenerative/reparative therapies for neurological disorders. However, a number of hurdles remain to be resolved. This thesis aims to address some of these issues by investigating the characteristics of bone marrow-derived human MSCs (hMSCs) during long-term culture, the potential of hMSCs to differentiate *in vitro* toward the neuronal lineage under the influence of cytokines, and the effects of intracerebral transplantation in the hemiparkinsonian rat model.

During expansion culture hMSCs were found to display the expected characteristics of MSC populations, and also constitutively expressed neural and pluripotency markers simultaneously with mesodermal markers. Analysis of hMSC long-term subcultivation revealed an optimal period for commencing neuronal differentiation (first 6-8 passages), and also showed the absence of spontaneous neural differentiation.

Application of neural-inducing cytokines and culture conditions resulted in the generation of an immature neuronal-like phenotype by hMSCs. Through live cell microscopy it was demonstrated for the first time that cytokine-based hMSC neuronal differentiation occurs through active and dynamic cellular processes involving outgrowth and motility of cellular extensions. In addition, singleand multiple-stage cytokine-based strategies for inducing dopaminergic neuronal-like cells from hMSCs were investigated. These studies revealed that FGF-2 and EGF exerted the greatest benefits for hMSC neuronal differentiation.

Undifferentiated and neuronal-primed hMSCs were transplanted intracerebrally into the striatum and substantia nigra of cyclosporine-treated hemiparkinsonian rats. Grafted hMSCs could be clearly identified at 1-day and 7-days post-transplantation; however, grafts were gradually lost over time, with complete absence by 21-days. Co-transplantation with olfactory ensheathing cells, neuronal-priming prior to grafting, and nigral as well as striatal grafting could not provide engraftment and differentiation advantages. Immunohistological analysis demonstrated the presence of innate inflammatory responses (microglia and astrocyte activation) at graft sites, fibronectin deposition by hMSCs, and lack of endogenous host neurogenesis.

The results of my PhD work indicate that cytokine-based culture methods are capable of differentiating hMSCs to an immature neuronal-like phenotype, and host-mediated innate inflammatory responses may be a key contributing factor for the failure of *in vivo* hMSC engraftment.

Declaration relating to disposition of project thesis/dissertation

I hereby grant to the University of New South Wales or its agents the right to archive and to make available my thesis or dissertation in whole or in part in the University libraries in all forms of media, now or here after known, subject to the provisions of the Copyright Act 1968. I retain all property rights, such as patent rights. I also retain the right to use in future works (such as articles or books) all or part of this thesis or dissertation.

I also authorise University Microfilms to use the 350 word abstract of my thesis in Dissertation Abstracts International (this is applicable to doctoral theses only).

Signature

Witness

Date

The University recognises that there may be exceptional circumstances requiring restrictions on copying or conditions on use. Requests for restriction for a period of up to 2 years must be made in writing. Requests for a longer period of restriction may be considered in exceptional circumstances and require the approval of the Dean of Graduate Research.

FOR OFFICE USE ONLY

Date of completion of requirements for Award:

THIS SHEET IS TO BE GLUED TO THE INSIDE FRONT COVER OF THE THESIS

ORIGINALITY STATEMENT

'I hereby declare that this submission is my own work and to the best of my knowledge it contains no materials previously published or written by another person, or substantial proportions of material which have been accepted for the award of any other degree or diploma at UNSW other educational institution, except or any where due acknowledgement is made in the thesis. Any contribution made to the research by others, with whom I have worked at UNSW or elsewhere, is explicitly acknowledged in the thesis. I also declare that the intellectual content of this thesis is the product of my own work, except to the extent that assistance from others in the project's design and conception or in style, presentation and linguistic expression is acknowledged.'

Signed

Date

ACKNOWLEDGEMENTS

First and foremost I would like to express my deep and sincere gratitude to my supervisor Professor David Ma, and co-supervisors Dr. Helen Tao and Professor Alan Mackay-Sim (National Centre for Adult Stem Cell Research* (NCASCR), Griffith University, QLD), for the opportunity to embark on such a fulfilling and memorable PhD journey. To David, thank you for your guidance, support and constructive comments, and most importantly for encouraging me to explore and pursue my ideas and for sharing your vision with me. To Helen, I am very grateful for your supervision, dedication and words of advice, and I thank you for sharing your expertise and knowledge with me. To Alan, thank you for your encouragement, constructive critical review of my work, and for welcoming me so warmly into your laboratory, it has been very much appreciated.

A special thanks to Dr. Adrian Meedeniya (NCASCR*, Griffith University, QLD), Maria Nguyen (NCASCR*, Griffith University, QLD), and their families, to whom I am greatly indebted for their continual kindness and generosity. Thank you for sharing your homes and lives with me, and for your scientific and personal support, both during my visits to Brisbane and also from afar when I'm in Sydney. It definitely would not have been as memorable or enjoyable without your help and company, and for this I am most grateful.

To my fellow MSC team mates, Jean Hsu, Dr. Bojiang Shen and Shane Whittaker, I am truly appreciative of your always willing and friendly assistance and advice. To the resident mollie bollies, Dr. To Ha Loi and Dr. Mark Lutherborrow, thank you very much for your help and advice with all things molecular biology-related. To the ever 'flow'-ing Dr. David Connor, I am thankful for your support and guidance with flow cytometry. Also, to Andrea Herbert, Dr. Stephen Carlin and Lyne Peebles, my warm and sincere thanks for your insightful comments and words of encouragement. And to all members of the Haematology Research Laboratory, both past and present, thank you for being there during the joy, laughter and frustration, and for the much-needed refreshment of good company with hot chocolate, coffee and gelato.

To my colleagues and friends at the Mackay-Sim Lab (NCASCR*, Griffith University, QLD), Mary Pagendam-Turner, Brenton Cavanagh, Nathalie Romond, Tavia Davenport, and Joseph Kan, for always being willing to help me at any time, and for keeping the laughter and smiles going over the long experiment-packed days in Brisbane. And to all other members of the Mackay-Sim Lab, thank you for your hospitality and support during my stays.

To Awachana Jiamsakul (Australian Bone Marrow Transplant Recipient Registry) and Anna Campain (Sydney University) for your friendly advice and guidance in statistical analysis. And also, to Dr. Will Hughes (Garvan Institute) for much appreciated help and guidance with all microscopy-related matters.

I would like to gratefully acknowledge the financial support of the Australian Government Postgraduate Award, The University of New South Wales, the Hawkesbury Canoe Classic and Arrow Bone Marrow Transplant Foundation, and St Vincent's Hospital Sydney. Also, special mention to the members of the Arrow Bone Marrow Transplant Foundation; your tireless dedication and generosity to this charitable cause has been truly inspiring.

To my dear friends, Sharon Chow, Stephanie Tsang, Sam Fan and David Liuwantara, for your constant friendship, support and encouragement. And most importantly for reminding me how much more there is to life, and for filling my days with much fun and happiness.

To my parents Ronald and Wendy, I am eternally grateful for your continuous and unconditional love, encouragement, comfort, patience, and kindness throughout every facet of my life. Thank you for supporting and encouraging me to pursue my dreams and goals, and for always believing in me. To my sister Janice, I am most thankful for your company, friendship and support over the years and for always being there to listen to my tales and to encourage me. Thanks for bringing the cheeriness, giggles and chocolate.

To my fiancé Will, your unbelievable patience, kindness and love have made a world of difference to me, and has made this entire journey more meaningful, memorable and rewarding. Thank you for continually supporting me and believing in me, and for always keeping things amusing with mischief, laughter and merriment. For this I am grateful, forever and always.

And finally, above all, thanks and praise to our gracious and loving heavenly Father and our Lord and Saviour Jesus Christ. 'Now to the King eternal, immortal, invisible, the only God, be honour and glory for ever and ever' (1 Timothy 1:17).

* NCASCR, Eskitis Institute for Cell and Molecular Therapies, Eskitis Building (N75), Griffith University, Brisbane Innovation Park, Don Young Road, Nathan QLD 4111

ABSTRACT

Since the identification of mesenchymal stem cells (MSCs) in the bone marrow (BM) much interest has been generated in these cells by the development of methods to isolate, expand and differentiate MSCs. Discovery of the ability of MSCs to differentiate into cells of non-mesodermal tissues, particularly neuronal cells, has also raised the possibility of utilising MSCs in regenerative and reparative therapies for neurological disorders. However, a number of hurdles remain to be resolved. This thesis aims to address some of these issues by investigating the characteristics of BM-derived human MSCs (hMSCs) during long-term culture, the potential of hMSCs to differentiate *in vitro* toward the neuronal lineage under the influence of cytokines, and the effects of intracerebral transplantation of hMSC-derived cells in the 6-OHDA lesioned hemiparkinsonian rat model.

During expansion culture hMSCs were found to display characteristics of MSC populations, as assessed by morphological appearance, proliferative ability, surface antigen expression, gene and protein expression, and production of multi-lineage mesodermal derivatives. In addition, hMSCs constitutively expressed neural and pluripotency markers simultaneously with the expected mesodermal markers, indicating the importance of determining baseline expression levels prior to neural differentiation. Analysis of long-term serial passaging of hMSCs revealed maintenance of a stable *in vitro* phenotype over the initial 6-8 passages of expansion culture, with cells cultured beyond this period showing alterations in key phenotypic attributes, such as morphology, proliferative ability and gene and protein expression. However, extensive subcultivation of hMSCs did not result in spontaneous neural differentiation. These results allowed determination of the optimal window for hMSC expansion and commencement of neuronal differentiation for all subsequent experiments in this thesis.

Application of neural-inducing cytokines and culture conditions resulted in the generation of an immature neuronal-like phenotype by hMSCs, in terms of cellular morphology, gene expression and protein expression. Through live cell microscopy it was demonstrated for the first time that cytokine-based hMSC neuronal differentiation occurs through active and dynamic cellular processes involving outgrowth and motility of cellular extensions. Long-term subcultivation of hMSCs did not appear to prevent cells from acquiring immature neuronal-like characteristics, although use of hMSCs within the optimal window was associated with advantages, including higher proliferative ability and greater phenotypic stability of undifferentiated hMSCs and a lower proportion of cells unresponsive to differentiation conditions. Sequential application of cytokines chosen for their known involvement in midbrain dopaminergic neuronal differentiation in vivo was examined (MultiDA method) and compared to single-stage cytokine-based approaches (SingleDA and Single ND methods). The 3 different procedures investigated were observed to elicit differing effects on cellular morphology and gene expression; however, all methods investigated produced immature neuronal-like cells. Nevertheless, the first stage of the MultiDA method yielded the highest expression of early neuronal and early dopaminergic neuronal markers, and FGF-2 and EGF were found to exert the greatest benefits for hMSC neuronal differentiation.

Undifferentiated and neuronal-primed (first stage of the MultiDA method) hMSCs were transplanted intracerebrally into the striatum and substantia nigra of cyclosporine-treated hemiparkinsonian rats. Grafted hMSCs could be clearly identified at 1-day post-transplantation and were still detectable after 7-days; however, hMSCs were gradually lost over time, with complete absence by 21-days post-transplantation. Further differentiation of neuronal-primed hMSCs *in vivo* was not evident at either the

vi

striatum or substantia nigra. In addition, transplantation of hMSCs did not elicit neurogenic effects on host endogenous neural stem cells or progenitors, as evidenced by lack of nestin expression. Co-transplantation of human olfactory ensheathing cells (OECs) with neuronal-primed hMSCs was unable to provide engraftment or differentiation advantages, and OECs also did not survive in the rat brain after transplantation. To my knowledge, it was shown for the first time that innate inflammatory responses (microglia and astrocytes) are activated following transplantation of hMSC-derived cells in the lesioned hemisphere of hemiparkinsonian rats. Continued fibronectin production by hMSCs was also found indicating retention of MSC characteristics, which are foreign to the host brain microenvironment. This suggests that further *in vitro* differentiation of hMSCs to the neuronal lineage may be required. The observed graft loss may therefore be the result of innate inflammatory responses, the xenogeneic nature of transplanted cells, the 'foreign' nature of hMSCs (which still expressed mesodermal/MSC traits), or the inability of the lesioned adult rat striatum and substantia nigra to support hMSC survival, engraftment and differentiation.

The results of my PhD work indicate that cytokine-based *in vitro* culture methods are capable of differentiating hMSCs to an immature neuronal-like phenotype, and that host-mediated innate inflammatory responses may be a key contributing factor for the failure of *in vivo* engraftment of hMSC-derived cells in a rodent model of Parkinson's disease. Further studies are required to gain a better understanding of this field and to improve the *in vitro* and *in vivo* procedures for directed neuronal differentiation and effective transplantation of hMSCs.

vii

PUBLICATIONS, PRESENTATIONS AND AWARDS

Publications

Khoo, M.L.M., Shen, B., Tao, H. and Ma, D.D.F. (2008). Long-Term Serial Passage and Neuronal Differentiation Capability of Human Bone Marrow Mesenchymal Stem Cells. *Stem Cells and Development* **17**: 883-896.

Tao, H., Khoo, M.L.M., Shen, B. and Ma, D.D.F. (2007). Expression of Neurofilament
Proteins in Adult Bone Marrow Mesenchymal Stem Cells and Their Clinical Potential.
pp. 1-45. In L.P. Davenport (Ed): *Stem Cell Research Advances*, Nova Science
Publishers, Inc., New York.

Khoo, M.L.M., Meedeniya, A.C.B., Tao, H., Pagendam-Turner, M., Nguyen, M., Mackay-Sim, A. and Ma, D.D.F. Dopaminergic Neuronal Differentiation Capacity of Human Bone Marrow-Derived Mesenchymal Stem Cells *In Vitro* and *In Vivo* (Manuscript to be submitted).

Presentations at Scientific Meetings

Khoo, M.L.M., Meedeniya, A.C.B., Tao, H., Pagendam-Turner, M., Nguyen, M., Mackay-Sim, A. and Ma, D.D.F. The Fate of Transplanted Neuronal-Primed Human Mesenchymal Stem Cells in the 6-OHDA Unilaterally Lesioned Parkinsonian Rat. *National Centre for Adult Stem Cell Research 'Adult Stem Cells: Biology and Clinical Applications' Conference, Brisbane, Australia, November 2008 (P01).*

Khoo, M.L.M., Tao, H. and Ma, D.D.F. Modelling Neuronal Differentiation in Human Bone Marrow-Derived Mesenchymal Stem Cells Using Cytokines. *Australian Health and Medical Research Congress, Melbourne, Australia, November 2006 (1422).*

Khoo, M.L.M., Shen, B.J., Tao, H. and Ma, D.D.F. Pluripotent Characteristics of Human Bone Marrow Derived Mesenchymal Stem Cells in Long Term Culture. *Annual Scientific Meeting of the Haematology Society of Australia and New Zealand, Sydney, Australia, October 2005.*

Khoo, M.L.M., Shen, B.J., Tao, H. and Ma, D.D.F. Stability of Characteristics (Stemness) of Human Bone Marrow Mesenchymal Stem Cells in Long Term Culture. International Society for Stem Cell Research Annual Meeting, San Francisco, CA, USA, June 2005 (412).

Shen, B.J., Khoo, M.L.M., Tao, H. and Ma, D.D.F. Pluripotent and Differentiation Potential of Human Bone Marrow Derived Mesenchymal Stem Cells in Prolonged Expansion Culture. *Annual Meeting of the Australian Stem Cell Centre, Gold Coast, Australia, November 2005.*

Awards

- 2004 2007: UNSW Faculty of Medicine Rising Star Postgraduate Award
- 2005: Australian Stem Cell Centre Conference Grant
- 2005 2007: Arrow Bone Marrow Transplant Foundation & Hawkesbury Canoe Classic Postgraduate Award
- 2006: Australian Stem Cell Centre Conference Grant

TABLE OF CONTENTS

ACKNOWLEDGEMENTSiii
ABSTRACTv
PUBLICATIONS, PRESENTATIONS AND AWARDSviii
TABLE OF CONTENTSx
LIST OF FIGURESxix
LIST OF TABLESxxii
LIST OF ABBREVIATIONSxxiii
CHAPTER 1 LITERATURE REVIEW1
1.1 BONE MARROW-DERIVED MESENCHYMAL STEM CELLS
1.1.1 Adult Bone Marrow-Derived Stem Cells2
1.1.2 The Mesenchymal Stem Cell Identity
1.1.3 Isolation and Expansion of Mesenchymal Stem Cells7
1.1.4 Characterisation and Function of Mesenchymal Stem Cells10
1.1.4.1 Initial Observations
1.1.4.2 Role of MSCs in Supporting Haematopoiesis In Vivo11
1.1.4.3 Role of MSCs In Supporting Haematopoiesis In Vitro12
1.1.5 Immunomodulatory Properties of MSCs13
1.1.5.1 Effects of MSCs on T cells14
1.1.5.2 Effects of MSCs on Professional Antigen-Presenting Cells
1.1.5.3 Effects of MSCs on B cells
1.1.5.4 Effects of MSCs on Natural Killer Cells17
1.1.5.5 MSC Immunomodulation in In Vivo Studies

1.1.5.6 Controversy Related to MSC Immunomodulation20
1.1.6 Classical Multi-Lineage Mesodermal Differentiation Capacity of
MSCs
1.2 LINEAGE CONVERSION POTENTIAL OF BM-DERIVED MSCs
1.2.1 Evidence for a Pluridifferentiated Stromal Progenitor Cell
1.2.2 Evidence for Non-Mesodermal Differentiation of MSCs In Vitro
1.2.3 In Vivo Transplantation Studies Supporting MSC Plasticity
1.2.3.1 MSC Transplantation at Early Stages of Development
1.2.3.2 MSC Transplantation into Neural Tissues
1.2.3.3 MSC Transplantation into Hepatic Tissues
1.2.3.4 MSC Transplantation into Epithelial Tissues
1.2.4 Alternative Explanations for MSC Plasticity
1.3 NEURAL DIFFERENTIATION OF BM-DERIVED MSCs <i>IN VITRO</i>
1.3.1 Induction of Neuronal Differentiation of MSCs
1.3.1.1 Chemical Induction
1.3.1.2 Cytokine and Signalling Molecule Induction
1.3.1.3 Conditioned Media or Co-Culture Systems
1.3.1.4 Genetic Engineering
1.3.1.5 Combined Approach
1.3.2 Evidence of Neuronal Differentiation of MSCs
1.3.3 Possible Molecular Control of MSC Neuronal Differentiation
1.4 PARKINSON'S DISEASE AND CELL-BASED THERAPIES
1.4.1 Overview of Parkinson's Disease
1.4.1.1 Neuropathological Features of Parkinson's Disease
1.4.1.2 Clinical Features of Parkinson's Disease

1.4.	1.3 Aetiology and Pathogenesis of Parkinson's Disease
1.4.	1.4 Treatments for Parkinson's Disease
1.4.	1.5 Experimental Animal Models of Parkinson's Disease
1.4.2	Cellular Replacement Therapies for Parkinson's Disease
1.4.	2.1 Foetal Neural Tissue Transplantation in PD
1.4.	2.2 Neural Stem Cell Transplantation in PD74
1.4.	2.3 Embryonic Stem Cell Transplantation in PD76
1.4.	2.4 Patient-Specific Cellular Transplantation in PD77
1.4.	2.5 Challenges in Cellular Replacement Therapies for PD78
1.4.3	Lessons from Midbrain Dopaminergic Neuron Development In Vivo 80
1.4.	3.1 Extrinsic Cues Involved in Midbrain Dopaminergic Neuron
	Specification
1.4.	3.2 Cell Intrinsic Signals Directing Dopaminergic Neuron
	Differentiation
1.4.4	Dopaminergic Neuronal Differentiation of MSCs In Vitro and
	Transplantation into Animal Models of Parkinson's Disease
1.5 Aims	OF THE CURRENT STUDY95
CHAPTER 2	MATERIALS AND METHODS98
2.1 MATE	ERIALS
2.1.1	Cell lines
2.1.2	Culture Media and Cultureware99
2.1.3	Enzymes and enzyme buffers100
2.1.4	Oligonucleotides100
2.1.5	Antibodies100
2.1.6	Equipment

2.1.7 Gen	eral materials and chemicals102
2.2 Methods	
2.2.1 Cell	Culture
2.2.1.1	Isolation of Human Bone Marrow-Derived Mesenchymal
	Stem Cells
2.2.1.2	Expansion of Human Bone Marrow-Derived Mesenchymal
	Stem Cells
2.2.1.3	Induction of Multi-lineage Mesodermal Differentiation105
2.2.1.4	Induction of Neuronal Lineage Differentiation – Single-Stage
	Cytokine-Based Method (SingleND)106
2.2.1.5	Induction of Dopaminergic Neuronal Differentiation -
	Multiple-Stage Cytokine-Based Method (MultiDA)106
2.2.1.6	Induction of Dopaminergic Neuronal Differentiation Via
	Published Protocol - Single-Stage Cytokine-Based Method
	(SingleDA)108
2.2.1.7	Cell Lines
2.2.2 Flow	v Cytometric Analysis108
2.2.2.1	Detection of Apoptosis108
2.2.2.2	Characterisation of Surface Antigen Expression110
2.2.2.3	Statistical Analysis110
2.2.3 RNA	A Extraction
2.2.4 Rea	l-Time RT-PCR Analysis111
2.2.4.1	Primer Design and Optimisation113
2.2.4.2	cDNA Synthesis
2.2.4.3	Real-Time RT-PCR Procedure

2.2.4	.4 Electrophoresis of PCR Products	114
2.2.4	.5 Statistical Analysis	115
2.2.5 In	n Vitro Morphological Analysis	115
2.2.5	.1 Routine Morphological Recordings	115
2.2.5	5.2 Live Cell Imaging	115
2.2.6 In	mmunofluorescence Microscopy of In Vitro Cultured Cells	116
2.2.7 0	Cell Preparation and Stereotactic Intracerebral Transplantation	117
2.2.7	7.1 Animal Welfare	117
2.2.7	2.2 Experimental Design	117
2.2.7	2.3 Preparation of hMSCs for Transplantation	118
2.2.7	2.4 Preparation of OECs for Transplantation	120
2.2.7	2.5 Stereotactic Intracerebral Transplantation in Rat PD Model	120
2.2.7	2.6 Tissue Processing	121
2.2.8 In	mmunohistological Analysis	122
2.2.9 A	Amphetamine-Induced Rotation Testing	123
CHAPTER 3	CHARACTERISATION OF HUMAN BONE MARROW-	
DERIVED MI	ESENCHYMAL STEM CELLS AND THE EFFECT OF	
LONG-TERM	SERIAL PASSAGE	124
3.1 Introi	DUCTION	125
3.2 Result	TS	128
3.2.1 N	ASC Morphology and Growth Kinetics	128
3.2.2 Ii	mmunophenotype of Human MSCs by Flow Cytometry	131
3.2.3 N	Aulti-Lineage Mesodermal Differentiation	131
3.2.4 E	Expression of Pluripotency and Germ Lineage Markers	135
3.2.4	.1 Gene Expression by Real-Time RT-PCR	135

3.	2.4.2 Protein Expression by Immunofluorescence Microscopy137
3.3 Dise	CUSSION
3.3.1	Undifferentiated hMSCs Possess Characteristic Properties of MSCs 140
3.3.2	Undifferentiated hMSCs Express Ectodermal and Mesodermal
	Lineage Markers
3.3.3	Undifferentiated hMSCs Express Markers of Pluripotency142
3.3.4	Effects of Long-term Serial Passage on hMSCs
3.3.5	Common Culture Conditions are Insufficient for hMSC Maintenance. 144
3.3.6	Limitations Associated with Interpretation of Marker Expression 145
3.3.7	Summary146
CHAPTER	4 NEURONAL DIFFERENTIATION OF HUMAN
MESENCH	YMAL STEM CELLS IN VITRO147
4.1 INT	20DUCTION 148
	$\mathbf{X} \mathbf{U} \mathbf{U} \mathbf{U} \mathbf{U} \mathbf{U} \mathbf{U} \mathbf{U} U$
4.2 Drg	152
4.2 Res	ULTS
4.2 Res 4.2.1	ULTS
4.2 Res 4.2.1 4.2.2	ULTS
4.2 Res 4.2.1 4.2.2	ULTS
4.2 Res 4.2.1 4.2.2 4.2.3	ULTS
4.2 Res 4.2.1 4.2.2 4.2.3 4.2.3	ULTS
4.2 Res 4.2.1 4.2.2 4.2.3 4.2.3	ULTS
4.2 Res 4.2.1 4.2.2 4.2.3 4.2.3 4.	ULTS
4.2 Res 4.2.1 4.2.2 4.2.3 4.2.3 4.	ULTS
4.2 Res 4.2.1 4.2.2 4.2.3 4. 4. 4. 4.	ULTS
4.2 RES 4.2.1 4.2.2 4.2.3 4. 4. 4. 4.	ULTS

4.3.1	Cytokine-Based In Vitro Neuronal Differentiation of hMSCs Elicits
	Dynamic Cellular Responses191
4.3.2	Long-Term Subcultivation of hMSCs Does Not Affect
	Differentiation Toward an Immature Neuronal-Like Phenotype
4.3.3	Undifferentiated hMSCs Express Neuronal and Dopaminergic
	Neuronal Markers
4.3.4	Sequential Application of Cytokines Involved in Midbrain
	Dopaminergic Neuronal Development Did Not Enhance
	Dopaminergic Neuronal Differentiation of hMSCs195
4.3.5	Beneficial Effect of FGF-2 and EGF in hMSC Neuronal
	Differentiation
4.3.6	Recently Published SingleDA Method Yielded Similar Immature
	Neuronal-Like Cells as the MultiDA and SingleND Procedures
4.3.7	Transcription Factor Expression During hMSC Dopaminergic
	Neuronal Differentiation
4.3.8	Continued hMSC Expression of Mesodermal and Pluripotency
	Markers Following In Vitro Neuronal Differentiation
4.3.9	Functional Assessment of hMSC-Derived Neuronal-Like Cells is
	Required
4.3.10	Contaminating Cells, Culture Artifacts or Cell-Cell Fusion Events
	Cannot Explain the Observed hMSC Neuronal Differentiation
4.3.11	Summary
CHAPTER	5 IN VIVO STUDIES IN AN ANIMAL MODEL OF
PARKINSO	N'S DISEASE
5.1 INTE	210 210
J.1 1111F	

5.2 Res	ULTS – PART I
5.2.1	Long-Term Transplantation of Undifferentiated hMSCs in PD Model 215
5.3 Res	ULTS – PART II
5.3.1	Graft Site Identification
5.3.2	Limited Survival of Neuronal-Primed hMSCs and OECs Post-
	Transplantation
5.3.3	Accumulation of Astrocytes and Microglia/Macrophages at Graft
	Sites
5.3.4	Endogenous Host Neuronal Differentiation Post-Transplantation230
5.3.5	Characterisation of Graft Sites for Neuronal Marker Expression 232
5.3.6	Characterisation of Graft Sites for Extracellular Matrix Deposition233
5.4 Disc	CUSSION
5.4.1	Activation of Innate Immune Responses with Intracerebral hMSC
	Transplantation
5.4.2	Absence of Further hMSC Neuronal or Dopaminergic Neuronal
	Differentiation at Both Striatal and Nigral Graft Sites Post-
	Transplantation
5.4.3	Further Commitment of hMSCs to the Neuronal Lineage Prior to
	Transplantation May Be Necessary for Intracerebral Engraftment246
5.4.4	Unresolved Technical Aspects of hMSC Transplantation into the
	<i>CNS</i>
5.4.5	Controversies Associated with Cell Identification Techniques250
5.4.6	Transplantation of hMSCs Did Not Elicit Endogenous
	Neurotrophic Effects

-	5.4.7	Co-Transplantation of OECs Did Not Yield Improved Engraftment	
		of Neuronal-Primed hMSCs	256
-	5.4.8	Summary	257
CHAPT	FER 6	CONCLUSIONS AND FUTURE DIRECTIONS	259
6.1	Con	CLUDING REMARKS	260
6.2	Futu	JRE DIRECTIONS	265
СНАРТ	FER 7	REFERENCES	268
CHAPT	FER 8	SUPPLEMENTARY FILES	309

LIST OF FIGURES

Figure 1.1 Stem cell developmental potential
Figure 1.2 Position and axonal projection of dopaminergic cell groups within the
adult rodent brain (sagittal view)
Figure 1.3 A model illustrating the mechanisms controlling the positions and cell
fates of DA neurons during early development
Figure 1.4 Intrinsic and extrinsic factors affecting the specification and
development of midbrain dopaminergic neurons
Figure 3.1 Morphology of hMSCs at early and late culture stages
Figure 3.2 Proliferation and viability of hMSCs at early and late stages of culture 130
Figure 3.3 Flow cytometric detection of cell surface markers of undifferentiated
hMSCs
Figure 3.4 Histochemical staining of hMSCs following in vitro multi-lineage
mesodermal differentiation
Figure 3.5 Gene expression analysis of hMSCs following in vitro multi-lineage
mesodermal differentiation
Figure 3.6 Gene expression analysis of undifferentiated hMSC cultures during
long-term serial passage
Figure 3.7 Protein expression analysis of undifferentiated hMSC cultures during
long-term serial passage
Figure 4.1 Diagram of the hMSC neuronal differentiation methods
Figure 4.2 Examination of cytokine-induced hMSC neuronal differentiation by
live cell imaging
Figure 4.3 Examination of cytokine-induced hMSC neuronal differentiation by
immunofluorescence microscopy

Figure 4.4 Morphology of early and late passage hMSC cultures undergoing
neuronal differentiation
Figure 4.5 Gene expression of early and late passage hMSC cultures post-
neuronal differentiation
Figure 4.6 Morphology of hMSCs during in vitro cytokine-induced multiple-
stage dopaminergic neuronal differentiation (MultiDA) 164
Figure 4.7 Morphology of hMSCs during in vitro cytokine-induced single-stage
dopaminergic neuronal differentiation (SingleDA) 166
Figure 4.8 Morphology of hMSCs during in vitro cytokine-induced single-stage
neuronal differentiation (SingleND)
Figure 4.9 Expression of neuronal genes in hMSC cultures undergoing in vitro
neuronal and dopaminergic neuronal differentiation
Figure 4.10 Expression of dopaminergic neuronal genes in hMSC cultures
undergoing <i>in vitro</i> neuronal and dopaminergic neuronal differentiation 175
Figure 4.11 Expression of glial genes in hMSC cultures undergoing in vitro
neuronal and dopaminergic neuronal differentiation
Figure 4.12 Expression of mesodermal and pluripotency genes in hMSCs
undergoing neuronal & dopaminergic neuronal differentiation 177
Figure 4.13 Protein expression analysis of hMSCs during in vitro cytokine-
induced multiple-stage dopaminergic neuronal differentiation (MultiDA) by
immunofluorescence microscopy (Part 1) 182
Figure 4.14 Protein expression analysis of hMSCs during in vitro cytokine-
induced single-stage neuronal differentiation (SingleND) by immunofluorescence
microscopy

Figure 4.15 Protein expression analysis of hMSCs during in vitro cytokine-
induced single-stage dopaminergic neuronal differentiation (SingleDA) by
immunofluorescence microscopy
Figure 5.1 Presence of astrogliosis around undifferentiated hMSC grafts and
needle tract regions at 2 months post-transplantation
Figure 5.2 Evidence of autofluorescent cellular debris and lipofuscin in
undifferentiated hMSC grafts at 2 months post-transplantation
Figure 5.3 Identification of neuronal-primed hMSC graft sites and comparison
with the contralateral hemisphere
Figure 5.4 Evidence of limited survival of neuronal-primed hMSCs and OECs in
the lesioned hemisphere of the hemiparkinsonian rat
Figure 5.5 Confocal z-series identification of human cells at graft sites
Figure 5.6 Increasing glial accumulation at graft sites during the initial 3 weeks
post-intracerebral transplantation in the hemiparkinsonian rat
Figure 5.7 Absence of endogenous neuronal differentiation in host brain after
transplantation of neuronal-primed hMSCs231
Figure 5.8 Matrix deposition and lack of dopaminergic neuronal differentiation
after transplanting neuronal-primed hMSCs and OECs
Figure 5.9 Absence of neuronal marker expression within graft sites following
transplantation of neuronal-primed hMSCs

LIST OF TABLES

Table 1.1 Genetics of Parkinson's Disease 67
Table 1.2 In Vitro Dopaminergic Neuronal Differentiation of MSCs
Table 2.1 Summary of the Neuronal Differentiation Procedures
Table 2.2 Primer Pairs for Real-Time PCR
Table 2.3 Experimental Plan for Part I
Table 2.4 Experimental Plan for Part II
Table 3.1 Phenotypic Characterisation of Undifferentiated Human MSCs by
Immunofluorescence
Table 4.1 Summary of Real-Time RT-PCR Results: Gene Expression Changes
During hMSC Cytokine-Based Neuronal Differentiation
Table 4.2 Summary of hMSC Cytokine-Based Neuronal Differentiation 190

LIST OF ABBREVIATIONS

#	Including cytokines from Stage 1 (FGF-2 and EGF)
6-OHDA	6-Hydroxydopamine
AA	Ascorbic Acid
ACAN	Aggrecan
ADIPOQ	Adiponectin
AECs	Airway Epithelial Cells
α-FP	α-Fetoprotein
ALB	Albumin
APCs	Antigen-Presenting Cells
BDNF	Brain-Derived Neurotrophic Factor
BGLAP	Osteocalcin
BHA	Butylated Hydroxyanisole
BM	Bone Marrow
BME	β-mercaptoethanol
BMP	Bone Morphogenetic Protein
BrdU	5-bromo-2-deoxyuridine
BSA	Bovine Serum Albumin
cAMP	Cyclic Adenosine Mono-Phosphate
CFU	Colony-Forming Unit
CFU-F	Colony-Forming Unit – Fibroblast
СК	Cytokeratin
CNPase	2',3'-Cyclic nucleotide 3'-phosphohydrolase
CNS	Central Nervous System
CNTF	Ciliary Neurotrophic Factor
COL1A1	Type I Collagen
COL2A1	Type II Collagen
c-RET	RET Receptor Tyrosine Kinase
CSF	Colony-Stimulating Factor
CSPG4	Chondroitin Sulfate Proteoglycan 4
CTL	Cytotoxic T Lymphocyte
d	day
DA	Dopamine
DAPI	4'6-diamidino-2-phenylindole
DAT	Dopamine Transporter
dbcAMP	Dibutyryl Cyclic Adenosine Mono-Phosphate
DC	Dendritic Cell
DMEM	Dulbecco's Modified Eagle's Medium
DMEM-LG	Dulbecco's Modified Eagle's Medium – Low Glucose
DMSO	Dimethylsulfoxide
dPBS	Dulbecco's Phosphate Buffered Saline
E	Embryonic Day
EDTA	Ethylenediamine Tetraacetic Acid
EGF	Epidermal Growth Factor
EN	Engrailed
ESC	Embryonic Stem Cell
FACS	Fluorescence Activated Cell Sorting
FBS	Foetal Bovine Serum

ECE	
FGF	Fibroblast Growth Factor
FOXA2	Forkhead Box A2 (also known as HNF-3 β)
Fzd	Frizzled
GABA	γ-Aminobutyric Acid
GALC	Galactosylceramidase
GAPDH	Glyceraldehyde-3-Phosphate Dehydrogenase
GDNF	Glial-Derived Neurotrophic Factor
GFAP	Glial Fibrillary Acidic Protein
GFP	Green Fluorescent Protein
GFRA1	GDNF Family Receptor Alpha 1
GM-CSF	Granulocyte-Macrophage-CSF
GVHD	Graft-Versus-Host Disease
HGF	Hepatocyte Growth Factor
HBSS	Hank's Balanced Salt Solution
HLA	Human Leukocyte Antigen
hMSC	Human Mesenchymal Stem Cell
HNA	Human Nuclear Antigen
HNF	Hepatocyte Nuclear Factor
HPRT1	Hypoxanthine Phosphoribosyltransferase 1
HSC	Haematopoietic Stem Cell
Iba1	Ionised Calcium Binding Adaptor Molecule 1
IBMX	Isobutyl Methyl Xanthine
IDO	Indoleamine 2.3-dioxygenase
IFN	Interferon
Ig	Immunoglobulin
IL	Interleukin
iPS	Induced Pluripotent Stem
ITS	Insulin Transferrin Selenium
iv	Intravenous
IAK	Janus Kinase
LDL	Low-Density Linoprotein
L-DOPA	3 4 -Dihydroxyphenylalanine
LMX	LIM homeobox transcription factor
LPL	Linonrotein Linase
MAP	Microtubule-Associated Protein
MAPC	Multipotent Adult Progenitor Cell
MAPK	Mitogen-Activated Protein Kinase
MRP	Muelin Basic Protein
MHC	Major Histocompatibility Complex
min	Minutes
MNC	Mononuclear Cell
МРТР	1-Methyl_4-Phenyl_1 2 3 6-Tetrahydronyridine
MSC	Mesonchymal Stem Cell
MultiDA	Multiple-Stage Dopaminergic Neuronal Differentiation
NANOG	Nanog Homeobox
NES	Nectin
NouN	Nouronal Nuclear Antigen
NELIROD1	Neurogenic Differentiation 1
NEUROG	Neurogenin
NEUKUU NE M	Neurofilament protein subunit of madium malagular weight
1 N 1 ' - 1 V 1	recuromament protein subunit of medium molecular weight

NCE	Namua Crowth Easter
NUCD	Notah Intración
NICD	Noten Intracenular Domain
NK	Natural Killer
NR4A2	Nuclear Receptor Subfamily 4, Group A, Member 2 (also known
	as NURR1)
NSC	Neural Stem Cell
NSE	Neuron-Specific Enolase
OECs	Olfactory Ensheathing Cells
OSM	Oncostatin M
OTX	Orthodenticle Homeobox
Р	Passage
PAX	Paired Box Gene
PBS	Phosphate Buffered Saline
PD	Parkinson's Disease
PDGF	Platelet-Derived Growth Factor
PEG	Polyethylene Glycol
PET	Positron Emission Tomography
PFA	Paraformaldehvde
PGE2	Prostaglandin E2
PI	Propidium Iodide
PITX3	Paired-Like Homeodomain Transcription Factor-3
PKA	Protein Kinase A
POU5F1	POU Class 5 Homeobox 1 (also known as $OCT3/4$)
PTC	Patched
R A	Retinoic Acid
RT DCP	Reverse Transcription Polymerose Chain Reaction
	Seconds
S S	Store
S SC	Stage
SU	Stelli Cell Sonia Hadashaa
SПП SinalaDA	Sonic Hedgenog
SingleDA	Single-Stage Dopaminergic Neuronal Differentiation
SingleND	Single-Stage Neuronal Differentiation
SMO	Smoothened
SOX2	Sex Determining Region Y-box 2
SSEA	Stage-Specific Embryonic Antigen
STAT	Signal Transducer, and Activator of Transcription
TGF	Transforming Growth Factor
TH	Tyrosine Hydroxylase
T _H	T Helper
TNF- α	Tumour Necrosis Factor α
TPA	4β-12-O-tetradecanoylphorbol 13-acetate
VMAT2	Vesicular Monoamine Transporter 2
VPA	Valproic Acid
VTA	Ventral Tegmental Area
WNT	Wingless-Type MMTV Integration Site Family

CHAPTER 1

LITERATURE REVIEW

1.1 Bone Marrow-Derived Mesenchymal Stem Cells

1.1.1 Adult Bone Marrow-Derived Stem Cells

Stem cells (SCs) are regarded as possessing a unique position in the scheme of cell biology, due to the defining properties of self-renewal, extensive proliferation, differentiated progeny generation, and broad developmental potential (Anderson et al., 2001; Blau et al., 2001). Traditionally, a hierarchical system based on developmental potential has been employed for the classification of vertebrate SCs (Figure 1.1). Embryonic SCs (ESCs) derived from the inner cell mass of the blastocyst possess the greatest developmental potential with the ability to generate all differentiated cell types in the body, and are termed pluripotent. Whereas, tissue-committed SCs are restricted to producing cell types found within that tissue, and are termed multipotent. A range of terminologies have been used in describing committed SCs, but for the purposes of this thesis tissue-committed SCs will be described using the tissue of origin as a prefix, and progenitor cells will refer to the more committed progeny of the tissue-committed SCs. Interest in this area of cell biology has been fuelled by discoveries of SC populations residing in multiple adult tissues and organs, including the bone marrow (BM), epidermis, gastrointestinal epithelium and nervous system. The first adult tissue described to contain a population of SCs was the BM (Cole et al., 1955; Lorenz et al., 1951; Nowell et al., 1956; Till et al., 1961).

In the early 1950's, several independent studies suggested the presence of a factor within BM that, upon transplantation, was capable of preventing death in laboratory animals after otherwise lethal whole body irradiation (Cole *et al.*, 1955; Lorenz *et al.*, 1951; Nowell *et al.*, 1956). This culminated in the Nobel prize-winning discovery, by Till and McCulloch, of BM-derived cells that were capable of

2



Figure 1.1 Stem cell developmental potential.

Stem cells (SCs) are typically classified according to developmental potential, with pluripotent embryonic SCs (ESCs) possessing the greatest capacity for differentiation. Multipotent tissue-committed SCs are restricted to producing cell types of the tissue from which the SCs are derived. A long-held belief of the process of SC commitment is that it proceeds unidirectionally from left to right, with decreasing potential and increasing lineage commitment. More recently, the extent of lineage commitment of tissue SCs has been challenged by findings of transdifferentiation of cells from one lineage to a different lineage under certain circumstances, giving rise to the concept of adult SC plasticity. BM, bone marrow.

giving rise to colonies of rapidly proliferating and differentiating haematopoietic tissue in the spleen (Till *et al.*, 1961). These populations of "colony-forming units (CFUs)" are today known to be haematopoietic SCs (HSCs), which have the ability to reconstitute cells of the entire haematopoietic spectrum. To this day, of all the SCs identified in adult tissues, HSCs are perhaps the most well characterised both *in vitro* and *in vivo*, and much knowledge has been attained of the hierarchical lineage progression from HSC to mature progeny. This is due in part to the relative ease of assaying HSCs *in vivo*, due to the ability of prospectively isolated HSCs to circulate and home to permissive niches, the continuous and rapid turnover of HSCs and HSC-derived progeny, and the systemic distribution of haematopoiesis in the body (Bianco *et al.*, 2008; Ma, 1990). BM is now known to consist of a heterogeneous population of cells, including two types of SCs, HSCs as well as mesenchymal stem cells (MSCs), also known as multipotent mesenchymal stromal cells.

1.1.2 The Mesenchymal Stem Cell Identity

Evidence for the existence of a cell population within the BM that produced non-haematopoietic progeny emerged in the mid-1960's with the pioneering work of Friedenstein and colleagues (Friedenstein *et al.*, 1970; Friedenstein *et al.*, 1968; Friedenstein *et al.*, 1966). These studies demonstrated the occurrence of *in vivo* osteogenesis in transplants of BM (Friedenstein *et al.*, 1968; Friedenstein *et al.*, 1966), as well as *in vitro* clonal fibroblast colony formation in monolayer culture (also known as CFU – fibroblastic (CFU-F) formation) (Friedenstein *et al.*, 1970). This provided the framework for subsequent studies, particularly by the development of a rudimentary technique to isolate the MSC fraction from BM by adherence to cultureware and removal of non-adherent haemopoietic cells through a complete medium change at 24-48 hours post-plating (Friedenstein *et al.*, 1970; Friedenstein *et al.*, 1974b). These

findings were extended by a number of independent investigators, who employed the isolation and culture techniques of Friedenstein, *et al.*, and together these studies demonstrated that MSCs were multipotential and differentiated into cells of the mesodermal lineage, including osteoblasts (Beresford *et al.*, 1992; Cheng *et al.*, 1994; Friedenstein *et al.*, 1987; Howlett *et al.*, 1986; Rickard *et al.*, 1994), chondroblasts (Friedenstein *et al.*, 1987; Johnstone *et al.*, 1998; Mackay *et al.*, 1998), adipocytes (Beresford *et al.*, 1992; Lanotte *et al.*, 1982) and myoblasts (Wakitani *et al.*, 1995). The marrow stromal SC concept was proposed in 1988 by Owen and Friedenstein (Owen *et al.*, 1988), whereas the currently popular term "MSC" was later coined by Caplan (Caplan, 1991). Collectively, these studies have paved the way for the development of the field of MSC research as we know it today.

Since these initial reports, scientific and clinical research interest in MSCs has increased exponentially. It is now generally recognised that MSCs possess the *in vitro* characteristics of SCs with the ability to proliferate, symmetrically divide, and produce multi-lineage mesodermal derivatives, making MSCs an attractive candidate for use in potential cellular therapies. In addition, MSCs exhibit further promising qualities for regenerative medicine, including relatively easy isolation from small aspirates of BM, as well as relatively easy expansion in culture with low tumorigenicity and teratoma formation. Furthermore, MSCs have been reported to display immunosuppressive properties that are advantageous for allogeneic transplantation (see Section 1.1.5), and in ideal settings, autologous transplantation is also possible. Subpopulations of MSCs have also been reported to be capable of differentiation to non-mesodermal lineages (see Section 1.2.2, 1.2.3, and 1.3), indicating the potential application of MSCs in a wider range of diseases.

The results of decades of research are beginning to elucidate the factors affecting the complex regulation of MSC differentiation and proliferation pathways. However, despite the intense interest and research activity surrounding MSCs, there remain large gaps in our knowledge and understanding of these cells. Currently, distinctive phenotypic markers of MSCs are still yet to be identified, and MSCs have primarily been characterised *in vitro*, with *in vivo* validation of SC qualities yet to be shown. In addition, laboratories investigating MSCs have employed different methods of isolation, expansion and characterisation of these cells (see Section 1.1.3). MSC-like cells have also been isolated from a variety of tissue sources, including adipose tissue, skeletal muscle and umbilical cord, leading to the suggestion that cells with MSC-like characteristics reside in virtually all postnatal organs and tissues. Cumulatively, this has led to difficulties in comparing and contrasting study outcomes, and has added to the controversy and confusion in the field.

Consequently, criteria for defining MSCs has increasingly been sought, and the International Society for Cellular Therapy has proposed minimal criteria for defining human MSCs (hMSCs) in an effort to promote more uniform characterisation of MSCs and facilitate data exchange amongst laboratories (Dominici *et al.*, 2006). This proposal was based on the best available data at the time, and recognised that the emergence of new data would require revisions of the criteria. Three criteria were proposed to define hMSCs: 1) MSCs must be adherent to plastic when cultured in standard conditions; 2) MSC populations must express specific surface antigens CD105, CD73 and CD90 (at \geq 95% purity), but lack expression of haematopoietic surface antigens (\leq 2% positive), such as pan-leukocyte marker CD45, primitive haematopoietic progenitor and endothelial marker CD34, monocyte and macrophage markers CD14 and CD11b, B lymphocyte markers CD79a and CD19, and human leukocyte antigen (HLA)-DR

molecules; and 3) MSCs must exhibit multipotent differentiation potential by differentiating to osteoblasts, adipocytes and chondroblasts under standard *in vitro* tissue culture-differentiating conditions with confirmation of differentiation using histochemical and immunohistochemical staining (Dominici *et al.*, 2006). However, this proposal only addresses the *in vitro* characteristics of MSCs. The reason for this is a limitation common across most fields of adult SC research, that of primary examination of adult SCs only in *in vitro* systems, with little known of the function or existence of these cells *in vivo* (HSCs excepted). In the case of MSCs, while it is accepted that *in vitro* CFU-F potential and multipotent mesodermal differentiation may indicate MSC activity, it has thus proven far more challenging for MSCs, than for HSCs, to demonstrate *in vivo* long-term survival with self-renewal capacity and tissue repopulation with multi-lineage differentiation (Horwitz *et al.*, 2005). Therefore, whilst it is common practice to refer to MSCs as a type of SC, and while it is also generally accepted that a bona fide MSC exists, it must be acknowledged that the *in vivo* evidence for this remains to be demonstrated.

The issue of terminology remains controversial, and further suggestions have recently been made, with Bianco and colleagues (Bianco *et al.*, 2008) suggesting that the term "skeletal SC" be used until it has been definitively proven that these cells can produce tissues other than skeletal cell types. Bianco, *et al.* (Bianco *et al.*, 2008) also suggest that the term CFU-F be used to describe a cell that is assayed as clonogenic in culture, with a prefix to denote the tissue of origin, e.g. BM-CFU-F for CFU-F derived from BM.

1.1.3 Isolation and Expansion of Mesenchymal Stem Cells

The identification of MSCs has been limited as the biologic property which most uniquely identifies MSCs, that of multi-lineage mesodermal differentiation (Pittenger *et* *al.*, 1999; Prockop, 1997), does not permit prospective isolation. The plastic-adherence method of isolation, described by Friedenstein, *et al.* (Friedenstein *et al.*, 1970) more than 35 years ago, is still commonly used for MSC isolation today. Additionally, in this time little progress has been made in further optimising MSC culture conditions, with the majority of studies depending on the addition of foetal bovine serum (FBS) for MSC expansion. The reported frequency of MSC colony formation from the mononuclear cell (MNC) fraction of BM is approximately 1 in 10^5 to 1 in 10^6 (Bruder *et al.*, 1997; Reyes *et al.*, 2001), highlighting the need for increased homogeneity in isolated cells and the necessity of efficient expansion methods to generate sufficient cells for use in downstream applications.

Since the 1990s, efforts have been made by independent laboratories to improve the purity of isolated cells and to achieve clonal MSC populations. The persistence of haematopoietic cells in MSC cultures (Cheng *et al.*, 1994; Friedenstein *et al.*, 1970) prompted some investigators to implement antibody-mediated negative selection procedures. Cells expressing CD14, CD31 and CD11a have been removed from adherence isolated BM stromal cells via magnetic Dynabeads and a magnetic particle concentrator (Rickard *et al.*, 1996). CD45-positive cells and glycophorin A-positive cells have been depleted from the MNC fraction by means of immunomagnetic microbeads and MACS separation after Ficoll-Paque density gradient centrifugation (Reyes *et al.*, 2001; Tondreau *et al.*, 2004a). Most recently, haematopoietic cells have also been removed using tetrameric antibody complexes, which link the target cells to erythrocytes for removal during Ficoll-Paque centrifugation (Tondreau *et al.*, 2004a).

Alternatively, antibody-mediated positive selection procedures for prospective MSC isolation have also been investigated; however, identification of the most appropriate markers for this purpose remains an issue. Markers that have been examined for positive selection include: STRO-1, which has been utilised after MNC isolation by Ficoll density gradient centrifugation (Gronthos *et al.*, 1994; Simmons *et al.*, 1991); low affinity nerve growth factor receptor, also known as p75 neurotrophin receptor or CD271 (Quirici *et al.*, 2002); CD49a, the α 1-integrin subunit of VLA-1 (very-late antigen 1) integrin, which is a receptor for both collagen and laminin (Deschaseaux *et al.*, 2000; Deschaseaux *et al.*, 2003); neural ganglioside GD2 (Martinez *et al.*, 2007); stage-specific embryonic antigen (SSEA)-4, an early embryonic glycolipid antigen (Gang *et al.*, 2007); and melanoma-associated cell adhesion molecule, MCAM/CD146, which was applied following CD45⁺ cell depletion (Sacchetti *et al.*, 2007). Nonetheless, despite intense research, no consensus has been reached on a definitive BM MSC marker for prospective isolation and *in situ* identification. However, the recent finding of CD146-expressing subendothelial cells within BM stroma, may provide a promising lead for future studies, as the CD146 marker identified self-renewing osteoprogenitors that were capable of establishing the haematopoietic microenvironment at heterotopic sites *in vivo* (Sacchetti *et al.*, 2007).

Attempts have been made to improve MSC culture conditions with the intent of generating more pluripotent cells or to meet standards for clinical use. However, these newer methods have yet to be implemented in mainstream use. Furthermore, the use of different tissue sources and isolation and culture techniques by different laboratories has resulted in the isolation of subpopulations of MSCs with different characteristics. This has added further complexity to the field, and highlights the need for independent examination of the baseline characteristics of MSC populations employed in different studies.
1.1.4 Characterisation and Function of Mesenchymal Stem Cells

Although the discovery of MSCs occurred several decades ago, the properties and function of these cells are still being elucidated today. The multi-lineage mesodermal differentiation capacity of MSCs has been known for some time, however, research is currently underway to determine the mechanisms underlying these processes and methods for directed differentiation towards functional mature cell types. The search for a definitive marker of MSCs is also ongoing, and much progress has been made in characterising the MSC surface marker phenotype. The migratory capacity, cytokine secretion and unique immunological features of MSCs are also under further investigation. Furthermore, research aimed at defining the microenvironmental niche and the role of MSCs *in vivo* continues to be examined.

1.1.4.1 Initial Observations

Initially, the adherent cells isolated by Friedenstein, *et al.* were described as a heterogeneous population, which contained a number of isolated large fibroblast-like cells by the third day post-explantation (Friedenstein *et al.*, 1970; Friedenstein *et al.*, 1976). Haematoxylin staining of these cultures revealed the fibroblast-like cells to display a pale cytoplasm and ovoid nuclei containing several distinct nucleoli. After 5-6 days, discrete foci of 4-8 fibroblasts were observed in the cultures, and the cells multiplied rapidly to form large discrete colonies of 10^2 - 10^3 fibroblasts by the 10^{th} day, with colonies fusing together on the 15^{th} day. Friedenstein, *et al.* also demonstrated the osteogenic ability of these colony-forming fibroblasts after transplantation in diffusion chambers *in vivo* (Friedenstein *et al.*, 1970), and importantly, showed retention of the capacity for osteogenesis and chondrogenesis after 20-30 cell doublings *in vitro* (Friedenstein *et al.*, 1987).

1.1.4.2 Role of MSCs in Supporting Haematopoiesis In Vivo

The essential role of BM stroma in supporting *in vivo* haematopoiesis was established in early experiments, which demonstrated successful formation of the haematopoietic microenvironment at ectopic sites following transplantation of BM fragments or BM stromal cells (Friedenstein *et al.*, 1974a; Friedenstein *et al.*, 1968; Maniatis *et al.*, 1971; Tavassoli *et al.*, 1968). Tavassoli and Crosby showed that transplantation of large autologous marrow fragments resulted in initial necrosis and phagocytosis of haematopoietic elements, leaving behind a network of mesenchymal cells (Tavassoli *et al.*, 1968). These cells proliferated and acquired osteoblastic morphology with development of a lattice of trabecular bone throughout the implant by day 6, and later also formed the lining layer of sinusoids. After establishment of the sinusoidal microcirculation, haematopoietic elements appeared around the sinusoids in the interstices of the bone, which was gradually resorbed, leaving only a shell of bone encapsulating the new haematopoietic tissue by the 5th week (Tavassoli *et al.*, 1968).

Semi-syngeneic BM transplantation experiments revealed that by 12-14 months the structures formed were chimaeric, with the haematopoietic tissue of recipient origin, whilst the osteogenic tissue was donor-derived (Friedenstein *et al.*, 1968). And subsequently, through irradiation of BM prior to transplantation, Maniatis, *et al.* found that the osteogenic precursor cells were associated with the transplanted BM stroma, and that donor-derived haematopoietic cells were not necessary for regeneration of marrow implants (Maniatis *et al.*, 1971). However, the precise identity and anatomical location of the cells within the BM fragments that established the heterotopic haematopoietic microenvironment remained unknown.

Recent evidence has emerged describing CD146 as a potential marker of these cells (Sacchetti *et al.*, 2007). CD146 was found to label subendothelial (adventitial)

reticular cells in sinusoidal walls, which projected reticular processes that associated with haematopoietic cells. Furthermore, subcutaneous transplantation of human CD146⁺ CFU-Fs in immunocompromised mice revealed a defined developmental sequence of bone formation followed by the appearance of a sinusoidal system, and finally colonisation of the ossicle with murine haematopoietic tissue by 8 weeks (Sacchetti *et al.*, 2007).

1.1.4.3 Role of MSCs In Supporting Haematopoiesis In Vitro

Early studies on HSCs also identified that BM-derived adherent cells provide an important microenvironment in *in vitro* culture systems for the long-term maintenance, proliferation and differentiation of HSCs. Dexter and colleagues found that the introduction of a BM-derived adherent cell layer significantly improved *in vitro* murine HSC culture, and allowed granulopoiesis and HSC production to be maintained for 2-3 months, whereas previous systems were only successful for several days to a few weeks (Dexter *et al.*, 1977). In addition, cultures established without the adherent cell layer showed rapid decline in the numbers of cells within 1-2 weeks, as well as disappearance of HSCs and granulocyte precursors within 2-3 weeks (Dexter *et al.*, 1977).

Gartner and Kaplan adapted this liquid medium and BM feeder layer culture system to the *in vitro* culture of human HSCs (Gartner *et al.*, 1980). They found the production of granulocytic-macrophage progenitor cells to be dependent on the BMderived adherent layer of cells, and the continuation of haematopoiesis for at least 20 weeks. Furthermore, active haematopoiesis was observed to occur in "cobblestone"like areas, which contained myeloid and monocytoid cells. The non-adherent cells were very closely associated with the stromal layer, and monocytoid cells were seen scattered throughout the stromal layer in islands of haematopoiesis and in large tight clusters within the stromal layer.

Later studies showed that haematopoiesis-supporting stromal cells secrete biologically functional cytokines (including colony-stimulating factor (CSF)-1, granulocyte-CSF, granulocyte-macrophage-CSF (GM-CSF), interleukin (IL)-6 and c-kit ligand), which are involved in the development of haematopoietic cells, in particular those of the myeloid lineage (Eaves *et al.*, 1991; Kittler *et al.*, 1992). In addition, the adherent stromal cells synthesise abundant extracellular matrix, consisting of basal lamina and interstitial collagen types, including collagen types I, III, IV and V, as well as fibronectin and proteoglycans, in particular chondroitin sulfate-containing proteoglycans (Clark *et al.*, 1995).

1.1.5 Immunomodulatory Properties of MSCs

Further clinical interest in MSCs has been fuelled by an emerging body of data indicating that MSCs possess immunomodulatory properties. MSCs had been long thought of as having an immune privileged status, but are now recognised to possess the dual ability to both suppress and/or activate immune responses, depending on the stimulus to which the cells have been exposed (Locatelli *et al.*, 2007; Stagg, 2007). Initially, investigations centred around the immunosuppressive nature of MSCs in relation to T lymphocytes (Di Nicola *et al.*, 2002; Krampera *et al.*, 2003; Le Blanc *et al.*, 2003). However, increasing evidence has revealed that MSCs may exert effects on a broad range of immune cells, including antigen-presenting cells (APCs) (Aggarwal *et al.*, 2005; Beyth *et al.*, 2005; Jiang *et al.*, 2005; Krampera *et al.*, 2006; Maccario *et al.*, 2004), natural killer (NK) cells (Aggarwal *et al.*, 2005; Krampera *et al.*, 2006; Maccario *et al.*, 2005; Rasmusson *et al.*, 2003; Sotiropoulou *et al.*, 2006; Spaggiari *et al.*, 2006) and

B cells (Augello *et al.*, 2005; Corcione *et al.*, 2006; Glennie *et al.*, 2005; Krampera *et al.*, 2006).

1.1.5.1 Effects of MSCs on T cells

In vitro studies with human, baboon and murine MSCs have demonstrated that MSCs are capable of inhibiting the proliferation of T cells induced by allogeneic cells or mitogens such as phytohemagglutinin and concanavalin A (Bartholomew *et al.*, 2002; Di Nicola *et al.*, 2002; Le Blanc *et al.*, 2003; Tse *et al.*, 2003), as well as the activation of T cells by CD3 and CD28 antibodies (Krampera *et al.*, 2003; Tse *et al.*, 2003), in a dose-dependent manner. Furthermore, MSCs have been reported to escape recognition by cytotoxic T lymphocytes (CTLs), inhibit the formation of CTLs, induce formation of regulatory CD8⁺ T cells, decrease the secretion of interferon (IFN)- γ by T helper (T_H) 1 cells and increase IL-4 secretion by T_H2 cells, favour the differentiation of CD4⁺CD25⁺ regulatory T cells and decrease the alloantigen-specific cytotoxic capacity of CTLs or NK cells (Aggarwal *et al.*, 2005; Djouad *et al.*, 2003; Maccario *et al.*, 2005; Rasmusson *et al.*, 2003).

The suppressive effect of MSCs on T cell proliferation was found to be independent of the major histocompatibility complex (MHC) (Krampera *et al.*, 2003; Le Blanc *et al.*, 2003), with the majority of studies showing involvement of soluble factors secreted by MSCs, such as hepatocyte growth factor (HGF), transforming growth factor (TGF)- β 1, prostaglandin E2 (PGE2), indoleamine 2,3-dioxygenase (IDO; which depletes tryptophan, an essential factor for lymphocyte proliferation), nitric oxide and IL-10 (Aggarwal *et al.*, 2005; Di Nicola *et al.*, 2002; Djouad *et al.*, 2003; Meisel *et al.*, 2004; Rasmusson *et al.*, 2003; Rasmusson *et al.*, 2005; Sato *et al.*, 2007; Tse *et al.*, 2003). However, due to conflicting data, there remains no consensus on which soluble factors are essential for suppression. In addition, a role for direct cell-to-cell contact

mechanisms in MSC-mediated suppression has also been described (Augello *et al.*, 2005; Krampera *et al.*, 2003).

Another possible mechanism of MSC-mediated immune suppression is induction of T cell anergy. MSCs express MHC class I molecules which may activate T cells, but lack surface expression of costimulatory molecules, such as CD80 and CD86, resulting in MSCs being unable to provide a secondary signal and thereby leaving T cells anergic (Chamberlain *et al.*, 2007; Nauta *et al.*, 2007; Stagg, 2007). MSCs have been demonstrated to induce division arrest anergy in T cells, in which T cell proliferation was inhibited (arrested at G₁ phase) and IFN- γ production was decreased (Glennie *et al.*, 2005). However, this condition differed to classical anergy, which can be reversed by exogenous IL-2, as T cell proliferation could not be restored upon addition of exogenous IL-2.

Overall, these studies have been interpreted as indicative of an immunosuppressive role of MSCs, however, it has been acknowledged that the current evidence only shows targeting of T cell proliferation *in vitro* rather than T cell effector function, and therefore may suggest a general non-specific anti-proliferative effect of MSCs (Nauta *et al.*, 2007; Ramasamy *et al.*, 2007; Ramasamy *et al.*, 2008).

1.1.5.2 Effects of MSCs on Professional Antigen-Presenting Cells

The differentiation, maturation and function of the initiators of the immune response, the professional APCs e.g. dendritic cells (DCs), are also disrupted by MSCs. Several studies have shown that the presence of MSCs significantly inhibited differentiation of both monocytes and CD34⁺ progenitors into DCs, and prevented the increase of CD1a, CD40, CD80, CD86 and HLA-DR expression. During maturation, MSCs prevented up-regulation of CD40 and CD86, while mature DCs treated with MSCs also showed reduced CD83 expression, suggesting skewing toward an immature

status. Furthermore, exposure to MSCs during differentiation and maturation resulted in cells with decreased capacity for inducing T cell proliferation, showing that DC phenotype and function are affected by MSCs (Jiang *et al.*, 2005; Nauta *et al.*, 2006a; Zhang *et al.*, 2004). It has also been reported that hMSCs induce APCs toward a regulatory phenotype, and that the cells produced possess T cell-suppressive properties (Beyth *et al.*, 2005). Additionally, alterations in DC cytokine production have been demonstrated, with hMSCs found to cause decreased secretion of pro-inflammatory cytokines tumour necrosis factor- α (TNF- α) and IL-12, and increased production of anti-inflammatory cytokine IL-10 (Aggarwal *et al.*, 2005; Beyth *et al.*, 2005; Jiang *et al.*, 2005; Zhang *et al.*, 2004). Together these results indicate that MSCs are capable of suppressing DC differentiation and cause the formation of immature DCs that display a suppressor or inhibitory phenotype (Nauta *et al.*, 2007).

1.1.5.3 Effects of MSCs on B cells

MSCs have also been shown to exert immunosuppressive activities on B cells. Corcione, *et al.* (Corcione *et al.*, 2006) demonstrated that hMSCs inhibited *in vitro* human B cell proliferation by arresting cells in the G_0/G_1 cell cycle phase, prevented differentiation to antibody-secreting cells as seen by impaired immunoglobulin (Ig)M, IgG and IgA production, and interfered with chemotaxis through down-regulation of CXCR4, CXCR5 and CCR7 expression. However, no effect on B cell costimulatory molecule expression and cytokine production was observed, while paracrine secretion of soluble factors was implicated in the mechanism of B cell suppression. Murine studies have also observed similar findings of inhibition of B cell proliferation following stimulation with anti-CD40 monoclonal antibody and IL-4 or pokeweed mitogen, with soluble factors playing a role in suppression (Augello *et al.*, 2005; Glennie *et al.*, 2005). Additionally, co-culturing of B cells from a murine model of

systemic lupus erythematosus with allogeneic MSCs resulted in inhibition of B cell proliferation, activation and IgG secretion (Deng *et al.*, 2005). However, contradictory data has been reported, which found that B cell proliferation could only be inhibited by MSCs in the presence of exogenous IFN- γ (Krampera *et al.*, 2006). Nonetheless, these studies suggest that MSCs are capable of modulating B cell functions at multiple levels.

1.1.5.4 Effects of MSCs on Natural Killer Cells

Several independent groups have shown that MSCs also inhibit NK cell proliferation in response to IL-2, IL-15 and alloantigens (Krampera *et al.*, 2006; Maccario *et al.*, 2005; Sotiropoulou *et al.*, 2006; Spaggiari *et al.*, 2006). However, proliferation of activated NK cells was only partially suppressed by MSCs (Spaggiari *et al.*, 2006). Additionally, co-culture with MSCs was found to suppress the effector functions of NK cells. Secretion of cytokines, including IFN- γ , IL-10 and TNF- α , were reduced after culturing in the presence of MSCs (Aggarwal *et al.*, 2005; Krampera *et al.*, 2006; Sotiropoulou *et al.*, 2006; Spaggiari *et al.*, 2006; Sotiropoulou *et al.*, 2006; Spaggiari *et al.*, 2006; Sotiropoulou *et al.*, 2005; Sotiropoulou *et al.*, 2006; Spaggiari *et al.*, 2006; Sotiropoulou *et al.*, 2006; Sotiropoulou *et al.*, 2006; Spaggiari *et al.*, 2008).

Spaggiari, *et al.* (Spaggiari *et al.*, 2008) showed that the suppression of cytotoxic activity was related to the down-regulation of surface expression of activating NK receptors NKp30 and NKG2D, which are involved in NK cell-activation and target cell killing, and absence of NKp44 activating receptor. However, other studies have found that cytolytic activity is not inhibited in freshly isolated NK cells (Rasmusson *et al.*, 2003), or was restricted to HLA-class I-expressing targets (Sotiropoulou *et al.*, 2006).

Initially, MSCs were thought to escape recognition by NK cells, since it was reported that MSCs were not lysed by killer cell immunoglobulin-like receptor ligand mismatched alloreactive NK cells (Rasmusson *et al.*, 2003). However, recent reports have shown that hMSCs are highly susceptible to lysis by activated NK cells, and that this is mediated by activating NK receptors NKp30, NKG2D and DNAM-1 and the expression of the corresponding ligands by MSCs (Sotiropoulou *et al.*, 2006; Spaggiari *et al.*, 2006). Interestingly, exposure of hMSCs to IFN- γ caused protection from NK cell-mediated lysis due to the up-regulation of surface HLA-class I expression (Spaggiari *et al.*, 2006).

The mechanisms of MSC-mediated NK cell suppression are still under investigation, but recent reports have found involvement of soluble factors IDO, PGE2 and TGF- β (Krampera *et al.*, 2006; Sotiropoulou *et al.*, 2006; Spaggiari *et al.*, 2008) as well as cell-to-cell contact (Sotiropoulou *et al.*, 2006), suggesting the existence of different mechanisms.

1.1.5.5 MSC Immunomodulation in In Vivo Studies

The immunosuppressive effects and immunogenicity of MSCs have primarily been demonstrated in *in vitro* studies, and only a few studies have investigated these properties in *in vivo* or clinical settings. However, the limited data available at present provides little evidence that donor MSCs are able to engraft after infusion or transplantation (Fibbe *et al.*, 2007; Le Blanc *et al.*, 2005).

In one of the earliest *in vivo* studies, Bartholomew and colleagues (Bartholomew *et al.*, 2002) showed that intravenous administration of allogeneic, MHC-mismatched MSCs prolonged the survival of allogeneic skin grafts in baboons to 11 days, in comparison with 7 days in control animals. Engraftment in multiple tissues and site-specific differentiation of hMSCs has also been reported after intrauterine transplantation in foetuses of sheep, even after development of immunocompetency (Airey *et al.*, 2004; Liechty *et al.*, 2000).

MSCs have also been found to exert therapeutic effects on severe graft-versushost disease (GVHD), which was first reported in a landmark case study showing that repeated infusion of MSCs from the patient's HLA-haploidentical mother with cyclosporine treatment, completely reversed grade IV acute GVHD of the gut and liver (Le Blanc *et al.*, 2004). Additionally, systemic infusion of adipose-derived MSCs has been found to control lethal GVHD in mice, with greater efficiency if MSCs were administered early after HSC transplantation, and possible requirement of repeated doses (Yanez *et al.*, 2006). However, another murine study failed to show any effect of MSC transplantation on the incidence and severity of GVHD, although this may be due to administration of a single MSC dose alone (Nauta *et al.*, 2007; Sudres *et al.*, 2006). Horwitz, *et al.* have also administered allogeneic MSCs to children with osteogenesis imperfecta via intravenous (i.v.) infusion and demonstrated potential engraftment, at least for short-term periods of 4-6 weeks (Horwitz *et al.*, 2002).

Contrary to these findings, MSCs have been found to be immunogenic in certain transplant scenarios, indicating that MSCs are not intrinsically immune privileged. Xenogeneic transplantation of hMSCs into myocardium of immunocompetent rats resulted in an intense cellular immune response (primarily macrophage infiltration) and graft rejection (Grinnemo *et al.*, 2004). In contrast, myocardial engraftment was seen after transplantation into RNU athymic or tacrolimus-immunosuppressed rats. Furthermore, significant rat lymphocyte proliferation was observed when hMSCs were co-cultured with lymphocytes of rats previously exposed to hMSCs, indicating the presence of a sensitisation reaction. Another study examining the effect of MSCs on allogeneic BM transplantation in sublethally irradiated mice found that addition of syngeneic host MSCs significantly enhanced long-term engraftment, whereas infusion of donor-derived MSCs not only failed to prevent rejection, but increased rejection

(Nauta *et al.*, 2006b). This study further showed that MSCs were capable of inducing a memory T cell response after injection into immunocompetent hosts. Together with a previous study showing rejection of allogeneic MSCs after transplantation in MHC-mismatched mice (Eliopoulos *et al.*, 2005), these findings indicate that MSCs can be immunogenic under appropriate circumstances.

1.1.5.6 Controversy Related to MSC Immunomodulation

The precise mechanisms underlying the immunomodulatory properties of MSCs remain largely unresolved, despite being of increasing importance. The currently available data is controversial, and the majority of reports including a clinical study performed by our group (Ma *et al.*, 1987), have not demonstrated long-term engraftment of donor MSCs. Le Blanc and Ringden (Le Blanc *et al.*, 2005) have commented that the diversity in findings could be attributed to the use of different techniques for obtaining MSCs, different stimuli, culture conditions, doses and kinetics, and different lymphocyte populations. Furthermore, after *in vivo* transplantation it has been difficult to detect or recover MSCs from the BM of recipients, which could be due to a number of reasons, including homing of MSCs to other sites for mediating immune suppression, non-specific lodging in capillary beds of other tissues (particularly pulmonary tissue), the possibility that MSCs exert effects by production of growth factors and die after completion of this role, or perhaps difficulty in detecting MSCs within BM aspirates due to the endosteal location of these cells (Le Blanc *et al.*, 2005; Nauta *et al.*, 2007).

1.1.6 Classical Multi-Lineage Mesodermal Differentiation Capacity of MSCs

Consistent with the germ layer origin of these cells, it was demonstrated that under specific induction conditions, MSCs were capable of differentiation into the mesodermal cell types typically found in bone, cartilage and adipose tissue. The ability of MSCs to generate cell types of multiple mesodermal lineages has now been wellestablished. Techniques for the *in vitro* differentiation of MSCs towards osteocytes, adipocytes and chondrocytes are widely used and highly reproducible amongst different laboratories.

Osteocytic differentiation is typically achieved through culturing a confluent layer of MSCs under the influence of dexamethasone, β -glycerol phosphate and ascorbate for 1-3 weeks (Pittenger *et al.*, 1999). This results in the formation of aggregates or nodules that stain with Alizarin Red S and Von Kossa techniques for detecting mineral deposition. In addition, increases in alkaline phosphatase activity and calcium accumulation are observed.

Adipogenesis of MSCs can be accomplished by treatment with isobutyl methyl xanthine (IBMX), dexamethasone, insulin and indomethacin (Pittenger *et al.*, 1999). Differentiation towards adipocytes is marked by expression of peroxisome proliferation-activated receptor $\gamma 2$, lipoprotein lipase and fatty acid-binding protein aP2, and accumulation of intracellular lipid-rich vacuoles, which coalesce with continued culture to eventually fill the cell. These lipid-rich vacuoles can be stained through histochemical procedures using Oil Red O.

Differentiation of MSCs into chondrocytes *in vitro* is induced through gentle pelleting of cells into a micromass by centrifugation and culturing with TGF- β in the absence of serum (Johnstone *et al.*, 1998; Mackay *et al.*, 1998; Pittenger *et al.*, 1999). Histological analysis of the cell pellets post-differentiation revealed strong staining with toluidine blue and Safranin O indicating production of a glycosaminoglycan- and proteoglycan-rich extracellular matrix. Additionally, the extensive extracellular matrix was found to be abundant in type II collagen (COL2A1), type X collagen, aggrecan (ACAN), link protein, biglycan, decorin, chondroitin-4-sulfate and keratan sulfate, which are typical of cartilaginous tissues (Johnstone *et al.*, 1998; Kopen *et al.*, 1999;

Mackay *et al.*, 1998; Pittenger *et al.*, 1999; Yoo *et al.*, 1998). Chondrogenic differentiation was further marked by the formation of large cells embedded within lacunae, which morphologically resembled hypertrophic chondrocytes.

Further research is currently being conducted to extend these findings and to uncover the mechanisms involved during differentiation. In particular, differentiation procedures for obtaining specific subtypes of mesodermal cells are being examined for transplantation into specific tissues. These procedures are also being optimised, in terms of isolation and culturing techniques as well as removal of xenogeneic elements, in preparation for potential clinical use. Recent reports from our laboratory have also demonstrated that supplementation with bone morphogenetic protein (BMP)-2 enhances TGF- β 3-mediated *in vitro* chondrocytic differentiation of BM-derived hMSCs (Shen *et al.*, 2009b), whereas BMP-13 was found to inhibit *in vitro* osteogenic differentiation of hMSCs (Shen *et al.*, 2009a).

Since the discovery that MSCs were capable of differentiation into multiple mesodermal lineages, numerous studies have now shown MSCs to also be capable of differentiation into non-mesodermal cell types under certain circumstances. The observation of MSC plasticity or multi-lineage differentiation has generated great interest in the biology of these cells, and will be the focus of the following section.

1.2 Lineage Conversion Potential of BM-Derived MSCs

The traditional model of embryonic development centres upon a strict hierarchy of stem cell subsets that display progressive and orderly differentiation in a unidirectional manner, finally culminating in a terminally differentiated cell type (Anderson *et al.*, 2001; Blau *et al.*, 2001; Eisenberg *et al.*, 2003; Goodell, 2003; Lemischka, 2002a; Morrison, 2001; Orkin *et al.*, 2002; Wagers *et al.*, 2004). Central to this view is the early specification of cells during development into the three embryonic germ layers in the course of gastrulation. The segregation of cells into each germ layer was thought to be an irreversible event, and subsequent tissue regeneration and cell replacement were believed to occur through tissue-resident SCs only capable of generating mature cell types of that tissue. However, these established paradigms have been challenged in the past decade with reports demonstrating the surprising ability of adult SCs to cross lineage boundaries, bringing about a remarkable evolution in the field of SC biology (Anderson *et al.*, 2001; Blau *et al.*, 2001; Eisenberg *et al.*, 2003; Goodell, 2003; Lemischka, 2002a; Morrison, 2001; Orkin *et al.*, 2002; Phinney *et al.*, 2007; Prockop, 2007; Wagers *et al.*, 2004).

In particular, an increasing number of reports, from our laboratory and others, have shown MSCs to be capable of differentiation into a wide range of non-mesodermal cell types both *in vitro* and *in vivo*, including ectodermal neuronal and glial cells (Chen et al., 2001; Hermann et al., 2004; Jiang et al., 2003; Jiang et al., 2002; Kopen et al., 1999; Munoz-Elias et al., 2004; Sanchez-Ramos et al., 2000; Tao et al., 2005; Woodbury et al., 2000; Zhao et al., 2002), as well as endodermal hepatic cells (Jiang et al., 2002; Schwartz et al., 2002; Wang et al., 2004). These findings have challenged the extent of lineage commitment of MSCs, and together with similar findings in other adult SC populations, have suggested the potential of adult SCs to be capable of transdifferentiation, which has consequently given rise to the concept of adult SC plasticity (Wagers et al., 2004). Wagers and Weissman (Wagers et al., 2004) have described transdifferentiation as "the conversion of a cell of one tissue lineage into a cell of an entirely distinct lineage, with concomitant loss of the tissue-specific markers and function of the original cell type, and acquisition of markers and function of the transdifferentiated cell type". Whereas for SC plasticity, Blau and colleagues (Blau et al., 2001) have proposed that the fate of SCs in adult tissues may be amenable to

change, rather than restricted to a specific lineage, depending on the type of microenvironmental cues present.

The observations of MSC plasticity or multi-lineage differentiation, while generating great interest in the potential therapeutic value of these cells, challenges the long-held dogma of irreversible tissue specification during development, and remains a highly controversial area. It is generally accepted that more rigorous and consistent scientific evidence is necessary to verify these claims before the basic tenets of developmental biology are discarded, in line with the ideology that extraordinary claims require extraordinary proof (Anderson *et al.*, 2001; Eisenberg *et al.*, 2003). It has also been proposed that higher standards of evidence should be established for demonstrating adult SC transdifferentiation (Anderson *et al.*, 2001). The following sections will discuss the recent advances in our understanding of MSC plasticity, as well as alternative arguments that have been put forward to explain transdifferentiation events and plasticity-related improvements observed in animal disease models.

1.2.1 Evidence for a Pluridifferentiated Stromal Progenitor Cell

The controversy surrounding the MSC plasticity debate has been further fuelled by recent findings that MSCs not only acquire non-mesodermal marker expression following exposure to differentiation conditions, but also express genes and proteins of heterologous lineages prior to differentiation, leading to the suggestion that MSCs possess a 'multidifferentiated' or 'pluridifferentiated' phenotype (Blondheim *et al.*, 2006; Deng *et al.*, 2006; Deng *et al.*, 2001; Lamoury *et al.*, 2006; Minguell *et al.*, 2005; Phinney *et al.*, 2006; Sanchez-Ramos *et al.*, 2000; Seshi *et al.*, 2003; Suon *et al.*, 2004; Tao *et al.*, 2005; Tondreau *et al.*, 2004b; Tremain *et al.*, 2001; Vogel *et al.*, 2003; Wislet-Gendebien *et al.*, 2003; Woodbury *et al.*, 2002; Woodbury *et al.*, 2000). Whilst, only one study has detected endodermal gene expression in undifferentiated MSCs (Woodbury *et al.*, 2002), numerous reports have demonstrated that rodent and human MSCs express ectodermal markers before differentiation, in particular those of the neural lineage (Blondheim *et al.*, 2006; Deng *et al.*, 2006; Deng *et al.*, 2001; Lamoury *et al.*, 2006; Minguell *et al.*, 2005; Phinney *et al.*, 2006; Sanchez-Ramos *et al.*, 2000; Seshi *et al.*, 2003; Tao *et al.*, 2005; Tondreau *et al.*, 2004b; Tremain *et al.*, 2001; Vogel *et al.*, 2003; Wislet-Gendebien *et al.*, 2003; Woodbury *et al.*, 2002; Woodbury *et al.*, 2000). In addition, rat and murine MSCs have recently been found to express germ-line specific and pluripotency markers, including protamine 2, POU5F1 (POU class 5 homeobox 1; also known as OCT3/4) and zinc finger protein 42 homolog (also known as REX1) (Jiang *et al.*, 2002; Lamoury *et al.*, 2006; Woodbury *et al.*, 2002).

The significance of these findings is three-fold: firstly, it reveals the necessity of ascertaining the basal levels of lineage marker expression by undifferentiated MSCs, and also uncovers possible repercussions for previous reports claiming transdifferentiation without examining expression levels prior to commencement of differentiation; secondly, these findings provide additional support for the propensity of MSCs to undergo neural differentiation, by demonstrating the ease with which neural properties are spontaneously expressed by MSCs; and thirdly, the constitutive expression of both mesodermal and ectodermal phenotypes suggests that MSCs have a 'multidifferentiated' (Woodbury *et al.*, 2002) or 'pluridifferentiated' (Seshi *et al.*, 2003) phenotype.

The detection and level of neural marker expression by MSCs appears to vary from laboratory to laboratory. This confounding variability possibly arises from the use of different isolation and culturing techniques, which may have led to the isolation of different subpopulations of cells, and which is partially due to the lack of a defined marker/s for the identification of MSCs. Previously, our laboratory has shown that

undifferentiated hMSCs are capable of expressing a wide range of neural genes and proteins, including NF-M (neurofilament protein subunit of medium molecular weight), nestin (NES), β tubulin III, microtubule-associated protein (MAP)-2, neuron-specific enolase (NSE), myelin basic protein (MBP) and myelin-associated enzyme 2',3'-Cyclic nucleotide 3'-phosphohydrolase (CNPase) (Tao *et al.*, 2005). These markers were generally expressed at low levels and, with the exception of MBP and CNPase, are typically markers of a primitive neural phenotype or early neural development. However, astrocyte and neuroglial precursor marker glial fibrillary acidic protein (GFAP), axonal marker Tau, and markers of mature neurons, including neurotransmitters tyrosine hydroxylase (TH), serotonin, glutamic acid decarboxylase, and γ -aminobutyric acid (GABA), were unable to be detected (Tao *et al.*, 2005).

In contrast, an early study by Woodbury, *et al.* initially reported lack of NF-M and Tau protein expression, with only low levels of NSE protein in undifferentiated rat MSCs (Woodbury *et al.*, 2000). In a later report, they then showed that rat MSCs also expressed neuroectodermal genes *GFAP*, *NEUROD1* (neurogenic differentiation 1), aldolase C, amyloid precursor protein, N-methyl D-aspartate glutamate binding subunit, and syntaxin 13, but lacked expression of mature neurotransmission markers synaptophysin and choline acetyltransferase(Woodbury *et al.*, 2002). Other early studies detected low levels of NES, neuronal nuclear antigen (NeuN) and GFAP protein expression in hMSCs prior to neural induction (Sanchez-Ramos *et al.*, 2000), or expression of vimentin, NSE, MAP-1B and β tubulin III in the absence of NF-M, MAP-2, Tau, S-100, GFAP and MBP (Deng *et al.*, 2001).

In recent times, a number of studies have further examined this apparent neural predisposition of MSCs (Blondheim *et al.*, 2006; Deng *et al.*, 2006; Minguell *et al.*, 2005; Tondreau *et al.*, 2004b). Tondreau, *et al.* observed constitutive expression of

immature neuronal markers NES and β tubulin III in the majority of hMSCs (greater than 80%; both gene and protein expression detected), with the presence of mature neuronal and glial proteins, MAP-2, TH and GFAP, at lesser degrees after 4-5 passages of culture (Tondreau et al., 2004b). Suon and colleagues also found low level expression of β tubulin III, neurofilament, NSE and GFAP mRNA and protein in hMSCs (Suon et al., 2004). Neural lineage markers, NES, NEUROD1, NeuN and β tubulin III were also detected in certain subpopulations of hMSCs by Minguell, *et al.* (Minguell et al., 2005). Deng and colleagues described spontaneous neural protein expression by murine MSCs, with nearly 100% of murine MSCs positive for intermediate filament protein NES (Deng et al., 2006). Additionally, mouse MSCs were shown to express β tubulin III (12%), NF-M (13.2%) and S100B (calcium binding protein; 15%), however, were negative for the polysialylated-neural cell adhesion molecule, GFAP and vimentin (Deng et al., 2006). Whereas, Blondheim, et al. (Blondheim et al., 2006) reported hMSC expression of 12 neural genes (including neurofilament protein subunit of heavy molecular weight, NF-M, NES, NSE and NeuN), 8 genes related to the neuro-dopaminergic system (including TH, dopamine transporter (DAT), aromatic L-amino acid decarboxylase, engrailed-1 (ENI), NR4A2 (nuclear receptor subfamily 4, group A, member 2; also known as Nurr1) and paired-like homeodomain transcription factor-3 (PITX3)), and 11 transcription factors with neural significance (including neural zinc fingure 3, paired box gene (PAX) 3 and PAX6). In addition, NES, NSE, NeuN, TH and CNPase proteins were detected in hMSCs by Western blot analysis (Blondheim et al., 2006). Overall, the commonly reported neural markers in undifferentiated MSCs include NES, NSE, β tubulin III, NF-M, NeuN and GFAP.

1.2.2 Evidence for Non-Mesodermal Differentiation of MSCs In Vitro

Of the known adult SC populations, BM-derived MSCs are one of the more extensively studied in terms of differentiation potential, as it is an easily accessible and expandable population with demonstrated multi-lineage differentiation potential. However, variability in the differentiation capacity of MSCs exists between laboratories, which may at least be partially accounted for by discrepancies in the isolation and culture expansion procedures of different laboratories. The majority of studies also require expansion of MSCs in culture, however, the expansion methods are often inconsistent and the effect of *in vitro* culture (sometimes long-term) on MSC plasticity has not been well-documented (e.g. possible introduction of epigenetic and genetic modifications, impact of cell density and confluence, cell senescence). Additionally, the conditions used to drive differentiation are not well-defined. Currently, the mechanisms involved in differentiation and development *in vivo* remain largely undetermined, adding to the difficulties of this field. In general, the differentiation methods employed manipulation involve the of culture microenvironments, mainly through addition of cytokines, signalling molecules, conditioned medium and/or chemicals, and alterations to the growth substrate.

The potential of MSCs to generate non-mesodermal cell types, such as neural, hepatic and epithelial cells, has been investigated in view of the possibility of utilising these cells in the treatment of degenerative diseases of these organs. The *in vitro* neurogenic potential of MSCs will be discussed in greater depth in Section 1.3. Generation of endodermal hepatocyte-like cells from MSCs *in vitro* has been achieved through culturing MSCs on Matrigel and supplementing media with fibroblast growth factor (FGF)-4 and HGF (Jiang *et al.*, 2002; Schwartz *et al.*, 2002), serum starvation of MSCs followed by HGF addition (Wang *et al.*, 2004), supplementation with HGF,

FGF-2 and nicotinamide, followed by a second step of oncostatin M (OSM), dexamethasone and insulin transferrin selenium (ITS) mix (Lee et al., 2004), addition of FGF-4 and OSM (Shi et al., 2008; Shi et al., 2005), co-culture with damaged liver tissue (Luk et al., 2005), pellet culture with HGF and FGF-2 supplementation followed by OSM (Ong et al., 2006), sequential addition of FGF-4, HGF, and HGF + ITS + dexamethasone (Snykers et al., 2006), hepatocyte-conditioned medium with type I collagen (COL1A1) culture surface coating (Chen et al., 2007c), and epigenetic modification through valproic acid (VPA) treatment followed by sequential addition of FGF-4, then HGF, and HGF + OSM + ITS + dexamethasone (Chen *et al.*, 2008). Cells obtained through these procedures were demonstrated to display epithelioid morphology, with some cells becoming binucleated, and expressed hepatocyte markers including hepatocyte nuclear factor (HNF)- 3β , transcription factor GATA-4, cytokeratin (CK) 19, transthyretin, α -fetoprotein (α -FP), CK18, HNF-4, HNF-1 α , albumin (ALB), glucose-6-phosphatase, and tyrosine-aminotransferase (Chen et al., 2007c; Chen et al., 2008; Jiang et al., 2002; Lee et al., 2004; Luk et al., 2005; Ong et al., 2006; Schwartz et al., 2002; Shi et al., 2008; Shi et al., 2005; Snykers et al., 2006; Wang et al., 2004). Functional characteristics of hepatocytes were also acquired by MSCs postdifferentiation. Schwartz and colleagues (Schwartz et al., 2002) showed hepatocyte-like MSCs were capable of secreting urea and ALB, taking up low-density lipoprotein (LDL), expressing functional hepatocyte-specific cytochrome p450 and storing glycogen. Lee, et al. (Lee et al., 2004) also demonstrated the properties of LDL uptake, cytochrome p450 enzyme presence, glycogen storage, urea production, as well as the expression of bile canaliculi-specific antigen in differentiated MSCs. Whilst other studies showed production of ALB and urea secretion (Shi et al., 2008), together with

glycogen accumulation (Chen *et al.*, 2007c; Ong *et al.*, 2006; Shi *et al.*, 2005), and inducible cytochrome p450-dependent activity (Snykers *et al.*, 2006).

MSCs have also been reported to differentiate in vitro into epithelial-like cells following co-culture with heat-shocked small airway epithelial cells (AECs) or in an airliquid interface co-culture system with AECs (Spees et al., 2003; Wang et al., 2005). The co-cultured MSCs gained epithelium-like characteristics, including a broad, flattened cytoplasm with an elevated perinuclear region, or an epithelial-like cuboidal or columnar morphology, expression of epithelial-specific markers such as keratin 17, 18 and 19, as well as CD24 (epithelial mucin-like glycoprotein), CC26 (marker of clara, serous and goblet cells in the lung) and occludin (tight junction protein), and a gene expression profile that closely resembled that of small AECs (Spees et al., 2003; Wang et al., 2005). Furthermore, the morphologically differentiated MSCs were found to form adherens junctions with neighbouring small AECs (Spees et al., 2003). Surprisingly, while direct differentiation of some MSCs into epithelial-like cells was observed after incorporation of MSCs into the epithelial monolayer, Spees, et al. also detected fusion of MSCs with adjacent epithelial cells (Spees et al., 2003). The frequency of this cell fusion was higher than previously reported, and in some cases nuclear fusion also occurred.

1.2.3 In Vivo Transplantation Studies Supporting MSC Plasticity

The prospect of using MSCs in autologous transplantation therapies provides the advantages of overcoming the ethical issues and constraints associated with the acquisition of other types of donor tissue and ESCs, and circumvents the need for antirejection drugs and the possibility of transplant rejection. Additionally, transplantation of MSCs is not associated with teratoma formation, unlike ESC transplantation. These factors, and the abilities to expand MSCs relatively easily in culture, as well as to differentiate MSCs into many cell types *in vitro*, clearly make these cells attractive for potential cell-based patient-specific therapies. The fate of MSCs after transplantation *in vivo* has been investigated widely, with much interest garnered through reports of MSC-mediated disease improvement in experimental animal models. However, despite recent progress in our understanding, much remains to be determined concerning the *in vivo* behaviour and function of MSCs.

1.2.3.1 MSC Transplantation at Early Stages of Development

Transplantation studies have found MSCs to be capable of *in vivo* differentiation across germinal boundaries, in a manner that appears dependent on cues provided by the microenvironment of different organs. This cross-lineage potential was observed after transplantation of hMSCs intraperitoneally in foetal sheep early in gestation (Liechty *et al.*, 2000). Engraftment was observed in multiple tissues, where hMSCs underwent sitespecific differentiation to form chondrocytes, adipocytes, myocytes, cardiomyocytes, BM stroma and thymic stroma. Furthermore, hMSCs persisted in tissue for 13 months post-transplantation, and long-term engraftment was observed even with transplantation after the development of immunocompetence. These results suggest that hMSCs are capable of migration, integration and survival in xenogeneic host tissue, and may possess unique immunologic characteristics.

Transplantation of a subset of MSCs, known as multipotent adult progenitor cells (MAPCs), into early blastocysts has resulted in the generation of chimaeric mice (Jiang *et al.*, 2002). Murine MAPCs were observed to contribute to many somatic tissues, including brain, retina, lung, myocardium, skeletal muscle, liver, intestine, kidney, spleen, BM, blood and skin. Furthermore, injection of a single MAPC could generate balanced chimaeras, and chimaerism was not only detected in mouse embryos, but also in 6-20 week old mice. This study additionally showed evidence of robust,

early and persistent engraftment of MAPCs after i.v. infusion into non-irradiated and minimally irradiated post-natal non-obese diabetic/severe combined immunodeficient mice. MAPC engraftment and differentiation were seen in haematopoietic tissues (blood, BM and spleen) and lung, liver, and intestinal epithelium of all recipient animals.

Plasticity of MSCs *in vivo* has also been demonstrated in a chick embryo model, which showed engraftment and partial differentiation of MSCs in the absence of cell fusion (Pochampally *et al.*, 2004a). Four days after infusion of rat MSCs into 1.5-2 day old chick embryos, cells were found to migrate into multiple host tissues, and histological analysis revealed the presence of MSCs in developing organs such as the heart, liver, brain, and spinal cord. Additionally, the number of MSCs increased 1.5-33 fold in one-third of surviving embryos, indicating engraftment and expansion of donor cells. MSCs were most commonly detected in the heart, perhaps due to infusion in the area just above the dorsal aorta, and some cells expressed markers of cardiomyocytes, including cardiotin and α -heavy-chain myosin, suggesting partial differentiation of MSCs.

1.2.3.2 MSC Transplantation into Neural Tissues

The ability of MSCs to contribute to particular organ systems has also been the focus of intense investigation. Studies have examined the potential of MSCs to integrate and differentiate after transplantation into normal tissue, as well as developing and injured post-natal tissue. Numerous reports have now described the remarkable efficacy of MSCs in ameliorating damage caused by injury and disease in a wide range of organs and tissues. The mechanisms underlying the observed improvements are yet to be clearly defined; however, transdifferentiation and replacement of damaged cells,

secretion of trophic factors, interaction with endogenous host cells, alteration of immune responses, or a combination of these, have been proposed to be involved.

The majority of initial reports demonstrating neural engraftment and restoration by MSCs examined direct intracerebral transplantation of cells into neural regions, including the corpus striatum of normal albino rats (Azizi et al., 1998), the lateral ventricles of neonatal mice (Kopen et al., 1999), the ventricles of embryonic rats in utero (Munoz-Elias et al., 2004), and the ischemic boundary zones in rat models of cerebral ischemia (Chen et al., 2001; Zhao et al., 2002). Injection of MSCs into uninjured animal models resulted in engraftment, migration along known neural SC (NSC) migratory pathways, incorporation into successive layers of the brain, and phenotypic expression of markers in a manner suggesting the involvement of regionspecific signals within the different neural microenvironments to produce characteristics of radial glia, subventricular zone progenitors, migratory cells, parenchymal neurons, and glia (Azizi et al., 1998; Kopen et al., 1999; Munoz-Elias et al., 2004). In addition, Azizi, et al. (Azizi et al., 1998) found an approximate MSC engraftment rate of 20%, loss of COL1A1 and fibronectin production by MSCs post-transplantation, and lack of an inflammatory response or rejection of grafted cells. MSCs also exhibited long-term survival at 1 and 2 postnatal months when infused in rat ventricles *in utero* at embryonic day 15.5, and were detected in distant locations throughout the brain, including the olfactory bulbs, rostral migratory stream, frontal, parietal and occipital cortices, hippocampus, dentate gyrus, and the telencephalic ventricular and subventricular zones (Munoz-Elias et al., 2004). In experimental models of stroke, intracerebral transplantation of MSCs was found to result in significant recovery of motor and somatosensory deficits (Chen et al., 2001; Li et al., 2000; Zhao et al., 2002). In these studies MSCs were observed to survive and migrate along white matter tracts, as well as

differentiate to express neural phenotypic proteins, including astrocytic marker GFAP, oligodendrocytic marker GALC (galactosylceramidase), and neuronal markers NeuN, MAP-2, β tubulin III, NSE, and neurofilament proteins, when grafted into the ischemic brain microenvironment. However, while behavioural improvement was detected, the grafted cells did not re-establish a normal tissue cytoarchitecture, with Zhao and colleagues noting that grafted cells displayed a spherical morphology with few processes. From these observations the authors suggested that cell replacement and integration into host circuitry may not be responsible for functional recovery, and this could instead be mediated by trophic factor secretion by MSCs, or by host cells in response to interaction with MSCs (Chen et al., 2001; Zhao et al., 2002). Li, et al. (Li et al., 2001a) also demonstrated functional improvement following administration of MSCs via an intracarotid arterial route in a rat middle cerebral artery occlusion model of cerebral ischemia. MSCs were observed throughout the ischemic region in middle cerebral artery territory and in multiple areas of the ipsilateral hemisphere, however, more than 90% of MSCs were localised in the ischemic core and boundary zone. In addition, approximately 10% of MSCs in the ipsilateral hemisphere were observed to express GFAP, and 1% expressed MAP-2, suggesting possible differentiation in vivo towards astrocytes and neurons respectively. Furthermore, the results of this study suggested that intra-arterial administration of MSCs was superior to using an intracerebral route, and transplantation of MSCs did not appear to evoke an inflammatory response. However, the mechanisms underlying the observed neurological recovery remained undefined, although the authors commented that it was unlikely to be due to the apparent adoption of a neural phenotype by MSCs, and could be related to trophic factors produced by MSCs or interaction of MSCs with the host brain (Li et al., 2001a). Intracerebral and i.v. transplantation of MSCs into rodent models of traumatic

brain injury have also yielded successful results with observations of neurological improvement in the recipient animals (Lu *et al.*, 2001a; Mahmood *et al.*, 2003; Mahmood *et al.*, 2002; Mahmood *et al.*, 2001). Promising outcomes have also been obtained in animal models of Parkinson's disease (PD) following transplantation of MSCs, and these reports will be discussed in greater detail in Section 1.4.4. Collectively these studies highlight MSCs as a promising therapeutic modality for degeneration, ischemia and injury of the nervous system, and are indicative of the ability of MSCs to respond to specific signals in different microenvironments *in vivo*.

1.2.3.3 MSC Transplantation into Hepatic Tissues

The possible contribution of BM cells to liver cell types was initially observed in studies investigating cross-gender BM or liver transplantation in rodent models with or without hepatic injury (Petersen et al., 1999; Theise et al., 2000a), as well as a retrospective study examining human archival autopsy and biopsy liver specimens from recipients of therapeutic BM transplantations or orthotopic liver transplantations (Theise et al., 2000b). However, transdifferentiation of MSCs toward hepatocyte-like cells in vivo remains controversial. Several studies have demonstrated engraftment of MSCs, induction of a hepatocyte-like phenotype and partial restoration of function after transplantation in the uninjured and injured liver of rodents (Aurich et al., 2007; Chen et al., 2007c; Chen et al., 2008; Luk et al., 2005; Oyagi et al., 2006; Sato et al., 2005). Whilst Fang and colleagues (Fang et al., 2004) reported functional improvement in rodent liver fibrosis models, but could only show a low frequency of donor cell engraftment and transdifferentiation, with alleviation of injury attributed to endogenous hepatic regeneration elicited through interaction between MSCs and host cells. In contrast, the results of other studies suggested that MSCs recruited to chronically injured liver may instead display pro-fibrogenic myofibroblast-like properties and contribute to liver fibrosis, with limited hepatocellular differentiation (di Bonzo *et al.*, 2008; Russo *et al.*, 2006). Nonetheless, marked differences exist between these studies, including the transplantation of uninduced MSCs or differentiated hepatocyte-like MSCs, the use of different routes or time points of administration, as well as different harvest time points, with some of these factors being touted as crucial to the success of the study in question.

1.2.3.4 MSC Transplantation into Epithelial Tissues

Reports have also documented the differentiation of MSCs in vivo toward a variety of epithelial cell types, including those of the lung (Kotton et al., 2001; Ortiz et al., 2003; Rojas et al., 2005), retina (Arnhold et al., 2007; Arnhold et al., 2006; Kicic et al., 2003), skin (Fu et al., 2006a; Nakagawa et al., 2005), and kidney (Herrera et al., 2004; Morigi et al., 2004). In the bleomycin-induced rodent pulmonary fibrosis model, transplanted MSCs were found to engraft in recipient lung parenchyma, localise to areas of injury, and differentiate into cells resembling type I pneumocytes (Kotton et al., 2001) and type II pneumocytes (Ortiz et al., 2003), as well as other lung cell types such as fibroblasts and endothelial cells (Rojas et al., 2005). The rescue effects of MSCs were also detected in rodent models of retinal degeneration, with donor cells expressing markers of photoreceptor cells, adopting morphology similar to retinal pigment epithelial cells, integrating into the host retinal pigment epithelial cell layer and neuroretina layers, and establishing tight junction contacts with adjacent host cells (Arnhold et al., 2007; Arnhold et al., 2006; Kicic et al., 2003). Animal grafting experiments with MSCs also revealed enhanced wound healing properties and possible generation of *de novo* intact skin, with transdifferentiation into epithelium (Nakagawa et al., 2005), and formation of epidermal cells and sebaceous duct cells (Fu et al., 2006a). Contribution of MSCs to renal repair was observed in murine models of acute renal

failure, in which i.v. transplantation of MSCs resulted in engraftment in the damaged kidney, differentiation into tubular epithelial cells, and restoration of morphological and functional alterations (Herrera *et al.*, 2004; Morigi *et al.*, 2004). Nonetheless, the majority of these studies recognised that the observed protective effects of MSCs may involve mechanisms other than transdifferentiation and replacement of lost cells, such as roles of MSCs in the suppression of inflammation, reduction of collagen deposition and production of reparative growth factors. Additionally, most were unable to rule out the possibility of cell fusion. Further controversies surround the reports describing MSC epithelial differentiation, with several studies arguing that MSCs show little evidence of differentiation into cells of the tubular epithelium after ischemic renal injury (Duffield *et al.*, 2005), or into corneal epithelial cells when grafted to chemically-damaged corneal surfaces (Ma *et al.*, 2006), or into photoreceptor cells in a rat model of retinal degeneration (Inoue *et al.*, 2007).

Taken together, these studies suggest that MSCs may have extraordinary potential for contributing to and restoring the structure and function of many different organ systems. However, the contradictory findings across the different studies, and the fact that the underlying mechanisms still remain largely undetermined, all point towards the need to consider alternative explanations to the observations of MSC plasticity and transdifferentiation. These possible alternatives and potential caveats will be examined in the following section.

1.2.4 Alternative Explanations for MSC Plasticity

Together with the numerous reports published demonstrating evidence for plasticity and transdifferentiation of MSCs, also come equally numerous studies containing contradictory findings that question these concepts. Several alternative explanations and mechanisms have been proposed to account for MSC plasticity, including the presence of multiple distinct SC or progenitor cells within the heterogeneous BM compartment, cell-cell fusion leading to the transfer of markers between unrelated cell types, acquisition of broader developmental potential through *in vitro* culturing techniques, difficulties in tracking transplanted SCs and the progeny of these cells, as well as difficulties in ascertaining cellular identity through examination of only a few or several markers (Anderson *et al.*, 2001; Goodell, 2003; Morrison, 2001; Orkin *et al.*, 2002; Wagers *et al.*, 2004; Wulf *et al.*, 2001). Furthermore, observations of functional improvement in animal models of disease and injury credited to MSC transdifferentiation and cellular replacement, could instead be due to other mechanisms, such as the production of soluble factors, modulation of immune responses, transfer of mitochondria, and interaction with, or stimulation of endogenous host responses (Phinney *et al.*, 2007; Prockop, 2007).

A significant caveat of all studies involving MSCs, whether *in vitro* or *in vivo*, is the heterogeneous nature of MSC populations. Furthermore, the BM origin of MSCs is known to possess pre-existing heterogeneity. The complexity of these cellular populations allows for alternative interpretations of the proposed transdifferentiation events, since multiple SC types could be present and therefore account for the generation of multiple cellular lineages (Orkin *et al.*, 2002). This has been shown in studies investigating the production of haematopoietic cells from muscle, which demonstrated that itinerant HSCs residing within muscle were responsible for all muscle-associated haematopoietic activity, negating earlier studies supporting the plasticity of muscle-derived cells (Issarachai *et al.*, 2002; Kawada *et al.*, 2001; McKinney-Freeman *et al.*, 2002). A recent study has also reported the presence of neural crest-derived SCs within a variety of adult mouse tissues, including the BM (Nagoshi *et al.*, 2008). These findings suggest that the observed neural differentiation potential of BM cells may actually reflect the potential of the neural crest-derived SCs that reside within BM. Further to this, Pierret and colleagues have put forward the hypothesis that adult SCs are all derived from the neural crest, which would then explain the reported plasticity seen in adult SC subsets (Pierret *et al.*, 2006). Phinney and Prockop (Phinney *et al.*, 2007) have also commented that the heterogeneity of MSCs and the complexity of the BM organ system may contribute to the therapeutic efficacy associated with these cell populations, but on the contrary, makes assessment of the transdifferentiation potential more difficult. Additionally, these authors proposed that evolutionary mechanisms have conferred plasticity to adult mesenchyme, in view of the fact that during development there are multiple transitions from epithelium to mesenchyme, and vice versa (Phinney *et al.*, 2007).

Reports of somatic SC plasticity often describe these events as infrequent or rare, leading to the possibility that alternative low-probability or aberrant mechanisms are responsible (Anderson *et al.*, 2001; Lemischka, 2002a), one of which is cell fusion (Alvarez-Dolado *et al.*, 2003; Terada *et al.*, 2002; Vassilopoulos *et al.*, 2003; Wang *et al.*, 2003; Weimann *et al.*, 2003; Ying *et al.*, 2002). Moreover, certain cell types, including myocytes and hepatocytes, have a natural propensity for cell fusion. Initially, cell fusion events were demonstrated in co-culture experiments with ESCs and BM cells or central nervous system (CNS) cells (Terada *et al.*, 2002; Ying *et al.*, 2002). The BM and CNS cells were found to undergo spontaneous fusion with the ESCs, as well as adoption of an ESC phenotype. Shortly after these studies, additional findings were published documenting fusion of MSCs or BM cells with hepatocytes, cardiomyocytes and Purkinje neurons post-transplantation to form multinucleated cells (Alvarez-Dolado *et al.*, 2003; Vassilopoulos *et al.*, 2003; Wang *et al.*, 2003; Weimann *et al.*, 2003). Furthermore Alvarez-Dolado, *et al.* (Alvarez-Dolado *et al.*, 2003) found no evidence of

MSC transdifferentiation or contribution to non-haematopoietic cell types without cell fusion *in vivo*. Importantly cell fusion does not appear to explain all instances of transdifferentiation. Transplanted BM-derived cells have been found to contribute *in vivo* to buccal epithelial cells of the cheek, insulin-producing pancreatic endocrine cells, and cardiomyocytes, in the absence of cell fusion (Ianus *et al.*, 2003; Pochampally *et al.*, 2004a; Tran *et al.*, 2003). Nonetheless a study investigating *in vitro* co-culture of MSCs with heat-shocked lung epithelial cells did observe the presence of fusion events; however, these phenomena could only account for a minority portion of the total transdifferentiation events (Spees *et al.*, 2003).

Technique-related factors have also been proposed to account for the discrepancies between studies examining SC plasticity and transdifferentiation. One such factor is the impact of *in vitro* culturing on developmental potential (Anderson *et al.*, 2001; Lemischka, 2002b; Morrison, 2001; Wagers *et al.*, 2004). The concern here arises from the possibility of cells accruing changes during extensive *in vitro* culture (such as reported for extensively cultured NSCs (Morshead *et al.*, 2002)) or after exposure to mitogens in culture, raising the question of whether the plasticity exhibited is a normal property possessed *in vivo*, or if cells are endowed with broader potential by the culture conditions employed. If this is indeed the case, then a greater concern emerges regarding the possibility that the manipulated cells could exhibit unpredictable behaviour if used therapeutically (Anderson *et al.*, 2001).

Another potential source of experimental variation lies in the difficulties with detecting donor cells post-transplantation and identifying the differentiation status of these cells (Goodell, 2003; Wulf *et al.*, 2001). Commonly used techniques for labelling or identifying donor cells are all subject to artifact and misinterpretation, for example, the high technical difficulty of *in situ* hybridisation techniques used for detecting the Y-

chromosome of male donor cells in female recipients, and limitation in the *lacZ* transgene reporter system due to the presence of endogenous β -galactosidase activity in lysosomes of mammalian cells, such as macrophages (Goodell, 2003; Wulf *et al.*, 2001). Labelling with green fluorescent protein (GFP) can also yield false positive results as a consequence of endogenous autofluorescence, which can be visualised at the same wavelengths used for detecting GFP fluorescence (Goodell, 2003; Wulf *et al.*, 2001). Furthermore, tissue resident macrophages of the host animal may phagocytose protein products from donor cells that could be detectable histochemically, again leading to erroneous results (Lemischka, 2002a). On the contrary, false negative results can also occur due to loss or reduction of transgene fluorescence. In addition, demonstration of donor cell differentiation is typically achieved through examining markers indicative of the tissue type, however, in most situations this does not allow definitive identification since most commonly used markers are expressed in a variety of tissues (Goodell, 2003; Wulf *et al.*, 2001).

Mechanisms for MSC-related *in vivo* functional improvement, other than transdifferentiation, have already been alluded to in Section 1.2.3. A number of studies have suggested that the alleviation of injury and disease in animal models following MSC transplantation could be attributed to the secretion of trophic factors, which could enhance repair by stimulating the regeneration of damaged cells or endogenous tissue SCs or progenitor cells (Chen *et al.*, 2001; Fang *et al.*, 2004; Inoue *et al.*, 2007; Li *et al.*, 2001a; Lu *et al.*, 2001a; Mahmood *et al.*, 2003; Mahmood *et al.*, 2002; Mahmood *et al.*, 2001; Munoz *et al.*, 2005; Phinney *et al.*, 2007; Prockop, 2007; Zhao *et al.*, 2002). MSCs have also been shown to promote functional recovery in a rat spinal cord injury model, by forming bundles that bridged the epicentre of the injury and that guided regeneration of host axons through the spinal cord lesion (Hofstetter *et al.*, 2002). These lesions were typically filled with debris and macrophages, therefore, the MSCs reestablished a degree of cellular organisation within the injury zone, and may have promoted nerve-fibre outgrowth by providing a physical growth-permissive surface (Hofstetter et al., 2002). In ischemic renal injury, resident epithelial cells were found to restore kidney epithelial integrity rather than the transplanted MSCs, however, MSCs may have played an indirect role by contributing to the repopulation of the injured vascular endothelium (Duffield et al., 2005). Modulation of immune responses by MSCs, discussed in Section 1.1.5, may also play a role in functional improvement in vivo. Support for this is seen in a recent study by Ohtaki and colleagues, which found improvement in neurological function and decreased neuronal death after transplantation of MSCs in a mouse model of transient forebrain ischemia (Ohtaki et al., 2008). The beneficial effects observed were attributed to the ability of MSCs to modulate inflammatory and immune responses through alternative activation of microglia and/or macrophages and subsequent establishment of a T_H2 cell immune bias, together with the up-regulation of factors that decrease apoptosis (Ohtaki et al., 2008). Another possible mode of tissue repair by MSCs could involve the transfer of intact mitochondria or mitochondrial DNA, as MSCs have been reported to rescue aerobic respiration in mammalian cells lacking functional mitochondria through this action, without evidence of cell fusion (Spees et al., 2006).

Given the wide variance in the prevalence of transdifferentiation, the rarity of such occurrences, and the continued controversy surrounding this topic, a call has been made for a higher standard of evidence to verify claims of plasticity (Anderson *et al.*, 2001; Blau *et al.*, 2001; Morrison, 2001; Wagers *et al.*, 2004). Stringent criteria have been proposed for the demonstration of a bona fide somatic SC transdifferentiation event, including the necessity for prospective isolation of SC populations and clonal

analysis, minimal manipulation for assessment of intrinisic developmental potential, evidence of morphologically indistinguishable integration into the host tissue structure, demonstration of function, exclusion of cell fusion, production of robust and sustained regeneration of target tissues, detailed examination of co-localised protein signals by laser scanning confocal or deconvolution microscopy, and independent replication of results in different laboratories and in different experimental models (Anderson *et al.*, 2001; Blau *et al.*, 2001; Morrison, 2001; Wagers *et al.*, 2004). However, there is recognition of the validity of at least some of the plasticity findings, although the frequency at which transdifferentiation occurs may only be exceedingly rare (Eisenberg *et al.*, 2003; Goodell, 2003). And, there is general agreement that given the potential importance of this phenomenon, efforts must be undertaken to further understand these properties of SCs.

1.3 Neural Differentiation of BM-Derived MSCs In Vitro

Upon the realisation of the MSC transdifferentiation phenomenon, an immense effort across the field was channeled into determining whether differentiation toward specific cell types could be achieved and controlled in culture. Additionally, endeavours were made to understand the mechanisms that govern such cell fate decisions. An aim of many of these studies was the derivation of neural cells from MSCs. A wide variety of methods for neural differentiation of MSCs have been investigated, with some studies yielding promising results, including the induction of neural gene and protein expression and, in some cases, action potential generation. However, like many areas in the field of SC research, much remains to be determined.

The generation of neuroectodermal-like derivatives from MSCs *in vitro* is typically achieved through manipulation of the culture microenvironment. In general, at the commencement of neural differentiation, the MSC expansion medium is altered or substituted with culture media, supplements and coating substrates characteristically used in neural cell culture. Depending on whether a specific neural cell type is desired, a range of different supplements and substrates have been employed. Surface coating substrates that have been utilised in MSC neural differentiation include fibronectin (Jiang *et al.*, 2003; Jiang *et al.*, 2002; Kim *et al.*, 2002a; Tao *et al.*, 2005), poly-D-lysine (Jin *et al.*, 2003; Kondo *et al.*, 2005), poly-L-lysine (Guo *et al.*, 2005; Hermann *et al.*, 2004; Long *et al.*, 2005), poly-L-lysine/laminin (Joannides *et al.*, 2003), poly-ornithine (Wislet-Gendebien *et al.*, 2003) and poly-L-ornithine/laminin (Suzuki *et al.*, 2004).

The main approaches taken for MSC neural induction primarily involve the use of chemicals, growth factors and/or signalling molecules, conditioned media, or coculture systems. On a lesser scale, some studies have employed a genetic engineering approach, whilst others have combined various aspects of the different approaches. The different methods that have been employed are further discussed in the following sections.

1.3.1 Induction of Neuronal Differentiation of MSCs

1.3.1.1 Chemical Induction

Rapid *in vitro* transformation of human and rat MSCs into neuronal-like cells using a simple chemical treatment protocol was first reported by Woodbury and colleagues (Woodbury *et al.*, 2000). This study proposed that treatment with preinduction media, containing Dulbecco's Modified Eagle's Medium (DMEM), 20% FBS, 1 mM β -mercaptoethanol (BME) and FGF-2 for 24 hours, followed by neuronal induction media composed of DMEM, 2% dimethylsulfoxide (DMSO), and 200 μ M butylated hydroxyanisole (BHA) produced cells exhibiting neuronal morphology and protein expression (NSE, NF-M and Tau) in a very short period (changes observed as early as 60 min to 3 hours) (Woodbury *et al.*, 2000). This procedure was also used by a separate group in a neurosphere formation step (Suzuki *et al.*, 2004), and was later optimised further through the addition of 10 μ M forskolin, N-2 supplement, 2 mM VPA, 5 nM K252A, and 10 mM KCl (Munoz-Elias *et al.*, 2003). The method incorporating these additional factors resulted in greater than 70% morphologic conversion and expression of Tau, NeuN, NSE, and TUC-4 within 24 hours in the absence of mitosis (Munoz-Elias *et al.*, 2003).

Induction of early neural progenitors from hMSCs has also been achieved through the use of conditions that increase intracellular cyclic adenosine monophosphate (cAMP; 0.5 mM IBMX / 1 mM dibutyryl cAMP (dbcAMP) in DMEM/20% FBS) (Deng *et al.*, 2001). Treatment with IBMX/dbcAMP for 6 days resulted in conversion of 20% of MSCs to neuronal-like morphology and increased expression of NSE and vimentin protein (Deng *et al.*, 2001). A study comparing the Woodbury and Deng methods found a higher percentage of neuron-like cells generated from the Woodbury protocol, however, this method also yielded a much higher degree of cell death at 53% of total cells, compared to the less than 5% cell death seen in the Deng protocol (Rismanchi *et al.*, 2003). Alternative techniques that have been reported for hMSC neural differentiation include the application of 10^{-3} M BME +/- 5 x 10^{-7} M all-*trans*-retinoic acid (RA) to attain NES, NSE, NeuN and β tubulin III protein expression within 5 hours (Hung *et al.*, 2002), and treatment with 3 μ M 5-bromo-2-deoxyuridine (BrdU) for 3 weeks to increase hMSC multi-lineage differentiation potential (Qu *et al.*, 2004).

Recent findings have raised doubts over the phenomenon observed during chemical induction and question whether these changes occur through true differentiation processes or through alternative cellular events. The factors underlying these concerns include the seemingly unrealistic time frame of the observed changes,
which occurred rapidly over several hours, as well as the lack of neurite growth cone formation with development of neuronal morphology, and the high degree of cell death accompanying the induction process. Recent independent studies using chemical neuronal induction techniques, in particular the DMSO/BHA and IBMX/dbcAMP methods, were unable to reproduce the results of the original studies (Bertani *et al.*, 2005; Choi *et al.*, 2006; Deng *et al.*, 2006; Lu *et al.*, 2004; Neuhuber *et al.*, 2004; Suon *et al.*, 2004; Tao *et al.*, 2005). Furthermore, the morphological changes and increases in certain neuronal markers elicited by chemical exposure were demonstrated to be due to cellular toxicity, cell shrinkage and disruption of the actin cytoskeleton in response to environmental stress, rather than complex regulated cellular differentiation processes (Bertani *et al.*, 2005; Lu *et al.*, 2004; Neuhuber *et al.*, 2004; Suon *et al.*, 2004). Observations of similar cellular responses by a range of cell types, including primary fibroblasts, HEK293 cells and NIH3T3 cells (Bertani *et al.*, 2005; Deng *et al.*, 2006; Lu *et al.*, 2004; Neuhuber *et al.*, 2005; Choi *et al.*, 2006), further substantiated these results.

1.3.1.2 Cytokine and Signalling Molecule Induction

The use of different combinations of cytokines and signalling molecules, such as RA, have been utilised for MSC neural differentiation in attempts to mimic the physiological environment thought to drive *in vivo* neural development. Additionally, clues garnered from the culture requirements of NSCs have also been implemented. Sanchez-Ramos and colleagues (Sanchez-Ramos *et al.*, 2000) conducted the first study examining cytokine and signalling molecule induction of MSCs toward neural-like cells. This work demonstrated that human and mouse MSCs cultured with 10 ng/mL epidermal growth factor (EGF) or 0.5 μ M RA and 10 ng/mL brain-derived neurotrophic factor (BDNF) could be induced toward a neuronal phenotype (Sanchez-Ramos *et al.*, 2000)

2000). These factors caused large flat fibronectin-positive MSCs to transform into smaller ovoid or spindle-shaped cells, which expressed neural markers and decreased expression of fibronectin (Sanchez-Ramos *et al.*, 2000). Shortly afterwards, Kim, *et al.* (Kim *et al.*, 2002a) showed that combined stimulation of hMSCs with FGF-2, RA and fibronectin substrate yielded a significant increase in expression of neuronal lineage marker NF-M from <1.0% to 40%. This combination proved to be the most effective in generating NF-M expression, over the other growth factors (RA alone, nerve growth factor (NGF) and EGF) and culture subtrates (laminin, gelatin, collagen and polyornithine) examined. However, the neurofilament protein-expressing cells obtained were considered to be immature neuronal lineage progenitors, since action potential-like responses could not be observed in whole cell patch-clamp studies (Kim *et al.*, 2002a).

Experiments conducted in our laboratory with hMSCs have utilised FGF-2, EGF, platelet-derived growth factor (PDGF) and fibronectin surface coating to produce a stable neuronal-like phenotype that is maintained for up to 3 months *in vitro* (Tao *et al.*, 2005). Changes in cell morphology were observed at 2 weeks post-cytokine induction, with the majority (80-95%) of cells displaying retraction of the cytoplasm towards the nucleus, formation of refractile cell bodies and development of long branching processes (Tao *et al.*, 2005). Additionally, increased expression of neuronal markers, including NF-M, NSE, β tubulin III, MAP-2 and Tau, were detected postinduction, and neuronal-like cells stained positively for neurotransmitters or associated proteins, such as GABA, TH and serotonin, suggesting generation of GABAergic, dopaminergic and serotonergic neuron-like cells *in vitro* (Tao *et al.*, 2005). Significantly, this work also showed in parallel experiments using sister cultures of MSCs that growth factors were capable of inducing a neuronal-like phenotype in MSCs, while chemical agents elicited changes that were not associated with physiological development and also caused cell death after 48 hours (Tao *et al.*, 2005).

A different combination that has been investigated in mouse MSCs by Jin and colleagues, uses EGF, FGF-2, RA and NGF, for production of cells with neuronal-like processes and neuronal marker expression (NeuN, MAP-2, Tau, and synaptic functionassociated proteins, including synaptophysin, GABA, $\alpha 1A$ and $\alpha 1B$ calcium channel subunits and NR2A glutamate receptor subunits), albeit in the absence of appropriate intracellular localisation (Jin et al., 2003). Whereas, the MAPC subset of the MSC population of rodents have been stimulated sequentially with FGF-2, followed by FGF-8, and finally BDNF, resulting in a more mature neural phenotype than stimulation with FGF-2 alone (Jiang et al., 2002). MSCs stained positively with markers for neurons (neurofilament-200; 68%), astrocytes (GFAP; 15%) and oligodendrocytes (GALC; 12%) when cultured with FGF-2 alone, but with the sequential cytokine induction approach, mature neuronal markers for dopamine (DA)-containing neurons (30% dopa decarboxylase and TH positive), serotonin-containing neurons (20% serotonin positive) and GABA-containing neurons (50% GABA positive), as well as MAP-2 and Tau could be detected (Jiang *et al.*, 2002). In a later study, sonic hedgehog (SHH) was added to the second step (FGF-8) of the neuronal differentiation process, and murine MAPC-derived neuron-like cells were demonstrated to be capable of acquiring functional voltage-gated sodium channels; however, this could only be achieved after co-culture with astrocytes (Jiang et al., 2003).

Interestingly, several studies have also reported the formation of neurospherelike aggregates from rat and human MSCs, that are morphologically similar to those produced by NSCs (Hermann *et al.*, 2004; Suon *et al.*, 2006; Suzuki *et al.*, 2004; Wislet-Gendebien *et al.*, 2003). Suzuki, *et al.* induced neurosphere formation in rat MSCs using FGF-2 alone, and neurosphere differentiation by plating spheres onto poly-L-ornithine/laminin coated surfaces with media containing FGF-2 for the first 1-2 days, followed by BDNF for up to 7 days (Suzuki et al., 2004). Whereas, Hermann and colleagues observed neurosphere-forming capacity with the addition of FGF-2 and EGF to hMSC cultures, and also induced terminal neural differentiation by plating onto poly-L-lysine-coated surfaces and treating with RA and BDNF for neuronal differentiation, or RA and PDGF for glial differentiation (Hermann et al., 2004). Furthermore after neuronal differentiation neurosphere-derived cells were found to be capable of DA production and potassium-dependent release. However, neither mature MAP-2abpositive neurons nor TH-positive/DA producing cells were obtained through this protocol, and cells continued to express pluripotency marker POU5F1 (Hermann et al., 2004). Sphere formation by hMSCs was initiated in a similar manner by Suon, et al., through suspension culture and supplementation with EGF and FGF-2 (Suon et al., 2006). However, for further differentiation neurosphere-like aggregates were plated onto poly-ornithine/laminin-coated surfaces in serum-free medium containing IBMX, forskolin and TPA (4 β -12-O-tetradecanoylphorbol 13-acetate) for 3 hours, followed by dbcAMP for 7 days, producing neuronal-like cells with GABAergic-like and dopaminergic-like traits (Suon et al., 2006). While NES expression was found by another study to be necessary for neurosphere formation by rat MSCs (Wislet-Gendebien et al., 2003).

1.3.1.3 Conditioned Media or Co-Culture Systems

Co-culturing of MSCs with a variety of neural cell types or treatment with neural cell-conditioned media has also been performed in attempts to achieve a more mature neuronal phenotype. The study by Sanchez-Ramos and colleagues (Sanchez-Ramos *et al.*, 2000) that reported the earliest cytokine- and signalling molecule-based neural induction of MSCs, also demonstrated that co-culturing murine MSCs with primary foetal murine midbrain cells provided enhanced differentiation in comparison with culturing in induction medium alone. After co-culturing, the number of MSCderived cells expressing NeuN and GFAP increased by at least two-fold, supporting the hypothesis that cell-to-cell contact plays an important role in MSC differentiation, in addition to signalling with cytokines and trophic factors (Sanchez-Ramos *et al.*, 2000).

Astrocytes have been utilised by Joannides, et al. (Joannides et al., 2003) and Jiang, et al. (Jiang et al., 2003) in co-culture and conditioned media experiments, since hippocampus-derived astrocytes have been shown to have a role in instructing neuronal fate specification of NSCs and promoting neuronal maturation and functional synapse formation (Song et al., 2002a; Song et al., 2002b). Application of hippocampus astrocyte-conditioned medium to hMSCs achieved neurofilament and ß tubulin IIIexpressing cells, which displayed greater process elaboration in comparison to DMSOtreated cells, and also fewer GFAP-positive astrocyte-like cells (Joannides et al., 2003). Meanwhile, Jiang, et al. did not observe significant enhancement of morphologic maturation following the addition of astrocyte-conditioned medium as a final differentiation step after sequential cytokine induction of mouse MAPCs (Jiang et al., 2003). However, co-culturing with foetal brain astrocytes resulted in prolonged survival, and cells acquired a much more mature neuronal morphology with a more elaborate array of axons. Continued expression of dopaminergic, serotonergic and GABA-ergic markers was also seen. In addition, electrophysiological properties typical of neurons were only observed following co-culturing with astrocytes, and not in MAPCs exposed to astrocyte-conditioned medium or cytokine induction alone (Jiang et al., 2003).

NSCs have also been employed in co-culture systems for rodent MSC neural differentiation (Alexanian, 2005; Wislet-Gendebien et al., 2003). In a study by Wislet-Gendebien, et al., NES-positive rat MSCs were co-incubated with mouse NSCs for 48 hours, resulting in aggregation of cells and sphere formation, followed by plating of cells on poly-ornithine substrate (Wislet-Gendebien et al., 2003). Under these conditions, NES-positive rat MSCs were able to express astrocytic markers GFAP and GLAST (glial high affinity glutamate transporter), but lacked expression of neuronal markers, NeuN, NSE, β tubulin III and MAP-2b, and oligodendroglial markers, O4 and A2B5 (Wislet-Gendebien et al., 2003). In a later study, Wislet-Gendebien, et al. further showed that rat MSCs co-cultured with murine cerebellar granule neurons express neuronal markers, including Tuj1 and NeuN, and display electrophysiological properties of immature neurons, such as generation of single-action potentials and response to several neurotransmitters (Wislet-Gendebien et al., 2005). Whereas, Alexanian, et al. co-cultured mouse MSCs with proliferating or fixed mouse NSCs, and generated NSClike cells with increased expression of NES and SOX2 (sex determining region Y-box 2); and after further differentiation, β tubulin III and GFAP expression could also be detected, along with neuronal and glial morphologies (Alexanian, 2005).

Kondo and colleagues have also cultured murine MSCs in the presence of an organ of Corti explant, which is the target tissue for developing cochlear ganglion neurons, with neural induction medium consisting of FGF-2, forskolin, IBMX, BDNF and BME (Kondo *et al.*, 2005). Additionally, mouse MSCs were also cultured in conditioned medium from mouse embryonic day (E)10 hindbrain/somite/otocyst and neural induction medium (Kondo *et al.*, 2005). Incubation of MSCs with neural induction medium and conditioned medium resulted in strong up-regulation of sensory neuron marker and POU-domain transcription factor Brn3a, as well as basic helix-loop-

helix genes NEUROD1 and neurogenin (NEUROG) 1, which are essential for cranial sensory neurogenesis (Kondo et al., 2005). When co-cultured with an E18 organ of Corti tissue, MSCs survived and propagated vigorously, and on exposure to neural induction reagents, ceased proliferation and gained neuronal-like morphology with extension of processes toward sensory hair cells in the explants, accompanied by expression of defined sensory neuron markers, such as Brn3a, glutamate receptor 4, and calretinin (Kondo et al., 2005). The microenvironment created by tissue explant cultures was utilised by another study examining murine MSC differentiation in an organotypic hippocampal slice co-culture model (Abouelfetouh et al., 2004). In this system, differentiation of mouse MSCs to neuron-like cells occurred within the hippocampal slice boundaries starting from day 3 of co-culture, whereas formation of network-like connections were observed at approximately day 14 (Abouelfetouh et al., 2004). Axonlike processes were also observed to penetrate into the hippocampal tissue, however migration of mouse MSCs was not observed (Abouelfetouh et al., 2004). Furthermore, it was shown that addition of RA to the culture system increased the number of differentiated cells (both neuron-like and non-neuron-like) and 9.6% of these cells were positive for NeuN, while no staining could be detected for astrocytic marker GFAP, or microglial marker Iba1 (ionised calcium binding adaptor molecule 1) (Abouelfetouh et al., 2004).

1.3.1.4 Genetic Engineering

A genetic engineering approach has also been employed to attain MSC-derived neural-like cells through the over-expression of specific genes known to be important for neural development and function. Genes that have been over-expressed in MSCs for generation of neural cells include: Noggin (Kohyama *et al.*, 2001), the Notch

52

intracellular domain (*NICD*) (Dezawa *et al.*, 2004), and the *BDNF* gene (Zhao *et al.*, 2004).

Transfection of MSCs with the Noggin gene caused cells to aggregate and detach from culture surfaces, and form non-adherent neurosphere-like clusters (Kohyama *et al.*, 2001). Plating of the cluster-forming cells onto poly-Lornithine/fibronectin-coated surfaces and growth in medium containing a cytokine cocktail, elicited differentiation into neuronal-like cells, with greater than 50% of cells expressing MAP-2, and evidence of calcium influx in response to depolarising stimuli (Kohyama *et al.*, 2001). Dezawa and colleagues (Dezawa *et al.*, 2004) employed *NICD* transfection of MSCs together with application of FGF-2, ciliary neurotrophic factor (CNTF), forskolin and in some cases glial-derived neurotrophic factor (GDNF) or BDNF/NGF, which resulted in generation of neuronal-like cells capable of electrophysiological activity, as well as function in an animal disease model. Whilst, combination of *BDNF* gene transfection with RA induction was found to produce cells with significantly increased expression of neural markers, such as NES, NeuN, O4 and GFAP, and the ability to generate voltage dependent K⁺/Ca²⁺ currents using the patch clamp technique (Zhao *et al.*, 2004).

Studies have also genetically engineered MSCs with the *TH* gene (Lu *et al.*, 2005), as well as the two genes required for 3,4,-dihydroxyphenylalanine (L-DOPA) synthesis (Schwarz *et al.*, 2001), to assess the ability of MSCs to act as gene delivery vehicles for gene therapy of PD. Upon transplantation of *TH*-engineered MSCs into a rat model of PD, behavioural improvement was observed (assessed by apomorphine-induced rotational behavior), TH expression was detected around the graft sites and the levels of DA in the lesioned striatum were elevated in comparison with levels in control *lacZ*-MSC treated rats (Lu *et al.*, 2005). Whereas, Schwarz, *et al.* introduced human TH

type I (rate limiting enzyme in DA biosynthesis) and rat GTP cyclohydrolase I (enzyme providing tetrahydropterin cofactor for TH) into rat MSCs and found synthesis and secretion of L-DOPA together with 1000-fold expansion over 3-4 weeks (Schwarz *et al.*, 2001).

1.3.1.5 Combined Approach

A number of studies have also combined elements from the different induction methods. Variations that have been investigated include: cytokine induction and addition of factors that increase intracellular cAMP concentrations (Barzilay *et al.*, 2008; Kan *et al.*, 2007a; Kondo *et al.*, 2005; Long *et al.*, 2005; Suon *et al.*, 2006; Tondreau *et al.*, 2008); cytokine induction with factors to increase intracellular cAMP and use of conditioned medium (Guo *et al.*, 2005; Kondo *et al.*, 2005); genetic manipulation and cytokine induction (Dezawa *et al.*, 2004; Kohyama *et al.*, 2001); and co-culture systems with RA or cytokine induction (Abouelfetouh *et al.*, 2004; Jiang *et al.*, 2003).

1.3.2 Evidence of Neuronal Differentiation of MSCs

The results of the above-mentioned studies together with those of *in vivo* MSC transplantation studies holds great promise for the eventual use of MSCs in cellular therapies for neurodegenerative diseases. However, recent inconsistent results and findings of MSC fusion with mature cell types have revealed the necessity for further detailed analyses in order to gain a greater understanding of the true nature of MSC multi-lineage differentiation events. Moreover, a number of factors complicate the interpretation of published results, as the methods employed for neural differentiation and assessment of produced cells are often fragmented and inconsistent (Phinney *et al.*, 2007). Additionally, many markers used for classifying neural cell types are not entirely

neural-specific, and *in vitro* expansion of MSCs in culture may introduce artifacts (Phinney *et al.*, 2007).

Currently, the majority of studies have relied on morphological, immunological and gene expression changes as evidence of neural plasticity of MSCs. However, several criteria have been proposed to define whether a functional neuron has indeed been generated (Reh, 2002; Svendsen et al., 2001). According to Reh (Reh, 2002), the proposed neuronal cell should be (1) post-mitotic, (2) polarised, with a single axon and multiple dendrites, (3) capable of firing voltage-gated action potentials, and (4) able to communicate with other neurons through synapses, requiring both neurotransmitter release and neurotransmitter receptors. Svendsen, et al. also believe that in defining a new neuron, not only should the correct anatomy of a neuron be displayed, but the cell should display appropriate developmental maturation, that is, the anatomical features of a maturing neuron should be matched with the expression of neuronal-specific markers (Svendsen *et al.*, 2001). However, the authors point out that since the specificity of the markers relies heavily on the quality of the antibody, markers should only be considered as indicative of neuronal identity, but cannot confirm neuronal identity on their own (Svendsen et al., 2001). Therefore, the ultimate tests for identification of neuronal cells are assessment of the ability to connect and form synapses with other neurons and to subsequently affect the behavioural function of an animal; these are typically examined through electrophysiological techniques and transplantation into animal disease models with demonstration of graft function through behavioural testing (Svendsen et al., 2001). To date, none of the studies examining MSC-derived neuronal-like cells have shown fulfillment of all these criteria.

The expression of neural genes and proteins by undifferentiated MSCs prior to application of differentiation stimuli (discussed in Section 1.2.1), emphasises the need

for examination of basal expression levels in MSCs, and for caution in interpretation of results. Therefore, in neuronal differentiation of MSCs it would be necessary to demonstrate significant up-regulation of an already expressed neuronal marker/s, and/or *de novo* expression of neuronal-specific marker/s. Additionally, if genes and proteins from multiple lineages are constitutively expressed in MSCs, it would be expected that concomitant down-regulation of non-neuronal markers should be observed. Expression of NES by MSCs has frequently been reported in the literature as evidence of a NSClike or neural progenitor-like phenotype, since NES was originally described as a marker of NSCs (Lendahl et al., 1990). However, NES expression has now been documented in a variety of cells and tissues, including developing and regenerating muscle (Kachinsky et al., 1994; Sejersen et al., 1993; Vaittinen et al., 2001), newly formed endothelial cells (Klein et al., 2003; Mokry et al., 1998a; Mokry et al., 1998b), epithelial cells of the developing lens (Mokry et al., 1998b), and activated hepatic stellate cells (Niki et al., 1999). In addition, non-neural cells such as chondrocytes, myoepithelial cells and certain fibroblast populations have been reported to express GFAP, neurofilament and/or NSE proteins (Egerbacher et al., 1995; Hainfellner et al., 2001). Phinney and Prockop (Phinney et al., 2007) have also commented that while many reports have shown expression of a wide range of neural proteins in MSC-derived neuron-like cells, only an unbalanced repertoire of neuronal markers has been described, with cells often lacking the functional properties of bona fide neurons. Furthermore, limitations with the current evidence of electrophysiological activity also exist, as those studies which have shown acquisition of neuronal functional properties have not verified the presence of all electrical characteristics of neurons, or are unable to show these attributes (Phinney et al., 2007).

Recently, time-lapse microscopy has been employed for examining the process of neural differentiation from rat and human MSCs using the chemical induction method and serum withdrawal (Bertani et al., 2005; Croft et al., 2006; Lu et al., 2004; Neuhuber et al., 2004). Using this technique it was shown that the acquired morphological changes after application of BME, DMSO/BHA or serum withdrawal were due to cellular shrinkage rather than new neurite outgrowth, and that there was no motility or further elaboration of the processes remaining after cytoplasmic retraction (Bertani et al., 2005; Croft et al., 2006; Lu et al., 2004; Neuhuber et al., 2004). Furthermore, these morphological changes could be reproduced in normal primary fibroblasts, and could be mimicked by drugs that elicit cytoskeletal collapse and disruption of focal adhesion contacts (Bertani et al., 2005; Croft et al., 2006; Neuhuber et al., 2004), or by various cellular stressors such as detergents, high-molarity sodium chloride, and extremes of pH (Lu et al., 2004). With respect to the use of morphological analysis as evidence of MSC neural differentiation, it appears that time-lapse microscopy is a useful technique for ascertaining the validity of morphological changes, and to ensure that any putative axons/dendrites were formed through outgrowth rather than cellular shrinkage. Therefore, this technique will be employed in my PhD studies to confirm the morphological observations during cytokine-based MSC neural differentiation.

Another challenging aspect in characterising MSC neuronal differentiation, involves the identification of MSC-derived neurons after grafting. Much of the difficulty with this is in distinguishing between host and graft cells. Currently, for grafting cells of the same species, identification of donor cells is mainly achieved through the use of genetic modification to label cells (e.g. with GFP or β -galactosidase), transplanting male donor cells to allow for Y chromosome detection, pre-labelling cells

57

with fluorescent dyes, or using BrdU or tritiated thymidine to label dividing cells before transplantation (Svendsen et al., 2001). In addition, if hMSCs are to be grafted into an animal model, then it is possible to stain for detection of human specific markers such as human nuclear antigen (HNA). Using the above cell identification techniques together with neural-specific markers, therefore, allows identification of grafted cells and an indication of whether neuronal differentiation events have occurred. However, caution should be taken in analysing tissue-sections using this method, as false doublepositive cells may be detected with standard fluorescence microscopic analysis (i.e. single-labelled cells juxtaposed onto one another). Instead, detailed analysis of tissue sections using high-powered analytical equipment, such as confocal microscopes, would be useful to reduce the likelihood of detecting false double-positive cells (Svendsen et al., 2001). The possibility of cell fusion must also be considered, particularly with in vivo transplantation experiments or *in vitro* co-culture systems, in which the MSCs are in direct contact with other cell types. The reported frequency of cell fusion events is typically low, however, it has been detected in lethally irradiated animals receiving BM grafts (Alvarez-Dolado et al., 2003; Weimann et al., 2003), and in mouse BM cells cocultured with mouse ESCs (Terada et al., 2002). Nevertheless, cell fusion cannot account for all MSC neural differentiation events, as in vitro studies have shown acquisition of neural phenotypes by MSCs that were not in contact with neural cells.

While much remains to be determined, continued investigations conducted in a meticulous and systematic manner in areas such as MSC prospective isolation, efficient clonal selection, optimal expansion and *in vitro* differentiation, will undoubtedly provide a great deal of knowledge and understanding for the proper development of therapeutic approaches involving MSCs. Advancements have already been made with an increasing number of studies reporting evidence for the acquisition of

58

electrophysiological properties by MSCs post-neuronal differentiation *in vitro* (Cho *et al.*, 2005; Greco *et al.*, 2007b; Kan *et al.*, 2007a; Trzaska *et al.*, 2007). Functionality has been shown in these studies by demonstrating neurotransmitter synthesis and packaging (Cho *et al.*, 2005; Trzaska *et al.*, 2007), neurotransmitter release in response to depolarising stimuli (Greco *et al.*, 2007b; Kan *et al.*, 2007a), presence of spontaneous post-synaptic currents (Cho *et al.*, 2005; Greco *et al.*, 2007b), constitutive DA release (Trzaska *et al.*, 2007), and up-regulation of voltage-gated potassium (Trzaska *et al.*, 2007) and sodium channels (Greco *et al.*, 2007b).

1.3.3 Possible Molecular Control of MSC Neuronal Differentiation

In recent years, increasing efforts have been directed towards identifying the molecular mechanisms underlying MSC plasticity. To date, the underlying mechanisms remain elusive, and the great variety of differentiation methods reported adds further complications. The critical transitions of MSCs to neuronal-like cells are controlled by signalling pathways and subject to subsequent transcriptional regulation (Blondheim et al., 2006). Jori and colleagues have recently investigated the biochemical pathways involved in neuron-like commitment and maturation of rat MSCs via induction with forskolin (which increases cAMP), DMSO, BHA, VPA and KCl (Jori et al., 2005). The findings of this study suggests that: (1) increases in cAMP induced by forskolin treatment, activates the classical protein kinase A (PKA) pathway, however not through the exchange protein directly activated by cAMP, a guanine nucleotide exchange factor for the small GTPase Rap1 and Rap2; (2) MEK-ERK signalling could contribute to neural commitment and differentiation; (3) CaM KII activity seems dispensable for neuron differentiation, but its inhibition could contribute to rescuing differentiating cells from death (Jori et al., 2005). Additionally, this study found that these in vitro differentiation agents are required for the early steps of neural differentiation, but are unable to further sustain this process and therefore are not useful for long-term *in vitro* survival of neurons (Jori *et al.*, 2005). Chu and colleagues also showed involvement of cAMP-PKA activation of cAMP response element binding protein for inducing neurite outgrowth and reducing astrocyte and oligodendrocyte differentiation in a neural induction protocol utilising IBMX or forskolin (Chu *et al.*, 2006).

Dezawa, *et al.* utilised a genetic engineering approach (transfection with *NICD* followed by application of cytokines) to obtain neuronal-like cells from rat MSCs and hMSCs, that were capable of action potential generation and behavioural improvement in a 6-hydroxydopamine (6-OHDA)-lesioned rat model of PD (Dezawa *et al.*, 2004). Analysis of the effect of *NICD* transfection showed that down-regulation of signal transducer, and activator of transcription (*STAT*) 1 and *STAT3* were tightly associated with *NICD*-mediated neuronal induction (Dezawa *et al.*, 2004). Additionally, treatment with Janus kinase (JAK)/STAT inhibitor alone, instead of *NICD*-transfection, was able to elicit neuron-like cells though to a lesser extent, emphasising the importance of STAT inhibition in this neuronal induction procedure (Dezawa *et al.*, 2004). It was also suggested that the mitogen-activated protein kinase (MAPK)-signalling pathway may be involved, since FGF-2, CNTF and forskolin (applied after *NICD*-transfection) activate the MAPK cascade, which may then activate the transcription of neuronal genes (Dezawa *et al.*, 2004).

Studies performed by Wislet-Gendebien and colleagues have led them to propose that NES expression by rat MSCs is a prerequisite for acquiring the capacity to progress toward the neural lineage (Wislet-Gendebien *et al.*, 2003). It was found that only NES-positive rat MSCs were capable of: (1) sphere formation when cultured in conditions typically used for NSC culture, and (2) differentiation into GFAP-positive cells when co-cultured with NSCs (Wislet-Gendebien *et al.*, 2003). In addition, they

60

have also recently shown that NES-positive rat MSCs over-expressed transcription factors SOX2, SOX10 and PAX6, WNT (wingless-type MMTV integration site family) receptors frizzled (Fzd)1, Fzd2 and Fzd5, and neuregulin receptors ErbB2 and ErbB4, in comparison to NES-negative rat MSCs (Wislet-Gendebien *et al.*, 2005).

As further investigations are being conducted into the molecular mechanisms underlying the neural induction of MSCs, a greater understanding of this complex process will be obtained. It is anticipated, however, that the regulation of MSC neural differentiation involves complex interactions between a range of signalling pathways, transcription factor cascades and microRNAs. Although some of the key studies that pioneered this area (Deng *et al.*, 2001; Woodbury *et al.*, 2000), have now been found to describe culture artifacts rather than true neuronal differentiation, promising results have recently been obtained, which demonstrate the acquisition of electrophysiological properties characteristic of neurons, as well as synthesis of neurotransmitters. While mature neuronal cell types have yet to be generated, the progress and knowledge achieved since the first reports of MSC neuronal differentiation continue to fuel and encourage this field of research.

1.4 Parkinson's Disease and Cell-Based Therapies

The development of methods to induce neuronal differentiation of hMSCs has opened the possibility of applying these cells in regenerative or reparative therapies of the CNS. The loss of neurons characterises many neurodegenerative diseases, such as PD, Alzheimer's disease, and Huntington's disease, and injuries such as stroke, traumatic brain injury, and spinal cord injury. Furthermore, the CNS is known to have a poor capacity for generating new neurons and oligodendrocytes after injury or degeneration, with the adult and aged CNS, in particular, possessing a prevailing nonneurogenic and growth-inhibitory milieu that negatively influences these processes (Horner *et al.*, 2002; Ourednik *et al.*, 2004). For these reasons, therapies endeavouring to replace the damaged or missing neurons and restore some degree of function have been the focus of much attention. Positive outcomes, including neurological recovery, have been observed following transplantation of MSCs into animal models of CNS disorders (some of which are discussed in Section 1.2.3.2), revealing MSCs to be a promising therapeutic modality for the treatment of CNS diseases and injuries.

1.4.1 Overview of Parkinson's Disease

PD is a chronic neurodegenerative disorder characterised by the progressive loss of nigrostriatal neurons that synthesise the neurotransmitter DA (Cotran *et al.*, 1999). The cell bodies of affected dopaminergic neurons reside within the substantia nigra pars compacta, while the nerve terminals project rostrally to innervate the striatum (also known as the caudate putamen; see Figure 1.2). Consequently, neuronal degeneration in PD is associated with a reduction in striatal DA content, with the severity of symptoms being proportional to the DA deficiency (Cotran *et al.*, 1999).

1.4.1.1 Neuropathological Features of Parkinson's Disease

A number of neuropathological features define PD, including the specific loss of pigmented catecholaminergic neurons of the substantia nigra (A9 group of dopaminergic neurons), the presence of Lewy bodies in remaining nigral neurons, and the normal appearance of the striatum (Gelb *et al.*, 1999; Klockgether, 2004; Schulz *et al.*, 2004; Teismann *et al.*, 2004). The classic macroscopic finding of substantia nigra depigmentation or pallor results from the depletion of neurons which normally contain conspicuous amounts of neuromelanin (Schulz *et al.*, 2004). Lewy bodies were first described by Lewy in 1912, and are single or multiple, eosinophilic, round to elongated inclusions in the cytoplasm of affected neurons, which are composed of numerous proteins, including misfolded synaptic protein α -synuclein, parkin, ubiquitin,



Figure 1.2 Position and axonal projection of dopaminergic cell groups within the adult rodent brain (sagittal view).

The cell bodies of mesencephalic dopaminergic neurons reside in the substantia nigra pars compacta (A9 group; red), ventral tegmental area (A10 group; green), and in the retrorubral field (A8 group). Different dopaminergic cell groups possess characteristic projection areas, with neurons of the A9 group innervating the dorsolateral striatum (caudate nucleus and putamen) and globus pallidus (GP), forming the nigrostriatal pathway, while A10 and A8 neurons project to the ventral striatum (nucleus accumbens (N Acc), amygdala (Amyg) and olfactory tubercle (O Tub)) as part of the mesolimbic system, and to the prefrontal cortex (mesocortical system). Additional connections from the A10 group are also established with the thalamic and hippocampal regions. Approximate locations are shown for the diencephalic groups (A11-15) of the hypothalamus/ventral thalamus (grey) and the associated projections into the brain stem/spinal cord and anterior pituitary (dashed lines). Periglomerular interneurons within the olfactory bulb form the telencephalic A16 group (yellow dashed line). The A8 and A17 (retina amacrine interneurons) groups have not been depicted. Figure from (Prakash et al., 2006).

synphilin and neurofilaments (Cotran et al., 1999; Dunnett et al., 1999; Schulz et al., 2004). Often, these protein aggregates have a dense hyaline core surrounded by a clear halo, and ultrastructurally, are composed of fine filaments densely packed together at the core, but loose at the rim (Cotran et al., 1999; Schulz et al., 2004). Additionally, Lewy neurites can be observed, which occur when proteinaceous aggregates form in dendrites producing a local swelling (Schulz et al., 2004). Interestingly, in PD the mesolimbic dopaminergic neurons, which reside in the ventral tegmental area (VTA) adjacent to the substantia nigra pars compacta, are much less affected (Schulz et al., 2004). Further neurodegeneration also exists in PD beyond that observed in dopaminergic neurons, with degeneration and Lewy body formation evident in noradrenergic (locus coeruleus), serotonergic (raphe), and cholinergic (dorsal motor nucleus of vagus, nucleus basalis of Meynert) systems, and in the cerebral cortex (especially the cingulate gyrus and parahippocampal gyrus), olfactory bulb and autonomic nervous system (Cotran et al., 1999; Gelb et al., 1999; Schulz et al., 2004). However, degeneration of dopaminergic neurons of the substantia nigra pars compacta remains the earliest and most consistent neuropathologic feature of PD.

1.4.1.2 Clinical Features of Parkinson's Disease

The clinical features of PD were initially described almost two centuries ago by James Parkinson in 1817 (Dunnett *et al.*, 1999; Langston, 1998; Orr *et al.*, 2002). The cardinal symptoms of PD include bradykinesia (slow movement), muscular rigidity and resting tremor, although postural instability, impaired gait, depression, dementia and other motor and non-motor symptoms may also be involved (Dunnett *et al.*, 1999; Gelb *et al.*, 1999; Klockgether, 2004). Of the cardinal symptoms, bradykinesia primarily contributes to the disability experienced by PD patients, whilst tremor is the most conspicuous feature and stigmatises patients as PD sufferers (Klockgether, 2004). Degeneration of dopaminergic neurons is well underway by the time symptoms become apparent, with loss of approximately 50% of nigral dopaminergic neurons and depletion of 70-80% of DA in the striatum (Bernheimer *et al.*, 1973; Dunnett *et al.*, 1999; Fearnley *et al.*, 1991). Estimates from the rates of substantia nigra cell loss suggest that disease onset in PD patients commences approximately 5 years prior to the onset of symptoms (Fearnley *et al.*, 1991).

Although the symptoms and pathology of PD have been long identified, the mechanisms underlying the dysfunction and progressive loss of dopaminergic neurons is still poorly defined. Idiopathic PD is usually sporadic, with major risk factors of aging and environmental exposure, but evidence of a genetic component has also been demonstrated with a 2-14 fold increase in relative risk found in close relatives of PD patients, and some concordance between identical twins who develop PD at less than 50 years of age, or when examining subclinical dysfunction of the nigrostriatal dopaminergic system by positron emission tomography (PET) imaging (Brooks, 1998; Dunnett et al., 1999; Gasser, 1998; Olanow et al., 1999; Veldman et al., 1998). Nonetheless, investigation of an environmental cause was the focus of much research, with epidemiological studies showing geographical variation in incidence that appeared to be associated with living in a rural environment, well-water consumption, and exposure to pesticides, herbicides, farming and industrial chemicals (Dick, 2006; Langston, 1998; Olanow et al., 1999; Veldman et al., 1998), and the finding of a PD syndrome resulting from a single exposure to a toxin, 1-methyl-4-phenyl-1,2,3,6tetrahydropyridine (MPTP) (Langston et al., 1983). Surprisingly, an inverse relationship has also been found between cigarette smoking and the risk for PD, which has been considered to provide support for an environmental cause of PD (Langston, 1998).

1.4.1.3 Aetiology and Pathogenesis of Parkinson's Disease

Advances in understanding the pathogenesis of this disease have recently been achieved through findings of causal genetic, molecular and cellular tissue changes. Mechanisms involving protein aggregation, dysfunction of the ubiquitin-proteasome pathway, mitochondrial dysfunction, oxidative stress, activation of the c-Jun N-terminal kinase system, and inflammation have all been implicated in PD pathogenesis, and may also further interact and amplify one another (Dunnett et al., 1999; Schulz, 2008; Schulz et al., 2004). A growing number of disease-related genes have been identified to cause hereditary forms of PD with autosomal dominant or autosomal recessive inheritance (see Table 1.1). While most single gene mutations are only responsible for a small number of patients with PD, these discoveries have provided important insights into the possible disease mechanisms underlying PD. One possible pathway involves disturbances in protein quality control, which could result from point mutations or gene duplications and triplications in the α -synuclein gene, or mutations in the Parkin or ubiquitin-C-terminal hydrolase-L1 genes that would cause dysfunction of the ubiquitinproteasome protein degradation system, therefore, leading to accumulation of aggregated α -synuclein (Schulz, 2008). Another potential mechanism includes altered mitochondrial function and oxidative stress, suggested by mutations in the PINK1 and DJ-1 genes, respectively, as well as polymorphisms in the Htra-2/Omi gene (Schulz, 2008). Further to this, dopaminergic neurons are known to be particularly susceptible to oxidative injury since DA production and use depends on oxidative mechanisms (Jenner et al., 1998). Aberrant kinase activities may also underlie PD pathogenesis with observations of c-Jun N-terminal kinase activation in animal studies, and mutations in LRRK2 resulting in increased kinase activity that may cause neuronal toxicity (Schulz, 2008). Cell-cell interactions and immune regulation have also been identified as being

Table 1.1 Genetics of Parkinson's Disease

Locus	Chromosomal Localization	Gene product	Inheritance	Lewy body pathology	Specific clinical symptoms
PARK 1	4q21	∝-Synuclein	AD	Yes	Dementia
PARK 2	6q25.2-27	Parkin	AR	No	Early onset, L-Dopa-induced dyskinesias, improvement during sleep, foot dystonia
PARK 3	2p13	?	AD	Yes	Dementia
PARK 4	4q21	∝-Synuclein	AD	Yes	Dementia, postural tremor
PARK 5	4p14	UCH-L1	AD	No report	Not described
PARK 6	1p35-36	PINK-1	AR	No report	Early onset, tremor dominant
PARK 7	1p36	DJ-1	AR	No report	Early onset, dystonia, psychiatric alterations
PARK 8	12cen	LRRK2/Dardarin	AD	Yes/but also Tau pathology	Tremor, late onset
PARK 9	1p36	ATP13A2	AR		Kufor-Rakeb syndrome, very early onset
PARK 10	1p32	?	AD (?)	No report	Late onset
PARK 11	2q34	?	AD (?)	No report	Late onset
	5q23	Synphilin-1	Susceptibility	No report	Late onset
PARK 13	2p13	HtrA2/Omi	Susceptibility	No report	Late onset

AD, autosomal dominant

AR, autosomal recessive

(Table from (Schulz, 2008))

significant in PD pathogenesis (Orr *et al.*, 2002). Nigral dopaminergic neurons are selectively vulnerable to inflammatory attack due to the presence of higher numbers of inflammatory microglia in the substantia nigra (Kim *et al.*, 2000; Lawson *et al.*, 1990), and evidence of a role for inflammation in PD is supported by findings that substantial microglial activation has been directly associated with dopaminergic neuronal death in PD (Orr *et al.*, 2002).

1.4.1.4 Treatments for Parkinson's Disease

The current treatment of PD involves alleviation of symptoms through pharmacologic replacement of DA with L-DOPA (the immediate precursor of dopamine; also referred to as Levodopa (International Nonproprietary Name)) (Cotran et al., 1999; Davie, 2008; Dunnett et al., 1999). Additionally, a peripheral DOPAdecarboxylase inhibitor is co-administered to control disabling side effects, and together these drugs effectively alleviate akinesia and rigidity in early- and middle-stage disease (Dunnett et al., 1999). This combination therapy has been the gold standard for symptomatic treatment of PD for four decades (Davie, 2008). Other drugs have also been developed such as dopamine receptor agonists, slow-release L-DOPA formulations, inhibitors of degrading enzymes catechol-O-methyltransferase and monoamine oxidase B, dopamine transport blockers, anti-cholinergic drugs, and antiglutamatergic drugs (Davie, 2008; Dunnett et al., 1999; Klockgether, 2004). However, these medications do little to prevent or reverse disease progression, and efficacy declines several years after disease onset (Cotran et al., 1999; Dunnett et al., 1999). Furthermore, after long-term levodopa therapy patients frequently develop severe side effects, particularly in the form of dyskinesias (Davie, 2008; Dunnett et al., 1999).

The use of neurosurgery as a therapy for PD has been revisited after originally being implemented over 50 years ago (Davie, 2008; Dunnett *et al.*, 1999). At that time,

68

surgical intervention was not very reliable, and could only address tremor symptoms. Presently, neurosurgery has shown promise with the advent of modern imaging techniques, electrophysiological monitoring during surgery, improved understanding of basal ganglia circuits and disturbances caused by PD that have allowed identification of critical target sites, as well as the introduction of chronic deep-brain stimulation for functional and reversible inhibition of target areas (Davie, 2008; Dunnett *et al.*, 1999; Klockgether, 2004). Now, rather than focusing on primary symptom relief, surgical intervention has been applied for controlling levodopa-induced dyskinesias, therefore extending pharmacotherapy to advanced patients (Dunnett *et al.*, 1999). New approaches are needed for pharmacological therapies for PD, and while drug trials concentrate primarily on motor symptoms and complications, studies have found that quality of life is, instead, determined by other factors, including depression and dementia (Klockgether, 2004). Another approach under investigation as a potential treatment for PD is the transplantation of DA-producing cells, which will be further discussed in Section 1.4.2.

1.4.1.5 Experimental Animal Models of Parkinson's Disease

A number of experimental models of PD are available, which have proven to be valuable tools in evaluating clinical approaches for symptomatic treatments. This has allowed the development of effective DA replacement therapies that are now routinely used in clinical settings, and is one of the major reasons for the advancement of PD therapies in comparison with other neurodegenerative diseases, which still lack effective symptomatic treatments (Jenner, 2008). Nevertheless, whilst many approaches have been taken for producing animal models of PD, including use of a range of toxins, transgenic expression of mutant proteins involved in familial PD, and deletion of genes affecting dopaminergic neuronal development (knock-out models), there are still no accepted experimental models that closely resemble the progression and pathogenesis of PD as observed in humans (Jenner, 2008). Additionally, while the existing models have demonstrated utility in identifying symptomatic treatments, these models have been much less successful in discovering neuroprotective therapies (Jenner, 2008). For this purpose, new models may be required which more accurately portray PD pathogenesis as it affects humans (Jenner, 2008).

Two of the most extensively studied models of PD are the 6-OHDA and the MPTP toxin-based models (Jenner, 2008; Schwarting et al., 1996; Teismann et al., 2004). Once it was found that 6-OHDA exhibited toxicity towards catecholaminergic neurons (dopamine and noradrenaline), it was applied in the depletion of these neurotransmitters in the forebrain and opened the way for toxin-based models of PD (Jenner, 2008; Teismann et al., 2004; Ungerstedt, 1968; Ungerstedt et al., 1970). Destruction of catecholaminergic neurons is believed to occur through the selective uptake of 6-OHDA by these neurons, and autooxidation of 6-OHDA following intracellular metabolism, resulting in the simultaneous formation of several cytotoxic products (Sachs et al., 1975). The unilateral 6-OHDA-lesioned rat has become the universal standard model for assessing potential PD pharmacotherapies (Teismann et al., 2004), and it is this model that will be used for the present study. Typically, the neurotoxin is stereotaxically injected into the medial forebrain bundle or substantia nigra of one hemisphere, causing unilateral degeneration of the nigrostriatal pathway and depletion of striatal DA (Jenner, 2008; Teismann et al., 2004). Unilateral lesioning results in a quantifiable functional asymmetry that is readily measurable by inducing rotational behaviour in rats with administration of direct or indirect DA agonists, apomorphine and amphetamine, respectively (Schwarting et al., 1996; Teismann et al., 2004; Ungerstedt et al., 1970). Furthermore, several other deficits exist in this model

that correspond with the degree of nigrostriatal loss, including abnormalities in sensorimotor performance, posture and paw usage (Henderson *et al.*, 2003; Schwarting *et al.*, 1996).

Although, the 6-OHDA model is useful for testing symptomatic treatments, there exist several discrepancies between this experimental model and PD in humans. Firstly, it only reproduces a single component of the cell death process rather than the complex cascade of events that occurs in PD; secondly, there is little evidence of Lewy body formation, despite this being a characteristic pathological hallmark of PD; thirdly, the pattern of pathology and cell loss in different neural regions is not reproduced; and fourthly, 6-OHDA administration leads to acute death of dopaminergic neurons in the substantia nigra, rather than the progressive degeneration usually observed in PD (Gerlach *et al.*, 1996; Jenner, 2008; Teismann *et al.*, 2004). Other drawbacks of this model include the need for stereotactic application of 6-OHDA to appropriate regions of the brain, and that using rotational behaviour as a measure of functional improvement is not relevant to the pathophysiology of PD symptoms (Gerlach *et al.*, 1996).

The discovery that MPTP causes an irreversible and severe parkinsonian syndrome in humans that is almost indistinguishable from PD provided perhaps the greatest advance in PD experimental models, and also led to the first effective non-human primate model of PD (Jenner, 2008). Generally, the MPTP model is produced with systemic administration of MPTP, and a murine model is widely utilised, but, this is not as robust and reproducible a model of PD as sometimes reported (Jenner, 2008). The mechanism of action of MPTP involves entry of the neurotoxin into the CNS, followed by metabolism into the active form (1-methyl-4-phenylpyridinium) by astrocytes, and entry into dopaminergic neurons via the dopamine transporter, where it inhibits complex I of the mitochondrial electron transport chain (Du *et al.*, 2001; Orr *et*

al., 2002). Important roles for glial activation and the accompanying up-regulation of inducible nitric oxide synthase in MPTP neurotoxicity have also been discovered (Du *et al.*, 2001; Orr *et al.*, 2002). Although, like most toxin-based PD models, the MPTP model is also divergent from PD in humans, for the same reasons that the 6-OHDA model differs from PD in humans (Gerlach *et al.*, 1996; Jenner, 2008). Additional drawbacks exist with the MPTP model, such as the lack of permanency of motor deficits, and tremor symptoms are not often observed (Gerlach *et al.*, 1996).

1.4.2 Cellular Replacement Therapies for Parkinson's Disease

The primary aim of cell-based treatments for PD is to restore neuronal function by transplanting cells with DA-producing capacity into the striatum in an attempt to replace the lost nigrostriatal neurons. Ideally, the implanted cells should also reconstitute neural networks and provide physiologically-regulated feedback-controlled DA release to prevent graft-related side effects (Isacson *et al.*, 2008). Cell transplantation therapy is believed to hold great potential for PD, due to the localised nature of this lesion, however, at present research in this area is still largely exploratory (Isacson *et al.*, 2008). In addition, several issues concerning cell transplantation exist that have limited research progress, including scientific, technical and ethical difficulties (Klockgether, 2004). Nonetheless, encouraging results have been reported in both animal models and human clinical trials, suggesting that further understanding of appropriate cell populations and target sites for transplantation, as well as improved surgical and technical approaches, could lead to the development of much needed neuroprotective or neurorestorative interventions for PD.

1.4.2.1 Foetal Neural Tissue Transplantation in PD

A variety of cellular sources have been investigated for the potential to act as a renewable supply of transplantable dopaminergic neurons for PD, including foetal brain tissue, embryonic and adult SC populations, and xenogeneic porcine foetal nigral tissue. In preclinical studies involving neurotoxin-induced parkinsonian rodents and nonhuman primates, transplantation of embryonic or foetal dopaminergic neuron-rich mesencephalic tissue resulted in the amelioration of at least some aspects of lesioninduced motor deficits (Bjorklund *et al.*, 1979; Bjorklund *et al.*, 1981; Dunnett *et al.*, 1981; Perlow *et al.*, 1979; Redmond *et al.*, 1986; Wuerthele *et al.*, 1981). Additionally, these studies revealed that grafted cells survived, produced DA, formed effective dopaminergic terminals and synapses, and re-innervated the striatum. Furthermore, graft survival and functional improvements were also reported in xenogeneic situations, with transplantation of human foetal mesencephalic tissue from 6.5-9 week old or first trimester donors into the striatum of Parkinsonian rats (Brundin *et al.*, 1986; Brundin *et al.*, 1988; Stromberg *et al.*, 1986).

These promising findings led to the initiation of clinical trials in the late 1980s, which employed transplantation of foetal ventral mesencephalic tissue from aborted human foetuses into patients with PD (Hitchcock *et al.*, 1988; Lindvall *et al.*, 1989; Lindvall *et al.*, 1988). The majority of the early studies were open-label, uncontrolled trials, which reported a remarkable ability of transplanted human foetal dopaminergic cells to survive and provide clinically meaningful improvement that in some cases could replace pharmacological therapies (Brundin *et al.*, 2000; Freed *et al.*, 1992; Freeman *et al.*, 1995; Hagell *et al.*, 1999; Hauser *et al.*, 1999; Kordower *et al.*, 1995; Lindvall *et al.*, 1999; However, subsequent double-blind, placebo-controlled trials were unable to replicate these findings, and failed to show evidence of significant clinical benefit, with a few patients also developing dyskinesias or abnormal movements (Freed *et al.*, 2001; Isacson *et al.*, 2008; Olanow *et al.*, 2003; Svendsen, 2008).

Recent clinical trial results have proven more optimistic, with most human dopaminergic neuron-containing transplants found to display functional activity for at least a decade (Kordower *et al.*, 2008; Mendez *et al.*, 2008; Piccini *et al.*, 1999). Moreover, Mendez and colleagues detected the presence of graft survival in the absence of pathology for at least 14 years after transplantation (Mendez *et al.*, 2008). Notably, other reports found that in a few long-term cases (11 years post-transplant), a minor proportion of surviving dopaminergic neurons showed signs of pathological changes associated with PD, including protein aggregation and fibrillar changes suggestive of Lewy body formation (Kordower *et al.*, 2008; Li *et al.*, 2008). The mechanism underlying this observed pathology remains unknown, and reflects the many biological and technical challenges that lie ahead in developing a cell-based therapy for PD.

1.4.2.2 Neural Stem Cell Transplantation in PD

One of the limitations of using foetal neural tissue for cellular therapies is the rarity of developing dopaminergic neurons appropriate for transplantation, which are only present in 6-8 week old embryos (Sayles *et al.*, 2004; Svendsen, 2008). This, together with the low post-operative survival of these cells, necessitates the sourcing of mesencephalic tissue from multiple aborted foetuses, as well as transplantation within hours of tissue harvesting (Sayles *et al.*, 2004). To address this problem, research efforts since the early 1990s have been directed towards investigating the potential of different SC populations to act as an alternative dopaminergic cell source (Isacson *et al.*, 2008; Svendsen, 2008).

At first, *in vitro* culture expanded human foetal neural precursor cells were employed by Svendsen and colleagues in the 6-OHDA unilaterally lesioned Parkinsonian rat model (Ostenfeld *et al.*, 2000; Svendsen *et al.*, 1997; Svendsen *et al.*, 1996). These studies revealed that upon transplantation into the lesioned adult striatum, expanded neural precursors were capable of survival, migration and differentiation towards glia and neurons, including TH-immunoreactive neurons. Although, in contrast with the large self-contained tissue masses formed by primary foetal neural tissue grafts, the expanded neural precursors formed only thin grafts containing small numbers of surviving cells, and did not migrate extensively into host tissues. Occasional THpositive cells were detected in low numbers, with the observed expression perhaps being a transient event. In most cases, the small number of these cells was unable to elicit any functional effects, except in 2 animals that displayed partial reversal of lesioninduced rotation deficits. While the generation of some TH-positive neurons in these studies provided encouragement, it was clear that improvements were necessary for increased graft survival and dopaminergic differentiation.

Studer, *et al.* (Studer *et al.*, 1998) addressed these issues through *in vitro* culture expansion and pre-differentiation of neural precursors toward dopaminergic neurons prior to implantation. Using this procedure the yield of dopaminergic neurons was increased and partial alleviation of behavioural deficits in hemiparkinsonian rats could be achieved. However, *in vitro* expansion was only performed for a short period (6-8 days) and the degree of expansion was small (Sayles *et al.*, 2004). Subsequent studies have continued to examine these issues, and have tested the use of NSCs from different neural regions, different expansion and pre-differentiation techniques, *ex vivo* genetic modification of cells, and administration of trophic factors together with cell transplantation, with varying degrees of success. Common to the majority of these studies is the limited access to autologous NSCs from patients, due to the location of NSCs within the brain. Interestingly, a possible autologous cell source for PD cellular therapy has recently been described by my co-supervisor's laboratory (Murrell *et al.*,

2008), in a study showing functional improvement in the 6-OHDA rat model using SCs from rat and human olfactory mucosa.

1.4.2.3 Embryonic Stem Cell Transplantation in PD

Another potential source of cells that have been investigated are ESCs, which also provide opportunities for the development of large quantities of therapeutically useful cells. The first demonstration of TH-positive neuronal cell derivation from ESCs was reported by Lee, *et al.* (Lee *et al.*, 2000). This study implemented a differentiation protocol that was comprised of multiple induction stages, including embryoid body generation, selection of CNS SCs with a defined medium, proliferation of these cells using mitogen FGF-2, together with SHH and FGF-8, and differentiation and maturation following removal of FGF-2 and supplementation with ascorbic acid (AA). Cells obtained through this procedure were found to express CNS- and midbrainspecific genes in a pattern suggestive of progressive restriction to a mesencephalic and metencephalic CNS SC fate. Furthermore, the murine ESC-derived neurons were shown to produce DA, respond to neurotransmitters, and exhibit spontaneous synaptic activity. By the end of the final stage, this system yielded >30% TH-positive neurons, giving strength to the possibility of using ESCs as a reliable donor cell source for dopaminergic neuron transplantation in PD (Lee *et al.*, 2000).

This prospect was further extended by studies investigating the *in vivo* transplantation potential of ESCs in the 6-OHDA hemiparkinsonian rodent model. Transplantation of low numbers of undifferentiated murine ESCs into the rat striatum resulted in spontaneous generation of dopaminergic neurons capable of gradual and sustained restoration of motor asymmetry from 5-9 weeks post-transplantation (Bjorklund *et al.*, 2002). In addition, PET and functional magnetic resonance imaging scans also confirmed that the ESC-derived dopaminergic neurons had become

76

integrated within the host brain circuitry and mediated functional improvement. Nevertheless, 20% of rats receiving grafts developed teratomas within 9 weeks at the site of transplantation(Bjorklund *et al.*, 2002), indicating the need for caution and removal of proliferating and non-neuronal cells prior to application of this technique in the clinic.

However, another study reported the absence of teratoma formation and proliferative marker Ki-67, when pre-differentiated *NR4A2*-transfected ESCs were grafted into the striatum of 6-OHDA lesioned rats (Kim *et al.*, 2002b). Graft analysis revealed the presence of TH-positive cells that possessed complex cellular morphologies, with the immunoreactive cell bodies restricted to the graft region, whereas TH-positive cell processes extended into the parenchyma of the host striatum up to 2 mm away from the graft site. Additionally, these cells were found to release DA, form functional synaptic connections and modulate spontaneous and pharmacologicallyinduced behaviours, strongly supporting the capability of ESC-derived neurons to survive and function after intrastriatal grafting (Kim *et al.*, 2002b). Following these findings, other studies have continued to improve these procedures by examining different methods of pre-differentiation (cytokines, inductive factors, genetic modification, co-culture systems, and combinations of these), selection and enrichment of dopaminergic neuron populations, and combining trophic factor administration with transplantation.

1.4.2.4 Patient-Specific Cellular Transplantation in PD

Recent advances in the understanding of pluripotent SC biology have extended this work to include the possibility of patient-specific dopaminergic neuron replacement therapies. Studer and colleagues (Tabar *et al.*, 2008) utilised therapeutic cloning strategies to generate nuclear transfer ESCs, using donor fibroblasts obtained from individual 6-OHDA lesioned hemiparkinsonian mice. The genetically matched ESCs were then differentiated into midbrain dopaminergic neurons and transplanted back into the original donor mouse. Significantly, this study found enhanced graft survival and amelioration of parkinsonian behaviour with autologous grafts, in comparison with allogenic grafts, which appeared to be related to decreased graft immunogenicity (Tabar *et al.*, 2008).

In another study by Jaenisch and colleagues (Wernig *et al.*, 2008), the need for nuclear transfer techniques was circumvented through the use of induced pluripotent stem (iPS) cells (Takahashi *et al.*, 2006; Yamanaka, 2007). Murine iPS cells were successfully differentiated into dopaminergic neurons of midbrain character through application of regional patterning factors (SHH and FGF-8) to FGF-2 responsive, iPS cell-derived neural precursor cells. Intrastriatal transplantation of these cells into the 6-OHDA unilaterally lesioned rat model of PD resulted in marked improvement in rotational behaviour at 4 weeks post-grafting. Attempts were also made to minimise the risk of teratoma formation by employing fluorescence-activated cell sorting (FACS) to remove contaminating pluripotent cells (SSEA1-positive fraction) from the cell suspension prior to transplantation (Wernig *et al.*, 2008).

A number of alternative non-neural adult SC sources have also been examined for potential in dopaminergic neuron replacement therapy for PD, in particular BMderived MSCs, which will be discussed in detail in Section 1.4.4.

1.4.2.5 Challenges in Cellular Replacement Therapies for PD

Many important biological, technical, and surgical hurdles and challenges need to be addressed before cellular replacement therapy for PD patients can be implemented in the clinic. Currently, detailed knowledge is still required concerning critical issues, such as: 1) the appropriate neuronal (and perhaps also glial) cell type for transplantation, including the cellular and biochemical characteristics necessary in the donor cell population; 2) the correct anatomical location for cell administration; 3) suitable selection criteria for identifying patients most likely to respond to cellular replacement therapy, which may be influenced by the stage of the disease and responsiveness to levodopa; 4) effective parameters for cell preparation and delivery to obtain optimal graft survival, including the optimal volume, dosage and format of cells; 5) mechanisms for limiting host immunological responses to donor cells; and 6) connectivity variables for functional reconstitution of neurocircuitry (Isacson et al., 2008). Another crucial challenge for clinical application is the optimisation of graft function, whilst preventing graft-related side effects, since midbrain dopaminergic neurons are highly specialised cell types that must be capable of physiologically appropriate DA release and establishment of specific short- and long-distance connections within a complex cellular network (Isacson et al., 2008). Additionally, investigation of cellular transplantation therapies must take into account the rate of cellular maturation when progenitor cell populations are used, since this is donor dependent and currently available data indicates that long-term clinical evaluation periods may be necessary (Isacson et al., 2008). Production of clinical grade, individual, patient-specific cell lines also presents a considerable logistical challenge (Svendsen, 2008). And, it must also be remembered that the pathological changes present in PD are not restricted to dopaminergic neurons alone, with degeneration occurring in other neural regions (Svendsen, 2008). However, while gaps exist in our current knowledge, renewed hope is provided through the major research efforts directed towards understanding and developing novel cellular therapies for PD.

1.4.3 Lessons from Midbrain Dopaminergic Neuron Development In Vivo

In order to achieve efficient directed differentiation of MSCs into the neural lineage, it is surmised that the *in vitro* differentiation procedures utilised should mimic neural development as it occurs *in vivo*. Consequently, for the generation of midbrain dopaminergic neuronal cell types from SC populations, it is expected that the growth factors and signalling molecules important for directing the development of these cell types will be of significant relevance. Additionally, when SC populations are subjected to these differentiation inducing signals, it is also anticipated that changes in gene and protein expression would occur in a manner that reflects the patterns observed *in vivo* in developing dopaminergic neurons. Although the exact mechanisms of these processes are yet to be fully elucidated, current knowledge indicates the involvement of several critical signalling molecules and transcription factor cascades in complex cell extrinsic and intrinsic mechanisms that govern midbrain dopaminergic neuron specification and differentiation (Abeliovich *et al.*, 2007; Prakash *et al.*, 2006; Smidt *et al.*, 2007).

Cell fate specification in the vertebrate nervous system is believed to occur in accordance with a grid of Cartesian co-ordinates established by gradients of signalling molecules secreted along the dorso-ventral and rostro-caudal (anterior-posterior) axes of the neural tube (Maxwell *et al.*, 2005; Ye *et al.*, 1998). Neural progenitors consequently assume distinct cell fates according to the positional identity conferred by the grid location. Additionally, refinement of cell fate is achieved through multiple local signalling centres at the rostro-caudal level, which specify the identity and stereotypic locations of mature neuronal cell types within the major brain subdivisions.

After restriction of neuronal fate by these extrinsic cues, further differentiation into mature post-mitotic neurons is directed by intrinsic signals, often transcription factors (Maxwell *et al.*, 2005; Ye *et al.*, 1998). The development of midbrain

80

dopaminergic neurons can be broadly divided into 4 stages, the first of which involves migration of neuroblasts from the ventricular zone to the central and lateral ventral mesencephalon (Abeliovich *et al.*, 2007; Isacson *et al.*, 2008). The second stage is comprised of specification of progenitors to a dopaminergic neuronal precursor cell fate, followed by the third step, in which dopaminergic precursors exit the cell cycle and commence expressing markers indicative of an early midbrain dopaminergic neuron phenotype. Lastly, functional maturation of early dopaminergic neurons takes place, which is marked by expression of mature midbrain dopaminergic neuronal markers, axonal pathfinding and establishment of specific connectivity (Abeliovich *et al.*, 2007).

1.4.3.1 Extrinsic Cues Involved in Midbrain Dopaminergic Neuron Specification

The specification of the permissive region for midbrain dopaminergic neuron generation is a fundamental event that is initiated by the positioning of 2 key signalling centres, the floorplate, which is present at the ventral midline of the neural tube, and the isthmic organiser (also known as the isthmus) at the midbrain-hindbrain boundary (Abeliovich *et al.*, 2007; Prakash *et al.*, 2006). Cells of the floorplate secrete SHH, which influences cell fate along the dorso-ventral axis, whereas FGF-8 is secreted by the isthmus and exerts patterning activity along the rostro-caudal axis. Midbrain dopaminergic neurons develop at sites of SHH and FGF-8 intersection (see Figure 1.3), and Ye and colleagues (Ye *et al.*, 1998) have shown that induction with these 2 extrinsic factors is necessary and sufficient for dopaminergic neuron induction in multiple locations in the neural tube. Other extrinsic factors that have been demonstrated to be significant in specifying the site of dopaminergic neuron development include TGF- β , Nodal, WNT signalling (WNT1 and WNT5A), and RA (Abeliovich *et al.*, 2007; Prakash *et al.*, 2007).


Figure 1.3 A model illustrating the mechanisms controlling the positions and cell fates of DA neurons during early development.

The specification of the region for midbrain DA neuron generation is controlled by 2 signalling centres, SHH expression in the floor plate, and FGF-8 expression along the isthmus. Initially, FGF-4 expression in the primitive streak influences the hindbrain region and prepatterns this area to block DA neuron formation. Following this SHH is expressed along the floor plate, and FGF8 is expressed at the isthmus and anterior neural ridge. Areas influenced by both SHH and FGF-8 signals produce DA neurons, while the influence of SHH and FGF-8 on the FGF-4-prepatterned hindbrain produces serotonin neurons. Anterior is to the left. F, forebrain; M, midbrain; H, hindbrain; is, isthmus; PS, primitive streak; ANR, anterior neural ridge; DA, dopaminergic; 5HT, serotonergic. Top panel, presomitic embryo; middle and bottom panel, late somitogenic embryo. Figure from (Ye *et al.*, 1998).

The migration of young midbrain dopaminergic neurons has been associated with a small number of molecules (Smidt *et al.*, 2007). However, little is known of the specific roles of these molecules in migration and some of the reports have presented conflicting findings. The signalling molecules that have been implicated in early dopaminergic neuronal migration include axon guidance cue netrin 1 and its receptor DCC (deleted in colorectal cancer), reelin, proteoglycan phosphacan 6B4 together with L1 cell adhesion molecule, neural cell adhesion molecule and polysialic acid expression, and PAX6 (Smidt *et al.*, 2007).

Several secreted factors have been found to be implicated in midbrain dopaminergic neuronal differentiation and functional maturation (Abeliovich *et al.*, 2007; Smidt *et al.*, 2007). These include RA, which appears to be locally synthesised during all steps of development, docosahexanoic acid as a ligand for retinoid X receptor, WNT family members, FGF family members, and TGF signalling (TGF- α and TGF- β), which appears to act in conjunction with SHH and FGF-8, as well as possibly inducing neurotrophic support. Currently, it is not known whether other extrinsic factors that have a role in earlier development (Abeliovich *et al.*, 2007; Smidt *et al.*, 2007).

Many factors have been identified as having a role in midbrain dopaminergic neuron maintenance and neurotrophic support (Smidt *et al.*, 2007). Proteins with known neurotrophic effects include GDNF, BDNF, TGF- α , TGF- β , neurotrophin 3, -4 and -5, TrkB and TrkC receptors for neurotrophins, neurturin, neublastin/artemin, persephin, growth/differentiation factor 5, neuregulin 1, BMPs, heparin-binding EGF, and FGF-2. Of these factors, GDNF has shown the most promise in increasing midbrain dopaminergic neuronal survival. A role for GDNF signalling in long-term maintenance was also suggested in a recent *in vivo* study (Kramer *et al.*, 2007), which examined mice with regionally selective ablation of the RET receptor tyrosine kinase (c-RET; receptor for GDNF). This work showed progressive, adult-onset loss of dopaminergic neurons specifically in the substantia nigra with c-RET ablation. Additionally, crosstalk between GDNF signalling and TGF- β signalling has also been described (Smidt *et al.*, 2007).

1.4.3.2 Cell Intrinsic Signals Directing Dopaminergic Neuron Differentiation

In conjunction with the cell extrinsic cues that instruct midbrain dopaminergic neuronal development, also exist cell intrinsic programs that act in concert with extrinsic factors to progressively restrict cell fate choices (Abeliovich *et al.*, 2007)(see Figure 1.4). Recent advances through *in vivo* gene ablation studies have elucidated networks of transcriptional factors with sequential and parallel action that are implicated in the induction of specific phenotypic characteristics of midbrain dopaminergic neurons (Isacson *et al.*, 2008; Smidt *et al.*, 2007). None of the identified transcription factors are sufficient for individually instructing the midbrain dopaminergic neuronal phenotype, providing further support for a network model of transcription factor involvement (Abeliovich *et al.*, 2007).

The permissive region formed early in development by organising centres is defined by a specific pattern of gene expression (Smidt *et al.*, 2007). The expression domains of two transcriptional repressors orthodenticle homeobox (OTX) 2, at the presumptive fore- and midbrain, and gastrulation brain homeobox 2, at the presumptive hindbrain and spinal cord, establish the midbrain-hindbrain boundary (Abeliovich *et al.*, 2007; Maxwell *et al.*, 2005; Prakash *et al.*, 2006; Smidt *et al.*, 2007). Subsequent transcriptional regulation through LIM homeobox transcription factor (LMX) 1A, msh homeobox 1, and NEUROG2 has been found to play an important role in specifying the midbrain dopaminergic neuronal phenotype (Abeliovich *et al.*, 2007; Andersson *et al.*, 2006; Smidt *et al.*, 2007). Studies have also found that the development of the final



Figure 1.4 Intrinsic and extrinsic factors affecting the specification and development of midbrain dopaminergic neurons.

The developmental program of midbrain dopaminergic neurons consists of cell intrinsic programs that act in concert with cell extrinsic cues to progressively restrict cell fate choices. Initially, neuronal stem cells are patterned toward a ventral cell fate, and away from a dorsal fate (grey). Subsequently, mitotic ventral mesencephalic precursors are specified toward a dopaminergic fate, and away from alternative fates including the serotonergic cell fate (grey). Networks of transcription regulatory factors with sequential and parallel action have been implicated in these processes. Transcription factors that promote a dopaminergic neuronal fate are shown in blue, whilst inhibitory transcription factors are depicted in green. Additionally, secreted factors known to be involved in the different stages of the dopaminergic developmental program are shown in red. E, embryonic day; mDN, midbrain dopaminergic neurons. Figure from (Abeliovich *et al.*, 2007).

adult dopaminergic phenotype is dependent on several critical transcription factors, including EN1, EN2, LMX1B, NR4A2 and PITX3 (Abeliovich et al., 2007; Maxwell et al., 2005; Prakash et al., 2006; Smidt et al., 2007). In particular, NR4A2 has been proven to be necessary for the proper differentiation and survival of dopaminergic neurons at later developmental stages, but does not appear to be essential for the specification and initial birth of midbrain dopmainergic precursors. The majority of neurotransmitter-related genes and some trophic signalling pathways are also regulated by NR4A2, including several proteins required in DA synthesis and regulation, such as TH, vesicular monoamine transporter 2 (VMAT2), DAT and c-RET. Additionally, PITX3 is specifically expressed in midbrain dopaminergic neurons of the CNS, and appears to be critical for the proper terminal differentiation or maturation, and/or early maintenance of the subset of dopaminergic neurons that form the substantia nigra. Furthermore, the TH gene may be directly regulated by PITX3, and PITX3-deficient Aphakia mice exhibit specific loss of TH-expressing cells in the substantia nigra, whereas the VTA dopaminergic neurons were less affected (Abeliovich et al., 2007; Maxwell et al., 2005; Prakash et al., 2006; Smidt et al., 2007).

In recent years, many of the molecular processes underlying midbrain dopaminergic neuronal development, maintenance and function have been elucidated (Smidt *et al.*, 2007). These advances in our understanding provide many opportunities for the development of strategies to induce efficient generation of midbrain dopaminergic neurons from SC populations. Since successful cell replacement therapies are believed to depend upon the knowledge of how to make appropriate dopaminergic neurons, it is hoped that further unravelling of these mechanisms and continued research efforts will eventually lead to the development of functional cell-replacement therapies for PD. The present study aims to employ extrinsic factors, which are known to be important in midbrain dopaminergic neuron specification and differentiation, in *in vitro* culture systems for the dopaminergic neuronal differentiation of hMSCs. Furthermore, cell intrinsic factors will be examined by investigating the gene expression changes elicited in hMSCs during *in vitro* dopaminergic neuronal differentiation.

1.4.4 Dopaminergic Neuronal Differentiation of MSCs *In Vitro* and Transplantation into Animal Models of Parkinson's Disease

A number of investigations have been performed to determine the potential of MSCs to differentiate into functional DA-producing neuronal-like cells, and to examine the capability of MSC grafts to re-innervate the striatum and ameliorate behavioural deficits in animal models of PD. To date, these studies have exhibited varying degrees of success, ranging from the generation of TH-positive cells, to observations of electrophysiological activity and DA secretion, to significant improvements in pharmacologically-induced rotational behaviour in hemiparkinsonian rodent models. Although, amongst the different studies a diverse array of methodologies have been employed and variations exist in the cell types obtained, which together have caused difficulty in the evaluation and comparison of experimental outcomes. Nevertheless, the promising results that have been reported provide much encouragement for the prospect of developing a MSC-based cellular therapy for PD.

Similar methods have been investigated for dopaminergic neuronal differentiation of MSCs as have been utilised in general MSC neuronal differentiation (discussed in Section 1.3.1), but with the inclusion of factors more specific to midbrain dopaminergic neuron development (see Section 1.4.3). At the time that the neuronal differentiation work in the present study was commenced, there were 6 reports in the literature describing dopaminergic neuronal differentiation of MSCs from rodents (Dezawa *et al.*, 2004; Guo *et al.*, 2005; Jiang *et al.*, 2003) and humans (Dezawa *et al.*,

2004; Fu *et al.*, 2006b; Hermann *et al.*, 2004; Suon *et al.*, 2006). The majority of these studies employed the addition of extrinsic factors to cell cultures, such as cytokines and small molecules; however, the effects of gene transfection, conditioned medium and co-culturing, as well as combinations of these methods were also investigated. These studies are summarised in Table 1.2.

The initial report by Jiang and colleagues (Jiang et al., 2003) described a multistage differentiation strategy capable of generating midbrain neuronal-like cells from the MAPC subset of murine MSCs. The first stage of differentiation consisted of supplementation with FGF-2 for 7 days, followed by a combination of FGF-8 and SHH for 7 days, then BDNF for 7 days, and finally 5-12 days of co-culture with foetal brain astrocytes. The resulting cells exhibited phenotypical and electrophysiological characteristics similar to midbrain neurons, with approximately 25% efficiency of dopaminergic neuron generation. Also, co-culturing with astrocytes for 7-12 days resulted in between 80% and 100% of cells acquiring functional voltage-gated sodium channels. In addition, activation of transcription factors known to be involved in neuroectodermal development, such as SOX1, OTX2, OTX1, PAX2, PAX5 and NR4A2, were detected in a similar developmental pattern as described for NSC and ESC dopaminergic differentiation (Jiang et al., 2003). However, a number of caveats are associated with this approach, including lack of specificity for dopaminergic neuronal production, since serotonergic (approximately 25%) and GABA-ergic (approximately 50%) neurons were also produced, and as a consequence of this, requirement for selection of desired dopaminergic cells. Additionally, astrocyte-conditioned medium or cytokines alone were unable to provide appropriate differentiation signals. Therefore, astrocyte co-culture is necessary for prolonged survival and maturation toward electrophysiologically active cells, which impedes clinical translation of this method.

Differentiation Method	Cell Type	Results (In vitro/In vivo)	References
Multi-stage method: 1) FGF-2 (7d) 2) FGF-8, SHH (7d) 3) BDNF (7d) 4) Astrocyte co-culture (5-12d)	Murine MAPC	25% efficiency of dopaminergic neuron generation (IF); electrophysiological characteristics similar to midbrain neurons	Jiang, <i>et al.</i> (2003)
NICD transfection, FGF-2+CNTF+ FBS +forskolin (5d), GDNF or BDNF+NGF (7-11d)	Human and rat MSCs	41% TH-positive (IF); electrophysiological properties; DA release after depolarisation; functional improvement in hemiparkin- sonian rats	Dezawa, <i>et al.</i> (2004)
Neurosphere formation with EGF+FGF-2 (2-10 weeks), BDNF (10-14d)	Human MSCs	11% TH-positive (IF); DA release after depolarisation; loss of osteogenic differentiation capacity	Hermann, <i>et al</i> . (2004)
Neurosphere formation with IBMX+forskolin+ TPA (3hrs), dbcAMP (7d)	Human MSCs	15% of β tubulin III-positive cells also TH-positive (IF); lack of graft survival and differentiation in PD model	Suon, <i>et al.</i> (2006)
IBMX (2d), GDNF+ IL-1β+mesencephalic glial-conditioned med- ium+flash-frozen mes- encephalic cell frag- ments (7-15d)	Rat MSCs	35% TH-positive (IF)	Guo, <i>et al.</i> (2005)
Expansion (10% FBS+ DMEM; 3d), neuron- conditioned medium alone (6-9d), SHH+ FGF-8+neuron-condit- ioned medium (3-12d)	Human MSCs from Wharton's jelly of umbil- ical cord	12.7% TH-positive (IF); DA secretion; partial correction of amphetamine- evoked rotational behaviour	Fu, <i>et al</i> . (2006)

Table 1.2 In Vitro Dopaminergic Neuronal Differentiation of MSCs

d, days IF, immunofluorescence staining

Furthermore, this study was conducted using murine cells, and whether the same results can be obtained with human cells must be confirmed.

The next key study demonstrating production of dopaminergic neuronal-like cells from MSCs was reported by Dezawa and colleagues (Dezawa et al., 2004), who employed transfection of the NICD gene and administration of trophic factors FGF-2 and CNTF with forskolin, followed by addition of GDNF. Importantly, it was shown that this approach could generate TH-positive cells with 41% efficiency, which was at the time the highest reported rate, and the cells obtained also displayed *in vitro* and *in* vivo functional properties of neurons, with little glial differentiation. Delayed rectifier potassium currents were recorded after treatment with FGF-2, CNTF and forskolin, whereas voltage-gated fast sodium currents and action potentials were only obtained with further administration of BDNF and NGF. Treatment with GDNF elevated the expression of transcription factors NR4A2, LMX1B, EN1 and PITX3, and also resulted in *in vitro* DA release in response to high K⁺ depolarising stimuli. Further evidence for the dopaminergic neuronal differentiation of MSCs was demonstrated through observations of significant improvements in hemiparkinsonian rats, in terms of apomorphine-induced rotational behaviour and non-pharmacological assessments, such as the step adjustment and paw-reaching tests, and also through examining DA production by the graft in cultured brain slices. Additionally, the authors confirmed that similar results could also be obtained when using rat or human MSCs (Dezawa et al., 2004). However, this study was not without limitations. The use of gene transfection may be problematic for application in the clinic, although the authors commented that this could be avoided through induction with a JAK/STAT inhibitor (Dezawa et al., 2004). Also, while this study employed the use of neural trophic factors, those typical of midbrain dopaminergic neuronal specification, SHH and FGF-8, were not included,

suggesting that the observed differentiation may be divergent from that which occurs typically during normal development.

Application of extrinsic factors alone have also resulted in the induction of MSCs toward a dopaminergic neuronal phenotype (Guo et al., 2005; Hermann et al., 2004; Suon et al., 2006). Hermann and colleagues (Hermann et al., 2004) devised a protocol for producing a NSC-like population from hMSCs, through the formation of neurosphere-like structures in low-attachment culture flasks in the presence of serumfree medium containing EGF and FGF-2. The majority of the neuroprogenitor-like cells were found to have upregulated expression of NES protein to high levels, while mesodermal protein fibronectin was downregulated. Quantitative PCR analysis showed decreases in SOX1, POU5F1, and neurotrophic tyrosine kinase receptor type 1, whereas neuroectodermal transcripts OTX1, NEUROD1, NEUROG2, musashi and NES were acquired with neurosphere formation. Further differentiation of MSC-derived neuroprogenitors was achieved using BDNF for neuronal differentiation (42% of resulting cells were positive for β tubulin III), or PDGF for glial differentiation (45%) GFAP-positive, and 27% GALC-positive). Interestingly, with this general neuronal differentiation procedure, 11% of cells were also shown to express TH, and were capable of releasing DA in response to membrane depolarisation in vitro. In addition, osteogenic differentiation capacity was lost after induction into neuroprogenitor-like cells (Hermann et al., 2004). However, the authors were unable to examine the electrophysiological function of neuronal-like cells due to technical constraints, and further assessment of function through in vivo transplantation is also required. This system does not appear capable of generating mature neuronal cells, although the authors suggest that immature cells are more suitable for transplantation, since fully differentiated mature neural cells poorly survive detachment and subsequent

91

transplantation (Hermann *et al.*, 2004). Additionally, MSC dopaminergic neuronal differentiation occurred at a low rate of 11%, although even this level of efficiency may be somewhat surprising given the lack of specific dopaminergic inductive cues in the culture media.

Neural progenitor-like sphere formation by hMSCs was also utilised by Suon, *et al.* (Suon *et al.*, 2006) prior to dopaminergic neuronal differentiation by plating spheres on a poly-ornithine/laminin-coated surface in serum-free medium containing IBMX, forskolin and TPA for 3 hours, followed by dbcAMP for 7 days. Down-regulation of mesenchymal genes was observed after differentiation, with concomitant up-regulation of neural-associated genes. Additionally, 15% of β tubulin III-positive cells also expressed TH. Surprisingly, all of the cells were found to express GABA, including those found to be TH-positive. Intrastriatal transplantation of MSC-derived spheres in 6-OHDA unilaterally lesioned rats resulted in short-term graft survival, however, no further differentiation into dopaminergic neuronal cells could be observed, and instead only GABA-positive cells were detected. The lack of survival of hMSCs transplanted in the rat PD model, despite administration of cyclosporine A, reveals a potential limitation in this method of assessment of hMSC *in vivo* function.

Yet another differentiation approach was taken by Guo, *et al.* (Guo *et al.*, 2005), which yielded a rate of 35% TH-positive cells from rat MSCs, after exposure to IBMX for 2 days, followed by application of GDNF, IL-1 β , mesencephalic glial cell-conditioned medium, and flash-frozen mesencephalic cellular fragments for 7-15 days. However, examination of DA secretion and excitability of the resulting cells was not performed, and assessment of this system using hMSCs remains to be determined. In addition, treatment with glial-conditioned medium and membrane fragments is unsuitable for use in clinical applications.

Fu and colleagues (Fu *et al.*, 2006b) isolated MSCs from Wharton's jelly of human umbilical cord and employed a multi-stage differentiation method consisting of expansion in 10% FBS-DMEM for stage 1, neuronal differentiation through incubation with neuron-conditioned medium in stage 2, and induction of a dopaminergic neuronal phenotype with application of neuron-conditioned medium, SHH and FGF-8. TH-positive neurons were generated with approximately 12.7% efficiency, and DA secretion into the culture medium was detectable. Furthermore, functional effects were observed *in vivo* following transplantation into the striatum of hemiparkinsonian rats, as seen by partial correction of amphetamine-evoked rotational behaviour (Fu *et al.*, 2006b). The primary caveat of this work was the low efficiency of dopaminergic neuronal differentiation, which may have contributed to the incomplete behavioural recovery. In addition, the electrophysiological characteristics of the resulting cell population were not examined, and it should be noted that umbilical cord-derived MSCs have been reported to possess different properties than adult BM-derived MSCs (Kern *et al.*, 2006).

The above studies that have examined the *in vivo* functional potential of MSCderived dopaminergic neuron-like cells have found behavioural improvement with transplantation of these cells over that of control undifferentiated MSCs. However, other studies have reported survival, migration, generation of TH-positive cells and functional improvement with transplantation of undifferentiated MSCs (Hellmann *et al.*, 2006; Li *et al.*, 2001b). Li and colleagues (Li *et al.*, 2001b) performed the initial study investigating intrastriatal MSC transplantation in a murine MPTP-induced bilateral model of PD. Behavioural recovery was analysed using the rotarod test, with results showing that mice with a MSC graft exhibited prolonged duration on the rotarod at day 35, when compared with control sham grafted mice. Additionally, MSCs were found to survive within graft sites for at least 4 weeks post-transplantation, and a small proportion of scattered MSCs were TH-immunoreactive. However, the authors hypothesised that the therapeutic benefits most likely originated from MSC secretion of growth factors capable of promoting survival and plasticity within the damaged brain, since improvement was observed rapidly and few MSCs were present in the target tissue (Li *et al.*, 2001b). Hellmann, *et al.* (Hellmann *et al.*, 2006) also transplanted undifferentiated MSCs into parkinsonian rodents (6-OHDA unilaterally lesioned model), but for the purposes of assessing MSC survival, migration and differentiation. It was found that MSCs exhibited higher survival in the 6-OHDA lesioned hemisphere, in comparison with the unlesioned side. In addition, MSC transplantation into the striatum the lasticity in migration of MSCs through the corpus callosum into the striatum, thalamic nuclei and substantia nigra of the 6-OHDA lesioned hemisphere (Hellmann *et al.*, 2006). These studies suggest that the therapeutic benefits observed following MSC transplantation in rodent PD models may be conferred through mechanisms other than cellular replacement and DA production.

A variety of distinct approaches have been implemented in the quest for MSC dopaminergic neuronal differentiation and application in potential PD therapies. Nonetheless, the reported findings possess a number of unifying elements, although these mainly lie in the realm of newly raised questions and issues that remain to be resolved. In the midst of these pertinent issues, are questions regarding the optimal developmental point at which to transplant differentiating cells, the relevance of animal models and studies to the situation in humans, the little-defined potential immunological responses to undifferentiated and differentiated MSC grafts, and the suitability for translation into the clinic. While much progress has been achieved and promising results have been obtained, the generation of mature functional midbrain dopaminergic neurons

from MSCs remains elusive. However, the current evidence of phenotypic conversion and acquisition of functional characteristics *in vitro* and *in vivo*, support the feasibility of this approach and continue to render research efforts in this field as worthwhile pursuits.

1.5 Aims of the Current Study

The overall hypothesis of the current study is that human adult BM-derived MSCs can be induced to differentiate toward the neuronal lineage and in particular into midbrain dopaminergic neuronal-like cells. The project aims to investigate this hypothesis through firstly establishing cultures of hMSCs and characterising these cell populations in the undifferentiated state. The functional capacity of hMSCs to differentiate *in vitro* towards the neural lineage and the expected mesodermal lineages would then be examined, and the resulting cells characterised. Subsequently, the function of MSCs *in vivo* will be explored through transplantation into a rodent model of PD, in order to determine the anatomical and functional effects of MSC transplantation into neural regions. Within these general aims, there are three specific aspects that are to be investigated, and the following outlines these in greater depth.

Human BM-derived MSCs have previously been demonstrated to be capable of transdifferentiation, but have also been shown to have a finite expansion potential. Therefore, it is hypothesised that hMSCs display a pluridifferentiated phenotype, and that long-term *in vitro* culture alters the hMSC phenotype. The first section of this project aims to test these hypotheses through the characterisation of undifferentiated human BM-derived MSCs, in terms of expression of markers of pluripotency and the three germ lineages, and to determine the baseline expression profile of these cells prior to commencing differentiation.

Our laboratory has previously shown that hMSCs are able to adopt a neuronallike phenotype upon cytokine exposure. In view of this, human BM-derived MSCs are hypothesised to be capable of responding to sequential stimuli with cytokines that have been reported to be physiologically important in midbrain dopaminergic neuron development. This project aims to differentiate hMSCs into dopaminergic neuronal-like cells through the manipulation of *in vitro* culture conditions. In addition, this work aims to examine the morphological changes associated with cytokine-induced neuronal differentiation, to determine whether the observed changes result from culture artifacts due to environmental stress during the differentiation procedure. Live cell imaging techniques will be employed to record changes in cellular morphology for these purposes. Human BM-derived MSCs undergoing neuronal differentiation will also be characterised in terms of gene and protein expression, to determine whether a neuronallike phenotype is acquired, together with a concomitant loss of mesodermal lineage and pluripotency markers. Furthermore, the effect of long-term serial passaging on the neuronal differentiation capability of hMSCs will also be investigated, as this has previously been reported to affect the mesodermal differentiation ability of MSCs.

Finally, it is also hypothesised that human BM-derived MSCs are capable of differentiation to dopaminergic neuronal-like cells *in vivo* when transplanted directly into the brains of 6-OHDA unilaterally-lesioned Parkinsonian rats, and also that partial-differentiation of hMSCs prior to transplantation enhances engraftment. This will be investigated through analysis of the survival, anatomical integration and functional effects of the hMSC graft. The specific aims of this final section are: a) To transplant hMSCs (undifferentiated and partially differentiated) into the hemiparkinsonian rat model, and investigate the survival, differentiation and migration of these cells in the brain; b) To determine whether the substantia nigra or the striatum is the optimal

96

transplantation site for hMSCs; c) To evaluate any effects of hMSCs on host cells, for example, neurotrophic effects, or immunomodulatory effects; and d) To discover if there exists an advantage or synergistic effect of co-transplanting hMSCs with olfactory ensheathing cells (OECs) into the hemiparkinsonian rat model. A significant part of this *in vivo* study was conducted at the National Centre for Adult Stem Cell Research (Griffith University, Brisbane, QLD), under the supervision of Prof. Alan Mackay-Sim.

CHAPTER 2

MATERIALS AND METHODS

2.1 Materials

2.1.1 Cell lines

SK-N-SH, HepG2, U-87 MG and MG-63 cell lines from the American Type Culture Collection (ATCC) Global Bioresource Center (Manassas, VA, USA). HUH7 cell line was a gift from Prof. Bernard Tuch (Prince of Wales Hospital, Sydney).

2.1.2 Cell Isolation and Culture Reagents

The tetrameric antibody mixture RosetteSep[™] Bone Marrow Progenitor Cell Pre-Enrichment Cocktail was obtained from StemCell Technologies (Vancouver, Canada) for depleting lineage-committed cells in BM expressing CD3, CD11b, CD14, CD16, CD19, CD56, CD66b and Glycophorin A/CD235a.

DMEM - Low Glucose (DMEM-LG) containing L-glutamine and sodium pyruvate, DMEM/nutrient mixture F12, DMEM/nutrient mixture F12 (without HEPES buffer), Neurobasal medium, DMEM/HAM F12, Dulbecco's Phosphate Buffered Saline (dPBS), 25 U/mL penicillin/25 µg/mL streptomycin, L-glutamine, sodium pyruvate, FBS (Lot No. 1236374), N-2 supplement, B-27 serum-free supplement (50x) liquid, and fibronectin (human plasma) were purchased from Gibco/Invitrogen (Carlsbad, California, USA). MCDB-201 medium, Hank's Balanced Salt Solution (HBSS), ITS liquid media supplement (100x), linoleic acid-albumin, AA 2-phosphate, dexamethasone, β-glycerophosphate, proline, bovine serum albumin (BSA), IBMX, insulin, indomethacin, poly-L-ornithine, and laminin (human placenta) were purchased from Sigma-Aldrich (St. Louis, MO, USA). Mouse laminin was obtained from BD Biosciences (Bedford, MA, USA). All cultureware (sterile, tissue culture-treated plastic flasks and chamber slides) was purchased from Falcon/Becton Dickinson Labware (Franklin Lakes, NJ, USA), except for chamber slides used in live cell imaging, which

were from Lab-Tek, Nunc (IL, USA). Glass coverslips were sourced from Mediglass/Lomb Scientific (NSW, Australia).

The following growth factors/cytokines were purchased from R&D Systems (Minneapolis, MN, USA): recombinant human EGF, recombinant human FGF basic (FGF-2), recombinant mouse FGF-8b, recombinant human GDNF, human PDGF, recombinant human SHH amino terminal peptide, and recombinant human TGF-β3. Recombinant human FGF-8 was also purchased from PeproTech, Inc. (NJ, USA). All cytokines were reconstituted in 0.2% BSA in dPBS and stored at -80°C until use.

2.1.3 Enzymes and enzyme buffers

Platinum[®] SYBR[®] Green qPCR SuperMix-UDG and SuperScript[™] III First-Strand Synthesis System for RT-PCR (reverse transcription-polymerase chain reaction) were obtained from Invitrogen. 0.5% trypsin-ethylenediamine tetraacetic acid (EDTA) was purchased from Gibco/Invitrogen.

2.1.4 Oligonucleotides

Sense and anti-sense primers for real-time RT-PCR were designed using the Primer3 program (<u>http://frodo.wi.mit.edu/cgi-bin/primer3/primer3_www.cgi</u>), and synthesised by Sigma-Aldrich.

2.1.5 Antibodies

Antibodies used in flow cytometry were CD29, CD73, CD45, CD14, CD34, CD166, HLA Class I and HLA Class II from BD Biosciences Pharmingen (San Jose, CA, USA), CD44 from Beckman Coulter (Fullerton, CA, USA) and CD105 from Chemicon (Millipore, NSW, Australia).

Primary antibodies used in immunofluorescent staining were mainly sourced from Chemicon (Millipore), including mouse monoclonal antibodies POU5F1, NES, SSEA-1, SSEA-4, TRA-1-60, A2B5, NeuN, and human nuclear antigen (HNA; clone 235-1; MAB1281), and rabbit polyclonal antibody Forkhead Box A2 (FOXA2; also known as HNF3 β). Other mouse monoclonal antibodies were obtained from BD Pharmingen (CD44), Sigma-Aldrich (MAP-2 and GFAP-Cy3) and Immunostar Incorporated (WI, USA; TH). Rabbit polyclonal primary antibodies were also sourced from Sigma-Aldrich (Fibronectin), Dako Cytomation (Glostrup, Denmark; GFAP), Covance (CA, USA; β tubulin III), and Wako Pure Chemical Industries Ltd (Osaka, Japan; Iba-1). Secondary antibodies used were Alexa Fluor 488 or 594 highly cross adsorbed goat anti-rabbit or anti-mouse antibodies from Molecular Probes (Invitrogen).

2.1.6 Equipment

For flow cytometric analysis, cells were acquired and analysed on a FACSCalibur flow cytometer (BD) with Cell-Quest software.

The NanoDrop 1000 spectrophotometer used to quantify total RNA was obtained from Thermo Scientific. A Rotor-Gene RG3000 machine (Corbett Research, NSW, Australia) was used for performing real-time RT-PCR reactions. The Gel Doc-1000 photographic system (Bio-Rad Laboratories, NSW, Australia) was used for agarose gel electrophoresis. Statistical analysis was performed using GraphPad Prism from GraphPad Software Incorporated (La Jolla, CA, USA) and Microsoft Office Excel.

Routine morphological observations of cultured cells were recorded using an inverted light microscope and digital camera from Olympus (Tokyo, Japan). Live cell imaging was performed with an inverted microscope equipped with CO₂- and temperature-controlled stage from Carl Zeiss (Germany). Zeiss inverted fluorescent microscope, Axiocam digital camera and Axiovision software used for imaging hMSCs cultured on slides and coverslips were also from Carl Zeiss.

The stereotaxic frame and microinjector unit used for intracerebral transplantation were obtained from Kopf Instruments (USA). The syringe and needle used for cell injection were from Hamilton Company (Reno, NV, USA).

Confocal imaging was performed using a Leica laser scanning confocal microscope (DM IRE2 TCS SP2 AOBS) from Leica Microsystems (Wetzlar, Germany). Images were prepared as figures using Adobe Photoshop (Adobe Systems Incorporated, CA, USA). Three-dimensional rendering of graft sites was prepared using Imaris software from Bitplane (Zurich, Switzerland). Images for rendering were captured using a Zeiss Axio Imager Z1 with ApoTome and Axiovision software (Carl Zeiss).

Amphetamine-induced rotational behaviour was measured using the Rotometer Activity System from San Diego Instruments (San Diego, CA, USA).

2.1.7 General materials and chemicals

All general reagents used in molecular biology were of molecular biology grade. TRIzol reagent was obtained from Invitrogen. RNeasy® Micro kit and Mini kit for total RNA isolation and RNase-free DNase I were obtained from Qiagen (Basel, Switzerland). Human brain total RNA was obtained from Stratagene (Texas, USA). Human ESC cDNA was kindly provided by Dr Justin Lees (Prince of Wales Hospital, NSW, Australia). Chloroform was obtained from Sigma-Aldrich. Agarose was obtained from Promega (Madison, USA). GeneRuler Ultra Low Range 10-300 bp DNA ladder was purchased from Fermentas (Ontario, Canada).

Sodium heparin tubes were obtained from Becton Dickinson. Ficoll-Paque[™] Plus solution was purchased from GE Healthcare Bio-Sciences AB (Uppsala, Sweden). Trypan blue solution (0.4%) was purchased from Sigma-Aldrich.

Annexin V-FITC Apoptosis Detection Kit I was purchased from BD Biosciences. Trisodium citrate was obtained from Sigma-Aldrich.

102

Paraformaldehyde (PFA) powder and Triton X-100 solution were obtained from Sigma-Aldrich. Normal goat serum was obtained from Chemicon. DAPI (4',6-diamino-2-phenylindole) was obtained from Molecular Probes (Invitrogen). Glycerol was sourced from Bacto Laboratories (NSW, Australia).

Cyclosporine A was obtained from Sandimmune, Sandoz Pharmaceutical (East Hanover, NJ, USA). Isofluorane gas was from Affane, Bomac (NSW, Australia). Lethabarb-Sodium Pentabarbitone was from Virbac Animal Health (TX, USA). Polyethylene glycol (PEG) and D-amphetamine were obtained from Sigma-Aldrich. Damphetamine was used in compliance with the regulations for Schedule 4 Controlled Substances and with approval from the Chief Pharmacist of QLD and NSW.

2.2 Methods

2.2.1 Cell Culture

2.2.1.1 Isolation of Human Bone Marrow-Derived Mesenchymal Stem Cells

Whole BM (1-2 mL) was collected into sodium heparin tubes from the posterior iliac crest of haematologically normal donors following written informed consent as approved by the Human Research Ethics Committee of our Institute (St Vincent's Hospital Sydney and The University of New South Wales). hMSCs were isolated using antibody-mediated negative selection and density gradient centrifugation, followed by plastic adherence. For negative selection, BM specimens were incubated for 20 minutes (min) with a tetrameric antibody mixture (RosetteSep Bone Marrow Progenitor Cell Pre-Enrichment Cocktail) for depleting lineage-committed cells expressing CD3, CD11b, CD14, CD16, CD19, CD56, CD66b and Glycophorin A/CD235a. Subsequently, MNCs were separated on a Ficoll-Paque density gradient (30 min at 400 g). Enriched cells were re-suspended in expansion medium (see next section) and

103

seeded in culture flasks at a density of less than 8×10^4 cells/cm². hMSCs were isolated from 12 BM specimens using this procedure.

2.2.1.2 Expansion of Human Bone Marrow-Derived Mesenchymal Stem Cells

hMSCs were expanded in medium (adapted from (Pittenger et al., 1999), (Reves et al., 2001), (Gronthos et al., 2003)) consisting of 60% DMEM-LG, 40% MCDB-201 medium, 1x ITS, 1x linoleic acid-albumin, 10⁻⁹ M dexamethasone, 10⁻⁴ M AA 2phosphate, 100 U/mL penicillin, 100 µg/mL streptomycin, and 10% heat-inactivated FBS. FBS was heat-inactivated by incubating room temperature FBS in a water bath at 56°C for 30 min. This medium was chosen as it was demonstrated to expand BM MSCs with multi-lineage differentiation potential (Jiang et al., 2003; Jiang et al., 2002; Reyes et al., 2001). Cultures were maintained at 37°C, 5% CO₂, and 20% O₂. Non-adherent cells were removed 3 days after initial plating, and medium replaced every 3-4 days. When cultures reached 80-90% confluency, hMSCs were harvested by incubating cultures with 0.05% trypsin/0.53 mM EDTA for 5 min at 37°C, followed by immediate inactivation of trypsin with serum-containing media. Exposure to trypsin was minimised and cells were thoroughly washed and re-plated as Passage (P) 1 in expansion medium at a density of $1.7-2.0 \times 10^3$ cells/cm². Cells were subcultured weekly, and population doubling and cell viability (assessed by trypan blue dye exclusion) were recorded. The formula used for calculating the number of population doublings was: $X \times 2^{N} = Y$ (where X is the number of cells initially seeded, N is the number of population doublings, and Y is the final cell number obtained after culturing. Cumulative doubling was calculated by adding together the calculated population doublings. For cryopreservation, cells were stored in cryovials with 10% DMSO, 20% FBS and 70% HBSS and cooled slowly to -80°C in an isopropanol tank, prior to storage in liquid nitrogen.

2.2.1.3 Induction of Multi-lineage Mesodermal Differentiation

Osteogenic, chondrogenic and adipogenic differentiation were induced in P4-6 hMSCs to confirm the MSC identity of isolated cells. All multi-lineage mesodermal differentiation experiments were performed by Dr. Bojiang Shen (St Vincent's Hospital Sydney, NSW Australia), according to published protocols, with some modifications (Abdallah et al., 2005; Khoo et al., 2008; Schutze et al., 2005). Briefly, for osteogenic differentiation, confluent cultures were exposed to DMEM-LG, 100 nM dexamethasone, 50 μ g/mL AA 2-phosphate and 10 mM β -glycerophosphate for 2 weeks. For chondrogenic differentiation, 1×10^6 hMSCs were placed in a 15 mL polypropylene tube and centrifuged to form 3-D pellet cultures. Pellets were cultured for 3 weeks with DMEM-LG, 100 nM dexamethasone, 50 µg/mL AA 2-phosphate, 100 µg/mL sodium pyruvate, 40 µg/mL proline, 1% ITS, 5.35 µg/mL linoleic acid, 1.25 mg/mL BSA, and 10 ng/mL TGF-β3. For adipogenic differentiation, confluent cultures were exposed to DMEM-LG, 10% FBS, 1 μ M dexamethasone, 0.5 mM IBMX, $10 \,\mu\text{g/mL}$ insulin and $100 \,\mu\text{M}$ indomethacin for 2 weeks. Standard histochemical staining procedures were performed on differentiated and undifferentiated hMSCs (cultured in parallel) to confirm mesodermal differentiation. Osteogenic differentiation was confirmed by Alkaline Phosphatase staining, chondrogenic differentiation by Alcian Blue staining, and adipogenic differentiation by Oil Red O staining. For osteogenic and chondrogenic differentiation, multiple fields were examined to determine the area of extracellular matrix deposition as a percentage of the total area of confluent cultures (as described (Digirolamo et al., 1999)). Adipogenic differentiation was assessed by performing cell counts of multiple fields to determine the percentage of positively stained cells.

2.2.1.4 Induction of Neuronal Lineage Differentiation – Single-Stage Cytokine-Based Method (SingleND)

Neuronal differentiation of early (P4-5) and late (P11-12) stage hMSC cultures was induced using a cytokine-based single-stage neuronal differentiation (SingleND) method as reported previously (Tao et al., 2005), with medium containing DMEM/nutrient mixture F12, 1x N-2 supplement, 100 U/mL penicillin, and 100 µg/mL streptomycin, supplemented with 10 ng/mL FGF-2, 10 ng/mL EGF and 1 ng/mL PDGF. hMSCs were plated in neuronal differentiation medium (0.11 mL/cm^2) at 6.7×10^3 cells/cm² (i.e. for a T75 flask, cells were seeded at 5.0×10^5 cells in 8 mL medium). Cultureware was pre-coated with 10 µg/mL Fibronectin to facilitate cell adhesion in serum-free neuronal differentiation conditions. Cultures were incubated at 37° C, 5% CO₂, and 20% O₂ with daily replenishment of growth factors and medium changed every 2 days. Cells were subcultured weekly with 0.05% trypsin/0.53 mM EDTA in the same way as for subculturing hMSCs (see Section 2.2.1.2), however, cells were only exposed to trypsin/EDTA for 2 min at 37° C, due to serum-free conditions. Proliferation and cell viability were assessed by trypan blue dye exclusion. In these experiments, neuronal differentiation conditions were applied for a maximum of 3 weeks, although cells have been shown to survive in these conditions for up to 3 months (Tao et al., 2005). Neuronal-induced cells were harvested for characterisation at 1, 2 and 3 weeks of differentiation. The SingleND method was used for the neuronal differentiation of 4 hMSC cultures obtained from different donors.

2.2.1.5 Induction of Dopaminergic Neuronal Differentiation – Multiple-Stage Cytokine-Based Method (MultiDA)

A multiple-stage protocol was designed with the purpose of using cytokines reported to be important in dopaminergic neuronal development to direct hMSCs more

specifically toward a dopaminergic neuronal fate. Cytokine-based multiple-stage dopaminergic neuronal differentiation (MultiDA) was induced in exponential growth phase hMSC cultures (P5-6) from 3 individuals after the usual MSC subculturing procedure. Cells were seeded into cultureware pre-coated with 2 µg/cm² Poly-Lornithine (overnight incubation at 37° C) and $1 \mu g/cm^2$ Mouse Laminin (>3 hour incubation at 37° C) at 6.0 x 10^{3} cells/cm² in 0.11 mL/cm² differentiation medium (i.e. for a T75 flask, cells were seeded at 4.5×10^5 cells in 8 mL medium). The medium used in all 3 stages of dopaminergic neuronal differentiation contained DMEM/nutrient mixture F12 (without HEPES buffer), 1x N-2 supplement, 100 U/mL penicillin, and 100 µg/mL streptomycin. The aim of Stage 1 of the MultiDA method was to prime hMSCs toward a neural fate. Stage 1 medium was supplemented with 10 ng/mL FGF-2 and 10 ng/mL EGF. The purpose of Stage 2 was to initiate midbrain specification of hMSCs, and Stage 2 medium was supplemented with 100 ng/mL SHH, 10 ng/mL FGF-8 (human) and 200 µM AA 2-phosphate. Stage 3 of the MultiDA method aimed to induce differentiation and maturation of hMSCs toward a dopaminergic neuronal phenotype. Stage 3 medium was supplemented with 50 ng/mL GDNF and 200 µM AA 2-phosphate. Each stage was applied for 1 week. Additionally, different combinations of the stages were also examined over the 3 week differentiation period. Cultures were incubated at 37°C, 5% CO₂, and 20% O₂ with daily replenishment of growth factors and medium changed every 2 days. Cells were subcultured at the end of each stage (weekly) with 0.05% trypsin/0.53 mM EDTA in the same way as for subculturing neuronalinduced hMSCs (Section 2.2.1.4). Cell counts and viability analysis were performed using trypan blue dye exclusion. Dopaminergic neuronal-induced hMSCs were harvested for characterisation at the end of each stage of differentiation. The MultiDA method was used for differentiation of 3 hMSC cultures obtained from different donors.

2.2.1.6 Induction of Dopaminergic Neuronal Differentiation Via Published Protocol – Single-Stage Cytokine-Based Method (SingleDA)

Differentiation was induced in exponential growth phase hMSCs (P5-6) from 3 individual donors, using a cytokine-based single-stage dopaminergic neuronal differentiation (SingleDA) protocol that had recently been published (Trzaska *et al.*, 2007). Briefly, hMSCs were seeded into 6 well plates at a density of 1 x 10⁵ cells per plate, with MSC expansion medium. After 1 day, the expansion medium was replaced with dopaminergic neuronal differentiation medium, containing Neurobasal medium, 0.25x B-27 supplement, 250 ng/mL SHH, 100 ng/mL FGF-8 (mouse), and 50 ng/mL FGF-2. Cells were cultured at 37°C, 5% CO₂, and 20% O₂, and media was not replaced during the induction period. Dopaminergic neuronal-induced hMSCs were harvested for characterisation after 6 days and 12 days of differentiation, and cell proliferation and viability were assessed by trypan blue dye exclusion. The SingleDA method was used for differentiation of 3 hMSC cultures obtained from different donors.

Table 2.1 summarises the different neuronal differentiation methods employed in the present study.

2.2.1.7 Cell Lines

The cell lines SK-N-SH, HUH7, U-87 MG, HepG2, and MG-63 were used as positive controls and were cultured according to the recommended protocols.

2.2.2 Flow Cytometric Analysis

2.2.2.1 Detection of Apoptosis

For detection of apoptosis, hMSCs (4 cultures) were harvested at P2, P5, P8 and P11 by trypsinisation, washed with 10% FBS in HBSS, and resuspended in cold dPBS. hMSCs were stained with Annexin V-FITC (early apoptosis) and Propidium Iodide (PI;

Procedure	Stage	Harvest*	Cytokines	Surface Coating	Medium
SingleND	1	D7, 14, 21	10 ng/mL FGF-2 + 10 ng/mL EGF + 1 ng/mL PDGF	Fibronectin	DMEM/F12 + N-2
MultiDA	1 2^ 3 [~]	D7 D7 D7	10 ng/mL FGF-2 + 10 ng/mL EGF 100 ng/mL SHH + 10 ng/mL FGF-8 50 ng/mL GDNF	Poly-L-ornithine + Laminin	DMEM/F12 w/o HEPES + N-2 AA (Stage 2 & 3)
SingleDA	1	D6, 12	250 ng/mL SHH + 100 ng/mL FGF-8 + 50 ng/mL FGF-2	None - hMSCs allowed to adhere prior to induction	Neurobasal + B-27

Table 2.1 Summary of the Neuronal Differentiation Procedures

D, day

*cells were harvested following application of the indicated cytokines for the indicated number of days

^Stage 2 applied following induction with Stage 1 conditions

[~]Stage 3 applied following induction with Stage 2 conditions

late apoptosis) using Annexin V-FITC Apoptosis Detection Kit I, and following the manufacturer's recommendations. Cells (>10,000) were acquired and analysed on a FACSCalibur flow cytometer with Cell-Quest software.

2.2.2.2 Characterisation of Surface Antigen Expression

For characterisation of surface antigen expression, hMSCs were harvested at P4 and P8 by trypsinisation, washed with 10% FBS in HBSS, and resuspended in cold dPBS as reported previously (Tao *et al.*, 2005). hMSCs were incubated with normal human AB plasma (1:10) at 4°C for 30 min, then washed with FACS buffer (dPBS, 13.6 mM tri-sodium citrate and 1% BSA), and labeled with FITC-, PE- or PerCPconjugated antibody in the dark at 4°C for 30 min. Antibodies used were CD29, CD44, CD73, CD105, CD166, CD14, CD34, CD45, HLA Class I and HLA Class II. Cells were washed, resuspended in FACS buffer and >10,000 cells were acquired and analysed on a FACSCalibur flow cytometer with Cell-Quest software. This experiment was performed twice (n = 2). Similar findings were observed in additional experiments performed by Jean Hsu (St Vincent's Hospital Sydney, NSW Australia).

2.2.2.3 Statistical Analysis

For the apoptosis detection assay, data are presented as mean \pm SEM. Statistical analysis was performed using the Wilcoxon Signed Ranks Test to determine the equality of the means between paired time points P2–P5, P2–P8 and P2–P11. The significance level was set at 0.05.

2.2.3 RNA Extraction

Total RNA was extracted from undifferentiated hMSC cultures at P2, P5, P8 and P11, and hMSC cultures pre- and post-differentiation using TRIzol reagent and RNeasy Micro kit. Cells were harvested by trypsinisation, as described previously, and

resuspended in cold dPBS, followed by centrifugation and removal of all the supernatant, including the last drop using a pipette. Cold TRIzol reagent was added to the cell pellet, at a volume of 500 µl TRIzol for every 1 x 10⁶ cells, and the solution was resuspended by pipetting. Samples were stored at -80°C or immediately utilised for RNA extraction. For RNA extraction, samples were incubated at room temperature for 2-3 min, before addition of 20% volume of chloroform and mixture by shaking. Samples were centrifuged at 13,000 rpm for 15 min at 4°C. The top supernatant layer was immediately removed and an equal volume of room temperature 70% ethanol was added, mixed by pipetting, and immediately transferred to an RNeasy spin column. The manufacturer's recommendations were followed for the remainder of this procedure. To exclude genomic DNA contamination, samples were also treated with RNase-free DNase I, in accordance with the manufacturer's protocol.

Total RNA was also isolated from cell lines SK-N-SH, U87, HUH7, HepG2, and MG-63, for use in primer optimisation and as positive controls in RT-PCR, since these cell types are known to express the genes of interest. Other positive control samples included RNA obtained from human brain, chondrocytes and ESCs.

The total RNA extracted from samples was quantified using a Nanodrop spectrophotometer, and samples were aliquoted into 1-2 μ g aliquots and stored at -80°C for use in cDNA synthesis reactions.

2.2.4 Real-Time RT-PCR Analysis

In order to examine the expression of markers of hMSCs, pluripotency, and the three germ lineages, as well as genes indicative of neural and dopaminergic neuronal development, real-time RT-PCR was performed. Table 2.2 lists the genes that were examined for mRNA expression by real-time RT-PCR, as well as the corresponding forward and reverse primers used.

Gene	GenBank	Sequence (5'-3')		Product	Positive				
	Accession			Size	Control				
	Number	Sense	Antisense	(bp)					
Pluripoten	cy Markers								
POU5F1	NM_002701	ctcaccctgggggttctatt	agettectecacecaettet	129	hESCs				
NANOG	NM_024865	aactggccgaagaatagcaa	catccctggtggtaggaaga	86	hESCs				
Ectoderma	Ectodermal Lineage Markers								
NES	NM_006617	tccaagacttccctcagctt	tcaggactgggagcaaagat	145	SK-N-SH				
MAP-2	NM_002374	agaccaccattgacgactcc	tctccgagcttccttttcag	134	SK-N-SH				
CSPG4	NM_001897	gaaggaggacggacctcaa	gatccatctcggaggcatta	143	U-87 MG				
GFAP	NM_002055	atcgagatcgccacctacag	caccacgatgttcctcttga	150	Brain				
ENO2	NM_001975	aggccagatcaagactggtg	caagcagaggaatcacagca	148	Brain				
Endoderm	al Lineage Ma	rkers							
FOXA2	NM_021784	cgctctccttcaacgactgt	tagcagccgttctcgaacat	115	HUH7				
ALB	BC039235	acattcaccttccatgcaga	aaagctgcgaaatcatccat	146	HepG2				
Mesodermal Lineage Markers									
CD44	NM_000610	aagacacattccaccccagt	ggttgtgtttgctccacctt	98	hMSCs				
COL1A1	NM_000088	gagagcatgaccgatggatt	atgtaggccacgctgttctt	149	MG-63				
BGLAP	NM_199173	ggcgctacctgtatcaatgg	tcagccaactcgtcacagtc	106	MG-63				
COL2A1	NM_033150	gtgacaaaggagaggctgga	acctctagggccagaaggac	146	Chondrocyte				
ACAN	NM_013227	tcaacaacaatgcccaagac	aaagttgtcaggctggttgg	128	Chondrocyte				
ADIPOQ	NM_004797	cctggtgagaagggtgagaa	ctcctttcctgccttggatt	124					
LPL	NM_000237	gtggccgagagtgagaacat	gaaggagtaggtcttatttgtggaa	66					
Dopamine	rgic Neuronal	Development Transcrip	tion Factors						
NR4A2	NM_006186	gctgttgggatggtcaaaga	ctgtgggctcttcggtttc	87	Brain				
PITX3	NM_005029	gctaccccgacatgagcac	gcgaagctgcctttgcat	148	Brain				
EN2	NM_001427	gcgtgggtctactgtacgc	tgtcctctttgttcgggttc	91	Brain				
Dopaminergic Neuronal Markers									
TH	NM_000360	cagccctaccaagaccagac	gtacgggtcgaacttcacg	129	Brain				
VMAT2	NM_003054	ctttttgcccctctctgctt	gggcagttgtgatccatga	83	Brain				
DAT	NM_001044	tcgagagaaactggcctacg	aggatgacttcctggggtct	138	Brain				
Ubiquitous Marker									
HPRT1	NM_000194	gaccagtcaacaggggacat	cctgaccaaggaaagcaaag	132					
GAPDH	NM_002046	aatcccatcaccatcttcca	tggactccacgacgtactca	82					

Table 2.2 Primer Pairs for Real-Time PCR

hESCs, human embryonic stem cells hMSCs, human mesenchymal stem cells

2.2.4.1 Primer Design and Optimisation

A three-step procedure was utilised for the design and optimisation of primers used in this study:

 Identification of coding sequences from NCBI (Entrez Nucleotide) – http://www.ncbi.nlm.nih.gov/.

2) Design of sense and anti-sense primers using the Primer3 program (<u>http://frodo.wi.mit.edu/cgi-bin/primer3/primer3_www.cgi</u>) with subsequent NCBI BLAST (blastn) performed for designed primers to check potential homology with other human genes. In addition, to exclude genomic DNA contamination, primers were designed to be intron-spanning.

3) Optimisation by real-time RT-PCR. Primer optimisation was carried out using cDNA from the positive control for the respective genes.

2.2.4.2 cDNA Synthesis

Total RNA (1 μ g) was reverse transcribed into cDNA using the SuperscriptTM III First Strand Synthesis Kit, with 25 ng random hexamer and 1.25 μ M Oligo(dT)₂₀ primers, according to manufacturer's recommendations. Following cDNA synthesis, samples were treated with RNase H for 20 min at 37°C. The cDNA was stored at -80°C and diluted 1:20 for use in real-time RT-PCR.

2.2.4.3 Real-Time RT-PCR Procedure

Real-time RT-PCR was performed to examine gene expression using the Platinum SYBR Green qPCR SuperMix-UDG on a Rotor-Gene RG3000 machine. The PCR reaction contained 9 μ L of diluted (1:20) cDNA, 1x Platinum SYBR Green qPCR SuperMix-UDG, 0.25 μ M of each primer (or 0.125 μ M of each primer for *PITX3*), and RNase-free water to make a final volume of 20 μ L. All reactions were performed in

duplicate within the same PCR plate. The thermal profile for all reactions was: 2 min at 50°C, 2 min at 95°C, 40 cycles of 30 seconds (s) at 95°C, 30 s at 60°C and 30 s at 73°C. Melt curve analysis was also performed with ramp from 72°C to 95°C, rising by 1°C in each step (the first step being 45 s and each step afterwards 5 s).

Quantitation analysis was performed using the Rotor-Gene software with Dynamic Tube and Slope Correct options selected, threshold set at 0.007, and elimination of cycles before 7. For experiments with control untreated and treated samples the ratio of the target gene expression in experimental/control ('fold change in target gene'/'fold change in reference gene') was determined using the $\Delta\Delta$ Ct method (Livak *et al.*, 2001). For other experiments target gene expression was expressed as a ratio relative to the reference gene. Melt curves were also assessed for the presence of possible non-specific products and primer dimer.

The reference genes used were *HPRT1* (hypoxanthine phosphoribosyltransferase 1) and *GAPDH* (glyceraldehyde-3-phosphate dehydrogenase). Water amplification controls were included in each reaction to control for contaminating cDNA and genomic DNA.

2.2.4.4 Electrophoresis of PCR Products

Following primer optimisation the size of PCR products were confirmed using gel electrophoresis. PCR products were visualised following separation in a 3% (w/v) agarose gel prepared in 1x TAE buffer (40 mM Tris-acetate, 1 mM EDTA) with 1 μ g/mL ethidium bromide. PCR product was mixed with 1/6th volume of 6x DNA loading dye (40% (w/v) sucrose, 0.25% (w/v) bromophenol blue, 0.25% (w/v) xylene cyanol in Baxter water). GeneRuler Ultra Low Range 10-300 bp DNA ladder was loaded in parallel with samples to approximate the size of PCR products.

Electrophoresis was conducted in 1x TAE buffer at 100 volts. Gel images were captured digitally by transillumination using the Gel Doc-1000 photographic system.

2.2.4.5 Statistical Analysis

Gene expression data were prepared as figures using computer package GraphPad Prism. Data obtained were expressed as mean fold change in mRNA expression + SEM relative to control undifferentiated hMSCs. The student's *t*-test was performed to determine significant differences using Microsoft Excel, with statistical significance set at p < 0.05.

2.2.5 In Vitro Morphological Analysis

2.2.5.1 Routine Morphological Recordings

For morphological analysis hMSCs cultured in expansion and differentiation conditions were examined on a daily basis and observations were recorded. Phasecontrast images were routinely captured prior to subculturing or harvesting using an Olympus upright light microscope and digital camera. Cells undergoing neuronal differentiation procedures were imaged more frequently at least every 2 days.

2.2.5.2 Live Cell Imaging

To further examine the morphological changes induced in hMSCs by neuronal differentiation, cells were differentiated for 3 weeks using the SingleND method, trypsinised, and cultured in chamber slides for examination by live cell imaging. Images were captured immediately after re-adherence of cells (commencing at 30 min post-trypsinisation) using a Zeiss inverted microscope equipped with a CO₂- and temperature-controlled stage. Images were recorded at 5 min intervals for periods longer than 18 hours.

2.2.6 Immunofluorescence Microscopy of In Vitro Cultured Cells

Human MSCs cultured on chamber slides (BD Falcon) or glass coverslips were fixed with 4% PFA (warmed to 37°C) at room temperature for 20 min. For intracellular antigens, cells were permeabilised using 100% DMSO for 10 min, and then washed 3x with 0.1% Triton X-100 in phosphate buffered saline (PBS; used for all washes unless otherwise stated; each wash was 5 min long for all washes). Slides/coverslips were blocked with normal goat serum (10%) in PBS for 1 hour, followed by incubation with primary antibodies at room temperature for 1 hour. Primary antibodies included: mouse monoclonal antibodies (all IgG unless specified) against POU5F1 (1:50), NES (1:200), SSEA-1 (1:50; IgM), SSEA-4 (1:50), TRA-1-60 (1:50; IgM), CD44 (1:100), MAP-2 (1:100), A2B5 (1:200; IgM), NeuN (1:200) and TH (1:1000); and rabbit polyclonal antibodies against Fibronectin (1:400), FOXA2 (1:100), GFAP (1:200) and β tubulin III (1:2000). Slides/coverslips were washed 3x and then incubated with Alexa Fluor 488 or 594 highly cross adsorbed goat anti-rabbit or anti-mouse secondary antibodies at 1:400 dilution for 1 hour. Negative controls were performed in all cases, consisting of secondary antibody application in the absence of primary antibody. Slides/coverslips were washed 3x, after which nuclei were counterstained with 300 nM DAPI in PBS for 10-15 min, followed by a final wash with PBS (1 min; 3x) and mounting in 100% glycerol. Cells were examined and imaged using a Zeiss inverted fluorescent microscope with Axiocam digital camera and Axiovision software. For undifferentiated hMSCs, immunofluorescence analysis was performed on cultures from 4 different donors. For all neuronal differentiation procedures, immunofluorescence analysis was performed on 3 hMSC cultures, with control undifferentiated hMSCs from the same cultures examined in parallel.

2.2.7 Cell Preparation and Stereotactic Intracerebral Transplantation

The *in vivo* experiments in my thesis were performed at the National Centre for Adult Stem Cell Research, Griffith University (Brisbane, QLD) in the laboratory of my co-supervisor Prof. Alan Mackay-Sim. Additional supervision was also provided by Dr. Adrian Meedeniya, and technical assistance was received from Maria Nguyen, Mary Pagendam-Turner, Brenton Cavanagh, Joseph Kan and Nathalie Romond. Preparation of undifferentiated and neuronal-primed (partially differentiated) hMSCs and OECs for transplantation, as well as immunohistological analysis of graft sites was performed by me, primarily at St Vincent's Hospital Sydney, and also at Griffith University.

2.2.7.1 Animal Welfare

Animal studies were performed under approval from the Animal Ethics Committee of Griffith University (Brisbane, QLD), and in accordance with the Australian Code of Practice for the Care and Use of Animals for Scientific Purposes. Animal subjects were housed in pairs with a 12 hour light/dark cycle and with food and water *ad libidum*.

2.2.7.2 Experimental Design

Part I of this study employed the use of 6-OHDA unilaterally-lesioned adult female Sprague-Dawley rats (Taconic Farms, NY, USA), weighing 250-350 g at the start of the experiment, for all surgeries. This source of animals was selected as successful published studies had also used these animals for the same purposes (Roy *et al.*, 2006; Sanchez-Pernaute *et al.*, 2001). Animal subjects received single intrastriatal grafts of undifferentiated hMSCs. Amphetamine-induced rotational behaviour was assessed once prior to transplantation and was intended to be assessed every 4 weeks after transplantation. Graft phenotype was intended to be examined at 3 months posttransplantation, however due to the early experimental endpoint immunohistological
analysis of graft sites was performed at a little over 2 months post-transplantation. Table 2.3 outlines the experimental plan for Part I.

Part II was performed to address issues raised in Part I. For Part II, an alternative source, sex and strain of animal subjects was utilised. Adult male 6-OHDA unilaterally-lesioned Wistar Ob rats weighing 250-350 g were obtained from a source within Australia, the Integrative Neuroscience Facility (Howard Florey Institute, VIC, Australia). Transplantation into the striatum, as well as the substantia nigra was performed, and hMSCs underwent cytokine-based neuronal-priming prior to transplantation (Stage 1 of the MultiDA method; see Section 2.2.1.5). The experiment was conducted over a shorter period of 21 days, and co-transplantation with OECs was also performed. Table 2.4 describes the experimental plan for Part II.

2.2.7.3 Preparation of hMSCs for Transplantation

Both undifferentiated and neuronal-primed hMSCs were prepared at St Vincent's Hospital Sydney and transported to Griffith University for further culturing and transplantation. Undifferentiated hMSCs were expanded in culture as described in Section 2.2.1.2, until the middle of P6 when cells were transported to Griffith University. For transportation, T25 flasks containing adherent hMSCs were filled to the neck with DMEM-LG medium containing 10% FBS, 100 U/mL penicillin and 100 μ g/mL streptomycin. Flasks were then sealed and stored upright in an esky during transportation (approximately 6 hours). Cells were returned to expansion culture conditions upon arrival at Griffith University.

Neuronal-primed hMSCs were obtained following exposure of undifferentiated P5 hMSCs to Stage 1 of the MultiDA method (Section 2.2.1.5) for 1 week and then harvested by trypsin/EDTA (described in Section 2.2.1.4) for transportation. Cells were resuspended in complete MultiDA – Stage 1 medium at approximately 50,000 cells

Table 2.3 Experimental Plan for Part I

Time Point	Undifferentiated hMSCs	Sham	
3 months	8	8	

NB: Transplantation into striatum only

Table 2.4 Experimental Plan for Part II

Time Point	PhMSCs	PhMSCs + OECs	Sham
Day 1	3	3	3
Day 7	3	3	3
Day 21	3	3	3

NB: Transplantation into both striatum and substantia nigra PhMSCs, Neuronal-Primed hMSCs per mL and transferred into T25 culture flasks, which were sealed and stored upright in an esky during transportation. Cells were returned to Stage 1 differentiation conditions upon arrival at Griffith University.

Prior to transplantation, cells were harvested by trypsinisation as described in Sections 2.2.1.2 and 2.2.1.4, and washed twice with HBSS. Undifferentiated hMSCs were resuspended at a concentration of 30,000 cells per μ L in HBSS, and 3 μ L of the cell suspension were injected into the lesioned striatum (total of 90,000 cells per rat). Neuronal-primed hMSCs were resuspended at 50,000 cells per μ L in HBSS, and 2 μ L of the cell suspension were injected into the striatum and substantia nigra of the lesioned hemisphere (total of 100,000 cells per site).

2.2.7.4 Preparation of OECs for Transplantation

OECs were prepared from frozen stocks available at the National Centre for Adult Stem Cell Research, generated with the previously described protocol (Bianco *et al.*, 2004). OECs were maintained in DMEM/HAM F12 medium containing 10% foetal calf serum, and penicillin/streptomycin (100 U/mL and 100 μ g/mL respectively), and passaged at 80% confluency. For transplantation, OECs were harvested by trypsinisation, using the same procedure as for trypsinising hMSCs, and washed twice with HBSS. Subsequently, OECs were resuspended at 25,000 cells per μ L in HBSS together with the neuronal-primed hMSCs (total of 50,000 OECs per site).

2.2.7.5 Stereotactic Intracerebral Transplantation in Rat PD Model

Surgical procedures were performed by Dr. Adrian Meedeniya and Mary Pagendam-Turner. Briefly, rats were anaesthetised using isofluorane gas in an induction chamber, and anaesthesia was maintained during surgery via a nose cone. Rats were placed in a small animal stereotaxic frame with microinjector unit, and cell suspensions were injected into the lesioned hemisphere of each subject according to the following co-ordinates: undifferentiated hMSCs into the striatum at AP: 1.2, L: -2.5, V: 5.0, and neuronal-primed hMSCs and OECs into the striatum at the same co-ordinates as above, as well as the substantia nigra at AP: -5.2, L: -3.2, V: -7.4. Cells were injected with a 27-gauge injection needle connected to a Hamilton syringe. The needle was inserted at the injection co-ordinates and a cavity was created for the injection through overshooting by 200 μ m. After waiting 2 min the cell suspension was injected at 1 μ L per min. Following infusion, the needle was left in place for 2 min and then slowly withdrawn. Sham-operated animals were treated in the same manner, but HBSS was infused alone without cells. Transplant groups and harvest time points are shown in Table 2.3 and 2.4. All subjects were immunosuppressed with daily injections of cyclosporine A (10 mg/kg; subcutaneous) from 3 days prior to grafting until harvest.

2.2.7.6 Tissue Processing

Tissue processing procedures were performed by Maria Nguyen and Joseph Kan. Subjects were anaesthetised with Lethabarb-Sodium Pentabarbitone (1.5 mL per animal) and perfused transcardially with PBS followed by Zambonie's Fixative. Brains were harvested and immersed in Zambonie's Fixative overnight under vacuum, then rinsed twice in PBS for 30 min, and placed in PBS-azide (0.1%). Specimens were dehydrated through graded alcohols (50% ethanol for 30 min, 70% ethanol for 30 min, and finally 100% ethanol for 60 min), permeabilised with DMSO for 60 min or until sunken (under vacuum), rinsed in 100% ethanol for 5 min, and placed in fresh 100% ethanol for a further 60 min. Brains were embedded in PEG by placing in PEG 400 overnight at room temperature under vacuum, PEG 1000 at 48°C under vacuum until sunk (approximately 60 min), and embedded in PEG 1450/1000 mix at a ratio of 4:1. Sectioning was performed at room temperature on a rotary

microtome to obtain 30 μ m thick sections, which were placed 10 per well in PBS-azide (0.1%) and stored at 4°C.

2.2.8 Immunohistological Analysis

Characterisation of graft sites was performed on free-floating brain sections in 24 well plates using the same procedure as described in Section 2.2.6, with some modifications. These modifications included: incubation and washing steps were all performed with agitation on a shaker; washes were all performed for 10 min; permeabilisation with 100% DMSO for 20 min; overnight incubation with primary antibodies at 4°C; and incubation with secondary antibodies for 3 hours. Additional primary antibodies used were rabbit anti-Iba-1 (1:2000), mouse anti-GFAP-Cy3 (1:400), and mouse anti-Human Nuclear Antigen (HNA; 1:100). Furthermore, the permeabilisation step for HNA staining was performed with 0.1% Triton X-100 in PBS, rather than 100% DMSO, for 20 min.

Confocal imaging was performed using a Leica laser scanning confocal microscope (DM IRE2 TCS SP2 AOBS). Individual colour channels were captured separately and merged in Adobe Photoshop. Co-localisation of HNA staining with DAPI-positive nuclei was confirmed through analysis of z-series confocal reconstructions and corresponding orthogonal planes under 40x and 100x. Contralateral brain regions served as internal controls to confirm primary antibody specificity. Images presented are 'maximum projection images' from z-stacks, with 10 sections captured over the thickness of the brain section (30-45 μ m). Three-dimensional rendering of graft sites in Part I was prepared by Brenton Cavanagh using Bitplane Imaris software from images captured using a Zeiss Axio Imager Z1 with ApoTome and Axiovision software.

2.2.9 Amphetamine-Induced Rotation Testing

Behavioural testing was conducted by Maria Nguyen. Briefly, rotational behaviour was measured prior to transplantation in Part I and II to confirm the lesioning of nigrostriatal dopaminergic neurons, and was intended to be assessed every 4 weeks post-transplantation of undifferentiated hMSCs in Part I. Rats were placed in a Rotometer and allowed to habituate prior to injection of D-amphetamine (4 mg/kg, intraperitoneal). Net rotation over a period of 90 min was recorded, commencing from 15 min post-injection. All behavioural testing was conducted in a closed room with a controlled environment and no exposure to loud or sudden noises or lighting changes.

CHAPTER 3

CHARACTERISATION OF HUMAN BONE MARROW-DERIVED MESENCHYMAL STEM CELLS AND THE EFFECT OF LONG-TERM SERIAL PASSAGE

3.1 Introduction

Since the identification of MSCs in the BM by Friedenstein and colleagues (Friedenstein *et al.*, 1970; Friedenstein *et al.*, 1968; Friedenstein *et al.*, 1966), increasing interest in these cells has been stimulated by the potential use of MSCs in regenerative and reparative therapies. The characteristic feature that most uniquely distinguishes MSCs is the ability to differentiate into multiple mesodermal cell types, typically osteocytes, adipocytes and chondrocytes (Pittenger *et al.*, 1999; Prockop, 1997). However, this property does not allow prospective MSC isolation, and consequently, the identification of MSCs has been restricted.

Due to the low frequency of MSCs in BM (Bruder *et al.*, 1997; Reyes *et al.*, 2001), it is necessary to expand MSCs extensively *in vitro* to acquire sufficient cells for use in research and clinical trials, and potentially for therapeutic applications. Various expansion methods exist, which are often inconsistent, and the effects of long-term serial passage are not well defined among the different protocols. Several reports on the extensive subcultivation of BM-derived MSCs have described changes in morphology, proliferation, telomere length and mesodermal differentiation capacity (Baxter *et al.*, 2004; Bruder *et al.*, 1997; Digirolamo *et al.*, 1999; Mets *et al.*, 1981). In particular, it was observed that while MSCs retained osteogenic differentiation potential through 10-15 passages, adipogenic potential was lost (Digirolamo *et al.*, 1999). In addition, spontaneous osteogenic differentiation with deposition of mineral was observed in some cultures (Digirolamo *et al.*, 1999). These findings suggest that the use of MSCs at different passages may contribute to discrepancies observed between studies.

In vitro and *in vivo* studies have shown that certain BM-derived MSC populations are capable of multi-lineage differentiation. Under select culture conditions, MSCs have been reported to differentiate into ectodermal derivatives including neuronal

and glial cells (Hermann *et al.*, 2004; Jiang *et al.*, 2003; Jiang *et al.*, 2002; Sanchez-Ramos *et al.*, 2000; Tao *et al.*, 2005; Woodbury *et al.*, 2000), as well as endodermal derivatives, such as hepatic cells (Jiang *et al.*, 2002; Lee *et al.*, 2004; Schwartz *et al.*, 2002; Shi *et al.*, 2005; Wang *et al.*, 2004). Transplantation of MSCs into post-natal animals has yielded engraftment and differentiation of donor-derived cells into lung, liver and intestinal epithelium, whereas introduction into early blastocysts has resulted in contribution to most somatic tissues (Jiang *et al.*, 2002). Furthermore, expression of genes indicative of the three germ lineages have been detected in undifferentiated MSCs (Minguell *et al.*, 2005; Seshi *et al.*, 2003; Tondreau *et al.*, 2004b; Woodbury *et al.*, 2002), providing further evidence for a 'multidifferentiated' (Woodbury *et al.*, 2002) or 'pluridifferentiated' (Seshi *et al.*, 2003) MSC state.

Recognition of the potential for BM-derived MSCs to act as a renewable cell source for cellular therapies has led to investigations aimed at defining the optimal conditions required for differentiation towards particular lineages. However, it is not known whether BM-derived hMSCs simultaneously express pluripotency and lineagespecific genes and proteins, or whether long-term *in vitro* expansion culture affects the 'pluridifferentiated' nature of hMSCs.

This study aimed to address these issues by investigating the effects of long-term serial passage on the characteristics of BM-derived hMSC populations by concurrently assessing growth kinetics, apoptosis, expression of cell surface antigens, and gene and protein expression. Since hMSCs have been reported to possess multi-lineage differentiation potential, the expression of markers of pluripotency and the three germ lineages will be examined to determine whether the expression profile is suggestive of a 'pluridifferentiated' phenotype, and also whether there exists an optimal period to initiate differentiation towards particular lineages. Furthermore, these studies would allow the establishment of a baseline expression profile of undifferentiated hMSCs prior to the commencement of differentiation.

3.2 Results

3.2.1 MSC Morphology and Growth Kinetics

Primary colonies of hMSCs at P0 contained small raised cells with a fibroblastlike appearance (Figure 3.1A). With continued growth *in vitro*, cells gradually gained a larger and more flattened morphology, but maintained a consistent appearance from P1 up to P6-8 (approximately 8-10 weeks; Figures 3.1B and C). Beyond this period (up to 19 weeks), the cells were large, flat, and difficult to harvest by trypsinisation. In addition, cultures contained fewer dividing cells and greater amounts of cell debris (Figure 3.1D).

The proliferative capability of hMSC populations was examined by calculating the number of population doublings at each subculture. A linear correlation between cumulative doubling and passage number was observed up to approximately P6-8 (8-10 weeks; Figure 3.2A), and in one case up to P10 (14 weeks; $R^2 = 0.9857$), indicating the presence of symmetric cell division. Beyond this period, a reduction in the proliferation rate was observed, which corresponded with the observed morphological changes, and suggested deviation toward asymmetric division.

Viability of hMSC populations at each subculture was consistently above 85% (Figure 3.2B). The Pearson correlation coefficient for viability versus passage number was 0.16 with a *p*-value of 0.31 (two-tailed test), indicating that cell viability was not affected by time in culture. This was confirmed using an apoptosis assay in which hMSC cultures were found to contain an average of 70-85% viable cells (Figure 3.2C) with no statistical difference observed across all time points (Wilcoxon Signed Ranks Test for paired data).



Figure 3.1 Morphology of hMSCs at early and late culture stages.

Representative phase-contrast micrographs showing morphology of hMSC cultures at (A) P0 (Day 14), (B) P5 (Day 64), (C) P8 (Day 85), and (D) P15 (Day 134). Cells gradually gain a larger and more flattened morphology with growth *in vitro*. A fairly consistent appearance is maintained from P1 to P8. Arrows indicate dividing cells. Scale bar: 100 μ m.



Figure 3.2 Proliferation and viability of hMSCs at early and late stages of culture.

(A) Cumulative population doubling of hMSCs from 2 representative donors. Cumulative doubling is linearly correlated to days in culture (Dn1: $R^2 = 0.9857$), indicating the presence of symmetric division. Each data point represents the number of doublings obtained at the end of the indicated passage. (B) Cell viability by trypan blue dye exclusion performed at each subculture. Viability of hMSC populations is consistently >85%, and is not correlated to days in culture (R = 0.159). (C) Detection of apoptosis by Annexin V-FITC and PI staining with flow cytometric analysis (mean ± SEM; n = 4 donors). P, passage; Dn, donor.

3.2.2 Immunophenotype of Human MSCs by Flow Cytometry

Undifferentiated hMSCs in expansion cultures were examined at P4 and P8 for the expression of cell surface antigens commonly used in the characterisation of hMSC populations. Cells expressed the expected pattern of surface antigens for human MSCs (CD44, CD73, CD105, CD166, CD29 and HLA Class I), and lacked expression of haematopoietic markers CD45, CD14 and CD34, as well as HLA Class II (Figure 3.3). The pattern of surface antigen expression was maintained consistently in cultures up to P8.

3.2.3 Multi-Lineage Mesodermal Differentiation

The MSC identity of cells isolated in this study was confirmed with the ability to differentiate into chondrocytic, adipocytic, and osteocytic mesodermal lineages. Chondrocytic differentiation was evident in micromass pellet cultures after 3 weeks in differentiation conditions, with proteoglycan deposition detected in >70% of the total area by Alcian Blue staining (Figure 3.4D; control in Figure 3.4A), and increased *COL2A1* and *ACAN* gene expression (Figure 3.5). Differentiation into the adipocyte lineage was apparent in >65% of cells from the intracellular accumulation of lipid-laden vacuoles that positively stained with Oil Red O (Figure 3.4E; control in Figure 3.4B). Expression of adipocytic genes, *ADIPOQ* (adiponectin) and *LPL* (lipoprotein lipase), were also greatly increased post-differentiation (Figure 3.5). Osteogenic differentiation was evident by Alkaline Phosphatase staining in >60% of the total area (Figure 3.4F; control in Figure 3.4C), as well as increased *COL1A1* and *BGLAP* (osteocalcin) gene expression (Figure 3.5).



Figure 3.3 Flow cytometric detection of cell surface markers of undifferentiated hMSCs.

Representative results demonstrating that hMSCs express the expected pattern of surface antigens for human BM-derived MSCs. In addition, the pattern of expression is consistent across P4 to P8 of culture.



Figure 3.4 Histochemical staining of hMSCs following *in vitro* multilineage mesodermal differentiation.

Images shown depict undifferentiated control hMSCs (A-C) and hMSCs cultured in mesodermal differentiation conditions (D-F). (**A** and **D**) Alcian Blue staining (with Fast Red nuclear counterstain) shows evidence of proteoglycan deposition after chondrogenic differentiation. (**B** and **E**) Adipogenic differentiation of hMSCs results in accumulation of lipid-laden vacuoles that stain with Oil Red O. (**C** and **F**) Increased expression of alkaline phosphatase was detected with osteogenic induction. Scale bar: 100 μ m.



Figure 3.5 Gene expression analysis of hMSCs following *in vitro* multi-lineage mesodermal differentiation.

Representative real-time RT-PCR results showing the fold change in gene expression relative to undifferentiated control hMSCs after treatment with osteogenic, adipogenic, and chondrogenic differentiation conditions. Ctrl, control; Chondro, chondrocytic; Adipo, adipocytic; Osteo, osteocytic.

3.2.4 Expression of Pluripotency and Germ Lineage Markers

Earlier reports of MSC multi-lineage differentiation potential prompted me to examine the expression of markers of pluripotency and the three germ lineages to determine whether the expression profile of undifferentiated hMSC cultures was consistent with a 'pluridifferentiated' phenotype. Cultures were also investigated at different stages (P2, P5, P8, and P11) in order to detect any spontaneous changes in expression during long-term culture.

3.2.4.1 Gene Expression by Real-Time RT-PCR

The hMSC cultures were found to express genes typical of MSCs, pluripotent cells and mesodermal and ectodermal lineages by real-time RT-PCR. Of the markers examined, mesodermal genes *COLIA1* and *CD44* were the most highly expressed (Figure 3.6A), followed by neuroectodermal genes *CSPG4* (chondroitin sulfate proteoglycan 4) and *NES* (Figure 3.6B). The decrease in *CSPG4* expression at P8 seen in this sample was not observed across other samples. Pluripotency genes, *POU5F1* and *NANOG* (Nanog homeobox), were also detected, with *NANOG* expressed at lower levels than *POU5F1* (Figure 3.6C). Morphologically undifferentiated hMSC cultures also expressed neuronal gene *MAP-2* at low levels (Figure 3.6C); astroglial gene *GFAP* was very weakly expressed or absent, whereas endodermal genes *FOXA2* and *ALB*, and mesodermal gene *COL2A1*, could not be detected (results not shown). Expression of these genes did not appear to vary greatly during long-term serial passage over P2 to P11.



Figure 3.6 Gene expression analysis of undifferentiated hMSC cultures during long-term serial passage.

Representative results of gene expression analysis by real-time RT-PCR relative to *HPRT1* expression (normalised to 100%; n = 4). The most highly expressed genes were (A) mesodermal *COL1A1* and *CD44*, followed by (B) ectodermal neural genes *CSPG4* and *NES*. (C) Weak/medium expression was detected for pluripotency genes *POU5F1* and *NANOG*, and neuronal gene *MAP-2*.

3.2.4.2 Protein Expression by Immunofluorescence Microscopy

The results of protein expression analysis supported those obtained by real-time RT-PCR. Human MSCs strongly expressed CD44 and fibronectin, markers of MSCs and the mesodermal lineage, consistently throughout expansion in culture (Figure 3.7 and Table 3.1). Pluripotent SC transcription factor POU5F1 and ectodermal neural progenitor marker NES were weakly expressed at both early and late stages of culture (Figure 3.7); however, the proportion of cells expressing these markers decreased over time (Table 3.1). Pluripotency markers SSEA-4 and TRA-1-60, and differentiation marker SSEA-1, were not detected at any stage of culture (results not shown). Endodermal markers, FOXA2 and ALB, were also undetectable (results not shown).



Figure 3.7 Protein expression analysis of undifferentiated hMSC cultures during long-term serial passage.

Images showing protein detection in hMSC cultures by immunofluoresence microscopy at P2 and P8 after labelling with primary antibodies against CD44, Fibronectin, NES, and POU5F1 (n = 4). Cells were only included as positive if pattern of staining was localised to the: cytoskeleton for NES; nucleus for POU5F1; cell surface for CD44; or extracellular and fibre-like in appearance for fibronectin. Negative controls consisted of secondary antibody application alone. Scale bar: 100 µm.

Table 3.1 Phenotypic Characterisation of Undifferentiated HumanMSCs by Immunofluorescence.

Antigenic Marker	Passage 2	Passage 5	Passage 8	Passage 11
CD44	+++++	+++++	+++++	+++++
Fibronectin	+++++	+++++	+++++	+++++
POU5F1	++++	+++	+	+
NES	+/++	+/++	+	+

Data shown summarises results for all hMSC cultures (n = 4). No expression of ALB, FOXA2, A2B5, MAP-2, TRA-1-60, SSEA-4 or SSEA-1 was detected at any passage.

+++++ : Intense positive staining (greater than 91% of cells positive).

- ++++ : Strong positive staining (71-90%).
- +++ : Intermediate positive staining (41-70%).
- ++ : Weak positive staining (21-40%).
- + : Very weak positive staining (0-20%).

3.3 Discussion

The findings of this study show that (1) human BM-derived MSC populations maintain a stable phenotype *in vitro* during the first 6-8 passages of expansion culture, as assessed by proliferative ability, morphological appearance and surface antigen, gene and protein expression; (2) hMSCs constitutively express mesodermal and neural lineage markers in culture, making it necessary to determine baseline expression levels prior to the commencement of differentiation; (3) long-term serial passage of hMSCs results in changes to some key phenotypic attributes of MSCs, including morphology, proliferative ability, and gene and protein expression; (4) extensive subcultivation of hMSC cultures does not result in spontaneous differentiation toward the neural lineage; and (5) hMSCs possess SC characteristics *in vitro*, with the ability to proliferate, symmetrically divide, and produce multi-lineage mesodermal derivatives.

3.3.1 Undifferentiated hMSCs Possess Characteristic Properties of MSCs

The cell population isolated in this study expressed cell surface antigens characteristic of MSCs and lacked HSC markers, in agreement with the standard MSC phenotype. As no single set of phenotypic markers is able to unequivocally identify MSCs, the cell population used in this study was shown to possess other MSC properties such as extensive proliferation, symmetric division, and multi-lineage mesodermal differentiation.

3.3.2 Undifferentiated hMSCs Express Ectodermal and Mesodermal Lineage Markers

Examination of germ lineage and pluripotency marker expression revealed that undifferentiated hMSC cultures constitutively expressed the expected mesodermal lineage markers, as well as ectodermal-specific markers of neurons (NES, *MAP-2*; also NF-M and NSE proteins were detected in a previous study by our group (Tao *et al.*, 2005)) and glia (*CSPG4*). Other recent studies have also shown that MSCs express markers characteristic of the neural lineages (Blondheim *et al.*, 2006; Deng *et al.*, 2006; Minguell *et al.*, 2005; Phinney *et al.*, 2006; Seshi *et al.*, 2003; Tondreau *et al.*, 2004b; Tremain *et al.*, 2001; Woodbury *et al.*, 2002), providing independent support of a pluripotent MSC state, or what some have named a 'multidifferentiated' (Woodbury *et al.*, 2002) or 'pluridifferentiated' (Seshi *et al.*, 2003) phenotype. In addition, Tremain and colleagues (Tremain *et al.*, 2001) have detected the expression of several neurotrophic factor mRNAs by hMSCs, and hypothesised that MSCs produce these factors normally to promote, maintain and augment the repair of neural networks within bone and BM. Collectively, these findings emphasise the need for examining neural marker expression prior to neural differentiation as well as post-differentiated hMSCs from four individual donors, which will be useful for comparison in later neuronal differentiation studies.

Expression of endodermal genes (Ceruloplasmin, IPP isomerase, Lanosterol-14a-demethylase) has also been reported in undifferentiated rat MSCs by microarray analysis (Woodbury *et al.*, 2002). However, in the present study expression of representative genes or proteins (FOXA2, ALB) from the endodermal lineage were undetectable or extremely low in most hMSC cultures. Similarly, a microarray analysis of human BM-derived MSCs (Seshi *et al.*, 2003) detected simultaneous expression of transcripts from multiple lineages and cell types, including osteoblast, muscle, fibroblast, adipocyte, epithelial cell, endothelial cell and neural cell, without mention of endodermal lineage expression. Therefore, it could be possible that this discrepancy

141

may be due to species-to-species variation or differences in culture maintenance and isolation.

3.3.3 Undifferentiated hMSCs Express Markers of Pluripotency

Human MSC cultures were also found to express POU5F1 and *NANOG*, both of which are transcription factors present in pluripotent ESCs. Expression of pluripotency and germline-specific genes (including protamine 2, *POU5F1*, *hTERT*, and ornithine decarboxylase antizyme) was recently reported in rat and human MSCs (Pochampally *et al.*, 2004b; Roche *et al.*, 2007; Tondreau *et al.*, 2005; Woodbury *et al.*, 2002). A high degree of cooperation has been found between POU5F1 and NANOG in regulating gene expression, and it is thought that these transcription networks contain common downstream targets important for the prevention of ESC differentiation (Loh *et al.*, 2006). In the present study other pluripotent SC markers SSEA-4 and TRA-1-60 were absent from hMSC cultures, and the levels of POU5F1 and *NANOG* expression were low.

The postnatal role of POU5F1 is still unclear, with a recent study showing that terminally differentiated human peripheral blood MNCs expressed POU5F1 (Zangrossi *et al.*, 2007), raising the possibility of a separate role for POU5F1 outside of embryonic tissue. Nonetheless, increased *POU5F1* and *REX-1* expression in hMSCs by culturing in low serum concentrations has been shown to result in increased differentiation efficiency toward osteogenic and adipogenic phenotypes (Roche *et al.*, 2007), suggesting that POU5F1 may indeed have a role in the multipotency of MSCs.

Further conflicting results have been reported showing expression of both *POU5F1* transcripts as well as pseudogenes by hMSCs in the absence of detectable protein expression, suggesting that POU5F1A does not play a major role in MSC multipotency (Kaltz *et al.*, 2008). More recent studies have demonstrated that different

142

isoforms of POU5F1 and a number of pseudogenes may exist in adult SCs, complicating the interpretation of POU5F1 expression as a marker of pluripotency in cells derived from adult tissues (Kotoula *et al.*, 2008; Liedtke *et al.*, 2008; Mueller *et al.*, 2009). Alternative splicing of *POU5F1* has been found to result in two main variants POU5F1A and POU5F1B (also known as OCT4A and OCT4B) that share a common DNA-binding POU-homeodomain, but differ in NH₂-termini (Takeda *et al.*, 1992). Importantly, the different isoforms differ in function and subcellular localisation. Only the longer POU5F1A variant shows localisation to the nucleus and the abilities to sustain SC properties and confer self-renewal, whereas POU5F1B is mainly found in the cytoplasm and has as yet unknown function (Cauffman *et al.*, 2006; Lee *et al.*, 2006).

To date, the majority of reports in the literature have not distinguished between the two isoforms, and this is also a caveat of our present study as the primers and commercial antibody used here also detect the presence of both isoforms. Although, immunofluorescent staining did reveal some localisation of POU5F1 to the nucleus of hMSCs suggesting that some POU5F1A protein was detected. However, further confirmation of these results is required through methods suggested by Liedtke and colleagues (Liedtke *et al.*, 2008), such as application of primers specific for exon 1 of *POU5F1*, which would detect expression of *POU5F1A* alone, as well as use of a monoclonal antibody recognising a single epitope in the N-terminal part of POU5F1, to exclude recognition of POU5F1B.

3.3.4 Effects of Long-term Serial Passage on hMSCs

While multi-lineage marker expression has been demonstrated previously in MSCs, to our knowledge, no studies have investigated changes in the constitutive expression of pluripotency and lineage-associated differentiation markers by

undifferentiated hMSC populations during long-term serial passage, at both gene and protein levels. We found hMSCs to be capable of maintaining a consistent morphological appearance, proliferative rate and apoptotic level, as well as the expected antigenic pattern of MSCs between P1 and P6-8. Beyond this period, gradual changes in morphological appearance and decreased proliferative capability were noted. Protein expression of POU5F1 and NES was observed to decrease in parallel with changes in hMSC morphology and growth, suggesting that the pluridifferentiated state of hMSCs is affected by *ex vivo* expansion. However, *POU5F1, NES, MAP-2* and *CSPG4* gene expression levels were maintained over the duration of culture. The incongruity between gene and protein expression highlights the need to examine protein expression.

Mesodermal lineage marker expression by hMSC populations over long-term culture was maintained at a fairly constant level. A dramatic increase in *COL1A1* gene expression was detected in late stage culture, at the time corresponding with the cessation of hMSC proliferation. These findings suggest an apparent deviation away from pluripotency, and imply the commitment of hMSCs to the mesodermal lineage with long-term culture. This is also supported by findings by DiGirolamo, *et al.* (Digirolamo *et al.*, 1999) of spontaneous mineral deposition in late passage cultures, which were found to retain osteogenic differentiation capability at the expense of adipogenic differentiation. In addition, we found that long-term subculture of hMSCs did not result in spontaneous neural differentiation, in contrast to a recent study on rat MSCs (Tseng *et al.*, 2007). These differences could be attributed to variation between species and different culturing conditions.

3.3.5 Common Culture Conditions are Insufficient for hMSC Maintenance

The deviation toward asymmetric division observed in hMSC cultures during long-term *in vitro* expansion suggests that the current culture methods are insufficient

for hMSC maintenance. The culture method chosen for this study was one demonstrated to isolate and expand MSCs with pluripotent potential (Jiang *et al.*, 2003; Jiang *et al.*, 2002; Reyes *et al.*, 2001), and has also been shown to induce higher expression of genes involved in metabolism, when compared with an alternative method that upregulates genes involved in development, morphogenesis, extracellular matrix and differentiation (Wagner *et al.*, 2006). Decreased proliferative capacity has been documented in other studies (Baxter *et al.*, 2004; Bruder *et al.*, 1997; Digirolamo *et al.*, 1999; Mets *et al.*, 1981); however, little progress has been made in further optimising MSC culture conditions over the last two decades. The dependence on FBS is problematic due to the large variation in batches of FBS and potential immunogenicity of xenogeneic FBS proteins (Spees *et al.*, 2004).

These MSC culture inadequacies and the possibility of subpopulations within the culture may account for the inconsistent results between studies and inter-laboratory variation. Future prospects for improving MSC culture methods include: culturing under low oxygen conditions, replacement of FBS with defined cytokines, human serum (Shahdadfar *et al.*, 2005; Spees *et al.*, 2004) or serum substitutes (Meuleman *et al.*, 2006), and reduction of calcium and addition of antioxidants (Lin *et al.*, 2005). Further improvements in MSC culture homogeneity could arise with the identification of marker/s for the prospective isolation of MSCs. The recent finding that CD146 could be a potential marker of MSCs (Sacchetti *et al.*, 2007; Sorrentino *et al.*, 2008) may provide a much-needed and long-awaited positive selection method, which will hopefully facilitate progress in this research field.

3.3.6 Limitations Associated with Interpretation of Marker Expression

Difficulties have also been encountered in the SC field through the use of markers to represent certain cell types or tissue lineages, which may not be entirely restricted to that cell type or tissue lineage, leading to potential misinterpretation of data. One such marker is NES, which is expressed in certain mesodermal cell types, including developing and regenerating muscle (Kachinsky *et al.*, 1994; Sejersen *et al.*, 1993; Vaittinen *et al.*, 2001), and endothelial cells (Klein *et al.*, 2003; Mokry *et al.*, 1998a; Mokry *et al.*, 1998b), as well as within neural tissue. Therefore, caution must be taken in the interpretation of these forms of data, and examination of multiple markers should be undertaken, as well as assessment of function.

3.3.7 Summary

In summary, the experiments described in this chapter have demonstrated that hMSC populations isolated from four individual donors were capable of maintaining a stable phenotype during the first 6-8 passages of culture, and also constitutively expressed pluripotency and neural lineage markers in the undifferentiated state. These results highlight the necessity of ascertaining the baseline expression profile of undifferentiated MSCs prior to commencing differentiation. Furthermore, examination of the hMSC phenotype and expression profile during long-term *in vitro* culture has enabled me to determine that optimal time frames may exist for hMSC expansion and lineage differentiation. For the hMSC populations studied here, the window for expansion and differentiation occurs before P6, and in all subsequent studies hMSCs were only used prior to this passage. In addition, the effects of long-term serial passaging on hMSCs have been investigated further as part of the neuronal differentiation studies discussed in the following chapter.

CHAPTER 4

NEURONAL DIFFERENTIATION OF HUMAN MESENCHYMAL STEM CELLS *IN VITRO*

4.1 Introduction

The first studies describing the *in vitro* neuronal differentiation of MSCs were reported in 2000 (Sanchez-Ramos *et al.*, 2000; Woodbury *et al.*, 2000), and utilised a cytokine and signalling molecule-based or chemical-based approach. Since then, a wide range of methods have been implemented, which have been discussed in greater depth in Section 1.3.1. Nevertheless, the current approaches used for MSC neural induction still primarily involve the use of a chemical or cytokine-based system, or a combination of both.

Recently, some concerns have been raised over the validity of chemically induced neuronal differentiation of MSCs, particularly with methods involving DMSO/BHA or IBMX/cAMP (Bertani *et al.*, 2005; Choi *et al.*, 2006; Deng *et al.*, 2006; Lu *et al.*, 2004; Neuhuber *et al.*, 2004; Suon *et al.*, 2004; Tao *et al.*, 2005). These independent studies could not replicate the findings of the original studies, and instead found that the rapid effects of chemical-induced neuronal differentiation resulted from culture artifacts due to environmental stress and toxicity caused by the chemical agents (further details in Section 1.3.1.1 and Section 1.3.2). While these findings cannot account for all reported instances of *in vitro* and *in vivo* MSC neural transdifferentiation, serious questions surfaced regarding the actual extent of MSC plasticity, and the initial optimism associated with the application of MSCs in treatments for neurological disorders was dampened. However, it remained undetermined whether culture artifacts were also responsible for the changes observed during cytokine-based neural differentiation of MSCs.

In addition, it has not been established whether long-term serial passaging affects the neural differentiation capability of MSCs. Digirolamo and colleagues (Digirolamo *et al.*, 1999) have reported a loss of MSC multipotentiality, in terms of *in*

vitro mesodermal differentiation, following extensive proliferation in culture. Differentiation toward the adipocytic lineage failed in late passage MSCs, and this was found to occur in conjunction with the loss of CFU-F forming ability. On the other hand, osteogenic differentiation appeared unaffected, and there was some evidence of spontaneous differentiation into osteoblasts in some late passage MSCs (Digirolamo *et al.*, 1999). The results described in Chapter 3 of this thesis showed that long-term serial passaging of hMSCs did result in a number of phenotypic alterations (Khoo *et al.*, 2008). However, long-term subcultivation of hMSCs was not accompanied by spontaneous differentiation toward the neural lineage. Changes in the neural differentiations for the use of *in vitro* cultured MSCs in neural differentiation procedures, and may be a factor contributing to inconsistencies in the results of different studies. Therefore, further investigations are warranted.

The generation of DA-producing neuronal-like cells from rodent and human MSCs have been reported in previous studies (Dezawa *et al.*, 2004; Fu *et al.*, 2006b; Guo *et al.*, 2005; Hermann *et al.*, 2004; Jiang *et al.*, 2003; Suon *et al.*, 2006), which were discussed in more depth in Section 1.4.4. A range of varying methodologies were employed in these studies, although a number of similarities existed in the use of extrinsic factors and manipulation of the culture environment to induce differentiation. Also, these studies only achieved fairly low efficiencies of dopaminergic differentiation, which ranged between 11% and 41%. Since the goal of directed differentiation of MSCs toward DA-producing cells is clinical application in cellular therapies for PD, it is desirable to obtain these cells through methods devoid of genetic engineering and animal-derived proteins or cellular components. Ideally, the induction procedure

employed should also cause the treated MSCs to respond in a manner that mimics *in vivo* dopaminergic neuronal development and differentiation.

This chapter aimed to investigate the abovementioned aspects of neuronal and dopaminergic neuronal differentiation of BM-derived hMSCs. Our laboratory has previously shown hMSCs to be capable of adopting a neuronal-like phenotype upon cytokine-exposure (Tao et al., 2005). The present study evaluated this method using live cell microscopy to determine whether the morphological changes associated with cytokine-induced neuronal differentiation were due to a differentiation response or due to epiphenomena mediated by environmental stress similar to what occurs during chemical induction. The effect of long-term serial passaging on the neuronal differentiation capability of hMSCs was also investigated to determine whether there exists an optimal period at which to commence neuronal differentiation from our hMSC populations, and whether long-term in vitro culture alters hMSC neuronal differentiation potential. Furthermore, it was hypothesised that dopaminergic differentiation of hMSCs could be induced using cytokines that have been reported to be physiologically important in *in vivo* midbrain dopaminergic neuron development (see Section 1.4.3.1). In addition, since during development cells are sequentially exposed to growth factors, I investigated whether sequentially stimulating hMSCs with cytokines in a multiple-stage approach might provide advantages over a single-step protocol.

During the course of these experiments, a report by Trzaska and colleagues (Trzaska *et al.*, 2007) was published describing a single-stage cytokine-based procedure for specifying a dopaminergic phenotype from hMSCs. Therefore, in the present study I have compared a novel multiple-stage dopaminergic neuronal differentiation (MultiDA) method (sequential cytokine stimulation), and a previously described single-stage neuronal differentiation (SingleND) procedure developed in our laboratory, with the

150

recently published single-stage dopaminergic neuronal differentiation (SingleDA) method (Trzaska *et al.*, 2007). The three different methodologies examined are summarised in Table 2.1 and Figure 4.1.



Figure 4.1 Diagram of the hMSC neuronal differentiation methods.

(A) Novel multiple-stage dopaminergic neuronal differentiation method (MultiDA). **(B)** Previously described single-stage neuronal differentiation method developed by our laboratory (SingleND). (C) Recently published (Trzaska et al., 2007) single-step dopaminergic neuronal differentiation method (SingleDA). Ctrl, control undifferentiated hMSCs; S, stage; #, cytokines from S1 included; ND, neuronal differentiation; p, passage; d, day.

4.2 Results

4.2.1 Live Cell Imaging of hMSC Neuronal Differentiation

Recent studies have demonstrated that the rapid effects of chemical-induced MSC neuronal differentiation result from culture artifacts due to environmental stress caused by the chemical agents (Bertani *et al.*, 2005; Choi *et al.*, 2006; Deng *et al.*, 2006; Lu *et al.*, 2004; Neuhuber *et al.*, 2004; Suon *et al.*, 2004; Tao *et al.*, 2005). In these studies, the neuron-like morphology was found to result from shrinkage of the cell body due to chemical toxicity, with small remnants of the cell periphery left behind to form pseudo-processes, causing a pseudo-neuronal appearance. In the present study, I examined whether this was also true of the morphological changes associated with cytokine-based neuronal differentiation through the SingleND procedure. For this purpose, live cell imaging techniques were employed to provide real-time recording of changes in cell morphology to determine whether the neurite-like processes were indeed extended outward from the cells, rather than being remnants of the initial cell periphery after shrinkage.

Live cell imaging of hMSC neuronal differentiation revealed cells to undergo a considerable degree of morphological change and movement (Figure 4.2; see video footage labelled Supplementary File 1 – Part 1 and Supplementary File 1 – Part 2 on Supplementary CD). Cells developed a neuronal-like bipolar or multipolar appearance by active extension and retraction of long thin cellular processes (arrowheads), development of refractile cell bodies (black arrows), and formation of short transient branches (asterisks). Minute protuberances were also seen along the course of some long cellular processes. In addition, a degree of dynamic cell-cell interaction was observed, during which, neurite-like processes touched, appeared to interact, and then
retracted or moved away (white arrow). At the end of the 3 week differentiation period, this morphology was observed in approximately 65-80% of cells. These images were captured post-trypsinisation; however, spreading phenomena associated with re-attachment does not solely explain the observed morphology, as this was continually altered throughout the monitoring period (>18 hours) and also encompassed migration, and retraction as well as extension of cellular processes. This period of time allows for initial re-attachment as well as any subsequent, but distinct, alterations in morphology. In contrast our previous work has shown that chemically induced hMSCs failed to re-attach following trypsinisation. Furthermore, these cells showed little survival beyond 3 days post-induction, and also detached spontaneously from the culture surface due to cell death (Tao *et al.*, 2005). The morphological changes associated with hMSC neuronal differentiation using the SingleND cytokine-based system are, therefore, the result of active and dynamic processes involving the outgrowth of cellular extensions or processes.

In support of the observed morphological changes, immunofluorescence staining showed that hMSCs also expressed increased levels of neuronal progenitor marker NES and neuronal marker β tubulin III from basal levels after neuronal differentiation (>90% and >95% respectively; Figure 4.3). Weak expression of MAP-2 and GFAP was also detected in some cells (Figure 4.3).



Figure 4.2 Examination of cytokine-induced hMSC neuronal differentiation by live cell imaging.

Analysis using phase-contrast live cell imaging techniques over an 18 hour period revealed cells to be active and to undergo considerable elongation and movement in neuronal differentiation conditions. Cells actively extended and retracted long thin cellular processes (arrowheads), developed refractile cell bodies (black arrows), and formed short transient branches (asterisks). A degree of dynamic cell-cell interaction was also observed, during which cellular processes touched, appeared to interact, and then retracted or moved away (white arrow). The morphological changes associated with cytokine-induced hMSC neuronal differentiation are, therefore, the result of active and dynamic processes, rather than consequences of cellular shrinkage or toxicity. ', minutes. Scale bar: 100 μ m.



Figure 4.3 Examination of cytokine-induced hMSC neuronal differentiation by immunofluorescence microscopy.

After 3 weeks in neuronal differentiation conditions, cells strongly expressed neuronal markers NES (>90% +ve) and β tubulin III (>95% +ve). Weak expression of GFAP and MAP-2 was also detected in some cells. Representative images are shown. Nuclei counterstained with DAPI. Negative controls consisted of secondary antibody application alone. Scale bar: 20 μ m.

4.2.2 Effect of Long-Term Serial Passage on hMSC Neuronal Differentiation

To assess whether prolonged culture of hMSCs in vitro alters the neuronal differentiation potential of hMSCs, cells from P4-5 and P11-12 were exposed to the cytokine-based SingleND procedure and compared for differences in morphology and expression of neural markers. Characteristic morphological changes were observed in both cultures as early as within 7 days of treatment with SingleND conditions (Figures 4.4B and F; control in Figures 4.4A and E, respectively), and continued over the 3 week differentiation period (Figures 4.4C, D, G, and H). Early and late passage cultures showed development of refractile cell bodies and long cellular processes/extensions with some branching. Interestingly, despite having undergone changes in the ability to proliferate and symmetrically divide as seen in the culture growth kinetics, P11-12 hMSC cultures did not appear to have a reduced propensity toward acquiring a neuronal-like morphology. However, P11-12 cultures did contain larger cells and greater amounts of cellular debris than P4-5 cultures, and also appeared to contain a higher proportion of cells that were unresponsive to the differentiation conditions, and to a degree, retained a large flat fibroblast-like appearance (Figure 4.4I). In addition, both early and late passage cultures gradually ceased proliferating upon differentiation, and had similar proportions of viable cells by trypan blue dye exclusion (consistently >85% viable; results not shown). As described earlier, long-term serial passage alone without exposure to neuronal differentiation conditions again did not result in spontaneous neural differentiation.

Analysis of gene expression by real-time RT-PCR showed similar changes after differentiation in both early and late passage cultures, with some donor variability (Figure 4.5). Both showed highly increased expression of neuronal progenitor marker *NES*, whereas other markers for the neural lineage (*CSPG4* and *MAP-2*), pluripotency

(*NANOG* and *POU5F1*), and mesodermal lineage (*COL1A1*) were not very differentially regulated. Expression of astrocyte marker *GFAP* was markedly upregulated in late passage cultures, unlike in early passage cultures, suggesting a possible bias toward astrocytic differentiation in late passage cultures. However, although the fold change appears greatly increased in late passage hMSCs, the level of *GFAP* expression in differentiated cells was similar to the level in early passage hMSCs post-differentiation.

Neural protein expression was examined by immunofluorescence staining for neuronal markers NES, β tubulin III and MAP-2, and astrocyte marker GFAP. Similar results were obtained for both early and late passage neuronal differentiated hMSCs, with cytoskeletal expression of NES and β tubulin III, and weak/no expression of MAP-2 and GFAP. On the other hand, undifferentiated hMSCs expressed NES, but showed weak/no expression of β tubulin III, MAP-2 and GFAP (results not shown).



Figure 4.4 Morphology of early and late passage hMSC cultures undergoing neuronal differentiation.

Representative phase-contrast microscopy images of control hMSCs and hMSCs treated with neuronal differentiation conditions. (**A-D**) Early/P4-5 stage hMSCs and (**E-I**) Late/P11-12 stage hMSCs: (**A** and **E**) Prior to treatment, (**B** and **F**) 1 week, (**C** and **G**) 2 weeks, and (**D**, **H**, and **I**) 3 weeks post-treatment. Following treatment, both P4-5 and P11-12 hMSCs developed refractile cell bodies, retracted cytoplasm and long branching processes. Larger cells and more cellular debris were observed in P11-12 cultures, and a greater number of unresponsive large flat fibroblast-like cells were seen (I). Scale bar: 100 µm.



Figure 4.5 Gene expression of early and late passage hMSC cultures post-neuronal differentiation.

Representative real-time RT-PCR results from undifferentiated control hMSC cultures and cultures treated with neuronal differentiation conditions (n = 2). Results are depicted as mean fold change in gene expression relative to the control, with baseline set at 1.0. Ctrl, control undifferentiated hMSCs; ND, neuronal differentiated hMSCs.

160

4.2.3 Cytokine-Based Dopaminergic Neuronal Differentiation of hMSCs

Previous results obtained by live cell imaging (Section 4.2.1) indicated that the morphological changes induced in hMSCs by cytokine-based neuronal differentiation were the result of active and dynamic processes rather than culture artifacts. In addition, cytokines were found to drive hMSCs toward an early neuronal fate. As a result, it was hypothesised that the application of cytokines relevant to midbrain dopaminergic neuronal differentiation may direct hMSCs toward this lineage. Furthermore, sequential application of cytokines was investigated to determine whether this provided advantages over single-step protocols. Figure 4.1 shows a diagram of the steps in each differentiation method.

4.2.3.1 Morphological Analysis of hMSC Dopaminergic Neuronal Differentiation

Two methods of cytokine-based hMSC dopaminergic neuronal differentiation were employed in this study, a novel multiple-stage procedure (MultiDA) and a recently published single-step protocol (SingleDA) (Trzaska *et al.*, 2007). These techniques were also compared with the SingleND method previously published by our group (Tao *et al.*, 2005). Representative phase-contrast images of cultures treated with these different methods are shown in Figures 4.6, 4.7 and 4.8 respectively. Prior to differentiation the hMSC cultures mainly contained cells with a fibroblast-like appearance (Figures 4.6A, 4.7A and 4.8A), and with continued growth *in vitro* the cells displayed an increasingly flattened morphology. The three different cytokine-based treatments all produced cells possessing a similar neuronal-like bipolar or multipolar morphology, with the formation of refractile cell bodies and long thin cellular processes, occasionally containing branching or minute protuberances. In addition, for all three methods the changes in morphology occurred gradually over the differentiation period.

Treatment with the MultiDA protocol yielded the most distinct morphological alterations. Following the application of FGF-2 and EGF in the first stage (S) of differentiation, hMSCs developed a bipolar, refractile appearance and formed neurosphere-like cell clusters (Figure 4.6B). While bipolar refractile morphology was also common to both the single-stage differentiation methods, neither was found to result in neurosphere-like formation. With subsequent culturing in S2 (SHH/FGF-8) and S3 (GDNF) many of the cells that had acquired neuronal-like morphology were observed to return toward a flatter appearance, and thin cellular processes were also lost from some of these cells (Figures 4.6C and 4.6E). In addition, cell survival was observed to decrease in MultiDA S2 and S3 conditions, with fewer adherent cells remaining in culture. However, combining S1 cytokines with later stages (denoted by #), permitted hMSCs to maintain a neuronal-like appearance throughout the second MultiDA stage (Figure 4.6D). Continued supplementation of cultures with S1 cytokines right through S3 (S3#) allowed the bipolar and refractile morphology to be maintained (Figure 4.6F), whereas upon the removal of FGF-2 and EGF after S2 (S3(2#)), cells were again observed to gain a flattish appearance (Figure 4.6G and H). Interestingly, although the addition of FGF-2 and EGF to later culture stages appeared to promote a continued bipolar and refractile neuronal-like morphology and increased cell survival, it was also found that uninterrupted treatment resulted in cells with less complex branching of cellular processes. A considerable decrease in cell number was also observed across all stages of the MultiDA method.

Application of the recently reported SingleDA method (Trzaska *et al.*, 2007) generated a neuronal-like morphology similar to the published results (Figure 4.7). In addition, the morphology of hMSCs treated with the SingleDA method resembled that obtained through the SingleND protocol developed by our group ((Tao *et al.*, 2005);

Figure 4.8), despite differences in the cytokines used. The SingleND method also elicited reduced cell numbers, while slight increases were seen using the SingleDA method.



Figure 4.6 Morphology of hMSCs during *in vitro* cytokine-induced **multiple-stage dopaminergic neuronal differentiation (MultiDA).** (Refer to next page for Figure 4.6 Legend text)

Figure 4.6 Morphology of hMSCs during *in vitro* cytokine-induced multiple-stage dopaminergic neuronal differentiation (MultiDA).

Images of hMSCs undergoing dopaminergic differentiation were captured at the end of each stage (1 week) of sequential cytokine treatment. Representative phase-contrast images are depicted of hMSCs: (A) Prior to treatment (d0); after (B) S1 (d7), (C) S2 (d14), (D) S2# (d14), (E) S3 (d21), (F) S3# (d21), and (G and H) S3(2#) (d21) of differentiation (n = 3). d, day; S, stage; #, cytokines from S1 included. Scale bar: 100 µm.



Figure 4.7 Morphology of hMSCs during *in vitro* cytokine-induced single-stage dopaminergic neuronal differentiation (SingleDA).

Dopaminergic neuronal differentiation of hMSCs was performed according to a previously published single-step method (Trzaska *et al.*, 2007) for comparison with our multiple-stage method. Representative images captured with a phase-contrast light microscope are depicted of hMSCs: (A) Prior to treatment, and (B) 6 days and (C) 12 days post-treatment (n = 3). Scale bar: 100 µm.



Figure 4.8 Morphology of hMSCs during *in vitro* cytokine-induced single-stage neuronal differentiation (SingleND).

Neuronal differentiation of hMSCs was performed according to a method previously published by our group (Tao *et al.*, 2005) for comparison with the dopaminergic neuronal differentiation methods. Representative phase-contrast images are shown of hMSCs: (**A**) Prior to treatment (same image shown in Figure 4.6), and at the end of (**B**) P1 (1 week), (**C**) P2 (2 weeks), and (**D**) P3 (3 weeks) post-treatment (n = 3). P, passage. Scale bar: 100 µm.

4.2.3.2 Gene Expression Analyses During In Vitro hMSC Dopaminergic Neuronal Differentiation

The gene expression of hMSCs subjected to *in vitro* neuronal and dopaminergic neuronal differentiation procedures were examined by real-time RT-PCR to examine alterations in neuronal, dopaminergic neuronal, glial, mesodermal, and pluripotency markers during cytokine-based differentiation. Cells were harvested for analysis prior to differentiation (control hMSCs) and at the end of each week of cytokine treatment for hMSCs undergoing the MultiDA and SingleND procedures. Alternatively, for the SingleDA procedure (Trzaska *et al.*, 2007), hMSCs were collected at days 6 and 12 in accordance with the time points used in the published study. Data are presented as gene expression of the sample relative to the gene expression of the control (all relative to the expression of house-keeping gene *HPRT*; n = 3). Statistical analysis was performed using the paired *t*-test, and significance was set at p < 0.05.

Expression of Neuronal Genes During *In Vitro* hMSC Dopaminergic Neuronal Differentiation: Expression of neuronal progenitor gene *NES* and neuronal genes *MAP-2* and *ENO2* were detected in all samples before and after *in vitro* cytokinebased differentiation (Figure 4.9). *NES* was found to be significantly up-regulated at all time points in all three differentiation methods, apart from S2 of the MultiDA method. The highest levels of *NES* expression were elicited by the MultiDA procedure, and the addition of S1 cytokines (FGF-2/EGF) to later differentiation stages appeared to be beneficial for *NES* expression. The first stage of the MultiDA method (S1) resulted in a 39.2-fold increase in *NES* (p = 0.012), while S2# resulted in a 175.5-fold increase (p = 0.022), and S3# showed 117.5-fold increased *NES* expression (p = 0.023). Whereas, the highest level of *NES* expression in the SingleND method was found after 3 weeks of differentiation (P3: 23.9-fold increase; p = 0.025), and the SingleDA method yielded a similar result, with the highest *NES* expression levels seen after 12 days of induction (d12: 21.9-fold increase; p = 0.005). The mRNA levels of *MAP-2* were not significantly different to the control in all methods examined, except at day 12 of the SingleDA method which displayed a 1.36-fold increase (p = 0.040). Likewise, the expression of *ENO2* did not vary significantly from the control, with the exception of cells grown in S2 and S3 of the MultiDA method, which showed a significant decrease in *ENO2* levels of 2.9- and 3.8-fold respectively (p = 0.040 and p = 0.038 respectively). However, this decrease in *ENO2* could be prevented by the addition of FGF-2/EGF to these stages of differentiation.

Expression of Dopaminergic Neuronal Genes During In Vitro hMSC Dopaminergic Neuronal Differentiation: Human MSCs were also found to express NR4A2, a transcription factor essential for midbrain dopaminergic neuronal differentiation, as well as classical dopaminergic neuronal marker TH, prior to and following the three different induction protocols (Figure 4.10). A trend of increasing NR4A2 expression could be observed in S1 and S2# of the MultiDA method (1.9- and 3.0-fold respectively); however, this did not reach a level of significance in comparison to the control. Absence of FGF-2/EGF from Stage 3 of the MultiDA method did result in significantly decreased NR4A2 levels (S3: 18.9-fold decrease, p = 0.00068; S3(2#): 10.8-fold decrease, p = 0.027), which was not observed with FGF-2/EGF supplementation (S3#(2#): 1.4-fold decrease, p = 0.26). On the other hand, both singlestep differentiation procedures exhibited a non-significant trend of decreasing NR4A2 expression. For all differentiation methods investigated, no significant changes were detected in TH expression levels compared to control undifferentiated hMSC cultures; however, the SingleND method yielded the highest levels of TH expression (3.8- to 5.6fold increase).

Expression of Glial Genes During *In Vitro* hMSC Dopaminergic Neuronal Differentiation: Expression of mRNA for glial markers *GFAP* and *CSPG4* was also detected using the real-time RT-PCR technique (Figure 4.11). Only slight expression of *GFAP* could be detected in hMSCs prior to and after the differentiation procedures, although significant up-regulation was observed in S1 and S2# of the MultiDA method (S1: 15.8-fold increase, p = 0.026; S2#: 20.4-fold increase, p = 0.026). The level of *GFAP* expression in all other differentiation conditions did not show significant difference to the control cultures; however, a trend was observed in the SingleND method, which showed initial increase in *GFAP* expression at 1 week post-differentiation, followed by a trend of decreasing expression over later time points. *CSPG4* mRNA was moderately expressed under all conditions examined, although significant changes from the control were not observed.

Expression of Mesodermal Gene *COL1A1* During *In Vitro* hMSC Dopaminergic Neuronal Differentiation: The expression of mesodermal marker *COL1A1* was also examined to determine whether a loss of mesodermal characteristic occurred in parallel with neuronal differentiation (Figure 4.12). On the contrary, significant down-regulation of *COL1A1* compared to the control could not be detected by real-time RT-PCR under any of the neuronal differentiation conditions examined. Instead, *COL1A1* expression was observed to significantly increase in hMSCs cultured in the absence of FGF-2/EGF in stages 2 and 3 of the MultiDA method (S2: 6.3-fold increase, p = 0.0029; S3: 14.6-fold increase, p = 0.0012; S3(2#): 5.6-fold increase, p = 0.0099), while the other conditions did not yield any significant changes from the control.

Expression of Pluripotency Genes During *In Vitro* hMSC Dopaminergic Neuronal Differentiation: Pluripotency markers *POU5F1* and *NANOG* were also

examined by real-time RT-PCR, with hMSCs prior to and following differentiation found to express weak levels of mRNA for both these markers (Figure 4.12). None of the neuronal differentiation methods investigated was found to induce significant changes in *POU5F1* expression compared to control hMSC cultures. In addition, expression of the *NANOG* gene was also unaffected by the MultiDA procedure and the SingleND procedure. However, the SingleDA method published by Trzaska and colleagues (Trzaska *et al.*, 2007), was found to result in significant up-regulation of *NANOG* expression at both day 6 (4.5-fold increase, p = 0.0074) and day 12 (4.8-fold increase, p = 0.041) post-induction.

Effect of FGF-2 and EGF Supplementation: In addition, supplementation with FGF-2 and EGF in the second and third stages of the MultiDA procedure resulted in significant differences in the gene expression of hMSCs. Expression of NES was found to be significantly up-regulated in S3#, when compared to hMSCs treated with S3 (GDNF alone; p = 0.050), or S3(2#) (p = 0.021). MAP-2 expression also showed a significant increase at S3# compared to S3(2#), which did not contain FGF-2 and EGF in the final stage (p = 0.005). Furthermore, neuronal marker ENO2 was also significantly up-regulated in S2# in comparison S2 (p = 0.005). The level of NR4A2 transcription was also affected by FGF-2/EGF addition, with up-regulation occurring in S2# and S3# compared with the corresponding stages without further supplementation (p = 0.006, and p = 0.012 respectively). Significant increases in astrocytic marker *GFAP* were observed with FGF-2/EGF addition in stage 2 (p = 0.027), and in stage 3 when hMSCs exposed to continual supplementation were compared with those only receiving FGF-2/EGF in stage 2 (p = 0.034). The final marker exhibiting significantly increased expression with the addition of S1 cytokines to later stages was NANOG, which showed significant up-regulation in S2# compared to S2 (p = 0.046), and in S3(2#) compared to S3 (p = 0.003). The only gene examined that showed significantly down-regulated expression with FGF-2/EGF supplementation was *COL1A1*. This was seen in S2# (p = 0.045), and also in S3# (p = 0.047), when compared with the corresponding stages cultured in the absence of FGF-2 and EGF. These results suggest an important role for FGF-2 and EGF in the *in vitro* neuronal differentiation of hMSCs.

Very slight mRNA expression of further dopaminergic neuronal markers and transcription factors were also detected in some hMSC cultures prior to and following differentiation. These markers included *PITX3*, *EN2*, *VMAT2*, and *DAT*. However, expression of these markers did not occur consistently in samples from different donors and the level of expression was fairly low. For these reasons, this data has not been shown here.

Overall, the real-time RT-PCR findings suggest that all three cytokine-based neuronal differentiation methods examined were only capable of eliciting an immature neuronal phenotype in hMSCs, with elevated expression of neuronal progenitor marker NES, and little change in mature neuronal markers MAP-2, ENO2, and TH. Mesodermal and pluripotency markers were not observed to be down-regulated, providing further evidence for incomplete neuronal differentiation. In addition. continued supplementation with FGF-2 and EGF was beneficial in the MultiDA procedure, as it resulted in enhancement of neural marker expression and prevented increases in mesodermal COL1A1 expression. A summary of the gene expression changes during in vitro culturing of hMSCs under the three different neuronal induction protocols is shown in Table 4.1.



Figure 4.9 Expression of neuronal genes in hMSC cultures undergoing *in vitro* neuronal and dopaminergic neuronal differentiation.

(Refer to next page for Figure 4.9 Legend text).

Figure 4.9 Expression of neuronal genes in hMSC cultures undergoing *in vitro* neuronal and dopaminergic neuronal differentiation.

Real-time RT-PCR results from control undifferentiated hMSC cultures and cultures treated with the MultiDA method shown in purple shades, the SingleND method shown in blue shades, and the SingleDA method published by Trzaska, *et al.* (Trzaska *et al.*, 2007) shown in orange shades. Significant increases in the expression of neuronal progenitor marker *NES* was observed in all conditions, except S2, whereas *MAP-2* and *ENO2* showed similar expression levels to the control. Results are depicted as mean fold change in mRNA expression + SEM relative to control undifferentiated hMSC cultures, with baseline set at 1.0 (n = 3). Separate control hMSC data were used for the SingleDA method, as this was not performed in parallel with the other methods. * p < 0.05compared with the control. Ctrl, control undifferentiated hMSCs; S, Stage; #, cytokines from S1 included; P, passage; d, day.





Real-time RT-PCR analyses of the MultiDA (purple shades), SingleND (blue shades), and SingleDA (orange shades) methods. Significant decrease in transcription factor *NR4A2* expression was seen when S1 cytokines were absent from S3 cultures. Otherwise, *NR4A2* and *TH* expression were similar to the control. Results are depicted as mean fold change in mRNA expression + SEM relative to control undifferentiated hMSC cultures (set at 1.0; n = 3). Separate control hMSC data were used for the SingleDA method, as this was not performed in parallel with the other methods. * p < 0.05 compared with the control. Ctrl, control undifferentiated hMSCs; S, Stage; #, cytokines from S1 included; P, passage; d, day.





Real-time RT-PCR analyses of the MultiDA (purple shades), SingleND (blue shades), and SingleDA (orange shades) methods. Significant increase in *GFAP* expression was seen in S1 and S2# cultures. Otherwise, *GFAP* and *CSPG4* expression were not significantly different from the control. Results are depicted as mean fold change in mRNA expression + SEM relative to control undifferentiated hMSC cultures (set at 1.0; n = 3). Separate control hMSC data were used for the SingleDA method, as this was not performed in parallel with the other methods. * p < 0.05 compared with the control. S, Stage; #, cytokines from S1 included; P, passage; d, day.



Neuronal Differentiation of hMSCs In Vitro

Figure 4.12 Expression of mesodermal and pluripotency genes in hMSCs undergoing neuronal & dopaminergic neuronal differentiation.

(Refer to next page for Figure 4.12 Legend text).

Figure 4.12 Expression of mesodermal and pluripotency genes in hMSCs undergoing neuronal & dopaminergic neuronal differentiation.

Real-time RT-PCR results from control undifferentiated hMSC cultures and cultures treated with the MultiDA method shown in purple shades, SingleND method shown in blue shades, and SingleDA method published by Trzaska, et al. (Trzaska et al., 2007) shown in orange shades. Expression of mesodermal marker COLIA1 was significantly increased when S1 cytokines were absent from S2 and S3 cultures. Pluripotency markers POU5F1 and NANOG were detectable in all samples; however, expression was not significantly different from the control, except for significantly increased NANOG expression in SingleDA samples. Results are depicted as mean fold change in mRNA expression + SEM relative to control undifferentiated hMSC cultures, with baseline set at 1.0 (n = 3). Separate control hMSC data were used for the SingleDA method, as this was not performed in parallel with the other methods. *p < 0.05 compared with the control. Ctrl, control undifferentiated hMSCs; S, Stage; #, cytokines from S1 included; P, passage; d, day.

	MultiDA					SingleND			SingleDA	
Genes	S1	S2	S3	S2#	S3#	P1	P2	P3	d6	d12
NES	1	††	t	111	†††	1	1	t	1	†
MAP-2	-	—	—	-/t	-/t	—	—	—	_	-/t
ENO2	—	++	<u>++</u>	ŧ	ŧ	—	ŧ	ŧ	—	—
NR4A2	1	-	++	t	_	-	ł	ŧ	ŧ	ŧ
тн	—	—	—	—	—	-/t	-/†	-/†	—	—
GFAP	††	-/t	_	††	††	††	†/††	t	t	Ť
CSPG4	-/ŧ	—	—	—	—	—	—	—	-/+	—
POU5F1	-/†	_	_	-/†	-/t	—	_	-/†	-	_
NANOG	Ť	t	t	†/††	†/††	†/††	†/††	†/††	†/ ††	†/ ††
COLI		†/††	††	—	_	-/+	-/+	-/ŧ	++	++

Table 4.1 Summary of Real-Time RT-PCR Results: Gene ExpressionChanges During hMSC Cytokine-Based Neuronal Differentiation

FGF-2	+			+	+	+	+	+	+	+
EGF	+			+	+	+	+	+		
PDGF						+	+	+		
SHH		+		+					+	+
FGF-8		+		+					+	+
GDNF			+		+					

Yellow: Significant change (p < 0.05 compared with control)

#: Cytokines from Stage 1 (FGF-2/EGF) included

- **†** : Slight increase
- **††** : Moderate increase
- **†††** : High increase
- ↓ : Slight decrease
- **↓↓** : Moderate decrease
- : No change
- + : Cytokine included in culture conditions
- ---: Cytokine excluded from culture conditions

4.2.3.3 Protein Expression Analyses During In Vitro hMSC Dopaminergic Neuronal Differentiation

The previous results obtained by real-time RT-PCR (Section 4.2.3.2) indicated that hMSCs were capable of up-regulating the expression of certain neural genes when exposed to all three cytokine-based neuronal differentiation procedures. So, to examine whether these changes also occurred at the protein level, immunofluorescence staining for neuronal, glial, and mesodermal markers were performed on hMSCs undergoing the MultiDA (Figure 4.13), SingleND (Figure 4.14) and SingleDA (Figure 4.15) procedures, at the same time points investigated by real-time RT-PCR (n = 3).

Consistent with the PCR findings neuronal progenitor marker NES was detected in both undifferentiated control hMSCs (<10% positive) and at all time points examined during cytokine-based neuronal differentiation (<20% positive). NES expression showed an intracellular cytoplasmic localisation that resembled a cytoskeletal pattern or filamentous network, and appeared to be slightly up-regulated following cytokine induction. Similar levels and patterns of NES staining were observed in cells subjected to all three neuronal differentiation methods. Some cells cultured under the neuronal differentiation conditions were noticed to be lost from the culture surface during the staining procedure, particularly those displaying a refractile cell body and possessing long thin cellular extensions, i.e. cells with a neuronal-like appearance. Therefore, it should be acknowledged that the percentage of cells positive for NES and the other markers examined here may be underestimated.

Prior to differentiation, some hMSCs expressed neuronal marker β tubulin III at low levels (<15% positive). Following exposure to neuronal differentiation conditions, the level of β tubulin III expression was dramatically increased, and a greater proportion of cells also exhibited the intracellular cytoskeletal staining pattern typical of this

180

structural protein. Generally, >80% of cells cultured in the neuronal differentiation conditions were positive for β tubulin III protein, with the exception of: the SingleND method, which generated >90% β tubulin III-positive cells; and S2 and S3 of the MultiDA method that excluded FGF-2/EGF, which produced a lower yield of 60-80% β tubulin III-positive cells.

As expected, extracellular matrix glycoprotein fibronectin was observed to be strongly expressed by control undifferentiated hMSCs (90-100% positive). This expression was also found to occur in a typical extracellular fibrous pattern. In addition, continued fibronectin expression was detected after implementation of the three neuronal differentiation methods, with >90% of cells generally positive for fibronectin. S3 of the MultiDA method showed slightly lower proportions of fibronectin-positive cells (>80%), and the intensity of fibronectin expression in these cells also appeared weaker.

The other markers examined by immunofluorescence staining included astroglial protein GFAP, and neuronal proteins NeuN and TH. In general, for all three cytokinebased neuronal differentiation procedures investigated, the expression of these markers was absent from the majority of cells and only occasionally could very weak expression be detected.

As a whole, the examination of protein expression by immunofluorescent staining confirmed the findings obtained by real-time RT-PCR. Following neuronal differentiation hMSCs were found to continue expressing NES and fibronectin proteins, but also up-regulated β tubulin III protein expression. Together these results indicate that the three different cytokine-based neuronal induction methods generated hMSC-derived cells possessing an immature neuronal phenotype. The findings of this chapter are summarised in Table 4.2.

181



Figure 4.13 Protein expression analysis of hMSCs during *in vitro* cytokine-induced multiple-stage dopaminergic neuronal differentiation (MultiDA) by immunofluorescence microscopy (Part 1).

(Refer to the following pages for Figure 4.13 Parts 2 and 3, and Figure 4.13 Legend text).



Figure 4.13 Protein expression analysis of hMSCs during *in vitro* cytokine-induced multiple-stage dopaminergic neuronal differentiation (MultiDA) by immunofluorescence microscopy (Part 2).

(Refer to the following pages for Figure 4.13 Part 3, and Figure 4.13 Legend text).



Figure 4.13 Protein expression analysis of hMSCs during *in vitro* cytokine-induced multiple-stage dopaminergic neuronal differentiation (MultiDA) by immunofluorescence microscopy (Part 3).

(Refer to the next page for Figure 4.13 Legend text).

Figure 4.13 Protein expression analysis of hMSCs during *in vitro*cytokine-inducedmultiple-stagedopaminergicneuronaldifferentiation (MultiDA) by immunofluorescence microscopy.

Representative images depicting immunofluorescence staining of control undifferentiated hMSC cultures and cultures treated with the MultiDA method (n = 3) at the end of each stage (1 week). Cells were examined for expression of neuronal progenitor marker NES, astroglial marker GFAP, neuronal markers NeuN and β tubulin III, dopaminergic neuronal marker TH, and mesodermal marker fibronectin. Cytokine-treated cells were observed to express NES, and up-regulated β tubulin III expression, but other neural markers were not detected. Cells also continued to strongly express fibronectin. These results suggest that only immature neuronal-like cells could be obtained through the MultiDA method. Nuclei were counterstained with DAPI. Negative controls consisted of secondary antibody application alone (not shown). S, Stage; # cytokines from S1 included. Scale bar: 20 µm.



Figure 4.14 Protein expression analysis of hMSCs during *in vitro* cytokine-induced single-stage neuronal differentiation (SingleND) by immunofluorescence microscopy.

(Refer to the next page for Figure 4.14 Legend text).

Figure 4.14 Protein expression analysis of hMSCs during *in vitro* cytokine-induced single-stage neuronal differentiation (SingleND) by immunofluorescence microscopy.

Representative images depicting immunofluorescence staining of hMSC cultures treated with the SingleND method (n = 3) at the end of each passage (1 week). Control undifferentiated hMSC cultures are shown in Figure 4.13 (Part 1). Cells were examined for expression of neuronal progenitor marker NES, astroglial marker GFAP, neuronal markers NeuN and β tubulin III, dopaminergic neuronal marker TH, and mesodermal marker fibronectin. The protein expression observed here was similar to that obtained using the MultiDA method. Therefore, both methods only generated immature neuronal-like cells. Nuclei were counterstained with DAPI. Negative controls were performed without primary antibody application (not shown). P, passage. Scale bar: 20 μ m.



Figure 4.15 Protein expression analysis of hMSCs during *in vitro* cytokine-induced single-stage dopaminergic neuronal differentiation (SingleDA) by immunofluorescence microscopy. (Refer to the next page for Figure 4.15 Legend text).

Figure 4.15 Protein expression analysis of hMSCs during *in vitro* cytokine-induced single-stage dopaminergic neuronal differentiation (SingleDA) by immunofluorescence microscopy.

Representative images depicting immunofluorescence staining of control undifferentiated hMSC cultures and cultures treated with the SingleDA method at d6 and d12 (n = 3). Cells were examined for expression of neuronal progenitor marker NES, astroglial marker GFAP, neuronal markers NeuN and β tubulin III, dopaminergic neuronal marker TH, and mesodermal marker fibronectin. Cells obtained through the SingleDA method displayed similar protein expression as those obtained using the MultiDA method and the SingleND method. Therefore, all three neuronal differentiation procedures investigated were only capable of generating immature neuronal-like cells. Nuclei were counterstained with DAPI. Negative controls were performed without primary antibody application (not shown). d, day. Scale bar: 20 µm.
Table4.2Summary of hMSCCytokine-BasedNeuronalDifferentiation

Analysis	MultiDA	MultiDA#	SingleND	SingleDA
Morphology	+/++	+++	+++	+++
Gene Expression	++	+++	++ / +++	++
Protein Expression	++	++	++	++
Cell Number	~	~~	~~~	~~~

#: Cytokines from Stage 1 of MultiDA method included in later stages

+ : Indicates the degree of neuronal-like characteristics obtained

 \sim : Indicates the degree of cell survival obtained

4.3 Discussion

The results presented in this chapter have demonstrated that (1) hMSCs are capable of acquiring an immature neuronal phenotype upon stimulation with cytokines; (2) cytokine-based neuronal differentiation of hMSCs involves active and dynamic processes in response to differentiation stimuli, including the outgrowth and motility of cellular extensions; (3) long-term subcultivation of hMSC cultures does not compromise the ability of hMSCs to differentiate toward an immature neural fate; (4) treatment of hMSCs with different cytokine combinations in the MultiDA, SingleND and SingleDA procedures elicits different effects on cellular morphology and gene expression, however, none of these methods is capable of generating a mature midbrain dopaminergic neuronal phenotype from hMSCs; and (5) continual supplementation with FGF-2, and perhaps also EGF, appears beneficial for the *in vitro* neuronal differentiation of hMSCs, at least in terms of cellular morphology and gene expression.

4.3.1 Cytokine-Based *In Vitro* Neuronal Differentiation of hMSCs Elicits Dynamic Cellular Responses

The current study demonstrates for the first time that *in vitro* neuronal differentiation of hMSCs using cytokines elicits an entirely distinct cellular response in comparison with the "simple chemical induction protocol" described by Woodbury and colleagues (Woodbury *et al.*, 2000). Cells exposed to the SingleND cytokine cocktail (FGF-2/EGF/PDGF) were highly active and motile, displaying dynamic extension and movement of neurite-like processes and structures similar to neuronal growth cones, as well as the formation of transient branching morphology, as recorded in live cell imaging videos. These observations differ greatly from recent reports describing the

changes in cellular morphology generated by chemical induction (DMSO/BHA or IBMX/cAMP) to be the result of cytotoxic effects of added chemicals leading to cell death, rapid disruption of the actin cytoskeleton and retraction of selected areas of the cell perimeter and cytoplasm, leaving behind filopodium-like processes formed from strong focal adhesions to the substrate (Bertani *et al.*, 2005; Choi *et al.*, 2006; Deng *et al.*, 2006; Lu *et al.*, 2004; Neuhuber *et al.*, 2004; Suon *et al.*, 2004).

In addition, cytokine-induced hMSCs progressively attained neuronal-like morphology over the first week of culture and maintained these changes for at least 1 month *in vitro*. Viability of cultures was also consistently maintained at >85% over the course of cytokine-induced neuronal differentiation. The time frame of these changes was in terms of days and weeks, rather than hours, and allowed sufficient time for changes in cellular organisation and gene/protein expression, unlike the rapid conversion (within hours) observed with chemical treatment. Furthermore, MSCs could not be maintained in DMSO/BHA serum-free conditions, because of increased cell death (50% death within 72 hours) and detachment from the culture surface (Lu *et al.*, 2004; Tao *et al.*, 2005). In addition, subculturing DMSO/BHA-treated cells was not possible due to the failure of these cells to re-attach following trypsinisation. Therefore, the alterations in cell phenotype induced by cytokine-based neuronal differentiation of hMSCs are not simply due to culture artifacts, but can be attributed to active and dynamic cellular processes.

4.3.2 Long-Term Subcultivation of hMSCs Does Not Affect Differentiation Towards an Immature Neuronal-Like Phenotype

The effect of long-term serial passage on the neural differentiation capability of hMSCs has not been previously reported. From the examination of hMSCs over long-term *in vitro* culturing (results presented in Chapter 3) it was determined that extensive

subcultivation did not cause spontaneous neural differentiation (Khoo *et al.*, 2008). In this chapter, we found that late passage cultures (P11–12) were able to undergo morphological and gene expression changes toward neuronal-like cells similar to early passage cultures (P4–5), despite loss of proliferation and symmetric division abilities. However, a trend of increased *GFAP* expression in late passage cultures was seen that may affect differentiation toward a mature neural phenotype.

It is not yet clear why late passage cultures retain this ability when other SC qualities are lost, but the retention of *POU5F1* and *NANOG* expression may play a role, since it has been shown that POU5F1 expression in mouse ESCs is involved in neurogenesis (Chen et al., 2007a). Maintenance of NES expression may also be involved, since induction of NES in rat BM MSCs by prolonged culture in serum-free conditions, has been reported to be a prerequisite for neural lineage differentiation (Wislet-Gendebien et al., 2003). However, this report differs from the study presented in this chapter as 10 passages, or more, in serum-free conditions were necessary for the expression of NES by rat MSCs (Wislet-Gendebien et al., 2003), whereas, the hMSCs employed here were found to constitutively express NES in the presence of serum. In addition, another study reported that long-term cultivation (6 weeks without subculturing) of P1 rat MSCs resulted in spontaneous enrichment of NES-positive cells, which exhibited greater sensitivity to neuronal induction by serum deprivation and growth factor supplementation (Tseng et al., 2007). A number of variations exist between these studies, including the species, isolation methods and culturing techniques employed, which may account for the differences observed.

Nevertheless, while early and late passage hMSC cultures were equally capable of undergoing cytokine-based neuronal differentiation, early passage cells have distinct advantages over late passage hMSCs. The benefits associated with early passage hMSCs were shown in Chapter 3, and include a higher proliferative ability, and maintenance of a more stable phenotype in terms of morphology and gene/protein expression (Khoo *et al.*, 2008). Therefore, in our subsequent neuronal differentiation studies we have only utilised hMSCs harvested from early passage cultures.

4.3.3 Undifferentiated hMSCs Express Neuronal and Dopaminergic Neuronal Markers

Consistent with the data presented in Chapter 3, hMSCs were found to express neuroectodermal markers prior to the commencement of neuronal differentiation procedures. The current findings extend these previous observations that hMSCs express NES, MAP-2, GFAP and CSPG4, by demonstrating the expression of further neuronal (ENO2 and β tubulin III) and dopaminergic neuronal markers (NR4A2 and TH) in undifferentiated hMSCs. These results reflect those of other recent studies, which have also shown expression of key markers of the neuro-dopaminergic system by rat and human MSCs, including TH, NR4A2, PITX3, EN1, aldehyde dehydrogenase 1, aromatic L-amino acid decarboxylase, catechol-o-methyltransferase, GTP cyclohydrolase-1, OTX-1, SHH receptor components (patched (PTC) and smoothened (SMO)), and GDNF family receptor alpha 1 (GFRA1) (Blondheim et al., 2006; Kramer et al., 2006).

Expression of these neuroectodermal and dopaminergic neuronal markers demands that caution be taken in the interpretation of hMSC neural differentiation, particularly through examination of neural markers prior to differentiation as well as post-differentiation. However, these observations also provide support for the neural and dopaminergic neuronal differentiation capacity of hMSCs (Hermann *et al.*, 2006).

Furthermore, the constitutive *NR4A2* expression present in hMSCs suggests that these cells may be viable candidates for cellular replacement therapies for PD,

particularly since transfection of ESCs with *NR4A2* was considered to be crucial for the successful generation of ESC-derived dopaminergic neurons (Kim *et al.*, 2002b). This study by Kim and colleagues demonstrated that stable transfection of murine ESCs with *NR4A2* in conjunction with a five-stage dopaminergic neuronal differentiation method could generate TH-expressing neurons with electrophysiological activity and the ability to promote significant behavioural recovery in transplanted Parkinsonian rats.

4.3.4 Sequential Application of Cytokines Involved in Midbrain Dopaminergic Neuronal Development Did Not Enhance Dopaminergic Neuronal Differentiation of hMSCs

In view of my experimental findings that cytokine-based neuronal differentiation of hMSCs yields actual changes in cellular morphology (Section 4.2.1) and gene/protein expression (Section 4.2.2), I endeavoured to determine whether cytokines involved in midbrain dopaminergic neuronal development *in vivo* could induce hMSCs toward this neuronal lineage. Furthermore, a multiple-stage approach (MultiDA) was employed to mimic the *in vivo* situation, and this method was compared to two single step neuronal differentiation procedures (SingleND (Tao *et al.*, 2005) and SingleDA (Trzaska *et al.*, 2007)). Despite the different cytokines, surface coatings and culturing procedures employed, all three methods were effective to a similar extent and were only capable of generating immature neuronal-like cells. Sequential supplementation with cytokines in the MultiDA procedure did not direct hMSCs toward a mature dopaminergic neuronal fate, however, neither did the published SingleDA method.

Interestingly, slightly different responses were elicited in hMSCs by the three different methods. Examination of cellular morphology revealed the acquisition of a neuronal-like bipolar or multipolar morphology with long thin cellular extensions and occasional branching in all methods studied, which resembled the morphology obtained

during the live cell imaging analysis. The first stage of the MultiDA method was unique in the spontaneous formation of cellular aggregates resembling neurosphere-like structures. In addition, investigation of gene expression showed significant upregulation of *NES* and *GFAP* in stages 1 and 2 of the MultiDA method, as well as a trend of higher *NR4A2* expression. However, protein expression analysis through immunofluorescent staining did not show clear distinctions between the three methods.

In contrast, a study comparing the effects of different neuronal differentiation protocols by Hermann and colleagues (Hermann *et al.*, 2006) reported that a sequential two-step protocol involving neurosphere formation could induce significant neural conversion of hMSCs, whereas other single-step direct neural conversion methods (published in (Sanchez-Ramos *et al.*, 2000; Woodbury *et al.*, 2000), and a modified method from (Hermann *et al.*, 2004)) showed no significant differences in comparison to undifferentiated hMSCs. Furthermore, a multi-step technique was also employed by Jiang, *et al.* (Jiang *et al.*, 2003) that was capable of successfully deriving a mature electrophysiologically-active neuronal phenotype from murine MAPCs following astrocyte co-culture. While the MultiDA procedure displayed promising results favouring neural conversion of hMSCs in the early phases of culture, subsequent to this the cells then appeared to revert slightly toward a MSC-like phenotype by the final MultiDA stage, in terms of morphology and gene expression.

Observations that cultures were not maintained well in the later phases of the MultiDA culture system raise possible questions regarding the responsiveness of hMSCs to SHH, FGF-8 and GDNF. Previous reports have established that addition of these cytokines is beneficial for the *in vitro* dopaminergic neuronal differentiation of MSCs (Dezawa *et al.*, 2004; Fu *et al.*, 2006b; Jiang *et al.*, 2003; Trzaska *et al.*, 2007). In addition, MSCs have been shown to express both components of the SHH receptor,

PTC and *SMO* (Kramer *et al.*, 2006; Trzaska *et al.*, 2007), as well as FGF receptors *FGFR1* and *FGFR4* (Kramer *et al.*, 2006). Furthermore, treatment of MSCs with SHH and FGF-8 revealed increased DNA synthesis indicating that these receptors were functional (Kramer *et al.*, 2006). However, it was also found that only the *GFRA1* component of the GDNF receptor was present on MSCs, whereas the *c-RET* component was absent, rendering MSCs incapable of responding to stimulation with GDNF (Kramer *et al.*, 2006).

These results suggest possible variability in MSC expression of cytokine receptors, which may account for later parts of the MultiDA method failing to elicit the expected differentiation effects. Underlying factors that should also be taken into consideration include differences in the isolation procedures, species, culturing methods, and cytokine concentrations used across the different studies.

4.3.5 Beneficial Effect of FGF-2 and EGF in hMSC Neuronal Differentiation

The reversion phenomenon observed in the MultiDA procedure could be prevented by continuous addition of S1 cytokines (FGF-2 and EGF) to all MultiDA stages, which maintained the expression of neural genes and a refractile neuronal-like morphology, as well as inhibited the up-regulation of mesodermal gene *COL1A1*. Moreover, significant differences were observed between the gene expression of cells cultured in the presence or absence of FGF-2/EGF in stages 2 and 3 of the MultiDA method, including significantly higher expression of *NES*, *NR4A2*, *MAP-2*, *NANOG*, *ENO2*, and *GFAP* in FGF-2/EGF-containing cultures, and significantly lower expression of *COL1A1*. Although continued FGF-2/EGF supplementation appeared capable of maintaining an immature neuronal-like phenotype in MultiDA-treated hMSCs, no further advancement of neuronal-like characteristics could be observed.

Many other studies have achieved neuronal differentiation of MSCs through the application of FGF-2 (Jiang *et al.*, 2002; Kim *et al.*, 2002a; Suzuki *et al.*, 2004), EGF (Sanchez-Ramos *et al.*, 2000), or FGF-2 and EGF (Hermann *et al.*, 2004; Jin *et al.*, 2003; Suon *et al.*, 2006; Tao *et al.*, 2005), in conjunction with other differentiation agents, providing independent support of the usefulness of these cytokines for this purpose (discussed further in Section 1.3.1). Nevertheless, prior to use in MSC differentiation protocols, FGF-2 and EGF were described as critical factors for stimulating survival and proliferation of rodent multipotential neural progenitor/stem cells in culture (Gritti *et al.*, 1999; Gritti *et al.*, 1996; Ray *et al.*, 1993; Reynolds *et al.*, 1992; Richards *et al.*, 1992), and *in vivo* (Kuhn *et al.*, 1997; Wagner *et al.*, 1999). This role in regulating neuronal progenitors and SCs, therefore, opens the possibility that the continued addition of these cytokines in later stages of the MultiDA method may cause the retention of cells in an immature neuronal-like or neuronal progenitor-like state.

4.3.6 Recently Published SingleDA Method Yielded Similar Immature Neuronal-Like Cells as the MultiDA and SingleND Procedures

A single-step cytokine-based procedure for inducing a dopaminergic neuronal phenotype from hMSCs was reported by Trzaska and colleagues during the course of our studies (Trzaska *et al.*, 2007). This method utilised a cocktail of cytokines similar to those which were implemented in our work, i.e. SHH, FGF-8, and FGF-2, and generated promising results including the highest reported efficiency of TH-positive cell production (~67%), expression of dopaminergic-specific genes, synthesis and secretion of DA, and the presence of electrophysiological properties comparable with immature neurons. The period of differentiation was brief and only encompassed 12 days, in comparison with the 3 week differentiation period of our MultiDA and SingleND methods. However, despite detection of DA release *in vitro*, this was not found to occur

in response to depolarisation through elevated extracellular K^+ , and instead DA was found to be constitutively released into the culture medium.

Examination of the excitable properties provided further confirmation that the cells obtained were not mature neurons, with the whole-cell patch clamp technique only detecting robust outward K⁺ currents, while inward Na⁺ and Ca⁺⁺ currents were only weakly exhibited in a small fraction of cells, and spontaneous or evoked action potentials were not detected. In addition, although Trzaska and colleagues demonstrated up-regulation of dopaminergic-specific markers, the examination of mesodermal and MSC markers was not conducted, and in order to show clear dopaminergic neuronal differentiation, one would expect to observe concomitant down-regulation of markers of MSCs and the mesodermal lineage.

Since this recent report still demonstrated the most promising results for the derivation of DA-producing cells from hMSCs via a cytokine-based procedure, the SingleDA method was also examined for comparison with the MultiDA and SingleND systems. Using the SingleDA method morphological changes and up-regulation of NES and β tubulin III expression similar to that reported by Trzaska, *et al.* could be obtained, and slight increases in *TH* expression were also detected; however, the majority of the results obtained were dissimilar. In addition, increased expression of dopaminergic neuronal genes (including *NR4A2*, *EN2*, and *DAT*), and neuronal proteins (including TH and NeuN) could not be detected. As a result, the SingleDA method could only generate immature neuronal-like cells from the hMSC populations studied, and did not appear to provide any significant advantages over the MultiDA and SingleND methods.

4.3.7 Transcription Factor Expression During hMSC Dopaminergic Neuronal Differentiation

According to current knowledge, the development of midbrain dopaminergic neurons *in vivo* is also dependent on cell intrinsic programs, of which several critical transcription factors have been identified (discussed further in Section 1.4.3.2). It is thought that MSCs undergoing dopaminergic neuronal differentiation would elicit transcription factor networks similar to those involved in dopaminergic neuron development *in vivo*. Therefore, I investigated the involvement of several of these crucial transcription factors (*NR4A2*, *PITX3*, and *EN2*) during the course of the MultiDA, SingleND and SingleDA procedures.

Of the transcription factors examined, *NR4A2* was the most highly and consistently expressed across the various culture methods. However, apart from S1 and S2# of the MultiDA method, all treatments showed a trend of decreased expression compared to the control. Whereas for *PITX3* and *EN2*, expression was not detected in all samples, and in general, there was little change in expression with the three different treatments, although a non-significant trend of increased expression was observed. Taken together, these results indicate that exposure to the three differentiation methods did not direct hMSCs toward a dopaminergic neuronal fate. However, of the different treatments examined the FGF-2/EGF-containing stages of the MultiDA method appeared to induce the greatest up-regulation of *NR4A2*, *EN2* and *PITX3* expression, although this did not reach a level of significance.

Since the transcription factors studied here are thought to be involved in the differentiation and survival of dopaminergic neurons, rather than the initial specification and generation of midbrain dopaminergic precursors, more information may be garnered from examining earlier transcription factors, particularly since the hMSC-

derived neuronal-like cells in the present study appear fairly immature. Earlier transcription factors that could be investigated include LMX1A, msh homeobox 1, and NEUROG2 (discussed in Section 1.4.3.2), as well as FOXA2, which has recently been described to regulate the specification, neurogenesis and differentiation of midbrain dopaminergic neurons (Arenas, 2008; Kittappa *et al.*, 2007).

4.3.8 Continued hMSC Expression of Mesodermal and Pluripotency Markers Following *In Vitro* Neuronal Differentiation

The cytokine induction methods employed in this chapter were only successful in driving hMSCs toward an early neuronal fate, with cells continuing to express neuronal progenitor marker NES and pluripotency markers *POU5F1* and *NANOG*. Furthermore, the persistence of mesodermal and glial markers, together with the pluripotency markers, suggests that additional differentiation signals are required for attaining proper neuronal differentiation and maturation. A comparative study conducted by Hermann and colleagues (Hermann *et al.*, 2006) also found continued expression of mesodermal marker fibronectin in certain neuronal differentiation methods investigated, including the initial methods reported for MSC neuronal differentiation using cytokine-based (Sanchez-Ramos *et al.*, 2000) and chemical-based systems (Woodbury *et al.*, 2000). Examination of mesodermal lineage markers was not performed in these initial reports, therefore, the findings of Hermann and colleagues highlight the importance of showing concomitant down-regulation of mesodermal and MSC markers with neuronal differentiation.

The findings of this thesis and those of Hermann, *et al.* suggest that further differentiation signals or factors may be required for the *in vitro* neural differentiation of MSCs, particularly since a complex milieu of cytokines and signalling molecules are believed to be present in neural development *in vivo*. A number of potential factors have

been identified to provide advantages for the neuronal and dopaminergic neuronal differentiation of MSCs. One such factor involves co-culturing MSCs together with neural support cells, such as astrocytes, which may function through direct cell-cell interactions, trophic factor production, or removal of glutamate from the culture environment (Jiang *et al.*, 2003). Support for this approach has also been shown in a study examining ESC and astrocyte co-culture systems, which found that the type of neurons obtained were dependent upon the regional origin of the astrocytes, with mesencephalic astrocytes strongly potentiating dopaminergic neurogenesis (Roy *et al.*, 2006). Another approach similar to the co-culturing method is the supplementation of cultures with conditioned medium from neural support cells (Joannides *et al.*, 2003).

Furthermore, other studies have described improved MSC neuronal differentiation through the addition of fatty acids, such as docosahexaenoic acid and arachidonic acid (Kan *et al.*, 2007b), and inflammatory mediator IL-1 α (Cho *et al.*, 2005; Greco *et al.*, 2007a); whilst, still others have suggested that beneficial effects could be produced through additional supplementation with RA, BDNF and/or NGF (Trzaska *et al.*, 2007), or through the optimisation of culture surface nanotopography (Yim *et al.*, 2007). Aside from the improvement and optimisation of culture components, the time required for *in vitro* dopaminergic neuronal differentiation requires careful consideration, as it may differ between species and may also differ to that required for other neuronal cell types.

In an earlier study, Bjorklund and colleagues transplanted ESCs into the 6-OHDA unilaterally-lesioned Parkinsonian rat and demonstrated that the time course of behavioural recovery corresponded with the developmental rate (i.e. length of gestational period) of the donor species (Bjorklund *et al.*, 2002). Thus, if dopaminergic differentiation of hMSCs does indeed mimic that which occurs during gestation, then it

would be expected that longer periods are required for the differentiation of human cells than for rodent cells. Evidence supporting this has been shown in a meta-analysis of studies which found that behavioural recovery in Parkinsonian rats occurred approximately 5 weeks after mouse foetal ventral mesencephalon transplantation, while for the equivalent human cells, recovery only occurred after approximately 20 weeks (Isacson *et al.*, 1997).

Longer periods of *in vitro* hMSC differentiation may therefore be required, particularly since most protocols, including our own, have only been conducted over several weeks, which is considerably shorter than the 20 week period reported for *in vivo* behavioural recovery using human foetal cells. However, it should be noted that neural differentiation of hMSCs may be divergent from that occurring in normal physiological development, particularly since undifferentiated hMSCs have been shown to constitutively express some neural markers.

While the current study was being completed, a number of reports were published describing the induction of a dopaminergic phenotype in MSCs. The procedures employed included: multiple-stage application of cytokines FGF-2 (1 week), SHH+FGF-8 (1 week) followed by BDNF (1 week) (Bouchez *et al.*, 2008); treatment with FGF-2/EGF for 48 hours as step 1, then application of BHA, dbcAMP, IBMX, RA and GDNF for 96 hours as step 2 (Kan *et al.*, 2007a), and a later study by the same group, which found improved TH expression with addition of dbcAMP, IBMX and AA with BDNF alone or GDNF/RA/TGF- β 3 in the second step (Barzilay *et al.*, 2008); application of CoCl₂ and Y-27632, to presumably cause hypoxia-inducible factor 1 activation and Rho kinase inhibition (Pacary *et al.*, 2006); and treatment with cytokines FGF-2 and GDNF, and 10% foetal calf serum, for 2 weeks (Zhang *et al.*, 2008). A varying degree of success was achieved, including depolarisation-induced secretion of DA (Barzilay *et al.*, 2008; Kan *et al.*, 2007a; Zhang *et al.*, 2008), and reduction of behavioural deficits in a Parkinsonian rat model (Bouchez *et al.*, 2008). Furthermore, it was demonstrated for the first time that hMSCs isolated from PD patients could be induced toward a DA-secreting neuronal phenotype, with expression of TH protein in approximately 30% of cells (Zhang *et al.*, 2008).

Another study examining a subpopulation of MSCs, termed marrow-isolated adult multilineage inducible (MIAMI) cells, also described an alternative multi-step procedure for the generation of TH-positive cells (Tatard *et al.*, 2007). This method involved neural specification (step 1) with FGF-2 and 5-20% FBS for 24 hours, neuronal commitment (step 2) using BME, neurotrophin-3, SHH, FGF-8, and RA for 2 days, and finally, neuronal differentiation (step 3) with BHA, KCl, VPA, forskolin, hydrocortisone, insulin, neurotrophin-3, NGF, BDNF, and GDNF for 3-7 days. However, while research efforts are continuing to reveal new insights into the induction of a dopaminergic phenotype from MSCs, further improvements and optimisation of established techniques is still very much desired.

4.3.9 Functional Assessment of hMSC-Derived Neuronal-Like Cells is Required

The present study focused on examining hMSC neuronal differentiation *in vitro* by morphological and gene and protein expression analysis; however, true neuronal differentiation must be confirmed by investigating the electrophysiological properties and *in vivo* function of neuronal-like cells. The function of immature neuronal cells could be investigated by analysing electrophysiological function (Ambrogini *et al.*, 2004), performing further *in vitro* differentiation to attain a mature phenotype capable of firing voltage-gated action potentials and neurotransmitter-mediated synaptic communication with other neurons, or by transplantation to demonstrate functional integration in the host brain and alteration of host behaviour.

The cells described in this chapter exhibit an immature neuronal phenotype, therefore, future experiments are warranted for investigating whether the electrophysiological properties of these neuronal-like cells resembles that of immature neurons. Rat MSC-derived neuronal-like cells obtained through co-culture with mouse cerebellar granule neurons have been shown to acquire the electrophysiological properties of maturing newborn adult neurons (Wislet-Gendebien *et al.*, 2005), according to the maturation stages described by Carleton, *et al.* (Carleton *et al.*, 2003). These cells appear to progress through the first stage of neuronal maturation with cells displaying a low resting membrane potential and response to some neurotransmitters, but absence of Na⁺ currents and spiking behaviour.

A second stage of maturation could also be reached with cells exhibiting both Na^+ and K^+ currents, a more negative resting membrane potential and the ability to fire single-spike action potentials. However, the cells obtained by Wislet-Gendebien, *et al.* were unable to reach the final stage of maturation, i.e. mature neurons capable of firing trains of spikes, exhibiting strong synaptic activities and possessing a very negative resting membrane potential (Wislet-Gendebien *et al.*, 2005). A number of other studies have also found evidence for the acquisition of electrophysiological properties by MSCs that supports the neuronal differentiation potential of these cells (Cho *et al.*, 2005; Dezawa *et al.*, 2004; Hung *et al.*, 2002; Jiang *et al.*, 2003; Kohyama *et al.*, 2001; Mareschi *et al.*, 2006; Song *et al.*, 2007; Tropel *et al.*, 2006).

Achievement of functional mature neurons would have potential clinical relevance for the treatment of neurodegenerative diseases, although, prior rigorous testing in animal models is necessary. In addition, the use of immature but committed neurons may prove advantageous for increasing survival and functional integration posttransplantation, particularly since terminally differentiated neurons with elaborate axonal connections are known to poorly survive detachment and subsequent transplantation procedures (Bjorklund *et al.*, 2000; Hermann *et al.*, 2004).

4.3.10 Contaminating Cells, Culture Artifacts or Cell-Cell Fusion Events Cannot Explain the Observed hMSC Neuronal Differentiation

As described earlier, transdifferentiation has recently been met with skepticism arising from experimental findings over the past few years that have questioned the authenticity of observations of transdifferentiation, especially neuronal differentiation from MSC populations. These findings dictate the need for caution when interpreting the results of transdifferentiation studies. The main areas of concern include the possibilities that MSC neural transdifferentiation events may actually be due to: (1) contaminating cells of the neural lineage present within heterogeneous MSC populations; (2) rare culture artifacts resulting from repeated *in vitro* passaging or toxicity of differentiation agents; and (3) possible fusion of MSCs with neural cells present in the culture system or *in vivo* following transplantation (Jin *et al.*, 2003).

In the present study hMSCs were isolated by negative selection from BM aspirates and cultured under conditions favouring MSC expansion. The BM is innervated by a rich network of sympathetic autonomic nerve fibers, but the cell bodies of these neurons resides in paravertebral sympathetic ganglia (Calvo, 1968; Yamazaki *et al.*, 1990), therefore, it is unlikely that neural cells would be present in these cultures. Furthermore, hMSCs were passaged 4-6 times prior to differentiation in MSC culture medium lacking the factors necessary for *in vitro* neuronal maintenance, so that even if neuronal cells were initially present, these contaminating cells would be lost over the expansion period.

In addition, following cytokine-based differentiation neuronal-like characteristics were acquired by the majority of cells, which is inconsistent with the possibility of a small contaminating subpopulation of cells being responsible for the detection of a neuronal phenotype. Moreover, I have also demonstrated that these findings are unlikely to be caused by culture artifacts through the real-time analysis of differentiation using live cell imaging techniques, as well as examination of long-term serial passaging, which showed no significant changes to the phenotype of MSCs during the first 6-8 passages of culture, and the absence of spontaneous neural differentiation. Finally, neural cells were not utilised in our differentiation systems, therefore, these findings cannot be the result of cell-cell fusion events.

4.3.11 Summary

In conclusion, this study has demonstrated that cytokine-based neuronal induction of hMSCs elicits changes resulting from active and dynamic processes that involve outgrowth and motility of cellular extensions, which to my knowledge has not been previously reported. These observations were found to be entirely distinct to the rapid epiphenomena of cytotoxicity and cytoskeleton disruption associated with chemical-based neuronal induction methods. Furthermore, it was shown that long-term subcultivation of hMSCs did not compromise the ability of these cells to differentiate toward an early neuronal fate. In addition, cytokine-induced neuronal differentiation of hMSCs through the MultiDA, SingleND and SingleDA procedures were associated with up-regulation of genes and proteins indicative of an immature neuronal phenotype. However, none of the methods investigated was capable of generating mature midbrain dopaminergic neurons, although, the MultiDA method was found to yield the highest increases in NES and *NR4A2* expression. As a result, further analysis was performed in Chapter 5 using hMSCs exposed to the first stage of the MultiDA method (FGF-2/EGF). Next, the functional capacity of these phenotypically immature hMSC-derived

neuronal cells, as well as undifferentiated hMSCs, were assessed through transplantation into the 6-OHDA unilaterally-lesioned rat model of PD.

CHAPTER 5

IN VIVO STUDIES IN AN ANIMAL MODEL OF

PARKINSON'S DISEASE

5.1 Introduction

In vitro and in vivo studies describing the neurogenic potential and disease ameliorating effects of MSCs have stimulated interest and support for applying MSCs in cellular treatments of CNS diseases and injuries. The use of MSCs for cell-based therapies is associated with further advantages, such as, overcoming of ethical and logistical limitations tied to the use of ESCs and foetal tissue-derived cells, and also avoidance of possibilities of transplant rejection and teratoma formation. However, the fate and function of MSCs following transplantation *in vivo* remains a topic of much debate and investigation.

The initial reports establishing the engraftment of MSCs and restoration of neural deficits following intracerebral, intravenous or intra-arterial transplantation were reviewed in Section 1.2.3. Engraftment and migration of MSCs was observed in neonatal mice and embryonic rats following injection into the lateral ventricles (Kopen et al., 1999; Munoz-Elias et al., 2004), as well as in adult rats after direct injection into the uninjured corpus striatum (Azizi et al., 1998). Furthermore, these studies demonstrated efficacy of MSC transplantation in promoting neurological recovery in rodent models of cerebral ischemia (Chen et al., 2001; Li et al., 2000; Zhao et al., 2002) and traumatic brain injury (Lu et al., 2001a; Mahmood et al., 2003; Mahmood et al., 2002; Mahmood et al., 2001). Initially it was believed that neurological recovery occurred as a result of conversion of MSCs to a neural phenotype and integration into host circuitry, although more recently it has been suggested that recovery may instead be related to the secretion of trophic factors by MSCs, or interaction of MSCs with host cells (Chen et al., 2001; Li et al., 2001a; Zhao et al., 2002). In addition, further mechanisms have been proposed to explain observations of MSC-mediated functional improvement in animal disease models, including modulation of immune responses,

mitochondrial transfer and stimulation of endogenous host responses (Phinney *et al.*, 2007; Prockop, 2007).

Cellular transplantation therapy has long been thought to hold great potential for the treatment of PD, since this chronic neurodegenerative disorder results from the selective loss of dopaminergic neurons in a distinct region of the brain (see Section 1.4.1 and 1.4.2 for further details). A variety of cell types have been examined in the search for a potential renewable source of DA-producing cells, including human foetal brain tissue (Brundin et al., 1986; Brundin et al., 1988; Stromberg et al., 1986), ESCs (Bjorklund et al., 2002; Kim et al., 2002b; Lee et al., 2000), and adult NSCs and neural progenitors (Ostenfeld et al., 2000; Studer et al., 1998; Svendsen et al., 1997; Svendsen et al., 1996). These studies have yielded encouraging findings in animal models including graft survival, DA production and alleviation of lesion-induced motor deficits, providing evidence supporting the concept that cellular transplantation therapy may be viable for PD. Furthermore, while clinical trials examining transplantation of human foetal mesencephalic tissue in PD patients have produced conflicting results in the past, the findings of recent studies have restored hope in the possibility of future cell-based therapies for PD (Kordower et al., 2008; Li et al., 2008; Mendez et al., 2008; Piccini et al., 1999).

As described earlier, MSCs appear to have great potential for application in cellular therapies. Support for the potential utility of MSCs in PD therapy has been shown through studies demonstrating the acquisition of dopaminergic neuronal traits by MSCs *in vitro*. The findings of our *in vitro* neuronal differentiation studies presented in Chapter 4 suggest that MSCs are capable of acquiring markers typical of immature dopaminergic neurons, and further reports discussed in Section 1.4.4 have also described the expression of a range of dopaminergic specific markers, presence of

electrophysiologically active voltage-gated sodium channels, and/or DA secretion in response to depolarising stimuli (Dezawa *et al.*, 2004; Fu *et al.*, 2006); Guo *et al.*, 2005; Hermann *et al.*, 2004; Jiang *et al.*, 2003; Suon *et al.*, 2006; Trzaska *et al.*, 2007). In addition, engraftment and functional improvement have been demonstrated following transplantation of undifferentiated MSCs (Hellmann *et al.*, 2006; Li *et al.*, 2001b) and differentiated MSCs (Dezawa *et al.*, 2004; Fu *et al.*, 2006b) in rodent models of PD. These studies have been reviewed in Section 1.4.4.

However, in these reports a range of approaches have been taken with some conflicting results obtained, and furthermore the studies are not agreed on which methodology produces the optimal outcomes. As a result, a number of pertinent issues remain to be resolved, including the engraftment and survival capability of undifferentiated and partially dopaminergic-differentiated MSCs post-transplantation, the optimal site for intracerebral transplantation, the potential immunological responses to undifferentiated and differentiated MSC grafts, and whether undifferentiated MSCs or partially dopaminergic-differentiated MSCs are optimal for transplantation into an experimental PD model. The present study aims to extend the findings discussed above by examining these issues.

Once considered as having the unique ability to suppress immune responses both *in vitro* and *in vivo*, MSCs are now recognised to possess immunomodulatory properties with the capacity to suppress and/or activate immune responses (Locatelli *et al.*, 2007; Stagg, 2007), as discussed in Section 1.1.5. In addition, while the brain has been traditionally viewed as an immunologically privileged organ, evidence suggesting otherwise is now apparent and reveals routine and effective immunological surveillance of the CNS (Hickey, 2001). Consequently, transplanted cells and tissues are exposed to host immunological responses, particularly since the transplantation procedure elicits

inflammatory processes due to surgical trauma (Barker *et al.*, 2004; Widner *et al.*, 1988). Contrary to earlier findings, recent evidence has also been reported during the course of this study that suggests lack of MSC survival following intracerebral transplantation into normal or hemiparkinsonian rats (Coyne *et al.*, 2006; Suon *et al.*, 2006), raising further questions concerning the ability of MSCs to engraft and mature after transplantation into the CNS.

Another cell type that has demonstrated potential for application in autologous cellular therapies for injuries and diseases of the CNS are OECs, which can be easily obtained through biopsy of the nasal olfactory mucosa, as described by investigators including my co-supervisor (Bianco et al., 2004; Mackay-Sim, 2005; Ruitenberg et al., 2006). OECs are known to possess unique properties that are advantageous in the transplantation setting, including the secretion of a variety of growth factors, such as NGF, BDNF, GDNF, and neuregulin (Boruch et al., 2001; Fairless et al., 2005; Woodhall et al., 2001), which are beneficial for increasing neurite formation and exerting neuroprotective effects. OECs also express extracellular matrix proteins with axon and neurite growth promoting properties, such as laminin, fibronectin, and cell adhesion molecules like neural cell adhesion molecule and L1 cell adhesion molecule (Doucette, 1990; Ramon-Cueto et al., 1992). Furthermore, transplantation of OECs has been shown to promote axonal regeneration and behavioural recovery in animal models of spinal cord injury (Lu et al., 2001b; Lu et al., 2002). In addition, co-transplantation of OECs with foetal ventral mesencephalic cells into the lesioned striatum of hemiparkinsonian rats resulted in significant functional restoration, in terms of amphetamine-induced rotational behaviour, spontaneous locomotor activity, DA levels, and TH immunoreactivity (Agrawal et al., 2004; Johansson et al., 2005). Therefore, in

light of these findings the present study also examined an OEC co-transplantation strategy.

The present study aimed to investigate the function of hMSCs through in vivo transplantation studies rather than examining *in vitro* function by electrophysiological testing, since the differentiation procedures employed in Chapter 4 were only capable of generating immature neuronal-like cells, and the CNS environment in vivo may provide additional signals allowing further maturation. The experimental model employed in this study was the 6-OHDA unilaterally-lesioned rat (discussed further in Section 1.4.1.5), as this is the accepted universal standard model for assessing PD therapies, and enables quantifiable assessment of behavioural function through amphetamine-induced rotational asymmetry. Initially, this study aimed to examine the effects of transplanting undifferentiated MSCs into the striatum of hemiparkinsonian rats, in terms of amphetamine-induced rotational behaviour and graft phenotype over 3 months post-transplantation (referred to as Part I; see Section 2.2.7.2). The experimental results generated from Part I led to a change in experimental approach. Therefore, Part II of this study was performed to examine the survival, differentiation, and anatomical integration of neuronal-primed hMSCs over a short-time course postintracerebral transplantation (see Section 2.2.7.2). In addition, neuronal-primed hMSCs were transplanted into the striatum and substantia nigra to determine the optimal site for transplantation, and co-transplantation with OECs was also examined to determine if this provides an advantage or synergistic effect. Furthermore, potential neurotrophic and immunomodulatory effects of hMSCs on endogenous host cells were investigated, since it has been suggested that these mechanisms may play a role in MSC-mediated improvement or restoration of neural deficits.

5.2 Results – Part I

5.2.1 Long-Term Transplantation of Undifferentiated hMSCs in PD Model

Initially, undifferentiated hMSCs were transplanted intrastriatally into 6-OHDA unilaterally lesioned Parkinsonian rats obtained from Taconic Farms Inc. (NY, USA), for assessing behavioural function and graft phenotype over 3 months post-transplantation. However, due to the poor health status of the animals the study was adjusted to have an earlier experimental endpoint at slightly over 2 months. This was most likely the result of a number of factors, including the impact of transportation from the USA, differences in animal strain, size and sex, and also the high severity of the lesion, which resulted in additional loss of VTA dopaminergic neurons together with the expected loss of nigral dopaminergic neurons (discussed in Section 1.4.1.1). In combination, these factors appear to have resulted in weaker than expected animals. Consequently, in Part II of this study the rat PD model was sourced from Integrative Neuroscience Facility (Melbourne, AUS) to minimise the impact of transportation.

Nevertheless, some information was obtained from these experiments. Immunofluorescent analysis of the graft regions revealed the absence of engraftment of undifferentiated hMSCs in the lesioned striatum of hemiparkinsonian rats. Instead, only evidence of astrocytic infiltration and/or proliferation could be seen at the graft site and along the needle tract (Figure 5.1). Furthermore, the core of the graft and needle tract was found to contain autofluorescent matter resembling cell debris and lipofuscin granules (Figure 5.2), suggesting that cellular destruction had occurred. Since the hMSCs had been cultured in phenol red-containing culture medium prior to transplantation, and phenol red uptake by cells can cause autofluorescence, confocal three-dimensional reconstruction analysis was performed to investigate this possibility.

Analysis of the autofluorescent core using confocal 3-dimensional reconstruction confirmed that the autofluorescent signal did not co-localise with DAPI-stained nuclei, therefore establishing that phenol red uptake was not responsible for the observed autofluorescence (see 'Supplementary File 2' video on Supplementary CD). This finding also provided additional support for the absence of hMSC engraftment. Examination of behavioural function by amphetamine-induced rotational testing showed no improvement at 1 month post-transplantation (data not shown). Assessment of later time points (2 months and 3 months post-transplantation) were intended, but could not be performed due to the early experimental endpoint.

These data suggested that the loss of donor hMSCs could be the result of: 1) the inability of hMSCs to survive in the host striatal environment, resulting in cell death and phagocytosis, or 2) rejection of hMSCs through a host-mediated immune response. Additionally, the health of the rats and high severity lesion may have contributed to the absence of engraftment.



Figure 5.1 Presence of astrogliosis around undifferentiated hMSC grafts and needle tract regions at 2 months post-transplantation.

Representative mosaic tile scan showing immunohistological examination of the graft site following intrastriatal transplantation into the lesioned hemisphere of the 6-OHDA-induced hemiparkinsonian rat (n = 8). GFAP-positive astrocytes (red) were found to be concentrated around the graft site and along the length of the needle tract, while this pattern of astrocytic distribution was absent from the contralateral hemisphere. Autofluorescent material could also be observed (green). Nuclei were counterstained with DAPI (blue).



Figure 5.2 Evidence of autofluorescent cellular debris and lipofuscin in undifferentiated hMSC grafts at 2 months post-transplantation.

Immunohistological analysis revealed the presence of autofluorescent material (cellular debris and lipofuscin granules; shown in green) within graft cores and along needle tracts, suggesting that cellular destruction had occurred. The representative image depicted here is a maximum projection z-stack of the graft site containing an autofluorescent core and surrounded by GFAP-positive astrocytes (red). Nuclei were counterstained with DAPI (blue). Scale bar: $25 \,\mu$ m. (For confocal 3-dimensional reconstruction see 'Supplementary File 2' video on Supplementary CD).

5.3 Results – Part II

As a consequence of the findings obtained in Part I, a new experimental approach was taken for Part II in order to evaluate the survival, engraftment and differentiation of neuronal-primed hMSCs within the initial 3 weeks post-intracerebal transplantation. The concerns raised in Part I were addressed through purchasing the rat PD model from a source within Australia, and also using older, male rats of a different strain. Other approaches taken in attempts to increase the likelihood of hMSC engraftment, included (1) priming hMSCs toward the neuronal lineage prior to transplantation; (2) injection into the substantia nigra as well as the striatum; and (3) co-transplantation with OECs.

5.3.1 Graft Site Identification

For the identification of graft sites, regions of the striatum and substantia nigra were screened for expression of astrocyte marker, GFAP, and neuronal progenitor marker, NES, using immunohistological techniques. In OEC co-transplanted animals, the GFP signal from the OECs also aided graft identification. A marked astrogliosis was observed in graft regions and along the needle tract as early as 7 days posttransplantation (Figure 5.3), similar to the observations in Part I at the much later timepoint of 2 months post-transplantation. However, at day 1 the human cell grafts only appeared as a dense mass or cluster of cells (DAPI-stained nuclei) along the needle tract, sometimes containing rare NES-positive cells. In addition, autofluorescent cell debris and lipofuscin-like material were found within the graft sites, as seen previously in Part I. Also, it was found that transplantation of neuronal-primed hMSCs into the striatum or substantia nigra in the presence or absence of OECs yielded comparable results.



Figure 5.3 Identification of neuronal-primed hMSC graft sites and comparison with the contralateral hemisphere.

Representative results of immunohistological screening of the striatum and substantia nigra for identification of graft sites, using expression of GFAP (red) and NES (green), as well as GFP signal detection for OEC co-transplanted recipients (n = 3). (A) Needle tract entry site in Day 7 nigral graft and (B) corresponding region in contralateral hemisphere. (C) Striatal graft site at Day 7, and (D) corresponding region in contralateral hemisphere. Similar to results in Part I, graft sites contained autofluorescent material and were surrounded by astrogliosis, which were absent in the contralateral hemisphere. Images are maximum projection z-stacks. Similar findings were obtained with transplantation into either site in the absence or presence of OECs. Nuclei were counterstained with DAPI (blue). Scale bar: 150 µm.

5.3.2 Limited Survival of Neuronal-Primed hMSCs and OECs Post-Transplantation

Apart from the autofluorescent material detected within graft sites, the presence of intact healthy nuclei was also found suggesting the possible survival of transplanted cells. Therefore, the survival and engraftment of neuronally-primed hMSCs and OECs following transplantation in the striatum and substantia nigra were examined by immunohistological staining for human nuclear antigen (HNA; Figure 5.4). Cells displaying co-localisation of HNA and DAPI signals were found within striatal and nigral graft sites at day 1 post-operatively (Figures 5.4A and G). Surviving human cells were typically visible as a mass of cells toward the end of the needle tract region, with few HNA-positive cells detected at the needle tract entry point (Figure 5.4F), and none seen in the contralateral hemisphere (Figure 5.4E). Over time, a decrease in HNA staining was apparent with day 7 grafts showing little remaining HNA signal (Figure 5.4B and H), and complete absence by day 21 (Figure 5.4C and I). In addition, autofluorescence within the graft core increased in parallel with HNA loss, and shrinking graft size and cellular density. Surprisingly, little differences were observed between grafts containing neuronal-primed hMSCs alone or in combination with OECs. Concomitant loss of the GFP signal from OECs provided further support for the inability of transplanted human cells to persist within the host environment.

Confocal z-series analysis of graft sites was also performed to verify the colocalisation of HNA and DAPI staining, and to exclude possible misinterpretation due to overlaying fluorescent signals. Through this analysis HNA signals were indeed found to be co-localised with DAPI-stained nuclei (arrows in Figure 5.5) for both hMSC only and OEC co-transplanted grafts. Therefore, the presence of surviving human cells was confirmed; however, only limited survival could be detected following intracerebral transplantation.



Figure 5.4 Evidence of limited survival of neuronal-primed hMSCs and OECs in the lesioned hemisphere of the hemiparkinsonian rat. (Refer to next page for Figure 5.4 Legend text)

Figure 5.4 Evidence of limited survival of neuronal-primed hMSCs and OECs in the lesioned hemisphere of the hemiparkinsonian rat.

The survival and engraftment of neuronal-primed hMSCs and GFPpositive OECs (green) were examined by immunohistological staining using an antibody against HNA (red). Representative images following transplantation of (A-F) neuronal-primed hMSCs alone, and (G-I) neuronal-primed hMSCs co-transplanted with OECs (n = 3). Surviving human cells could be detected at day 1 post-transplantation (A and G). However, by day 7 little HNA signal remained (B and H), and this could no longer be detected by day 21 (C and I). Higher magnification view of (A) is shown in (D), with human nuclear staining seen to be co-localised with DAPI-stained nuclei. Increasing presence of autofluorescent matter within the graft core was observed in parallel with the loss of human cells. This autofluorescence and human nuclear staining were not observed in the corresponding regions in the contralateral hemisphere (representative image shown in (E), which is contralateral to (A)). The majority of human cells were concentrated in the needle tract with few, if any, detected at the needle tract entry site (F). Similar findings were observed in both striatal and nigral graft sites. Images shown are maximum projection z-stacks. Nuclei were counterstained with DAPI (blue). Scale bar: 150 µm in (I) for all images, except (D) which has scale bar: 30 µm.



Figure 5.5 Confocal z-series identification of human cells at graft sites.

Immunohistological examination of graft sites with antibodies to HNA (red) demonstrates co-localisation with DAPI-stained nuclei (blue) within the nigral graft site at day 1 (arrows). Similar results obtained for striatal graft sites and OEC co-transplanted animals, as well as at the day 7 time point. Scale bar: $30 \mu m$.
5.3.3 Accumulation of Astrocytes and Microglia/Macrophages at Graft Sites

To investigate the loss of neuronal-primed hMSCs and OECs following transplantation, graft sites from both the striatum and substantia nigra were characterised for the presence of glial cells by immunohistological analysis (Figure 5.6). Microglia and monocytes/macrophages were identified using the Iba-1 marker and astrocytes were detected by GFAP expression.

Both striatal and nigral grafts displayed massive infiltration by Iba-1-positive microglia/macrophages and GFAP-positive astrocytes that increased over the 3 week examination period (Figure 5.6 A-C). Initially, little glial accumulation was detected at day 1 post-operatively. However, by day 7 microglia/macrophages could be detected in a radial pattern around the graft core, which contained autofluorescent material. Increased astrocytic presence was also found surrounding the graft site and extending beyond the microglia/macrophage layer. Glial accumulation was observed to persist until 21 days post-transplantation, which was the last time point examined. Furthermore, the microglia/macrophages found at graft sites displayed an activated appearance, with increased intensity of Iba-1 immunoreactivity (Ito *et al.*, 1998), and alterations in morphology including retraction and thickening of processes, and increased cytoplasmic area (Ito *et al.*, 1998; Streit *et al.*, 1988).

The corresponding regions in the contralateral hemisphere did not exhibit glial accumulation (Figure 5.6 D-F). The microglia here possessed a quiescent morphology with small cell bodies and fine cytoplasmic ramifications. Interestingly, examination of injection sites in the sham controls also revealed evidence of increasing glial accumulation over the 3 week period (Figure 5.6 G-I), suggesting that this may be a response to injury caused by needle insertion during surgery. These results suggest that

intracerebral transplantation of hMSCs and OECs may have elicited an inflammatory response that led to the destruction of donor cells.



Figure 5.6 Increasing glial accumulation at graft sites during the initial 3 weeks post-intracerebral transplantation in the hemiparkinsonian rat.

(Refer to next page for Figure 5.6 Legend text)

Figure 5.6 Increasing glial accumulation at graft sites during the initial 3 weeks post-intracerebral transplantation in the hemiparkinsonian rat.

Immunohistological analysis revealed the accumulation of GFAPpositive astrocytes (red) and Iba-1-positive microglia/macrophages (green) at graft sites (n = 3). Representative images depicting (A-C) neuronal-primed hMSC graft sites in the striatum, (D-F) corresponding regions in the contralateral hemisphere, and (G-I) striatal injection sites in sham controls, at day 1 (left column), day 7 (middle column) and day 21 (right column). (A) Initially, at day 1 post-transplantation little GFAP or Iba-1 was detected. (B) By day 7, grafts were densely surrounded by a layer of GFAP-positive and Iba-1-positive cells, and contained an autofluorescent core (white arrow). (C) At the end of the 21 day period a marked astrogliosis was present in the graft region. Both striatal and nigral grafts exhibited this glial accumulation, which suggests the presence of an inflammatory response at the graft sites. (D-F) This pattern of glial distribution was absent from the contralateral hemisphere. (G-I) Whereas, a similar response is observed in the sham controls, which also exhibited autofluorescent cellular debris at injection sites (aqua colour). Images shown are maximum projection z-stacks. Nuclei have been counterstained with DAPI (blue). Scale bar: 150 µm.

5.3.4 Endogenous Host Neuronal Differentiation Post-Transplantation

The current study has also characterised the graft regions for host neuronal progenitor activity, since transplantation of hMSCs has been proposed to elicit endogenous host responses that may account for the functional improvements observed in animal models of neurological disease. Immunohistological analysis of graft sites for the expression of neuronal progenitor marker NES (Figure 5.7) revealed little, if any, NES immunoreactivity in the graft regions at any period. The predominant host-derived cells that were found to express NES were GFAP-positive astrocytes located immediately surrounding the grafted cells. Expression of NES and GFAP by astrocytes is an indicator of activation (Clarke et al., 1994), which is consistent with the increased accumulation of astrocytes around the graft sites. A few rare NES-positive cells could be detected within the graft core and at the astrocytic border of the graft at day 7 (Figure 5.7D). However, these NES-positive cells were not detected often, and were typically found in the graft core, therefore, it is unlikely that these NES-positive cells represented endogenous host neuronal progenitor involvement. Instead, this NES expression may be derived from grafted neuronal-primed hMSCs, which have been shown to express the NES gene and protein (see Chapter 4). Furthermore, this explanation is consistent with the lack of NES expression at day 21, since the transplanted neuronal-primed hMSCs were absent from this final time point. In addition, no differences were observed between striatal and nigral grafts of neuronalprimed hMSCs, or with grafting in the presence or absence of OECs. Interestingly, similar to previous findings in Part I, cavitations and autofluorescence were seen within the graft cores suggesting the presence of cell destruction.



Figure 5.7 Absence of endogenous neuronal differentiation in host brain after transplantation of neuronal-primed hMSCs.

Striatal and nigral grafts were also examined for the presence of NES (green) and GFAP (red) at (A) day 1, (B) day 7 and (C) day 21 posttransplantation (n = 3). Higher magnification view of (B) shown in (D). Rare NES-positive cells were only detected within the graft core or in the immediate astroglial boundary at day 7, suggesting that neuronalprimed hMSCs do not elicit host neuronal differentiation, but express some NES protein. NES-positive activated astrocytes could also be detected. Similar results were observed in OEC co-transplanted recipients. Representative images shown are maximum projection zstacks. Nuclei were counterstained with DAPI (blue). Scale bar: 150 μ m in (C) for all images, except (D) which has scale bar: 75 μ m.

5.3.5 Characterisation of Graft Sites for Neuronal Marker Expression

Since neuronal progenitor marker NES was detected in a small proportion of cells within the graft site, further immunohistological characterisation of grafts for the expression of neuronal markers TH and β tubulin III was also conducted. Staining for dopaminergic neuronal marker TH confirmed the unilateral expression of TH in the unlesioned hemisphere of the 6-OHDA-induced hemiparkinsonian rats. Dopaminergic neurons of the VTA were also found to be intact. TH immunoreactivity was predominantly absent from the graft sites (Figure 5.8 A-C), regardless of whether neuronal-primed hMSCs were transplanted alone or in combination with OECs, and regardless of the site of transplantation. Although, strong TH signal could be detected in the corresponding unlesioned contralateral hemisphere (Figure 5.8 D-F). Sham-treated animals also did not exhibit TH expression at injection sites (Figure 5.8 K and L). However, it was observed that the human cell grafts were not always administered into the correct regions, particularly for the substantia nigra. This could be due to the small size of the substantia nigra, and the difficulty in accessing this region of the CNS in stereotactic surgical procedures.

Striatal and nigral graft sites containing neuronal-primed hMSCs were also found to lack expression of neuronal marker β tubulin III (Figure 5.9 A-C), and similar results were observed with OEC co-transplantation. In addition, β tubulin III immunoreactivity was absent from the entire transplantation region, whereas the corresponding contralateral region showed consistent and evenly distributed expression of β tubulin III (Figure 5.9 D), suggesting that trauma associated with the surgical procedures caused loss of endogenous host β tubulin III protein. The findings obtained through immunohistological analysis of TH and β tubulin III expression indicate that neuronal-primed hMSCs do not undergo further differentiation or maturation *in vivo* after intracerebral transplantation into the striatum or substantia nigra of hemiparkinsonian rats.

5.3.6 Characterisation of Graft Sites for Extracellular Matrix Deposition

The presence of fibronectin deposition at striatal and nigral graft sites was also examined using immunohistological techniques (Figure 5.8). A dense fibronectinpositive matrix was observed surrounding neuronal-primed hMSC grafts as early as day 1 post-transplantation, however, immunoreactivity appeared to decrease over time with very little fibronectin remaining at day 21 (Figure 5.8 A-C). Similar findings were obtained with the transplantation of neuronal-primed hMSCs in combination with OECs (Figure 5.8 G-I), and with both striatal and nigral transplantation. Only slight fibronectin expression could be detected in the corresponding contralateral regions, at much lower levels than that present around and within graft sites (Figure 5.8 D-F). Furthermore, analysis of striatal and nigral injection sites in sham controls revealed very little, if any, fibronectin matrix deposition at all periods examined (Figure 5.8 J-L). These results suggest that the dense fibronectin matrix present at graft sites was produced by the transplanted human cells, rather than being deposited by host cells as part of a scar formation process. Observations that fibronectin was drastically decreased in parallel with the loss of hMSCs and OECs provide further support for this explanation, and also suggest a potential role for microglia/macrophages in removing fibronectin deposits. Additional evidence supporting graft-derived fibronectin production can be found in the results of Section 4.2.3.3, which demonstrate continued expression of fibronectin by neuronal-primed hMSCs in vitro.

233



Figure 5.8 Matrix deposition and lack of dopaminergic neuronal differentiation after transplanting neuronal-primed hMSCs and OECs.

(Refer to next page for Figure 5.8 Legend text)

Figure 5.8 Matrix deposition and lack of dopaminergic neuronal differentiation after transplanting neuronal-primed hMSCs and OECs.

Fibronectin and TH expression were assessed in striatal and nigral grafts at day 1 (left column), day 7 (middle column) and day 21 (right column). Representative images are shown of (A-C) neuronal-primed hMSC graft sites in the striatum, (D-F) corresponding regions in the contralateral hemisphere, (G-I) co-transplanted hMSC and OEC nigral graft sites, and (J-L) injection sites in sham controls (n = 3). Immunohistological analysis revealed a fibronectin-positive matrix surrounding hMSC and OEC-containing graft sites from day 1 post-transplantation (A, B, G, and H), however, this was drastically reduced by day 21 (C and I). The concomitant loss of transplanted cells suggests that the fibronectin matrix was deposited by the graft. Furthermore, sham injected animals did not exhibit dense fibronectin expression (J-L). TH expression was not detected in any graft sites or sham-injected sites, and was only expressed in the unlesioned hemisphere (D-F). Images shown are maximum projection z-stacks. Nuclei were counterstained with DAPI (blue). Scale bar: 150 µm.



Figure 5.9 Absence of neuronal marker expression within graft sites following transplantation of neuronal-primed hMSCs.

Striatal and nigral grafts were also examined for the presence of β tubulin III (green) and HNA (red) at (A) day 1, (B) day 7 and (C) day 21 post-transplantation (n = 3). Expression of β tubulin III was missing from the region containing the graft sites, and was primarily found surrounding the graft core. While, the corresponding regions in the contralateral hemisphere exhibited evenly distributed β tubulin III expression (representative image shown in (D), which is contralateral to (A)). Similar results were also obtained in OEC co-transplanted recipients. The representative images shown are maximum projection z-stacks. Nuclei were counterstained with DAPI (blue). Scale bar: 150 µm.

5.4 Discussion

The central findings of the present study have shown (1) evidence of limited survival and engraftment of undifferentiated and neuronal-primed hMSCs in the lesioned hemisphere of the 6-OHDA-induced hemiparkinsonian rat, with loss of hMSCs as early as 7 days post-grafting; (2) absence of further differentiation of neuronalprimed hMSCs in vivo in both the striatum and substantia nigra, suggesting that neither site contains factors involved in promoting the survival and dopaminergic neuronal hMSCs; differentiation of (3) accumulation of activated astrocytes and microglia/macrophages around graft sites, indicating the presence of an inflammatory response; (4) co-transplantation of OECs in combination with neuronal-primed hMSCs does not confer graft survival or differentiation advantages, and instead, OECs are also lost early after transplantation; (5) absence of endogenous host neuronal differentiation in response to hMSC transplantation; and (6) evidence suggesting continued production of fibronectin extracellular matrix by hMSCs after intracerebral transplantation.

The majority of published reports in this field have demonstrated some level of MSC-induced restoration of neurodegeneration in Parkinsonian animal models, whether it be renewed presence of immunoreactivity to dopaminergic markers within the lesioned hemisphere or functional improvement in behavioural asymmetry (reviewed in Section 1.4.4). In contrast, the current study has found limited survival and engraftment of hMSCs together with the continued absence of TH-positive neurons from the lesioned hemisphere following hMSC transplantation. A number of possible explanations may account for these observations, including the rejection of hMSCs by a host immune response, lack of appropriate factors for supporting hMSC survival and engraftment within the environment of the adult lesioned brain, insufficient differentiation of hMSCs toward a midbrain dopaminergic neuronal phenotype,

237

inadequate techniques for cellular transplantation, and misleading results of previous studies that employed thymidine analog labelling of donor cells as the sole method for graft identification.

5.4.1 Activation of Innate Immune Responses with Intracerebral hMSC Transplantation

In the present study evidence was found of vigorous microglia/macrophage infiltration in response to grafted neuronal-primed and undifferentiated hMSCs in an animal disease model of PD. This response occurred in the presence of cyclosporine immunosuppression and was also detected in sham control animals. In contrast, MSCs have long been considered to possess the ability to suppress adaptive immune responses. More recently, however, evidence to the contrary has arisen, which indicates that MSCs may not be intrinsically immune privileged (discussed in greater depth in Section 1.1.5). Furthermore, the CNS is now understood to be capable of effective immunological surveillance (Barker *et al.*, 2004; Hickey, 2001).

In the wake of these findings, reports published during the course of this study have described similar findings to that presented in this chapter, that is the rejection of transplanted MSCs following intracerebral transplantation in rodents (Camp *et al.*, 2009; Coyne *et al.*, 2006; Suon *et al.*, 2006). Coyne and colleagues demonstrated that transplantation of MSCs into the intact adult rat hippocampus or striatum elicited an immediate inflammatory response involving massive infiltration of activated microglia/macrophages and astrocytes in the presence of cyclosporine-mediated immunosuppression. This led to the rapid rejection of hMSC grafts with near complete rejection by 7 days, a much briefer period than rejection through adaptive immune responses, which generally commence 10-14 days post-transplantation (Coyne *et al.*, 2006). The inflammatory response was ascribed to the mechanical trauma produced by

intracerebral transplantation. Prior to the findings from this thesis and from Coyne and colleagues, the inflammatory response elicited during MSC transplantation had not been evaluated. Subsequently, Camp and colleagues (Camp *et al.*, 2009) also demonstrated the generation of robust cellular immune responses (MHC class I- and class II-expressing cells and $CD4^+$ and $CD8^+$ lymphocytes) following the injection of undifferentiated rat MSCs into the striatum of allogeneic hemiparkinsonian recipients. However, despite the presence of a marked immune response, some grafted MSCs were still detectable at 22-24 days post-transplantation; although MSC administration was not found to prevent behavioural deficits or DA depletion, and MSCs did not acquire the ability to synthesise TH (Camp *et al.*, 2009).

Furthermore these findings are in accordance with the current understanding of the role of microglia/macrophages in orchestrating the inflammatory response of CNS innate immunity (Giulian, 1987; Perry et al., 1995). Previous studies have reported that mechanical injury to the brain provoked rapid and focal microglial activation to shield the injured area, with possible immediate phagocytic engulfment and removal of damaged tissue (Davalos et al., 2005; Nimmerjahn et al., 2005). Microglia-mediated inflammation has also been associated with astrogliosis (Giulian et al., 1994; Matsumoto et al., 1992), and Giulian and colleagues have described the production of pro-inflammatory cytokine IL-1 by microglia, which is capable of inducing astrogliosis (Giulian et al., 1994). In addition, the pattern of microglial and astrocytic distribution observed in the present study bore resemblance to the pattern of microglial and reactions inflammatory astroglial to lesions in experimental autoimmune encephalomyelitis described by Matsumoto and colleagues (Matsumoto et al., 1992). This pattern consisted of (1) early response of microglia to inflammatory lesions; (2) proliferation of microglia close to lesions, with astrocytes encasing the lesions; and

(3) formation of micro-astroglial scars at later stages that were composed of residual inflammatory cell aggregates and dense microglial and astrocytic gliosis (Matsumoto *et al.*, 1992). Additional evidence for the induction of a microglial-mediated innate inflammatory response with SC transplantation was found in a recent study examining transplantation of Müller SCs into an experimental model of retinal degeneration (Singhal *et al.*, 2008). Extensive microglial accumulation was found in this study, in association with poor migration, integration and survival of Müller SCs (Singhal *et al.*, 2008).

In the current study, microglial- and astroglial-reactivity were found to coincide with fibronectin-reactivity. However, the majority of fibronectin deposition was observed to occur at hMSC graft sites with very little fibronectin present at sham injection sites, suggesting that the primary source of fibronectin may be the transplanted hMSCs. Similarly, Coyne, et al. have described the presence of fibronectin-reactivity in association with the marked astrogliosis surrounding MSC graft sites in the intact rat striatum and hippocampus (Coyne et al., 2006). On the contrary, the authors ascribed the presence of the dense fibronectin-positive matrix at the transplantation site to be solely indicative of scar formation. Another study by Singhal and colleagues also reported the deposition of chondroitin sulfate proteoglycans at Müller SC transplantation sites, which were thought to have been released by microglia and possibly contributed to the inhibition of cell migration and integration (Singhal et al., 2008). However, in the context of this thesis, the deposition of fibronectin at graft sites by neuronal-primed hMSCs may have also contributed to the lack of survival and engraftment, and may have resulted from the incomplete neuronal differentiation of hMSCs.

240

Further support for the rejection of hMSC transplants by an inflammatory response in this thesis is found in another study examining the early cellular responses mediating rejection of xenogeneic porcine endothelial cells and foetal neurons transplanted into the rat striatum (Remy et al., 2001). Surprisingly, the work by Remy, et al. demonstrated rapid rejection of endothelial cell grafts within 11 days through an inflammatory response mediated by microglia/macrophages, whereas rejection of foetal mesencephalic neurons occurred after 3 weeks and was typical of an adaptive immune response involving T cells. The differences in rejection kinetics were suggested to arise from the intrinsic characteristics of the xenogeneic donor cells, such as differences in MHC-I surface expression and production of pro-inflammatory cytokines. These findings also indicate the possibility that transplantation of non-neural cell types into neural environments elicits an immunological response distinct to that resulting from transplantation of neural cell types. In addition, since MSCs are known to secrete a wide variety of growth factors, including GM-CSF and others that support granulocyte and macrophage proliferation and maturation (Guba et al., 1992; Kittler et al., 1992), Coyne and colleagues have commented that the production of these factors by MSCs posttransplantation could augment post-operative inflammation, as well as aid graft destruction (Coyne et al., 2006). Further to this, GM-CSF has been described to be a potent mitogen for microglia, with the ability to induce reactive microglia capable of scavenging and removal of damaged tissue (Giulian et al., 1994).

In the present study, and in the study by Coyne, *et al.* (Coyne *et al.*, 2006), immunosuppression using cyclosporine was employed yet MSC graft rejection still occurred. Czech, *et al.* have also found that cyclosporine administration could not prevent or delay the rejection of xenogeneic bovine endothelial cells grafted into the rat CNS (Czech *et al.*, 1997). Since cyclosporine treatment acts primarily to inhibit T lymphocyte-mediated immunity (Borel et al., 1976; Borel et al., 1977), these findings indicate the involvement of other arms of the immune system, such as innate immunity, in MSC graft rejection. In the current study xenogeneic hMSCs were transplanted into the rat CNS, and other studies have reported that cyclosporine treatment for neural xenotransplantation is suboptimal (Larsson et al., 2000; Larsson et al., 2001). Furthermore, reports have suggested the involvement of complement factors, immunoglobulins and macrophages in immune responses against discordant xenografts, which are not affected in a crucial way by cyclosporine treatment (Barker *et al.*, 2000; Larsson et al., 1999; Wallgren et al., 1995). Therefore, in this thesis the observations of of transplanted xenogeneic neuronal-primed hMSCs, rapid loss massive microglial/macrophage and astrocyte accumulation at the graft site, and inability of cyclosporine treatment to avert these outcomes, together support the notion of graft rejection through host innate immune responses.

Administration of further immunosuppression may therefore be required for improved MSC graft survival. Enhanced microglial suppression was employed by Singhal and colleagues for Müller SC transplantation into degenerated retina through application of oral cyclosporine A, azathioprine, prednisolone, and indomethacin (Singhal *et al.*, 2008). This immunosuppressive regimen yielded decreased microglial accumulation and increased survival of Müller SCs. Other studies investigating the improvement of xenograft survival in the CNS have found promising outcomes through the use of triple drug therapy (cyclosporine, methylprednisolone, azathioprine), complete complement blockade, and absence of an induced primate anti-mouse antibody response (Cicchetti *et al.*, 2003); tacrolimus or cyclosporine treatment with additional inductive treatment using prednisolone or mycophenolate mofetil (Wennberg *et al.*, 2001); or local immunosuppression by co-transplantation of liposomal tacrolimus (Alemdar *et al.*, 2007).

Another potential factor affecting neuronal-primed hMSC engraftment *in vivo* is the immunogenicity of hMSCs following *in vitro* differentiation. At present, the majority of studies have examined the immunomodulatory properties of MSCs in the undifferentiated state, and little is known of whether this is affected by *in vitro* differentiation. Recently, *in vitro* chondrogenic differentiation has been reported to increase the immunogenicity of rat MSCs, resulting in the stimulation and maturation of human DCs, possibly due to MSC up-regulation of costimulatory B7 molecules after chondrogenic differentiation (Chen *et al.*, 2007b). Furthermore, the presence of stimulated DCs was required for chondrogenic-differentiated MSC induction of human peripheral blood lymphocyte proliferation and cytotoxicity. On the other hand, osteogenic and adipogenic differentiation did not appear to alter the immunological properties of MSCs (Chen *et al.*, 2007b). Therefore, the immunological properties of MSCs (Chen *et al.*, 2007b). Therefore, the immunological following *in vitro* differentiation may depend on the resulting cell type obtained through differentiation.

Another recent study investigated the effects of *in vitro* cytokine-based neuronal differentiation on hMSC expression of immunogenic molecules and stimulation of allogeneic peripheral blood lymphocyte proliferation (Liu *et al.*, 2006). While, increased hMSC expression of HLA molecules and costimulatory CD80 molecules was detected over the course of neuronal differentiation, the proliferation of allogeneic peripheral blood lymphocytes could not be induced. Instead, neuronal differentiated hMSCs suppressed the proliferation of peripheral blood lymphocytes induced by allogeneic lymphocytes or mitogens, through mechanisms involving cell-cell contact and soluble factors, such as TGF- β 1 and IL-10. In addition, IFN- γ treatment enhanced the

expression of HLA molecules by neuronal differentiated hMSCs, suggesting that an inflammatory environment *in vivo* could elicit increased immunogenicity of hMSCs. However, IFN-γ exposure was unable to significantly up-regulate the expression of CD40, CD80 or CD86, indicating that functional costimulatory signalling on neuronal differentiated hMSCs was incomplete. Also, peripheral blood lymphocyte proliferation was suppressed to a similar extent even after IFN-γ treatment of hMSCs cultured in neuronal differentiation medium (Liu *et al.*, 2006). Further studies are required to more clearly elucidate the immunogenicity and immunomodulatory properties of hMSCs after neuronal differentiation, and it will be of interest to evaluate this in the neuronal-primed hMSCs in the present study to determine whether these factors contributed to the rapid demise of hMSC grafts after *in vivo* transplantation.

5.4.2 Absence of Further hMSC Neuronal or Dopaminergic Neuronal Differentiation at Both Striatal and Nigral Graft Sites Post-Transplantation

Examination of striatal and nigral graft sites containing neuronal-primed hMSCs with or without OEC co-transplantation revealed the absence of further neuronal or dopaminergic neuronal differentiation and maturation. In the same way, comparable absence of neural maturation was described by Coyne and colleagues following transplantation of undifferentiated MSCs into the hippocampus and striatum of normal adult rats (Coyne *et al.*, 2006). This was attributed to the intact adult brain possessing limited capacity to direct the differentiation of transplanted SCs. It is understood that the survival and establishment of functional connections by neuronal grafts is dependent on host variables, occurring maximally during the foetal period and diminishing with postnatal development (Bjorklund *et al.*, 2000). Functional integration of donor neurons can still occur in the adult brain, however, the capacity for this is increased by damage to the CNS (Bjorklund *et al.*, 2000).

Similar lines of thought have also been applied to the transplantation of SCs into the CNS. Suon and colleagues transplanted hMSC-derived neural progenitor spheres into the lesioned striatum of 6-OHDA-induced hemiparkinsonian rats, but were unable to detect graft survival beyond 4 weeks post-transplantation (Suon et al., 2006). In addition, although in vitro differentiation systems had yielded cells capable of expressing dopaminergic and GABAergic traits, after in vivo transplantation hMSCderived neuroprogenitor spheres were unable to differentiate further into TH-expressing cells, although many were found to express GABA. The authors proposed that hMSCs transplanted into the striatum had preferentially differentiated into GABA-expressing neurons due to the local differentiation factors present in the striatum, which is intrinsically comprised of 90% GABA neurons (Suon et al., 2006). In the present study, neuronal-primed hMSCs were also transplanted into the striatum as this is the region requiring DA provision, and analogous results of poor graft survival and absence of TH expression were also observed. We also performed transplantation into the substantia nigra to determine whether graft survival and differentiation could be improved, since the cell bodies of midbrain dopaminergic neurons are contained in the substantia nigra and the specific cues for dopaminergic differentiation may be present here rather than in the striatum. However, no differences were seen between nigral grafts and striatal grafts, and neuronal-primed hMSCs were lost from both regions by 21 days posttransplantation.

In contrast, evidence supporting the *in vivo* and *in vitro* transdifferentiation of hMSCs into functional dopaminergic neurons was reported in a recent study by Shetty and colleagues (Shetty *et al.*, 2009). In this study, undifferentiated hMSCs ($2-3 \times 10^5$) cultured under xenofree conditions were transplanted into the substantia nigra of 6-OHDA unilaterally-lesioned rats. Significant motor improvement was observed from

4 weeks post-transplantation onwards and reduced apomorphine-induced rotations were seen after 12 weeks. Furthermore, histological analysis revealed survival of grafted cells and differentiation into dopaminergic neurons. *In vitro* differentiation of hMSCs using a 2-step procedure (pre-induction with 10% cord blood serum, 2% B27, 2 ng/mL FGF-2, 100 ng/mL NGF, and 50 ng/mL Noggin for 1 week, followed by 200 μM BHA for 4-5 hours) also yielded cells with neuronal morphology and gene and protein expression, which were capable of secreting DA into the culture medium (Shetty *et al.*, 2009). Therefore, despite difficulties encountered in this field, some hope remains in the ability of MSCs to be implemented in cell-based therapies for PD.

5.4.3 Further Commitment of hMSCs to the Neuronal Lineage Prior to Transplantation May Be Necessary for Intracerebral Engraftment

Both the present study and that by Suon, *et al.* (Suon *et al.*, 2006) had utilised partially neuronal-differentiated or neuronal progenitor-like cells for transplantation, since it is believed that immature neurons are more suitable for transplantation (Bjorklund *et al.*, 2000; Hermann *et al.*, 2004), and confirmation of this was seen in pilot studies conducted by Suon and colleagues (Suon *et al.*, 2006). Yet, neither study could obtain long-term graft survival. The stage of SC differentiation that is optimal for transplantation purposes remains to be determined, although, in general it is still thought that immature neuronal cells possess the greatest potential for successful engraftment and integration into the host circuitry.

Continued deposition of fibronectin matrix was observed at graft sites in the current study, perhaps indicating that hMSCs still possessed primitive properties and had not been directed far enough along the neuronal differentiation pathway. It is possible that excessive production of fibronectin by hMSCs may have elicited a stronger inflammatory reaction at graft sites by stimulating the removal of unphysiological amounts of fibronectin by microglia/macrophages. Furthermore, transplantation of undifferentiated MSCs into the hemiparkinsonian rat model resulted in graft loss by 3 weeks, presumably due to the non-neural/foreign nature of these cells within the CNS, whereas MSC-derived mature dopaminergic neurons were capable of engraftment and functional improvement (personal communication, M. Dezawa). Nevertheless, other studies have also reported fibronectin production by hMSCs following transplantation into the ischemic and normal rat brain, without adverse effects on survival and engraftment (Azizi *et al.*, 1998; Zhao *et al.*, 2002).

A narrow window of opportunity may exist during the neuronal differentiation of SCs that permits successful transplantation and integration into the CNS, but outside of this period cells may be either too mature to survive transplantation or not yet committed enough to complete differentiation *in vivo*. Therefore, additional studies are necessary to further define this optimal window for transplantation.

5.4.4 Unresolved Technical Aspects of hMSC Transplantation into the CNS

A number of technical aspects remain undetermined concerning the transplantation of MSCs into experimental models of CNS diseases. One of the primary concerns is the improvement of cell survival and differentiation post-transplantation. Graft survival can be linked to a number of factors, including the quantity of SCs used for transplantation. In the present study, undifferentiated and neuronal-primed hMSCs were transplanted at a dose of 100,000 cells per site, in accordance with the majority of reported studies examining MSC transplantation in Parkinsonian rodent models (Dezawa *et al.*, 2004; Fu *et al.*, 2006b; Hellmann *et al.*, 2006). However, the absence of engraftment we observed was not seen in these studies. For NSCs, transplantation of between 150,000 and 300,000 cells per site has been reported to be adequate for yielding graft survival, however, it was still suggested that use of higher numbers of

247

SCs may result in higher survival rates (Svendsen *et al.*, 1996). Whereas, a separate report comparing the transplantation of 200,000, 1 million or 2 million human neural precursors into 6-OHDA hemiparkinsonian rats found that smaller grafts elicited greater neuronal fibre extension and less immunological rejection in comparison with the larger grafts (Ostenfeld *et al.*, 2000). For MSCs, it was found that primate MSC engraftment into the primate CNS was enhanced when low doses (500,000) rather than high doses (2.5 million) of MSCs were injected intracerebrally (Isakova *et al.*, 2007). Therefore, the results of the present study may be improved through increasing the number of hMSCs transplanted to a quantity more similar to that used in successful neural SC transplantation.

Few transplanted hMSCs or OECs were observed to migrate out of graft sites and needle tract regions raising the question of whether migration and integration were prevented or inhibited. The glial scar formed at graft sites contains an inhibitory proteoglycan component, and the presence of inhibitory proteins such as the chondroitin sulfate proteoglycans ACAN, versican, and neurocan has been reported to inhibit axon guidance and prevent neurite outgrowth and axon regeneration (Fitch *et al.*, 1997; Horner *et al.*, 2002; Jones *et al.*, 2003; McKeon *et al.*, 1999). Recent studies have attempted to improve graft survival and integration by creating a more permissive microenvironment at graft sites through chondroitinase ABC treatment to reduce inhibitory proteoglycans (Fouad *et al.*, 2005; Singhal *et al.*, 2008). Singhal and colleagues employed combined transplantation of Müller SCs together with chondroitinase ABC for promotion of matrix degradation and cell migration in the degenerated retina (Singhal *et al.*, 2008). Chondroitinase ABC treatment in conjunction with enhanced microglial suppression (oral cyclosporine A, azathioprine, prednisolone, indomethacin) was found to result in dramatic improvement in migration of Müller SCs into the retina, decreased microglial accumulation, significantly higher numbers of transplanted Müller SCs, and a characteristic neuronal morphology in the migrating cells. These results suggest that abnormal extracellular matrix deposition and activation of innate inflammatory responses may constitute major barriers to retinal SC transplantation (Singhal et al., 2008). Furthermore, combination of OEC and Schwann cell transplantation with chondroitinase ABC resulted in improved axonal regeneration and locomotor recovery after complete transection of the spinal cord (Found et al., 2005). Alternatively, MSCs have been transplanted in the presence of heparin and DNase to improve cell survival and engraftment in experimental murine models of acute myocardial infarction (personal communication, K. Atkinson). Difficulties with migration through inhibitory proteoglycans and extracellular matrices may hold true for the current study, and as a consequence, the application of these treatments may be beneficial for the transplantation of undifferentiated and neuronal-primed hMSCs into the 6-OHDA hemiparkinsonian rat model. In addition, the findings of a recent study suggested that manipulation of the graft niche to facilitate cross-talk between transplanted SCs and the diseased brain may be beneficial for more efficacious transplantation in chronic lesions or advanced stages of neurodegenerative diseases in the aged CNS (Ourednik et al., 2009). Evidence of this was observed through the overexpression of neural cell adhesion molecule L1 in donor NSCs and in recipient transgenic MPTP-lesioned mice (over-expression in host astrocytes under the GFAP promoter), which resulted in rapid and extensive distribution of donor NSCs, rescue of dysfunctional host dopaminergic neurons, and enhanced differentiation of donor NSCs into TH-expressing neurons (Ourednik et al., 2009). These findings offer further options for more efficacious transplantation of SCs into the diseased CNS, and further studies are warranted.

Other strategies aimed at improving the survival and differentiation of SC and neuronal progenitor grafts have been suggested, including the addition of trophic factors, such as FGF-2 (Bhang *et al.*, 2007; Takayama *et al.*, 1995), or antioxidants (Nakao *et al.*, 1994) to the cell suspension prior to transplantation. Furthermore, little is known of the optimal timing or therapeutic window for SC transplantation post-injury. A study examining systemic infusion of MSCs for the treatment of liver fibrosis found that immediate MSC transplantation post-injury was able to significantly reduce liver damage and collagen deposition, whereas delaying MSC infusion by 1 week hampered the prevention of disease progression (Fang *et al.*, 2004). Unfortunately, no conclusions have been reached on which techniques or combinations of techniques provide the best outcomes and further investigation into these aspects is necessary.

5.4.5 Controversies Associated with Cell Identification Techniques

Further confusion has emerged through recent unexpected findings indicating that commonly used cell-labelling techniques were not as reliable as originally thought, and could result in the transfer of donor labels to host cells (Burns *et al.*, 2006; Coyne *et al.*, 2006). Burns and colleagues were the first to describe the *in vivo* release of thymidine analogs, BrdU, chlorodeoxyuridine and tritiated thymidine, from transplanted MAPCs and subsequent incorporation into the DNA of dividing host neural cells (Burns *et al.*, 2006). Added confirmation of thymidine analog transfer was demonstrated through the transplantation of labelled dead cells (by repeated freeze-thawing) or fibroblasts, leading the authors to suggest that rigorous controls or alternative cell identification techniques are required when transplanting into highly proliferative environments, such as the developing, neurogenic or injured brain (Burns *et al.*, 2006). Shortly thereafter Coyne, *et al.* also reported the transfer of BrdU and bis benzamide from donor MSCs to host phagocytes, astrocytes and neurons, which falsely suggested

MSC engraftment and differentiation up to 12 weeks, despite histological evidence of graft destruction by 14 days (Coyne *et al.*, 2006). Therefore, these cell-labelling techniques may provide misleading indication of donor cell survival and differentiation, and cast doubts on the findings of earlier studies that solely used these labelling techniques for graft identification.

In the present study, we avoided the use of these cell labels, and instead employed techniques of immunohistological detection through human-specific antibodies, as well as GFP-transgene fluorescence in OECs. Furthermore, our results were consistent with those reported by Burns, *et al.* and Coyne, *et al.* in that only limited survival of hMSCs could be detected post-transplantation. The discrepancy in results between these studies and the earlier studies demonstrating MSC engraftment and integration may therefore be explained by the methods of cell detection utilised in the various studies.

Although GFP fluorescence was not found to be transferable to host cells (Burns *et al.*, 2006), caution must also be taken in interpreting cell identification through GFP fluorescence techniques alone, since loss or reduction of transgene expression can also lead to false negative results. Further difficulties with the detection of donor cells post-transplantation have been discussed in Section 1.2.4, and include issues arising from the presence of endogenous autofluorescence, particularly from dying cells, which can be observed at similar wavelength as used for GFP detection (Goodell, 2003). Such autofluorescence was detected in the present study.

5.4.6 Transplantation of hMSCs Did Not Elicit Endogenous Neurotrophic Effects

An increasing number of studies have now described the ability of MSCs to promote tissue repair and functional improvement, despite little evidence of differentiation and only low or transient levels of *in vivo* engraftment (Phinney *et al.*, 2007; Prockop, 2007). These studies have suggested that MSCs may not repair tissues solely through differentiation and replacement of lost cells, but may possess a more prominent role in the secretion of factors with the ability to alter the host tissue microenvironment. The absence of engraftment observed in this thesis reflects certain aspects of this recent perspective. However, in the current study investigation of endogenous host NSC activation by immunohistological detection of NES expression found no response of host NSCs to hMSC transplantation. Staining indicative of NSCs could not be detected around graft regions, and host NES expression could only be detected in activated GFAP-positive astrocytes surrounding the needle tract and graft site (see Figure 5.7). In addition, transplantation of hMSCs did not result in differences in microglia/macrophage and astrocyte accumulation around the needle tract and graft regions when compared with the sham-treated controls (see Figure 5.6).

While we were unable to find a neurotrophic response in host cells following intracerebral hMSC transplantation, this has been reported in a number of recent studies. Munoz and colleagues demonstrated that hMSC transplantation into the murine hippocampus stimulated the proliferation, migration and differentiation of endogenous NSCs into neural precursors and mature neural cells (Munoz *et al.*, 2005). The mechanism underlying the increased neurogenesis was suggested to involve secretion of NGF, VEGF, CNTF, FGF-2, and other chemokines by hMSCs, which may have acted directly on NSCs, or could have activated astrocytes that then produced the increase in neurogenesis (Munoz *et al.*, 2005).

Another recent study showed that hMSC transplantation in an experimental model of transient forebrain ischemia led to improved neurologic function through hMSC expression of anti-immune, anti-inflammatory and anti-apoptotic-related factors and establishment of a Th2-immune bias through alternative activation of microglia

252

and/or macrophages by hMSCs (Ohtaki *et al.*, 2008). These studies, together with findings from many different sources have indicated the involvement of multiple mechanisms in MSC-mediated tissue repair, including the stimulation of survival and proliferation of endogenous host cells, modulation of immune responses, induction of angiogenesis and reduction of apoptosis (Phinney *et al.*, 2007; Prockop, 2007).

Although difficulties were encountered in the present study with the application of hMSCs in PD cellular therapy, efforts in this field have continued and promising results have been reported during the course of our study. The strategy employed in this thesis of *in vitro* dopaminergic neuronal differentiation of MSCs prior to transplantation has also been examined in 2 recent studies, which implemented *in vitro* differentiation procedures containing cytokines similar to those utilised in our methods (Bouchez *et al.*, 2008; Levy *et al.*, 2008). Apart from differences in the *in vitro* differentiation techniques employed, the *in vivo* studies in this thesis also differed from these two reports in that higher numbers of MSCs were transplanted in both these studies (180,000 and 500,000 respectively). In addition, unlike these reports we were unable to detect graft survival even with neuronal-priming of hMSCs prior to transplantation.

The study by Bouchez and colleagues employed a 3 week sequential culture protocol involving FGF-2, FGF-8, SHH and BDNF, which had been adapted from a previous study (Jiang *et al.*, 2002). The differentiation method resulted in incomplete engagement of MSCs through the neural pathway, but intrastriatal transplantation of these cells still yielded a stable 50% reduction in rotational asymmetry, and partial restoration of dopaminergic markers and the vesicular striatal pool of DA. Surprisingly, it was also found that transplantation of undifferentiated MSCs resulted in equivalent behavioural improvement. Recovery was observed to occur rapidly by 1 week postgrafting, and was maintained during the 6 week study period. This finding, together

with the detection of BDNF, GDNF, FGF-2 and FGF-8 expression *in vitro* by undifferentiated and neuronal-differentiated MSCs, suggested that neurotrophic factor secretion could be the mechanism behind recovery (Bouchez *et al.*, 2008).

The study conducted by Levy, *et al.* utilised a 2-stage method of neuronal differentiation containing growth factors (FGF-2, EGF), chemicals (BHA, dbcAMP, IBMX), RA, and docosahexaenoic acid (Levy *et al.*, 2008). Transplantation of these differentiated hMSCs also resulted in improvement in rotational behaviour, and was of superior benefit when compared to grafting of undifferentiated hMSCs. In contrast to the previous study by Bouchez, *et al.*, the authors of this study suggested that recovery was due in part to dopaminergic cell replacement, since donor hMSCs were found to survive more than 130 days post-transplantation and migrated to the damaged nigra, with a small proportion of cells also expressing TH (Levy *et al.*, 2008).

As previously mentioned, the majority of recently published studies have concentrated on the growth factor secretory effects of MSCs in Parkinsonian rodents *in vivo* (Bahat-Stroomza *et al.*, 2009; Sadan *et al.*, 2009; Shintani *et al.*, 2007), or in neural co-culture systems *in vitro* (Jin *et al.*, 2008; Sadan *et al.*, 2009; Shintani *et al.*, 2007). Shintani and colleagues reported the release of BDNF, GDNF and FGF-2 by murine MSCs into culture media, and found that treatment of embryonic dopaminergic neurons with MSC-conditioned medium prior to intrastriatal transplantation yielded significantly enhanced graft survival and more rapid recovery from rotational deficits in the 6-OHDA hemiparkinsonian rat model (Shintani *et al.*, 2007).

Two studies conducted by the Offen laboratory used a different strategy of directing hMSCs toward an astrocyte-like, neurotrophic factor-secreting phenotype (Bahat-Stroomza *et al.*, 2009; Sadan *et al.*, 2009). These cells were then found to be capable of eliciting behavioural improvement in pharmacologically-induced rotation

tests and non-pharmacological motor tests, as well as enhanced striatal DA levels, and increased TH immunoreactivity after intrastriatal transplantation. Interestingly, a different 6-OHDA-induced hemiparkinsonian experimental model was used in these studies. Instead of the complete lesion model used in our studies and in the majority of published reports, a partial lesion model with residual surviving DA terminals was employed, which was chosen as the authors were interested in examining the protectionrestoration effect of trophic factor-secreting MSCs. The authors also commented that neurotrophic factor-secreting cells are not expected to be beneficial in models with complete lesioning, and that these models may only show improvement with dopaminergic cell replacement (Bahat-Stroomza *et al.*, 2009). Similar to our findings, absence of graft survival was observed by Sadan, *et al.*, with the majority of transplanted cells rejected within a week post-transplantation, despite cyclosporine immunosuppression and transplantation of higher numbers of cells than used in our study (Sadan *et al.*, 2009).

In vitro effects of trophic factor secretion by MSCs have also been described, with findings of increased expression of TH and DA by ventral mesencephalic cells cocultured with undifferentiated rat MSCs (Jin *et al.*, 2008). Also, treatment with MSCconditioned medium was found to increase the survival of embryonic mesencephalic dopaminergic neurons in *in vitro* neuronal injury models of serum deprivation and exposure to 6-OHDA (Shintani *et al.*, 2007). Furthermore, marked protection was demonstrated when conditioned medium from astrocyte-like, neurotrophic factorsecreting MSCs or undifferentiated MSCs were applied to neuroblastoma cell line, SH-SY5Y, one hour prior to exposure to neurotoxin 6-OHDA (Sadan *et al.*, 2009).

Together these studies provide support for another potential mechanism by which MSCs may effect functional improvement in Parkinsonian models. Further evaluation of both the neuroprotective and cell regenerative/replacement properties of MSCs will be required in order to harness these effectively in cellular therapies for PD and other neurological diseases.

5.4.7 Co-Transplantation of OECs Did Not Yield Improved Engraftment of Neuronal-Primed hMSCs

In this thesis, OEC co-transplantation with neuronal-primed hMSCs was, to our knowledge, investigated for the first time, to determine whether OECs could aid hMSC engraftment into the lesioned hemisphere of the 6-OHDA-induced hemiparkinsonian rat. Since the commencement of this work, several studies examining OEC transplantation for PD and nigrostriatal dopaminergic axon transection have been reported (Dewar *et al.*, 2007; Shukla *et al.*, 2009; Teng *et al.*, 2008). Overall, these recent studies indicate that co-grafting with OECs is beneficial in cellular therapies for these disorders.

OEC co-transplantation with olfactory nerve fibroblasts was shown to promote axonal regeneration, and prevent fibrotic scar formation and type IV collagen deposition at lesions caused by unilateral transection of nigrostriatal dopaminergic axons (Teng *et al.*, 2008). Furthermore, Shukla and colleagues found that co-grafting of OECs significantly enhanced the survival of NSC-derived dopaminergic neurons and striatal reinnervation in 6-OHDA lesioned hemiparkinsonian rats (Shukla *et al.*, 2009). Cotransplanted animals also displayed significant reduction in amphetamine-induced rotational behaviour, increased spontaneous locomotor activity, and neurochemical recovery of DA and its metabolite DOPAC (Shukla *et al.*, 2009). Whereas, Dewar, *et al.* demonstrated that transplantation of OECs alone into the striatum of hemiparkinsonian rats is insufficient for the promotion of tissue repair and functional recovery (Dewar *et* *al.*, 2007), suggesting that in cellular therapies for PD, OECs play a supportive role for neuronal progenitors or SCs that act to replace damaged or lost dopaminergic neurons.

In contrast to these reports, co-transplantation of OECs in the present study did not appear to provide advantages for the survival or differentiation of neuronal-primed hMSCs in the striatum or substantia nigra of 6-OHDA lesioned hemiparkinsonian rats. Furthermore, transplanted OECs also did not survive or engraft into the host brain. A major difference in our study is the use of xenogeneic hMSCs and OECs for transplantation into the rat CNS, and although immunosuppression with cyclosporine was implemented strong glial accumulation was observed at graft sites, which may account for the graft loss. Other potential factors contributing to the loss of hMSCs have been discussed previously, and these may also have influenced the survival of OECs. Therefore, further studies are required to determine the capabilities of hMSC and OEC transplantation into experimental models of PD.

5.4.8 Summary

In conclusion, this chapter has shown that transplantation of hMSCs into the lesioned hemisphere of hemiparkinsonian rats results in limited graft survival and absence of differentiation *in vivo*, regardless of the differentiation state of hMSCs (undifferentiated or neuronal-primed), the transplantation site (striatum or substantia nigra), the presence or absence of OECs, and the use of cyclosporine immunosuppression. In addition, hMSC transplantation was not found to result in neurotrophic or immunomodulatory effects on endogenous host cells. These findings are contradictory to the majority of initial reports describing functional improvement in neurological diseases resulting from cellular replacement by MSCs. The discrepancies may be related to differences in the differentiation procedures, transplantation techniques and cell labelling/identification methods implemented. On the other hand,

257

similar findings have been reported in a recent publication describing limited MSC engraftment and the presence of inflammatory responses and glial accumulation around graft sites post-transplantation (Coyne *et al.*, 2006). The mechanisms underlying the rapid loss of transplanted hMSCs and OECs remain to be determined, but may involve innate immune response targeting of xenogeneic cells, inadequate neuronal differentiation of hMSCs prior to transplantation, insufficient immunosuppression, and the possible need for improved transplantation strategies. While improvements to our current understanding of MSC-based cellular therapy for PD are still required, the progress in this field to date validates the significance of this work and warrants continued research efforts in this field.

CHAPTER 6

CONCLUSIONS AND FUTURE DIRECTIONS

6.1 Concluding Remarks

The results of this thesis illustrate the complexity of the cellular therapy discipline, and emphasise the need for a greater understanding of each facet of this field. Promising outcomes and continued research progress have fuelled enthusiasm and hope within the scientific community and the general public, however, much remains to be understood concerning the biology of the donor cell population, the optimal procedures for directed differentiation toward the desired cell phenotype, effective strategies for transplantation into sites of injury/degeneration, the relevancy of animal models to the disease in humans, and techniques for accurate analysis of transplantation outcomes. This thesis aimed to investigate the baseline characteristics of BM-derived hMSCs over long-term subcultivation, the potential of hMSCs to differentiate *in vitro* toward the neuronal lineage under the influence of cytokines, and the effects of intracerebral hMSC transplantation in the 6-OHDA lesioned hemiparkinsonian rat model.

Analysis of undifferentiated hMSCs confirmed that the cell populations exhibited the expected characteristics of MSCs, in terms of proliferative ability, morphological appearance, expression of surface antigens, genes and proteins, and multi-lineage mesodermal differentiation capacity. Furthermore, through examination of these characteristics over the course of long-term serial passaging, it was found that hMSCs do not spontaneously differentiate toward the neural lineage, and optimal periods for the commencement of neuronal differentiation were determined. Interestingly, hMSCs cultured beyond the optimal time frame (first 6-8 passages of expansion culture) were still able to acquire an immature neuronal-like phenotype. Nevertheless, obvious differences were seen between neuronal-differentiated early passage and late passage hMSCs, with late passage cultures containing larger cells,

260

greater amounts of cellular debris, and a higher proportion of cells retaining a flat, fibroblast-like appearance that were unresponsive to cytokine treatment.

Through this work evidence was provided supporting the *in vitro* generation of an immature neuronal-like phenotype in hMSCs through the application of cytokines and culture conditions that are known to have neural-inducing capabilities. It was demonstrated for the first time that cytokine-based neuronal differentiation of hMSCs is the result of active and dynamic cellular processes, involving migration, as well as outgrowth, motility and retraction of cell extensions. These observations showed clear differences to chemical-based neuronal differentiation systems, which have been found to induce toxicity-related morphological changes causing a false pseudo-neuronal appearance (Bertani *et al.*, 2005; Choi *et al.*, 2006; Deng *et al.*, 2006; Lu *et al.*, 2004; Neuhuber *et al.*, 2004; Suon *et al.*, 2004). It was also established that prior to differentiation hMSCs expressed neural and pluripotency markers together with mesodermal markers, providing support for MSC neural differentiation capacity and also confirming the need to ascertain baseline expression levels in order to accurately assess the outcomes of differentiation procedures.

Since cytokine-based neuronal differentiation was able to generate immature neuronal properties in hMSCs, this study also endeavoured to induce a dopaminergic neuronal phenotype through the use of cytokines known to be important in midbrain dopaminergic neuronal development *in vivo*. In addition, a strategy of sequential cytokine application was also investigated, as this would more closely mimic the *in vivo* situation. However, none of the differentiation strategies investigated was capable of eliciting enhanced dopaminergic neuronal differentiation from hMSCs. Although, of the 3 methods examined, the first stage of the MultiDA method yielded the greatest increase in expression of neuronal progenitor marker *NES* and dopaminergic neuronal
transcription factor *NR4A2*. Furthermore, through evaluating the effects of the different cytokine combinations it was determined that FGF-2 and EGF exerted the greatest benefits for the *in vitro* neuronal differentiation of hMSCs.

In view of the fact that immature neuronal cells survive intracerebral transplantation procedures better than mature neurons, the present thesis investigated the effects of transplanting hMSCs that had undergone neuronal-priming (first stage of the MultiDA method), as well as undifferentiated hMSCs, on the lesioned hemisphere of hemiparkinsonian rats. Grafted hMSCs were found to survive transplantation and could be detected initially at days 1 and 7 post-transplantation through immunohistological staining for HNA expression. However, only limited graft survival could be detected, with loss of HNA signal apparent at day 7 post-transplantation and complete absence of grafted cells by day 21. Moreover, strategies that were implemented to improve engraftment and differentiation, such as administration into the substantia nigra as well as the striatum, neuronal-priming in vitro prior to transplantation, and co-transplantation with OECs, were unable to provide any significant advantages. The observed absence of engraftment may be explained by: the presence of an innate inflammatory response at graft and sham sites, which may have been initiated by surgical trauma during transplantation; the xenogeneic nature of the graft, since both MSCs and OECs were lost from graft sites; the 'foreign' nature of the hMSCs, which had not gained a completely mature neuronal phenotype as evidenced by the continued production of fibronectin extracellular matrix; or the lack of appropriate factors within the lesioned adult rat striatum and substantia nigra for supporting hMSC survival, engraftment and differentiation. Additionally, transplantation of hMSCs and OECs did not appear to elicit neurogenic effects on endogenous host NSCs or neural progenitors. These findings contradict those predominantly reported in the literature,

however, shows similarity with recent reports describing absence of MSC engraftment and presence of glial accumulation at graft sites (Coyne *et al.*, 2006; Suon *et al.*, 2006). The results of this thesis also show for the first time the surprising presence of microglial/macrophage accumulation, suggesting activation of innate inflammatory responses after hMSC transplantation into the lesioned hemiparkinsonian rat brain.

Therefore, taken together, the study presented in this thesis demonstrates that cytokine-based neuronal differentiation of hMSCs elicits authentic and dynamic cellular responses, and induces the acquisition of an immature neuronal phenotype. However, difficulties were encountered in achieving in vivo engraftment and differentiation after intracerebral transplantation into the lesioned brain. The results obtained in the *in vivo* studies of this thesis differ from some published reports; nevertheless, the results of those reported studies also display conflicting data and remain a topic of ongoing debate. Central to this controversy is the mechanism by which MSCs exert functional improvements in vivo in experimental animal models of disease. Pioneering studies suggested that tissue repair occurred via the broad developmental plasticity and transdifferentiation capacity of MSCs; however, an emerging line of evidence has arisen suggesting that MSCs exhibit low or transient levels of in vivo engraftment and transdifferentiation, and instead promote tissue repair through enhancing the regeneration of injured cells, preserving surviving cells, or stimulating endogenous host SC activity (Chen et al., 2006; Phinney et al., 2007; Prockop, 2007). In the context of this thesis, transient survival post-transplantation may therefore be expected. It is also possible that the absence of an endogenous host response to hMSC and OEC transplantation may be related to the severe, complete lesion hemiparkinsonian disease model employed in this study, in which there is no chance for rescuing damaged dopaminergic neurons since the process of neuronal death has been completed.

However, a MSC-induced protection-restoration effect on endogenous host neural cells may be possible in a less severe PD model, such as the partial lesion model used by Bahat-Stroomza and colleagues (Bahat-Stroomza *et al.*, 2009).

It is unclear whether the discrepancies observed between studies result from actual differences in outcome, or are representative of variations in experimental techniques. Interpretation of data across the different studies is also complicated by the vast array of procedures that have been utilised for the *in vitro* neuronal differentiation and *in vivo* transplantation and identification of MSCs. Furthermore, these diverse methodologies often result in supposedly similar outcomes, which is even more surprising given that different measures of that outcome have been used for assessment purposes. Taken together, this has led to little consistency across the field and difficulties in comparing data obtained by different studies. A meticulous and systematic approach is therefore required for the proper development of reproducible and reliable methods for MSC neuronal differentiation and application in cellular therapies for neurological diseases. Inherent to this situation is the reliance of the majority of studies on the use of heterogeneous populations of MSCs, raising the question of whether different studies are possibly using 'different' MSC populations, which could potentially account for the diverse findings. Nevertheless, current data still suggests that MSC-based cellular therapies for neurological diseases and disorders are possible, although very much more complicated than initially believed with potential involvement of a number of distinct underlying mechanisms.

The issues discussed above emphasise the present need for a more comprehensive understanding of the biology and identity of the bona fide MSC. Considering the potential significance of this research, further studies are necessary to achieve a better understanding of this field and to identify the optimal *in vitro* and *in* *vivo* procedures for directed neuronal differentiation and effective transplantation of hMSCs.

6.2 Future Directions

In order to advance the experimental observations related to the cytokine-based neuronal differentiation of hMSCs and the absence of *in vivo* engraftment found in this thesis, further studies are required. Efforts directed towards advancing our understanding of the isolation and expansion of undifferentiated hMSCs *in vitro* would allow the establishment of consistent methods across different studies and may lead to less variation between results from different laboratories. CD146 has been proposed as a potential positive selection marker for MSCs (Sacchetti *et al.*, 2007), therefore, CD146⁺ MSCs could be characterised according to the parameters examined in the present study to determine any similarities or differences with the MSC population utilised in our study.

To confirm the immature neuronal-like phenotype obtained in hMSCs through cytokine induction, *in vitro* functional analysis could be performed to detect the presence of electrophysiological activity. In addition, this analysis would reveal whether the hMSC-derived neuronal-like cells possess functional properties that resemble those of immature neurons.

Since hMSCs appeared to respond poorly to the second (SHH/FGF-8) and third (GDNF) stages of the MultiDA method, it would be interesting to examine the growth factor receptors present on hMSCs during the different stages of neuronal differentiation. Furthermore, the functionality of these receptors could be confirmed through performing DNA synthesis studies by incubating hMSCs with [³H] thymidine following cytokine exposure to examine stimulation of cells into the S-phase of the mitotic cycle (as performed by Kramer and colleagues (Kramer *et al.*, 2006)).

The *in vitro* neuronal differentiation procedure for hMSCs could be improved through the addition of neural support cells, such as astrocytes. In particular, it has been shown that the regional origin of astrocytes was the determining factor for the neuronal fate of co-cultured ESCs (Roy *et al.*, 2006), therefore potential lies in the co-culturing of hMSCs with midbrain astrocytes.

Further characterisation of the immune response to hMSCs after intracerebral transplantation should be undertaken to assess the involvement of other inflammatory cell types, as well as T and B lymphocytes. This investigation of innate and adaptive immune responses should also be conducted for the period between 1 and 7 days post-transplantation, to further elucidate the mechanisms involved in hMSC and OEC graft loss.

Since microglia/macrophages appear to be activated in the regions surrounding the graft site and needle tract, additional immunosuppression could be implemented as reported by Singhal and colleagues, who utilised a combination of oral cyclosporine A, azathioprine, prednisolone and indomethacin for enhanced immunosuppression (Singhal *et al.*, 2008).

Immunofluorescence staining could also be performed to examine the deposition of inhibitory proteoglycans, such as CSPGs, at the graft site and needle tract to reveal whether proteoglycan deposition encases the grafted cells from host brain tissue and prevents hMSC migration and integration. If this is indeed the case, then transplantation can be performed in the presence of chondroitinase ABC, or heparin and DNase to facilitate graft survival, engraftment and integration.

Additional improvements could be attempted for the *in vivo* neural transplantation of MSCs. Transplantation of a greater number of hMSCs may provide improved donor cell survival, since recent studies have used higher MSC dosages and

obtained promising results (Bouchez *et al.*, 2008; Levy *et al.*, 2008). Difficulties relating to the xenogeneic nature of the grafts employed in this study can be confirmed and avoided through the use of rat MSCs and OECs when transplanting into the hemiparkinsonian rat model. Furthermore, transplantation into a partial lesion hemiparkinsonian model could be performed to allow examination of endogenous host NSC responses and protection-restoration effects of MSC transplantation. Alternate routes of MSC delivery to the CNS could also be investigated, including intravenous or intrathecal administration of MSCs, which would avoid most of the trauma associated with direct intracerebral transplantation.

Finally, little is known of the effects of cytokine-based neuronal differentiation procedures on the immunomodulatory properties of hMSCs. To address these concerns, it will be of benefit to examine the hMSC surface antigen expression profile, capability for stimulating peripheral blood lymphocyte proliferation and cytotoxicity, and potential for interacting with DCs (rosette binding assay and chemotaxis assay) following the *in vitro* neuronal and dopaminergic neuronal differentiation procedures investigated in this thesis.

CHAPTER 7

REFERENCES

- Abdallah, B. M., Haack-Sorensen, M., Burns, J. S., Elsnab, B., Jakob, F., Hokland, P., and Kassem, M. (2005) Maintenance of differentiation potential of human bone marrow mesenchymal stem cells immortalized by human telomerase reverse transcriptase gene despite [corrected] extensive proliferation. *Biochem Biophys Res Commun* 326; 527-38.
- Abeliovich, A., and Hammond, R. (2007) Midbrain dopamine neuron differentiation: factors and fates. *Dev Biol* **304**; 447-54.
- Abouelfetouh, A., Kondoh, T., Ehara, K., and Kohmura, E. (2004) Morphological differentiation of bone marrow stromal cells into neuron-like cells after co-culture with hippocampal slice. *Brain Res* **1029**; 114-9.
- Aggarwal, S., and Pittenger, M. F. (2005) Human mesenchymal stem cells modulate allogeneic immune cell responses. *Blood* **105**; 1815-22.
- Agrawal, A. K., Shukla, S., Chaturvedi, R. K., Seth, K., Srivastava, N., Ahmad, A., and Seth, P. K. (2004) Olfactory ensheathing cell transplantation restores functional deficits in rat model of Parkinson's disease: a cotransplantation approach with fetal ventral mesencephalic cells. *Neurobiol Dis* 16; 516-26.
- Airey, J. A., Almeida-Porada, G., Colletti, E. J., Porada, C. D., Chamberlain, J., Movsesian, M., Sutko, J. L., and Zanjani, E. D. (2004) Human mesenchymal stem cells form Purkinje fibers in fetal sheep heart. *Circulation* 109; 1401-7.
- Alemdar, A. Y., Sadi, D., McAlister, V., and Mendez, I. (2007) Intracerebral cotransplantation of liposomal tacrolimus improves xenograft survival and reduces graft rejection in the hemiparkinsonian rat. *Neuroscience* 146; 213-24.
- Alexanian, A. R. (2005) Neural stem cells induce bone-marrow-derived mesenchymal stem cells to generate neural stem-like cells via juxtacrine and paracrine interactions. *Exp Cell Res* 310; 383-91.
- Alvarez-Dolado, M., Pardal, R., Garcia-Verdugo, J. M., Fike, J. R., Lee, H. O., Pfeffer, K., Lois, C., Morrison, S. J., and Alvarez-Buylla, A. (2003) Fusion of bonemarrow-derived cells with Purkinje neurons, cardiomyocytes and hepatocytes. *Nature* 425; 968-73.
- Ambrogini, P., Lattanzi, D., Ciuffoli, S., Agostini, D., Bertini, L., Stocchi, V., Santi, S., and Cuppini, R. (2004) Morpho-functional characterization of neuronal cells at different stages of maturation in granule cell layer of adult rat dentate gyrus. *Brain Res* 1017; 21-31.
- Anderson, D. J., Gage, F. H., and Weissman, I. L. (2001) Can stem cells cross lineage boundaries? *Nat Med* 7; 393-5.

Andersson, E., Tryggvason, U., Deng, Q., Friling, S., Alekseenko, Z., Robert, B., Perlmann, T., and Ericson, J. (2006) Identification of intrinsic determinants of midbrain dopamine neurons. *Cell* **124**; 393-405.

Arenas, E. (2008) Foxa2: the rise and fall of dopamine neurons. Cell Stem Cell 2; 110-2.

- Arnhold, S., Absenger, Y., Klein, H., Addicks, K., and Schraermeyer, U. (2007) Transplantation of bone marrow-derived mesenchymal stem cells rescue photoreceptor cells in the dystrophic retina of the rhodopsin knockout mouse. *Graefes Arch Clin Exp Ophthalmol* 245; 414-22.
- Arnhold, S., Heiduschka, P., Klein, H., Absenger, Y., Basnaoglu, S., Kreppel, F., Henke-Fahle, S., Kochanek, S., Bartz-Schmidt, K. U., Addicks, K., and Schraermeyer, U. (2006) Adenovirally transduced bone marrow stromal cells differentiate into pigment epithelial cells and induce rescue effects in RCS rats. *Invest Ophthalmol Vis Sci* 47; 4121-9.
- Augello, A., Tasso, R., Negrini, S. M., Amateis, A., Indiveri, F., Cancedda, R., and Pennesi, G. (2005) Bone marrow mesenchymal progenitor cells inhibit lymphocyte proliferation by activation of the programmed death 1 pathway. *Eur J Immunol* 35; 1482-90.
- Aurich, I., Mueller, L. P., Aurich, H., Luetzkendorf, J., Tisljar, K., Dollinger, M. M., Schormann, W., Walldorf, J., Hengstler, J. G., Fleig, W. E., and Christ, B. (2007) Functional integration of hepatocytes derived from human mesenchymal stem cells into mouse livers. *Gut* 56; 405-15.
- Azizi, S. A., Stokes, D., Augelli, B. J., DiGirolamo, C., and Prockop, D. J. (1998) Engraftment and migration of human bone marrow stromal cells implanted in the brains of albino rats--similarities to astrocyte grafts. *Proc Natl Acad Sci U S* A 95; 3908-13.
- Bahat-Stroomza, M., Barhum, Y., Levy, Y. S., Karpov, O., Bulvik, S., Melamed, E., and Offen, D. (2009) Induction of Adult Human Bone Marrow Mesenchymal Stromal Cells into Functional Astrocyte-Like Cells: Potential for Restorative Treatment in Parkinson's Disease. *J Mol Neurosci* **39**; 199-210.
- Barker, R. A., Ratcliffe, E., McLaughlin, M., Richards, A., and Dunnett, S. B. (2000) A role for complement in the rejection of porcine ventral mesencephalic xenografts in a rat model of Parkinson's disease. *J Neurosci* **20**; 3415-24.
- Barker, R. A., and Widner, H. (2004) Immune problems in central nervous system cell therapy. *NeuroRx* 1; 472-81.
- Bartholomew, A., Sturgeon, C., Siatskas, M., Ferrer, K., McIntosh, K., Patil, S., Hardy, W., Devine, S., Ucker, D., Deans, R., Moseley, A., and Hoffman, R. (2002)

Mesenchymal stem cells suppress lymphocyte proliferation in vitro and prolong skin graft survival in vivo. *Exp Hematol* **30**; 42-8.

- Barzilay, R., Kan, I., Ben-Zur, T., Bulvik, S., Melamed, E., and Offen, D. (2008) Induction of human mesenchymal stem cells into dopamine-producing cells with different differentiation protocols. *Stem Cells Dev* 17; 547-54.
- Baxter, M. A., Wynn, R. F., Jowitt, S. N., Wraith, J. E., Fairbairn, L. J., and Bellantuono, I. (2004) Study of telomere length reveals rapid aging of human marrow stromal cells following in vitro expansion. *Stem Cells* 22; 675-82.
- Beresford, J. N., Bennett, J. H., Devlin, C., Leboy, P. S., and Owen, M. E. (1992) Evidence for an inverse relationship between the differentiation of adipocytic and osteogenic cells in rat marrow stromal cell cultures. *J Cell Sci* 102; 341-51.
- Bernheimer, H., Birkmayer, W., Hornykiewicz, O., Jellinger, K., and Seitelberger, F. (1973) Brain dopamine and the syndromes of Parkinson and Huntington. Clinical, morphological and neurochemical correlations. *J Neurol Sci* 20; 415-55.
- Bertani, N., Malatesta, P., Volpi, G., Sonego, P., and Perris, R. (2005) Neurogenic potential of human mesenchymal stem cells revisited: analysis by immunostaining, time-lapse video and microarray. *J Cell Sci* **118**; 3925-36.
- Beyth, S., Borovsky, Z., Mevorach, D., Liebergall, M., Gazit, Z., Aslan, H., Galun, E., and Rachmilewitz, J. (2005) Human mesenchymal stem cells alter antigenpresenting cell maturation and induce T-cell unresponsiveness. *Blood* 105; 2214-9.
- Bhang, S. H., Lee, Y. E., Cho, S. W., Shim, J. W., Lee, S. H., Choi, C. Y., Chang, J. W., and Kim, B. S. (2007) Basic fibroblast growth factor promotes bone marrow stromal cell transplantation-mediated neural regeneration in traumatic brain injury. *Biochem Biophys Res Commun* 359; 40-5.
- Bianco, J. I., Perry, C., Harkin, D. G., Mackay-Sim, A., and Feron, F. (2004) Neurotrophin 3 promotes purification and proliferation of olfactory ensheathing cells from human nose. *Glia* 45, 111-23.
- Bianco, P., Robey, P. G., and Simmons, P. J. (2008) Mesenchymal stem cells: revisiting history, concepts, and assays. *Cell Stem Cell* **2**; 313-9.
- Bjorklund, A., and Lindvall, O. (2000) Cell replacement therapies for central nervous system disorders. *Nat Neurosci* **3**; 537-44.
- Bjorklund, A., and Stenevi, U. (1979) Reconstruction of the nigrostriatal dopamine pathway by intracerebral nigral transplants. *Brain Res* **177**; 555-60.

- Bjorklund, A., Stenevi, U., Dunnett, S. B., and Iversen, S. D. (1981) Functional reactivation of the deafferented neostriatum by nigral transplants. *Nature* 289; 497-9.
- Bjorklund, L. M., Sanchez-Pernaute, R., Chung, S., Andersson, T., Chen, I. Y., McNaught, K. S., Brownell, A. L., Jenkins, B. G., Wahlestedt, C., Kim, K. S., and Isacson, O. (2002) Embryonic stem cells develop into functional dopaminergic neurons after transplantation in a Parkinson rat model. *Proc Natl Acad Sci U S A* **99**; 2344-9.
- Blau, H. M., Brazelton, T. R., and Weimann, J. M. (2001) The evolving concept of a stem cell: entity or function? *Cell* **105**; 829-41.
- Blondheim, N. R., Levy, Y. S., Ben-Zur, T., Burshtein, A., Cherlow, T., Kan, I., Barzilai, R., Bahat-Stromza, M., Barhum, Y., Bulvik, S., Melamed, E., and Offen, D. (2006) Human mesenchymal stem cells express neural genes, suggesting a neural predisposition. *Stem Cells Dev* 15; 141-64.
- Borel, J. F., Feurer, C., Gubler, H. U., and Stahelin, H. (1976) Biological effects of cyclosporin A: a new antilymphocytic agent. *Agents Actions* **6**; 468-75.
- Borel, J. F., Feurer, C., Magnee, C., and Stahelin, H. (1977) Effects of the new antilymphocytic peptide cyclosporin A in animals. *Immunology* **32**; 1017-25.
- Boruch, A. V., Conners, J. J., Pipitone, M., Deadwyler, G., Storer, P. D., Devries, G. H., and Jones, K. J. (2001) Neurotrophic and migratory properties of an olfactory ensheathing cell line. *Glia* 33; 225-9.
- Bouchez, G., Sensebe, L., Vourc'h, P., Garreau, L., Bodard, S., Rico, A., Guilloteau, D., Charbord, P., Besnard, J. C., and Chalon, S. (2008) Partial recovery of dopaminergic pathway after graft of adult mesenchymal stem cells in a rat model of Parkinson's disease. *Neurochem Int* 52; 1332-42.
- Brooks, D. J. (1998) The early diagnosis of Parkinson's disease. Ann Neurol 44; S10-8.
- Bruder, S. P., Jaiswal, N., and Haynesworth, S. E. (1997) Growth kinetics, self-renewal, and the osteogenic potential of purified human mesenchymal stem cells during extensive subcultivation and following cryopreservation. *J Cell Biochem* **64**; 278-94.
- Brundin, P., Nilsson, O. G., Strecker, R. E., Lindvall, O., Astedt, B., and Bjorklund, A. (1986) Behavioural effects of human fetal dopamine neurons grafted in a rat model of Parkinson's disease. *Exp Brain Res* 65; 235-40.
- Brundin, P., Pogarell, O., Hagell, P., Piccini, P., Widner, H., Schrag, A., Kupsch, A., Crabb, L., Odin, P., Gustavii, B., Bjorklund, A., Brooks, D. J., Marsden, C. D., Oertel, W. H., Quinn, N. P., Rehncrona, S., and Lindvall, O. (2000) Bilateral

caudate and putamen grafts of embryonic mesencephalic tissue treated with lazaroids in Parkinson's disease. *Brain* **123** (**Pt 7**); 1380-90.

- Brundin, P., Strecker, R. E., Widner, H., Clarke, D. J., Nilsson, O. G., Astedt, B., Lindvall, O., and Bjorklund, A. (1988) Human fetal dopamine neurons grafted in a rat model of Parkinson's disease: immunological aspects, spontaneous and drug-induced behaviour, and dopamine release. *Exp Brain Res* 70; 192-208.
- Burns, T. C., Ortiz-Gonzalez, X. R., Gutierrez-Perez, M., Keene, C. D., Sharda, R., Demorest, Z. L., Jiang, Y., Nelson-Holte, M., Soriano, M., Nakagawa, Y., Luquin, M. R., Garcia-Verdugo, J. M., Prosper, F., Low, W. C., and Verfaillie, C. M. (2006) Thymidine analogs are transferred from prelabeled donor to host cells in the central nervous system after transplantation: a word of caution. *Stem Cells* 24; 1121-7.
- Calvo, W. (1968) The innervation of the bone marrow in laboratory animals. *Am J Anat* **123**; 315-28.
- Camp, D. M., Loeffler, D. A., Farrah, D. M., Borneman, J. N., and LeWitt, P. A. (2009) Cellular immune response to intrastriatally implanted allogeneic bone marrow stromal cells in a rat model of Parkinson's disease. *J Neuroinflammation* 6; 17.

Caplan, A. I. (1991) Mesenchymal stem cells. J Orthop Res 9; 641-50.

- Carleton, A., Petreanu, L. T., Lansford, R., Alvarez-Buylla, A., and Lledo, P. M. (2003) Becoming a new neuron in the adult olfactory bulb. *Nat Neurosci* **6**; 507-18.
- Cauffman, G., Liebaers, I., Van Steirteghem, A., and Van de Velde, H. (2006) POU5F1 isoforms show different expression patterns in human embryonic stem cells and preimplantation embryos. *Stem Cells* **24**; 2685-91.
- Chamberlain, G., Fox, J., Ashton, B., and Middleton, J. (2007) Concise review: mesenchymal stem cells: their phenotype, differentiation capacity, immunological features, and potential for homing. *Stem Cells* **25**; 2739-49.
- Chen, J., Li, Y., Wang, L., Lu, M., Zhang, X., and Chopp, M. (2001) Therapeutic benefit of intracerebral transplantation of bone marrow stromal cells after cerebral ischemia in rats. *J Neurol Sci* 189; 49-57.
- Chen, S., Choo, A. B., Nai-Dy, W., Heng-Phon, T., and Oh, S. K. (2007a) Knockdown of Oct-4 or Sox-2 attenuates neurogenesis of mouse embryonic stem cells. *Stem Cells Dev* **16**; 413-20.
- Chen, X., McClurg, A., Zhou, G. Q., McCaigue, M., Armstrong, M. A., and Li, G. (2007b) Chondrogenic differentiation alters the immunosuppressive property of bone marrow-derived mesenchymal stem cells, and the effect is partially due to the upregulated expression of B7 molecules. *Stem Cells* 25; 364-70.

- Chen, Y., Dong, X. J., Zhang, G. R., Shao, J. Z., and Xiang, L. X. (2007c) In vitro differentiation of mouse bone marrow stromal stem cells into hepatocytes induced by conditioned culture medium of hepatocytes. *J Cell Biochem* 102; 52-63.
- Chen, Y., Pan, R. L., Zhang, X. L., Shao, J. Z., Xiang, L. X., Dong, X. J., and Zhang, G. R. (2008) Induction of hepatic differentiation of mouse bone marrow stromal stem cells by the histone deacetylase inhibitor VPA. *J Cell Mol Med*.
- Chen, Y., Teng, F. Y., and Tang, B. L. (2006) Coaxing bone marrow stromal mesenchymal stem cells towards neuronal differentiation: progress and uncertainties. *Cell Mol Life Sci* 63; 1649-57.
- Cheng, S. L., Yang, J. W., Rifas, L., Zhang, S. F., and Avioli, L. V. (1994)
 Differentiation of human bone marrow osteogenic stromal cells in vitro: induction of the osteoblast phenotype by dexamethasone. *Endocrinology* 134; 277-86.
- Cho, K. J., Trzaska, K. A., Greco, S. J., McArdle, J., Wang, F. S., Ye, J. H., and Rameshwar, P. (2005) Neurons derived from human mesenchymal stem cells show synaptic transmission and can be induced to produce the neurotransmitter substance P by interleukin-1 alpha. *Stem Cells* 23; 383-91.
- Choi, C. B., Cho, Y. K., Prakash, K. V., Jee, B. K., Han, C. W., Paik, Y. K., Kim, H. Y., Lee, K. H., Chung, N., and Rha, H. K. (2006) Analysis of neuron-like differentiation of human bone marrow mesenchymal stem cells. *Biochem Biophys Res Commun* 350; 138-46.
- Chu, M. S., Chang, C. F., Yang, C. C., Bau, Y. C., Ho, L. L., and Hung, S. C. (2006) Signalling pathway in the induction of neurite outgrowth in human mesenchymal stem cells. *Cell Signal* 18; 519-30.
- Cicchetti, F., Fodor, W., Deacon, T. W., van Horne, C., Rollins, S., Burton, W., Costantini, L. C., and Isacson, O. (2003) Immune parameters relevant to neural xenograft survival in the primate brain. *Xenotransplantation* **10**; 41-9.
- Clark, B. R., and Keating, A. (1995) Biology of bone marrow stroma. *Ann N Y Acad Sci* **770**; 70-8.
- Clarke, S. R., Shetty, A. K., Bradley, J. L., and Turner, D. A. (1994) Reactive astrocytes express the embryonic intermediate neurofilament nestin. *Neuroreport* **5**; 1885-8.
- Cole, L. J., Habermeyer, J. G., and Bond, V. P. (1955) Recovery from acute radiation injury in mice following administration of rat bone marrow. *J Natl Cancer Inst* 16; 1-9.

- Corcione, A., Benvenuto, F., Ferretti, E., Giunti, D., Cappiello, V., Cazzanti, F., Risso, M., Gualandi, F., Mancardi, G. L., Pistoia, V., and Uccelli, A. (2006) Human mesenchymal stem cells modulate B-cell functions. *Blood* 107; 367-72.
- Cotran, R. S., Kumar, V., and Collins, T. (1999) *Robbins Pathologic Basis of Disease*. W.B. Saunders Company. Philadelphia.
- Coyne, T. M., Marcus, A. J., Woodbury, D., and Black, I. B. (2006) Marrow stromal cells transplanted to the adult brain are rejected by an inflammatory response and transfer donor labels to host neurons and glia. *Stem Cells* **24**; 2483-92.
- Croft, A. P., and Przyborski, S. A. (2006) Formation of neurons by non-neural adult stem cells: potential mechanism implicates an artifact of growth in culture. *Stem Cells* **24**; 1841-51.
- Czech, K. A., Ryan, J. W., Sagen, J., and Pappas, G. D. (1997) The influence of xenotransplant immunogenicity and immunosuppression on host MHC expression in the rat CNS. *Exp Neurol* 147; 66-83.
- Davalos, D., Grutzendler, J., Yang, G., Kim, J. V., Zuo, Y., Jung, S., Littman, D. R., Dustin, M. L., and Gan, W. B. (2005) ATP mediates rapid microglial response to local brain injury in vivo. *Nat Neurosci* 8; 752-8.
- Davie, C. A. (2008) A review of Parkinson's disease. Br Med Bull 86; 109-27.
- Deng, J., Petersen, B. E., Steindler, D. A., Jorgensen, M. L., and Laywell, E. D. (2006) Mesenchymal stem cells spontaneously express neural proteins in culture and are neurogenic after transplantation. *Stem Cells* 24; 1054-64.
- Deng, W., Han, Q., Liao, L., You, S., Deng, H., and Zhao, R. C. (2005) Effects of allogeneic bone marrow-derived mesenchymal stem cells on T and B lymphocytes from BXSB mice. DNA Cell Biol 24; 458-63.
- Deng, W., Obrocka, M., Fischer, I., and Prockop, D. J. (2001) In vitro differentiation of human marrow stromal cells into early progenitors of neural cells by conditions that increase intracellular cyclic AMP. *Biochemical & Biophysical Research Communications*. 282; 148-52.
- Deschaseaux, F., and Charbord, P. (2000) Human marrow stromal precursors are alpha 1 integrin subunit-positive. *J Cell Physiol* **184**; 319-25.
- Deschaseaux, F., Gindraux, F., Saadi, R., Obert, L., Chalmers, D., and Herve, P. (2003) Direct selection of human bone marrow mesenchymal stem cells using an anti-CD49a antibody reveals their CD45med,low phenotype. *Br J Haematol* 122; 506-17.

- Dewar, D., Bentley, D., and Barnett, S. C. (2007) Implantation of pure cultured olfactory ensheathing cells in an animal model of parkinsonism. *Acta Neurochir* (*Wien*) 149; 407-14.
- Dexter, T. M., Allen, T. D., and Lajtha, L. G. (1977) Conditions controlling the proliferation of haemopoietic stem cells in vitro. *J Cell Physiol* **91**; 335-44.
- Dezawa, M., Kanno, H., Hoshino, M., Cho, H., Matsumoto, N., Itokazu, Y., Tajima, N., Yamada, H., Sawada, H., Ishikawa, H., Mimura, T., Kitada, M., Suzuki, Y., and Ide, C. (2004) Specific induction of neuronal cells from bone marrow stromal cells and application for autologous transplantation. *J Clin Invest* 113; 1701-10.
- di Bonzo, L. V., Ferrero, I., Cravanzola, C., Mareschi, K., Rustichell, D., Novo, E., Sanavio, F., Cannito, S., Zamara, E., Bertero, M., Davit, A., Francica, S., Novelli, F., Colombatto, S., Fagioli, F., and Parola, M. (2008) Human mesenchymal stem cells as a two-edged sword in hepatic regenerative medicine: engraftment and hepatocyte differentiation versus profibrogenic potential. *Gut* 57; 223-31.
- Di Nicola, M., Carlo-Stella, C., Magni, M., Milanesi, M., Longoni, P. D., Matteucci, P., Grisanti, S., and Gianni, A. M. (2002) Human bone marrow stromal cells suppress T-lymphocyte proliferation induced by cellular or nonspecific mitogenic stimuli. *Blood* **99**; 3838-43.
- Dick, F. D. (2006) Parkinson's disease and pesticide exposures. *Br Med Bull* **79-80**; 219-31.
- Digirolamo, C. M., Stokes, D., Colter, D., Phinney, D. G., Class, R., and Prockop, D. J. (1999) Propagation and senescence of human marrow stromal cells in culture: a simple colony-forming assay identifies samples with the greatest potential to propagate and differentiate. *British Journal of Haematology*. **107**; 275-81.
- Djouad, F., Plence, P., Bony, C., Tropel, P., Apparailly, F., Sany, J., Noel, D., and Jorgensen, C. (2003) Immunosuppressive effect of mesenchymal stem cells favors tumor growth in allogeneic animals. *Blood* **102**; 3837-44.
- Dominici, M., Le Blanc, K., Mueller, I., Slaper-Cortenbach, I., Marini, F., Krause, D., Deans, R., Keating, A., Prockop, D., and Horwitz, E. (2006) Minimal criteria for defining multipotent mesenchymal stromal cells. The International Society for Cellular Therapy position statement. *Cytotherapy* 8, 315-7.
- Doucette, R. (1990) Glial influences on axonal growth in the primary olfactory system. *Glia* **3**; 433-49.
- Du, Y., Ma, Z., Lin, S., Dodel, R. C., Gao, F., Bales, K. R., Triarhou, L. C., Chernet, E., Perry, K. W., Nelson, D. L., Luecke, S., Phebus, L. A., Bymaster, F. P., and Paul, S. M. (2001) Minocycline prevents nigrostriatal dopaminergic

neurodegeneration in the MPTP model of Parkinson's disease. *Proc Natl Acad Sci U S A* **98**; 14669-74.

- Duffield, J. S., Park, K. M., Hsiao, L. L., Kelley, V. R., Scadden, D. T., Ichimura, T., and Bonventre, J. V. (2005) Restoration of tubular epithelial cells during repair of the postischemic kidney occurs independently of bone marrow-derived stem cells. J Clin Invest 115; 1743-55.
- Dunnett, S. B., and Bjorklund, A. (1999) Prospects for new restorative and neuroprotective treatments in Parkinson's disease. *Nature* **399**; A32-9.
- Dunnett, S. B., Bjorklund, A., Stenevi, U., and Iversen, S. D. (1981) Behavioural recovery following transplantation of substantia nigra in rats subjected to 6-OHDA lesions of the nigrostriatal pathway. I. Unilateral lesions. *Brain Res* 215; 147-61.
- Eaves, C. J., Cashman, J. D., Kay, R. J., Dougherty, G. J., Otsuka, T., Gaboury, L. A., Hogge, D. E., Lansdorp, P. M., Eaves, A. C., and Humphries, R. K. (1991) Mechanisms that regulate the cell cycle status of very primitive hematopoietic cells in long-term human marrow cultures. II. Analysis of positive and negative regulators produced by stromal cells within the adherent layer. *Blood* 78; 110-7.
- Egerbacher, M., Krestan, R., and Bock, P. (1995) Morphology, histochemistry, and differentiation of the cat's epiglottic cartilage: a supporting organ composed of elastic cartilage, fibrous cartilage, myxoid tissue, and fat tissue. *Anat Rec* **242**; 471-82.
- Eisenberg, L. M., and Eisenberg, C. A. (2003) Stem cell plasticity, cell fusion, and transdifferentiation. *Birth Defects Res C Embryo Today* **69**; 209-18.
- Eliopoulos, N., Stagg, J., Lejeune, L., Pommey, S., and Galipeau, J. (2005) Allogeneic marrow stromal cells are immune rejected by MHC class I- and class IImismatched recipient mice. *Blood* 106; 4057-65.
- Fairless, R., and Barnett, S. C. (2005) Olfactory ensheathing cells: their role in central nervous system repair. *Int J Biochem Cell Biol* **37**; 693-9.
- Fang, B., Shi, M., Liao, L., Yang, S., Liu, Y., and Zhao, R. C. (2004) Systemic infusion of FLK1(+) mesenchymal stem cells ameliorate carbon tetrachloride-induced liver fibrosis in mice. *Transplantation* 78; 83-8.
- Fearnley, J. M., and Lees, A. J. (1991) Ageing and Parkinson's disease: substantia nigra regional selectivity. *Brain* **114** (**Pt 5**); 2283-301.
- Fibbe, W. E., Nauta, A. J., and Roelofs, H. (2007) Modulation of immune responses by mesenchymal stem cells. *Ann N Y Acad Sci* **1106**; 272-8.

- Fitch, M. T., and Silver, J. (1997) Activated macrophages and the blood-brain barrier: inflammation after CNS injury leads to increases in putative inhibitory molecules. *Exp Neurol* **148**; 587-603.
- Fouad, K., Schnell, L., Bunge, M. B., Schwab, M. E., Liebscher, T., and Pearse, D. D. (2005) Combining Schwann cell bridges and olfactory-ensheathing glia grafts with chondroitinase promotes locomotor recovery after complete transection of the spinal cord. *J Neurosci* 25; 1169-78.
- Freed, C. R., Breeze, R. E., Rosenberg, N. L., Schneck, S. A., Kriek, E., Qi, J. X., Lone, T., Zhang, Y. B., Snyder, J. A., Wells, T. H., and et al. (1992) Survival of implanted fetal dopamine cells and neurologic improvement 12 to 46 months after transplantation for Parkinson's disease. N Engl J Med 327; 1549-55.
- Freed, C. R., Greene, P. E., Breeze, R. E., Tsai, W. Y., DuMouchel, W., Kao, R., Dillon, S., Winfield, H., Culver, S., Trojanowski, J. Q., Eidelberg, D., and Fahn, S. (2001) Transplantation of embryonic dopamine neurons for severe Parkinson's disease. *N Engl J Med* 344; 710-9.
- Freeman, T. B., Olanow, C. W., Hauser, R. A., Nauert, G. M., Smith, D. A., Borlongan, C. V., Sanberg, P. R., Holt, D. A., Kordower, J. H., Vingerhoets, F. J., and et al. (1995) Bilateral fetal nigral transplantation into the postcommissural putamen in Parkinson's disease. *Ann Neurol* 38; 379-88.
- Friedenstein, A. J., Chailakhjan, R. K., and Lalykina, K. S. (1970) The development of fibroblast colonies in monolayer cultures of guinea-pig bone marrow and spleen cells. *Cell & Tissue Kinetics* 3; 393-403.
- Friedenstein, A. J., Chailakhyan, R. K., and Gerasimov, U. V. (1987) Bone marrow osteogenic stem cells: in vitro cultivation and transplantation in diffusion chambers. *Cell Tissue Kinet* 20; 263-72.
- Friedenstein, A. J., Chailakhyan, R. K., Latsinik, N. V., Panasyuk, A. F., and Keiliss-Borok, I. V. (1974a) Stromal cells responsible for transferring the microenvironment of the hemopoietic tissues. Cloning in vitro and retransplantation in vivo. *Transplantation* 17; 331-40.
- Friedenstein, A. J., Deriglasova, U. F., Kulagina, N. N., Panasuk, A. F., Rudakowa, S. F., Luria, E. A., and Ruadkow, I. A. (1974b) Precursors for fibroblasts in different populations of hematopoietic cells as detected by the in vitro colony assay method. *Exp Hematol* 2; 83-92.
- Friedenstein, A. J., Gorskaja, J. F., and Kulagina, N. N. (1976) Fibroblast precursors in normal and irradiated mouse hematopoietic organs. *Exp Hematol* **4**; 267-74.

- Friedenstein, A. J., Petrakova, K. V., Kurolesova, A. I., and Frolova, G. P. (1968) Heterotopic of bone marrow. Analysis of precursor cells for osteogenic and hematopoietic tissues. *Transplantation* 6; 230-47.
- Friedenstein, A. J., Piatetzky, S., II, and Petrakova, K. V. (1966) Osteogenesis in transplants of bone marrow cells. *J Embryol Exp Morphol* **16**; 381-90.
- Fu, X., Fang, L., Li, X., Cheng, B., and Sheng, Z. (2006a) Enhanced wound-healing quality with bone marrow mesenchymal stem cells autografting after skin injury. *Wound Repair Regen* 14; 325-35.
- Fu, Y. S., Cheng, Y. C., Lin, M. Y., Cheng, H., Chu, P. M., Chou, S. C., Shih, Y. H., Ko, M. H., and Sung, M. S. (2006b) Conversion of human umbilical cord mesenchymal stem cells in Wharton's jelly to dopaminergic neurons in vitro: potential therapeutic application for Parkinsonism. *Stem Cells* 24; 115-24.
- Gang, E. J., Bosnakovski, D., Figueiredo, C. A., Visser, J. W., and Perlingeiro, R. C. (2007) SSEA-4 identifies mesenchymal stem cells from bone marrow. *Blood* 109; 1743-51.
- Gartner, S., and Kaplan, H. S. (1980) Long-term culture of human bone marrow cells. *Proc Natl Acad Sci U S A* **77**; 4756-9.
- Gasser, T. (1998) Genetics of Parkinson's disease. Ann Neurol 44; S53-7.
- Gelb, D. J., Oliver, E., and Gilman, S. (1999) Diagnostic criteria for Parkinson disease. Arch Neurol 56; 33-9.
- Gerlach, M., and Riederer, P. (1996) Animal models of Parkinson's disease: an empirical comparison with the phenomenology of the disease in man. *J Neural Transm* **103**; 987-1041.
- Giulian, D. (1987) Ameboid microglia as effectors of inflammation in the central nervous system. *J Neurosci Res* 18; 155-71, 132-3.
- Giulian, D., Li, J., Li, X., George, J., and Rutecki, P. A. (1994) The impact of microglia-derived cytokines upon gliosis in the CNS. *Dev Neurosci* 16; 128-36.
- Glennie, S., Soeiro, I., Dyson, P. J., Lam, E. W., and Dazzi, F. (2005) Bone marrow mesenchymal stem cells induce division arrest anergy of activated T cells. *Blood* 105; 2821-7.
- Goodell, M. A. (2003) Stem-cell "plasticity": befuddled by the muddle. *Curr Opin Hematol* **10**; 208-13.

- Greco, S. J., and Rameshwar, P. (2007a) Enhancing effect of IL-1alpha on neurogenesis from adult human mesenchymal stem cells: implication for inflammatory mediators in regenerative medicine. *J Immunol* **179**; 3342-50.
- Greco, S. J., Zhou, C., Ye, J. H., and Rameshwar, P. (2007b) An interdisciplinary approach and characterization of neuronal cells transdifferentiated from human mesenchymal stem cells. *Stem Cells Dev* **16**; 811-26.
- Grinnemo, K. H., Mansson, A., Dellgren, G., Klingberg, D., Wardell, E., Drvota, V., Tammik, C., Holgersson, J., Ringden, O., Sylven, C., and Le Blanc, K. (2004) Xenoreactivity and engraftment of human mesenchymal stem cells transplanted into infarcted rat myocardium. *J Thorac Cardiovasc Surg* 127; 1293-300.
- Gritti, A., Frolichsthal-Schoeller, P., Galli, R., Parati, E. A., Cova, L., Pagano, S. F., Bjornson, C. R., and Vescovi, A. L. (1999) Epidermal and fibroblast growth factors behave as mitogenic regulators for a single multipotent stem cell-like population from the subventricular region of the adult mouse forebrain. J Neurosci 19; 3287-97.
- Gritti, A., Parati, E. A., Cova, L., Frolichsthal, P., Galli, R., Wanke, E., Faravelli, L., Morassutti, D. J., Roisen, F., Nickel, D. D., and Vescovi, A. L. (1996)
 Multipotential stem cells from the adult mouse brain proliferate and self-renew in response to basic fibroblast growth factor. *J Neurosci* 16; 1091-100.
- Gronthos, S., Graves, S. E., Ohta, S., and Simmons, P. J. (1994) The STRO-1+ fraction of adult human bone marrow contains the osteogenic precursors. *Blood* 84; 4164-73.
- Gronthos, S., Zannettino, A. C., Hay, S. J., Shi, S., Graves, S. E., Kortesidis, A., and Simmons, P. J. (2003) Molecular and cellular characterisation of highly purified stromal stem cells derived from human bone marrow. *Journal of Cell Science* 116; 1827-35.
- Guba, S. C., Sartor, C. I., Gottschalk, L. R., Jing, Y. H., Mulligan, T., and Emerson, S. G. (1992) Bone marrow stromal fibroblasts secrete interleukin-6 and granulocyte-macrophage colony-stimulating factor in the absence of inflammatory stimulation: demonstration by serum-free bioassay, enzyme-linked immunosorbent assay, and reverse transcriptase polymerase chain reaction. *Blood* 80; 1190-8.
- Guo, L., Yin, F., Meng, H. Q., Ling, L., Hu-He, T. N., Li, P., Zhang, C. X., Yu, S., Duan, D. S., and Fan, H. X. (2005) Differentiation of mesenchymal stem cells into dopaminergic neuron-like cells in vitro. *Biomed Environ Sci* 18, 36-42.
- Hagell, P., Schrag, A., Piccini, P., Jahanshahi, M., Brown, R., Rehncrona, S., Widner,H., Brundin, P., Rothwell, J. C., Odin, P., Wenning, G. K., Morrish, P., Gustavii,B., Bjorklund, A., Brooks, D. J., Marsden, C. D., Quinn, N. P., and Lindvall, O.

(1999) Sequential bilateral transplantation in Parkinson's disease: effects of the second graft. *Brain* **122** (**Pt 6**); 1121-32.

- Hainfellner, J. A., Voigtlander, T., Strobel, T., Mazal, P. R., Maddalena, A. S., Aguzzi, A., and Budka, H. (2001) Fibroblasts can express glial fibrillary acidic protein (GFAP) in vivo. J Neuropathol Exp Neurol 60; 449-61.
- Hauser, R. A., Freeman, T. B., Snow, B. J., Nauert, M., Gauger, L., Kordower, J. H., and Olanow, C. W. (1999) Long-term evaluation of bilateral fetal nigral transplantation in Parkinson disease. *Arch Neurol* 56; 179-87.
- Hellmann, M. A., Panet, H., Barhum, Y., Melamed, E., and Offen, D. (2006) Increased survival and migration of engrafted mesenchymal bone marrow stem cells in 6hydroxydopamine-lesioned rodents. *Neurosci Lett* 395; 124-8.
- Henderson, J. M., Watson, S., Halliday, G. M., Heinemann, T., and Gerlach, M. (2003) Relationships between various behavioural abnormalities and nigrostriatal dopamine depletion in the unilateral 6-OHDA-lesioned rat. *Behavioural Brain Research* 139; 105-13.
- Hermann, A., Gastl, R., Liebau, S., Popa, M. O., Fiedler, J., Boehm, B. O., Maisel, M., Lerche, H., Schwarz, J., Brenner, R., and Storch, A. (2004) Efficient generation of neural stem cell-like cells from adult human bone marrow stromal cells. J Cell Sci 117; 4411-22.
- Hermann, A., Liebau, S., Gastl, R., Fickert, S., Habisch, H. J., Fiedler, J., Schwarz, J., Brenner, R., and Storch, A. (2006) Comparative analysis of neuroectodermal differentiation capacity of human bone marrow stromal cells using various conversion protocols. *J Neurosci Res* 83; 1502-14.
- Herrera, M. B., Bussolati, B., Bruno, S., Fonsato, V., Romanazzi, G. M., and Camussi, G. (2004) Mesenchymal stem cells contribute to the renal repair of acute tubular epithelial injury. *Int J Mol Med* 14; 1035-41.
- Hickey, W. F. (2001) Basic principles of immunological surveillance of the normal central nervous system. *Glia* **36**; 118-24.
- Hitchcock, E. R., Clough, C., Hughes, R., and Kenny, B. (1988) Embryos and Parkinson's disease. *Lancet* 1; 1274.
- Hofstetter, C. P., Schwarz, E. J., Hess, D., Widenfalk, J., El Manira, A., Prockop, D. J., and Olson, L. (2002) Marrow stromal cells form guiding strands in the injured spinal cord and promote recovery. *Proc Natl Acad Sci U S A* **99**; 2199-204.
- Horner, P. J., and Gage, F. H. (2002) Regeneration in the adult and aging brain. *Arch Neurol* **59**; 1717-20.

- Horwitz, E. M., Gordon, P. L., Koo, W. K., Marx, J. C., Neel, M. D., McNall, R. Y., Muul, L., and Hofmann, T. (2002) Isolated allogeneic bone marrow-derived mesenchymal cells engraft and stimulate growth in children with osteogenesis imperfecta: Implications for cell therapy of bone. *Proc Natl Acad Sci U S A* 99; 8932-7.
- Horwitz, E. M., Le Blanc, K., Dominici, M., Mueller, I., Slaper-Cortenbach, I., Marini, F. C., Deans, R. J., Krause, D. S., and Keating, A. (2005) Clarification of the nomenclature for MSC: The International Society for Cellular Therapy position statement. *Cytotherapy* 7; 393-5.
- Howlett, C. R., Cave, J., Williamson, M., Farmer, J., Ali, S. Y., Bab, I., and Owen, M.
 E. (1986) Mineralization in in vitro cultures of rabbit marrow stromal cells. *Clin Orthop Relat Res*; 251-63.
- Hung, S. C., Cheng, H., Pan, C. Y., Tsai, M. J., Kao, L. S., and Ma, H. L. (2002) In vitro differentiation of size-sieved stem cells into electrically active neural cells. *Stem Cells.* 20; 522-9.
- Ianus, A., Holz, G. G., Theise, N. D., and Hussain, M. A. (2003) In vivo derivation of glucose-competent pancreatic endocrine cells from bone marrow without evidence of cell fusion. *J Clin Invest* 111; 843-50.
- Inoue, Y., Iriyama, A., Ueno, S., Takahashi, H., Kondo, M., Tamaki, Y., Araie, M., and Yanagi, Y. (2007) Subretinal transplantation of bone marrow mesenchymal stem cells delays retinal degeneration in the RCS rat model of retinal degeneration. *Exp Eye Res* 85; 234-41.
- Isacson, O., and Deacon, T. (1997) Neural transplantation studies reveal the brain's capacity for continuous reconstruction. *Trends Neurosci* **20**; 477-82.
- Isacson, O., and Kordower, J. H. (2008) Future of cell and gene therapies for Parkinson's disease. *Ann Neurol* 64 Suppl 2; S122-38.
- Isakova, I. A., Baker, K., DuTreil, M., Dufour, J., Gaupp, D., and Phinney, D. G. (2007) Age- and dose-related effects on MSC engraftment levels and anatomical distribution in the central nervous systems of nonhuman primates: identification of novel MSC subpopulations that respond to guidance cues in brain. *Stem Cells* 25; 3261-70.
- Issarachai, S., Priestley, G. V., Nakamoto, B., and Papayannopoulou, T. (2002) Cells with hemopoietic potential residing in muscle are itinerant bone marrow-derived cells. *Exp Hematol* **30**; 366-73.
- Ito, D., Imai, Y., Ohsawa, K., Nakajima, K., Fukuuchi, Y., and Kohsaka, S. (1998) Microglia-specific localisation of a novel calcium binding protein, Iba1. Brain Res Mol Brain Res 57; 1-9.

- Jenner, P. (2008) Functional models of Parkinson's disease: a valuable tool in the development of novel therapies. *Ann Neurol* **64 Suppl 2**; S16-29.
- Jenner, P., and Olanow, C. W. (1998) Understanding cell death in Parkinson's disease. Ann Neurol 44; S72-84.
- Jiang, X. X., Zhang, Y., Liu, B., Zhang, S. X., Wu, Y., Yu, X. D., and Mao, N. (2005) Human mesenchymal stem cells inhibit differentiation and function of monocyte-derived dendritic cells. *Blood* 105; 4120-6.
- Jiang, Y., Henderson, D., Blackstad, M., Chen, A., Miller, R. F., and Verfaillie, C. M. (2003) Neuroectodermal differentiation from mouse multipotent adult progenitor cells. *Proc Natl Acad Sci U S A*. 100; 11854-60.
- Jiang, Y., Jahagirdar, B. N., Reinhardt, R. L., Schwartz, R. E., Keene, C. D., Ortiz-Gonzalez, X. R., Reyes, M., Lenvik, T., Lund, T., Blackstad, M., Du, J., Aldrich, S., Lisberg, A., Low, W. C., Largaespada, D. A., and Verfaillie, C. M. (2002) Pluripotency of mesenchymal stem cells derived from adult marrow. *Nature* 418; 41-9.
- Jin, G. Z., Cho, S. J., Choi, E. G., Lee, Y. S., Yu, X. F., Choi, K. S., Yee, S. T., Jeon, J. T., Kim, M. O., and Kong, I. K. (2008) Rat mesenchymal stem cells increase tyrosine hydroxylase expression and dopamine content in ventral mesencephalic cells in vitro. *Cell Biol Int* 32; 1433-8.
- Jin, K., Mao, X. O., Batteur, S., Sun, Y., and Greenberg, D. A. (2003) Induction of neuronal markers in bone marrow cells: differential effects of growth factors and patterns of intracellular expression. *Exp Neurol* 184; 78-89.
- Joannides, A., Gaughwin, P., Scott, M., Watt, S., Compston, A., and Chandran, S. (2003) Postnatal astrocytes promote neural induction from adult human bone marrow-derived stem cells. *Journal of Hematotherapy & Stem Cell Research* 12; 681-688.
- Johansson, S., Lee, I. H., Olson, L., and Spenger, C. (2005) Olfactory ensheathing glial co-grafts improve functional recovery in rats with 6-OHDA lesions. *Brain* 128; 2961-76.
- Johnstone, B., Hering, T. M., Caplan, A. I., Goldberg, V. M., and Yoo, J. U. (1998) In vitro chondrogenesis of bone marrow-derived mesenchymal progenitor cells. *Exp Cell Res* 238; 265-72.
- Jones, L. L., Margolis, R. U., and Tuszynski, M. H. (2003) The chondroitin sulfate proteoglycans neurocan, brevican, phosphacan, and versican are differentially regulated following spinal cord injury. *Exp Neurol* **182**; 399-411.

- Jori, F. P., Napolitano, M. A., Melone, M. A., Cipollaro, M., Cascino, A., Altucci, L., Peluso, G., Giordano, A., and Galderisi, U. (2005) Molecular pathways involved in neural in vitro differentiation of marrow stromal stem cells. *J Cell Biochem* 94; 645-55.
- Kachinsky, A. M., Dominov, J. A., and Miller, J. B. (1994) Myogenesis and the intermediate filament protein, nestin. *Dev Biol* **165**; 216-28.
- Kaltz, N., Funari, A., Hippauf, S., Delorme, B., Noel, D., Riminucci, M., Jacobs, V. R., Haupl, T., Jorgensen, C., Charbord, P., Peschel, C., Bianco, P., and Oostendorp, R. A. (2008) In vivo osteoprogenitor potency of human stromal cells from different tissues does not correlate with expression of POU5F1 or its pseudogenes. *Stem Cells* 26; 2419-24.
- Kan, I., Ben-Zur, T., Barhum, Y., Levy, Y. S., Burstein, A., Charlow, T., Bulvik, S., Melamed, E., and Offen, D. (2007a) Dopaminergic differentiation of human mesenchymal stem cells--utilization of bioassay for tyrosine hydroxylase expression. *Neurosci Lett* **419**; 28-33.
- Kan, I., Melamed, E., Offen, D., and Green, P. (2007b) Docosahexaenoic acid and arachidonic acid are fundamental supplements for the induction of neuronal differentiation. J Lipid Res 48; 513-7.
- Kawada, H., and Ogawa, M. (2001) Bone marrow origin of hematopoietic progenitors and stem cells in murine muscle. *Blood* **98**; 2008-13.
- Kern, S., Eichler, H., Stoeve, J., Kluter, H., and Bieback, K. (2006) Comparative analysis of mesenchymal stem cells from bone marrow, umbilical cord blood, or adipose tissue. *Stem Cells* 24; 1294-301.
- Khoo, M. L., Shen, B., Tao, H., and Ma, D. D. (2008) Long-term serial passage and neuronal differentiation capability of human bone marrow mesenchymal stem cells. *Stem Cells Dev* 17; 883-96.
- Kicic, A., Shen, W. Y., Wilson, A. S., Constable, I. J., Robertson, T., and Rakoczy, P. E. (2003) Differentiation of marrow stromal cells into photoreceptors in the rat eye. *J Neurosci* 23; 7742-9.
- Kim, B. J., Seo, J. H., Bubien, J. K., and Oh, Y. S. (2002a) Differentiation of adult bone marrow stem cells into neuroprogenitor cells in vitro. *Neuroreport* 13; 1185-8.
- Kim, J. H., Auerbach, J. M., Rodriguez-Gomez, J. A., Velasco, I., Gavin, D., Lumelsky, N., Lee, S. H., Nguyen, J., Sanchez-Pernaute, R., Bankiewicz, K., and McKay, R. (2002b) Dopamine neurons derived from embryonic stem cells function in an animal model of Parkinson's disease. *Nature*. 418; 50-6.

- Kim, W. G., Mohney, R. P., Wilson, B., Jeohn, G. H., Liu, B., and Hong, J. S. (2000) Regional difference in susceptibility to lipopolysaccharide-induced neurotoxicity in the rat brain: role of microglia. *J Neurosci* 20; 6309-16.
- Kittappa, R., Chang, W. W., Awatramani, R. B., and McKay, R. D. (2007) The foxa2 gene controls the birth and spontaneous degeneration of dopamine neurons in old age. *PLoS Biol* 5; e325.
- Kittler, E. L., McGrath, H., Temeles, D., Crittenden, R. B., Kister, V. K., and Quesenberry, P. J. (1992) Biologic significance of constitutive and subliminal growth factor production by bone marrow stroma. *Blood* **79**; 3168-78.
- Klein, T., Ling, Z., Heimberg, H., Madsen, O. D., Heller, R. S., and Serup, P. (2003) Nestin is expressed in vascular endothelial cells in the adult human pancreas. J Histochem Cytochem 51; 697-706.
- Klockgether, T. (2004) Parkinson's disease: clinical aspects. *Cell Tissue Res* **318**; 115-20.
- Kohyama, J., Abe, H., Shimazaki, T., Koizumi, A., Nakashima, K., Gojo, S., Taga, T., Okano, H., Hata, J., and Umezawa, A. (2001) Brain from bone: efficient "metadifferentiation" of marrow stroma-derived mature osteoblasts to neurons with Noggin or a demethylating agent. *Differentiation* 68; 235-44.
- Kondo, T., Johnson, S. A., Yoder, M. C., Romand, R., and Hashino, E. (2005) Sonic hedgehog and retinoic acid synergistically promote sensory fate specification from bone marrow-derived pluripotent stem cells. *Proc Natl Acad Sci U S A* 102; 4789-94.
- Kopen, G. C., Prockop, D. J., and Phinney, D. G. (1999) Marrow stromal cells migrate throughout forebrain and cerebellum, and they differentiate into astrocytes after injection into neonatal mouse brains. *Proc Natl Acad Sci U S A* 96; 10711-6.
- Kordower, J. H., Chu, Y., Hauser, R. A., Freeman, T. B., and Olanow, C. W. (2008) Lewy body-like pathology in long-term embryonic nigral transplants in Parkinson's disease. *Nat Med* 14; 504-6.
- Kordower, J. H., Freeman, T. B., Snow, B. J., Vingerhoets, F. J., Mufson, E. J., Sanberg, P. R., Hauser, R. A., Smith, D. A., Nauert, G. M., Perl, D. P., and et al. (1995) Neuropathological evidence of graft survival and striatal reinnervation after the transplantation of fetal mesencephalic tissue in a patient with Parkinson's disease. *N Engl J Med* 332; 1118-24.
- Kotoula, V., Papamichos, S. I., and Lambropoulos, A. F. (2008) Revisiting OCT4 expression in peripheral blood mononuclear cells. *Stem Cells* **26**; 290-1.

- Kotton, D. N., Ma, B. Y., Cardoso, W. V., Sanderson, E. A., Summer, R. S., Williams, M. C., and Fine, A. (2001) Bone marrow-derived cells as progenitors of lung alveolar epithelium. *Development* 128; 5181-8.
- Kramer, B. C., Woodbury, D., and Black, I. B. (2006) Adult rat bone marrow stromal cells express genes associated with dopamine neurons. *Biochem Biophys Res Commun* 343; 1045-52.
- Kramer, E. R., Aron, L., Ramakers, G. M., Seitz, S., Zhuang, X., Beyer, K., Smidt, M. P., and Klein, R. (2007) Absence of Ret signaling in mice causes progressive and late degeneration of the nigrostriatal system. *PLoS Biol* 5; e39.
- Krampera, M., Cosmi, L., Angeli, R., Pasini, A., Liotta, F., Andreini, A., Santarlasci, V., Mazzinghi, B., Pizzolo, G., Vinante, F., Romagnani, P., Maggi, E., Romagnani, S., and Annunziato, F. (2006) Role for interferon-gamma in the immunomodulatory activity of human bone marrow mesenchymal stem cells. *Stem Cells* 24; 386-98.
- Krampera, M., Glennie, S., Dyson, J., Scott, D., Laylor, R., Simpson, E., and Dazzi, F. (2003) Bone marrow mesenchymal stem cells inhibit the response of naive and memory antigen-specific T cells to their cognate peptide. *Blood* 101; 3722-9.
- Kuhn, H. G., Winkler, J., Kempermann, G., Thal, L. J., and Gage, F. H. (1997)
 Epidermal growth factor and fibroblast growth factor-2 have different effects on neural progenitors in the adult rat brain. *J Neurosci* 17; 5820-9.
- Lamoury, F. M., Croitoru-Lamoury, J., and Brew, B. J. (2006) Undifferentiated mouse mesenchymal stem cells spontaneously express neural and stem cell markers Oct-4 and Rex-1. *Cytotherapy* 8; 228-42.
- Langston, J. W. (1998) Epidemiology versus genetics in Parkinson's disease: progress in resolving an age-old debate. *Ann Neurol* 44; S45-52.
- Langston, J. W., Ballard, P., Tetrud, J. W., and Irwin, I. (1983) Chronic Parkinsonism in humans due to a product of meperidine-analog synthesis. *Science* **219**; 979-80.
- Lanotte, M., Scott, D., Dexter, T. M., and Allen, T. D. (1982) Clonal preadipocyte cell lines with different phenotypes derived from murine marrow stroma: factors influencing growth and adipogenesis in vitro. *J Cell Physiol* 111; 177-86.
- Larsson, L. C., Czech, K. A., Brundin, P., and Widner, H. (2000) Intrastriatal ventral mesencephalic xenografts of porcine tissue in rats: immune responses and functional effects. *Cell Transplant* 9; 261-72.
- Larsson, L. C., Czech, K. A., Widner, H., and Korsgren, O. (1999) Discordant neural tissue xenografts survive longer in immunoglobulin deficient mice. *Transplantation* 68; 1153-60.

- Larsson, L. C., Frielingsdorf, H., Mirza, B., Hansson, S. J., Anderson, P., Czech, K. A., Strandberg, M., and Widner, H. (2001) Porcine neural xenografts in rats and mice: donor tissue development and characteristics of rejection. *Exp Neurol* 172; 100-14.
- Lawson, L. J., Perry, V. H., Dri, P., and Gordon, S. (1990) Heterogeneity in the distribution and morphology of microglia in the normal adult mouse brain. *Neuroscience* 39; 151-70.
- Le Blanc, K., Rasmusson, I., Sundberg, B., Gotherstrom, C., Hassan, M., Uzunel, M., and Ringden, O. (2004) Treatment of severe acute graft-versus-host disease with third party haploidentical mesenchymal stem cells. *Lancet* **363**; 1439-41.
- Le Blanc, K., and Ringden, O. (2005) Immunobiology of human mesenchymal stem cells and future use in hematopoietic stem cell transplantation. *Biol Blood Marrow Transplant* 11; 321-34.
- Le Blanc, K., Tammik, L., Sundberg, B., Haynesworth, S. E., and Ringden, O. (2003) Mesenchymal stem cells inhibit and stimulate mixed lymphocyte cultures and mitogenic responses independently of the major histocompatibility complex. *Scand J Immunol* **57**; 11-20.
- Lee, J., Kim, H. K., Rho, J. Y., Han, Y. M., and Kim, J. (2006) The human OCT-4 isoforms differ in their ability to confer self-renewal. *J Biol Chem* **281**; 33554-65.
- Lee, K. D., Kuo, T. K., Whang-Peng, J., Chung, Y. F., Lin, C. T., Chou, S. H., Chen, J. R., Chen, Y. P., and Lee, O. K. (2004) In vitro hepatic differentiation of human mesenchymal stem cells. *Hepatology* 40; 1275-84.
- Lee, S. H., Lumelsky, N., Studer, L., Auerbach, J. M., and McKay, R. D. (2000) Efficient generation of midbrain and hindbrain neurons from mouse embryonic stem cells. *Nature Biotechnology*. 18; 675-9.
- Lemischka, I. (2002a) A few thoughts about the plasticity of stem cells. *Exp Hematol* **30**; 848-52.
- Lemischka, I. (2002b) Rethinking somatic stem cell plasticity. Nat Biotechnol 20; 425.
- Lendahl, U., Zimmerman, L. B., and McKay, R. D. (1990) CNS stem cells express a new class of intermediate filament protein. *Cell* **60**; 585-95.
- Levy, Y. S., Bahat-Stroomza, M., Barzilay, R., Burshtein, A., Bulvik, S., Barhum, Y., Panet, H., Melamed, E., and Offen, D. (2008) Regenerative effect of neuralinduced human mesenchymal stromal cells in rat models of Parkinson's disease. *Cytotherapy* 10; 340-52.

- Li, J. Y., Englund, E., Holton, J. L., Soulet, D., Hagell, P., Lees, A. J., Lashley, T., Quinn, N. P., Rehncrona, S., Bjorklund, A., Widner, H., Revesz, T., Lindvall, O., and Brundin, P. (2008) Lewy bodies in grafted neurons in subjects with Parkinson's disease suggest host-to-graft disease propagation. *Nat Med* 14; 501-3.
- Li, Y., Chen, J., Wang, L., Lu, M., and Chopp, M. (2001a) Treatment of stroke in rat with intracarotid administration of marrow stromal cells. *Neurology* 56; 1666-72.
- Li, Y., Chen, J., Wang, L., Zhang, L., Lu, M., and Chopp, M. (2001b) Intracerebral transplantation of bone marrow stromal cells in a 1-methyl-4-phenyl-1,2,3,6tetrahydropyridine mouse model of Parkinson's disease. *Neurosci Lett* 316; 67-70.
- Li, Y., Chopp, M., Chen, J., Wang, L., Gautam, S. C., Xu, Y. X., and Zhang, Z. (2000) Intrastriatal transplantation of bone marrow nonhematopoietic cells improves functional recovery after stroke in adult mice. *J Cereb Blood Flow Metab* 20; 1311-9.
- Liechty, K. W., MacKenzie, T. C., Shaaban, A. F., Radu, A., Moseley, A. M., Deans, R., Marshak, D. R., and Flake, A. W. (2000) Human mesenchymal stem cells engraft and demonstrate site-specific differentiation after in utero transplantation in sheep. *Nat Med* 6; 1282-6.
- Liedtke, S., Stephan, M., and Kogler, G. (2008) Oct4 expression revisited: potential pitfalls for data misinterpretation in stem cell research. *Biol Chem* **389**; 845-50.
- Lin, T. M., Tsai, J. L., Lin, S. D., Lai, C. S., and Chang, C. C. (2005) Accelerated growth and prolonged lifespan of adipose tissue-derived human mesenchymal stem cells in a medium using reduced calcium and antioxidants. *Stem Cells Dev* 14; 92-102.
- Lindvall, O., Brundin, P., Widner, H., Rehncrona, S., Gustavii, B., Frackowiak, R., Leenders, K. L., Sawle, G., Rothwell, J. C., Marsden, C. D., and et al. (1990) Grafts of fetal dopamine neurons survive and improve motor function in Parkinson's disease. *Science* 247; 574-7.
- Lindvall, O., Rehncrona, S., Brundin, P., Gustavii, B., Astedt, B., Widner, H.,
 Lindholm, T., Bjorklund, A., Leenders, K. L., Rothwell, J. C., and et al. (1989)
 Human fetal dopamine neurons grafted into the striatum in two patients with
 severe Parkinson's disease. A detailed account of methodology and a 6-month
 follow-up. Arch Neurol 46; 615-31.
- Lindvall, O., Rehncrona, S., Gustavii, B., Brundin, P., Astedt, B., Widner, H., Lindholm, T., Bjorklund, A., Leenders, K. L., Rothwell, J. C., and et al. (1988) Fetal dopamine-rich mesencephalic grafts in Parkinson's disease. *Lancet* 2; 1483-4.

- Lindvall, O., Sawle, G., Widner, H., Rothwell, J. C., Bjorklund, A., Brooks, D., Brundin, P., Frackowiak, R., Marsden, C. D., Odin, P., and et al. (1994) Evidence for long-term survival and function of dopaminergic grafts in progressive Parkinson's disease. *Ann Neurol* 35; 172-80.
- Liu, C. T., Yang, Y. J., Yin, F., Wang, X., Yu, X. H., Wang, Q. H., Wang, X. L., and Xie, M. (2006) The immunobiological development of human bone marrow mesenchymal stem cells in the course of neuronal differentiation. *Cell Immunol* 244; 19-32.
- Livak, K. J., and Schmittgen, T. D. (2001) Analysis of relative gene expression data using real-time quantitative PCR and the 2(-Delta Delta C(T)) Method. *Methods* **25**; 402-8.
- Locatelli, F., Maccario, R., and Frassoni, F. (2007) Mesenchymal stromal cells, from indifferent spectators to principal actors. Are we going to witness a revolution in the scenario of allograft and immune-mediated disorders? *Haematologica* **92**; 872-7.
- Loh, Y. H., Wu, Q., Chew, J. L., Vega, V. B., Zhang, W., Chen, X., Bourque, G., George, J., Leong, B., Liu, J., Wong, K. Y., Sung, K. W., Lee, C. W., Zhao, X. D., Chiu, K. P., Lipovich, L., Kuznetsov, V. A., Robson, P., Stanton, L. W., Wei, C. L., Ruan, Y., Lim, B., and Ng, H. H. (2006) The Oct4 and Nanog transcription network regulates pluripotency in mouse embryonic stem cells. *Nat Genet* 38; 431-40.
- Long, X., Olszewski, M., Huang, W., and Kletzel, M. (2005) Neural cell differentiation in vitro from adult human bone marrow mesenchymal stem cells. *Stem Cells Dev* 14; 65-9.
- Lorenz, E., Uphoff, D., Reid, T. R., and Shelton, E. (1951) Modification of irradiation injury in mice and guinea pigs by bone marrow injections. *J Natl Cancer Inst* 12; 197-201.
- Lu, D., Mahmood, A., Wang, L., Li, Y., Lu, M., and Chopp, M. (2001a) Adult bone marrow stromal cells administered intravenously to rats after traumatic brain injury migrate into brain and improve neurological outcome. *Neuroreport* 12; 559-63.
- Lu, J., Feron, F., Ho, S. M., Mackay-Sim, A., and Waite, P. M. (2001b) Transplantation of nasal olfactory tissue promotes partial recovery in paraplegic adult rats. *Brain Res* 889; 344-57.
- Lu, J., Feron, F., Mackay-Sim, A., and Waite, P. M. (2002) Olfactory ensheathing cells promote locomotor recovery after delayed transplantation into transected spinal cord. *Brain* 125; 14-21.

- Lu, L., Zhao, C., Liu, Y., Sun, X., Duan, C., Ji, M., Zhao, H., Xu, Q., and Yang, H. (2005) Therapeutic benefit of TH-engineered mesenchymal stem cells for Parkinson's disease. *Brain Res Brain Res Protoc* 15, 46-51.
- Lu, P., Blesch, A., and Tuszynski, M. H. (2004) Induction of bone marrow stromal cells to neurons: differentiation, transdifferentiation, or artifact? *J Neurosci Res* 77; 174-91.
- Luk, J. M., Wang, P. P., Lee, C. K., Wang, J. H., and Fan, S. T. (2005) Hepatic potential of bone marrow stromal cells: development of in vitro co-culture and intra-portal transplantation models. *J Immunol Methods* 305; 39-47.
- Ma, D. D. (1990) Hematopoietic Reconstitution Following Bone Marrow Transplantation, pp. 111-139. In F. L. Johnson, and C. Pochedly (Eds): Bone Marrow Transplantation in Children, Raven Press, Ltd., New York.
- Ma, D. D., Da, W. M., Purvis-Smith, S., and Biggs, J. C. (1987) Chromosomal analysis of bone marrow stromal fibroblasts in allogeneic HLA compatible sibling bone marrow transplantations. *Leuk Res* 11; 661-3.
- Ma, Y., Xu, Y., Xiao, Z., Yang, W., Zhang, C., Song, E., Du, Y., and Li, L. (2006) Reconstruction of chemically burned rat corneal surface by bone marrowderived human mesenchymal stem cells. *Stem Cells* 24; 315-21.
- Maccario, R., Podesta, M., Moretta, A., Cometa, A., Comoli, P., Montagna, D., Daudt, L., Ibatici, A., Piaggio, G., Pozzi, S., Frassoni, F., and Locatelli, F. (2005)
 Interaction of human mesenchymal stem cells with cells involved in alloantigen-specific immune response favors the differentiation of CD4+ T-cell subsets expressing a regulatory/suppressive phenotype. *Haematologica* 90; 516-25.
- Mackay, A. M., Beck, S. C., Murphy, J. M., Barry, F. P., Chichester, C. O., and Pittenger, M. F. (1998) Chondrogenic differentiation of cultured human mesenchymal stem cells from marrow. *Tissue Eng* 4; 415-28.
- Mackay-Sim, A. (2005) Olfactory ensheathing cells and spinal cord repair. *Keio J Med* **54**; 8-14.
- Mahmood, A., Lu, D., Lu, M., and Chopp, M. (2003) Treatment of traumatic brain injury in adult rats with intravenous administration of human bone marrow stromal cells. *Neurosurgery* 53; 697-702; discussion 702-3.
- Mahmood, A., Lu, D., Wang, L., and Chopp, M. (2002) Intracerebral transplantation of marrow stromal cells cultured with neurotrophic factors promotes functional recovery in adult rats subjected to traumatic brain injury. *J Neurotrauma* 19: 1609-17.

- Mahmood, A., Lu, D., Wang, L., Li, Y., Lu, M., and Chopp, M. (2001) Treatment of traumatic brain injury in female rats with intravenous administration of bone marrow stromal cells. *Neurosurgery* 49; 1196-203; discussion 1203-4.
- Maniatis, A., Tavassoli, M., and Crosby, W. H. (1971) Origin of osteogenic precursor cells in extramedullary marrow implants. *Blood* **38**; 569-75.
- Mareschi, K., Novara, M., Rustichelli, D., Ferrero, I., Guido, D., Carbone, E., Medico, E., Madon, E., Vercelli, A., and Fagioli, F. (2006) Neural differentiation of human mesenchymal stem cells: Evidence for expression of neural markers and eag K+ channel types. *Exp Hematol* 34; 1563-72.
- Martinez, C., Hofmann, T. J., Marino, R., Dominici, M., and Horwitz, E. M. (2007) Human bone marrow mesenchymal stromal cells express the neural ganglioside GD2: a novel surface marker for the identification of MSCs. *Blood* **109**; 4245-8.
- Matsumoto, Y., Ohmori, K., and Fujiwara, M. (1992) Microglial and astroglial reactions to inflammatory lesions of experimental autoimmune encephalomyelitis in the rat central nervous system. *J Neuroimmunol* **37**; 23-33.
- Maxwell, S. L., and Li, M. (2005) Midbrain dopaminergic development in vivo and in vitro from embryonic stem cells. *J Anat* **207**; 209-18.
- McKeon, R. J., Jurynec, M. J., and Buck, C. R. (1999) The chondroitin sulfate proteoglycans neurocan and phosphacan are expressed by reactive astrocytes in the chronic CNS glial scar. J Neurosci 19; 10778-88.
- McKinney-Freeman, S. L., Jackson, K. A., Camargo, F. D., Ferrari, G., Mavilio, F., and Goodell, M. A. (2002) Muscle-derived hematopoietic stem cells are hematopoietic in origin. *Proc Natl Acad Sci U S A* **99**; 1341-6.
- Meisel, R., Zibert, A., Laryea, M., Gobel, U., Daubener, W., and Dilloo, D. (2004) Human bone marrow stromal cells inhibit allogeneic T-cell responses by indoleamine 2,3-dioxygenase-mediated tryptophan degradation. *Blood* **103**; 4619-21.
- Mendez, I., Vinuela, A., Astradsson, A., Mukhida, K., Hallett, P., Robertson, H., Tierney, T., Holness, R., Dagher, A., Trojanowski, J. Q., and Isacson, O. (2008) Dopamine neurons implanted into people with Parkinson's disease survive without pathology for 14 years. *Nat Med* 14; 507-9.
- Mets, T., and Verdonk, G. (1981) In vitro aging of human bone marrow derived stromal cells. *Mech Ageing Dev* 16; 81-9.
- Meuleman, N., Tondreau, T., Delforge, A., Dejeneffe, M., Massy, M., Libertalis, M., Bron, D., and Lagneaux, L. (2006) Human marrow mesenchymal stem cell

culture: serum-free medium allows better expansion than classical alpha-MEM medium. *Eur J Haematol* **76**; 309-16.

- Minguell, J. J., Fierro, F. A., Epunan, M. J., Erices, A. A., and Sierralta, W. D. (2005) Nonstimulated human uncommitted mesenchymal stem cells express cell markers of mesenchymal and neural lineages. *Stem Cells Dev* 14; 408-14.
- Mokry, J., and Nemecek, S. (1998a) Angiogenesis of extra- and intraembryonic blood vessels is associated with expression of nestin in endothelial cells. *Folia Biol* (*Praha*) 44; 155-61.
- Mokry, J., and Nemecek, S. (1998b) Immunohistochemical detection of intermediate filament nestin. *Acta Medica (Hradec Kralove)* **41**; 73-80.
- Morigi, M., Imberti, B., Zoja, C., Corna, D., Tomasoni, S., Abbate, M., Rottoli, D., Angioletti, S., Benigni, A., Perico, N., Alison, M., and Remuzzi, G. (2004)
 Mesenchymal stem cells are renotropic, helping to repair the kidney and improve function in acute renal failure. *J Am Soc Nephrol* 15, 1794-804.
- Morrison, S. J. (2001) Stem cell potential: can anything make anything? *Curr Biol* **11**; R7-9.
- Morshead, C. M., Benveniste, P., Iscove, N. N., and van der Kooy, D. (2002) Hematopoietic competence is a rare property of neural stem cells that may depend on genetic and epigenetic alterations. *Nature Medicine*. **8**; 268-73.
- Mueller, T., Luetzkendorf, J., Nerger, K., Schmoll, H. J., and Mueller, L. P. (2009) Analysis of OCT4 expression in an extended panel of human tumor cell lines from multiple entities and in human mesenchymal stem cells. *Cell Mol Life Sci* 66; 495-503.
- Munoz, J. R., Stoutenger, B. R., Robinson, A. P., Spees, J. L., and Prockop, D. J. (2005) Human stem/progenitor cells from bone marrow promote neurogenesis of endogenous neural stem cells in the hippocampus of mice. *Proc Natl Acad Sci U* SA 102; 18171-6.
- Munoz-Elias, G., Marcus, A. J., Coyne, T. M., Woodbury, D., and Black, I. B. (2004) Adult bone marrow stromal cells in the embryonic brain: engraftment, migration, differentiation, and long-term survival. *J Neurosci* 24; 4585-95.
- Munoz-Elias, G., Woodbury, D., and Black, I. B. (2003) Marrow stromal cells, mitosis, and neuronal differentiation: stem cell and precursor functions. *Stem Cells.* **21**; 437-48.
- Murrell, W., Wetzig, A., Donnellan, M., Feron, F., Burne, T., Meedeniya, A., Kesby, J., Bianco, J., Perry, C., Silburn, P., and Mackay-Sim, A. (2008) Olfactory mucosa

is a potential source for autologous stem cell therapy for Parkinson's disease. *Stem Cells* **26**; 2183-92.

- Nagoshi, N., Shibata, S., Kubota, Y., Nakamura, M., Nagai, Y., Satoh, E., Morikawa, S., Okada, Y., Mabuchi, Y., Katoh, H., Okada, S., Fukuda, K., Suda, T., Matsuzaki, Y., Toyama, Y., and Okano, H. (2008) Ontogeny and multipotency of neural crest-derived stem cells in mouse bone marrow, dorsal root ganglia, and whisker pad. *Cell Stem Cell* 2; 392-403.
- Nakagawa, H., Akita, S., Fukui, M., Fujii, T., and Akino, K. (2005) Human mesenchymal stem cells successfully improve skin-substitute wound healing. *Br J Dermatol* 153; 29-36.
- Nakao, N., Frodl, E. M., Duan, W. M., Widner, H., and Brundin, P. (1994) Lazaroids improve the survival of grafted rat embryonic dopamine neurons. *Proc Natl Acad Sci U S A* 91; 12408-12.
- Nauta, A. J., and Fibbe, W. E. (2007) Immunomodulatory properties of mesenchymal stromal cells. *Blood* **110**; 3499-506.
- Nauta, A. J., Kruisselbrink, A. B., Lurvink, E., Willemze, R., and Fibbe, W. E. (2006a) Mesenchymal stem cells inhibit generation and function of both CD34+-derived and monocyte-derived dendritic cells. *J Immunol* 177; 2080-7.
- Nauta, A. J., Westerhuis, G., Kruisselbrink, A. B., Lurvink, E. G., Willemze, R., and Fibbe, W. E. (2006b) Donor-derived mesenchymal stem cells are immunogenic in an allogeneic host and stimulate donor graft rejection in a nonmyeloablative setting. *Blood* 108; 2114-20.
- Neuhuber, B., Gallo, G., Howard, L., Kostura, L., Mackay, A., and Fischer, I. (2004) Reevaluation of in vitro differentiation protocols for bone marrow stromal cells: disruption of actin cytoskeleton induces rapid morphological changes and mimics neuronal phenotype. J Neurosci Res 77; 192-204.
- Niki, T., Pekny, M., Hellemans, K., Bleser, P. D., Berg, K. V., Vaeyens, F., Quartier, E., Schuit, F., and Geerts, A. (1999) Class VI intermediate filament protein nestin is induced during activation of rat hepatic stellate cells. *Hepatology* 29: 520-7.
- Nimmerjahn, A., Kirchhoff, F., and Helmchen, F. (2005) Resting microglial cells are highly dynamic surveillants of brain parenchyma in vivo. *Science* **308**; 1314-8.
- Nowell, P. C., Cole, L. J., Habermeyer, J. G., and Roan, P. L. (1956) Growth and continued function of rat marrow cells in x-radiated mice. *Cancer Res* 16; 258-61.

- Ohtaki, H., Ylostalo, J. H., Foraker, J. E., Robinson, A. P., Reger, R. L., Shioda, S., and Prockop, D. J. (2008) Stem/progenitor cells from bone marrow decrease neuronal death in global ischemia by modulation of inflammatory/immune responses. *Proc Natl Acad Sci U S A* **105**; 14638-43.
- Olanow, C. W., Goetz, C. G., Kordower, J. H., Stoessl, A. J., Sossi, V., Brin, M. F., Shannon, K. M., Nauert, G. M., Perl, D. P., Godbold, J., and Freeman, T. B. (2003) A double-blind controlled trial of bilateral fetal nigral transplantation in Parkinson's disease. *Annals of Neurology* 54; 403-14.
- Olanow, C. W., and Tatton, W. G. (1999) Etiology and pathogenesis of Parkinson's disease. *Annu Rev Neurosci* 22; 123-44.
- Ong, S. Y., Dai, H., and Leong, K. W. (2006) Inducing hepatic differentiation of human mesenchymal stem cells in pellet culture. *Biomaterials* 27; 4087-97.
- Orkin, S. H., and Zon, L. I. (2002) Hematopoiesis and stem cells: plasticity versus developmental heterogeneity. *Nat Immunol* **3**; 323-8.
- Orr, C. F., Rowe, D. B., and Halliday, G. M. (2002) An inflammatory review of Parkinson's disease. *Prog Neurobiol* **68**; 325-40.
- Ortiz, L. A., Gambelli, F., McBride, C., Gaupp, D., Baddoo, M., Kaminski, N., and Phinney, D. G. (2003) Mesenchymal stem cell engraftment in lung is enhanced in response to bleomycin exposure and ameliorates its fibrotic effects. *Proc Natl Acad Sci U S A* **100**; 8407-11.
- Ostenfeld, T., Caldwell, M. A., Prowse, K. R., Linskens, M. H., Jauniaux, E., and Svendsen, C. N. (2000) Human neural precursor cells express low levels of telomerase in vitro and show diminishing cell proliferation with extensive axonal outgrowth following transplantation. *Exp Neurol* **164**; 215-26.
- Ourednik, V., and Ourednik, J. (2004) Multifaceted dialogue between graft and host in neurotransplantation. *J Neurosci Res* **76**; 193-204.
- Ourednik, V., Ourednik, J., Xu, Y., Zhang, Y., Lynch, W. P., Snyder, E. Y., and Schachner, M. (2009) Cross-talk between stem cells and the dysfunctional brain is facilitated by manipulating the niche: evidence from an adhesion molecule. *Stem Cells* **27**; 2846-56.
- Owen, M., and Friedenstein, A. J. (1988) Stromal stem cells: marrow-derived osteogenic precursors. *Ciba Found Symp* **136**; 42-60.
- Oyagi, S., Hirose, M., Kojima, M., Okuyama, M., Kawase, M., Nakamura, T., Ohgushi, H., and Yagi, K. (2006) Therapeutic effect of transplanting HGF-treated bone marrow mesenchymal cells into CCl4-injured rats. *J Hepatol* 44; 742-8.

- Pacary, E., Legros, H., Valable, S., Duchatelle, P., Lecocq, M., Petit, E., Nicole, O., and Bernaudin, M. (2006) Synergistic effects of CoCl(2) and ROCK inhibition on mesenchymal stem cell differentiation into neuron-like cells. *J Cell Sci* 119; 2667-78.
- Perlow, M. J., Freed, W. J., Hoffer, B. J., Seiger, A., Olson, L., and Wyatt, R. J. (1979) Brain grafts reduce motor abnormalities produced by destruction of nigrostriatal dopamine system. *Science* 204; 643-7.
- Perry, V. H., Bell, M. D., Brown, H. C., and Matyszak, M. K. (1995) Inflammation in the nervous system. *Curr Opin Neurobiol* 5; 636-41.
- Peschanski, M., Defer, G., N'Guyen, J. P., Ricolfi, F., Monfort, J. C., Remy, P., Geny, C., Samson, Y., Hantraye, P., Jeny, R., and et al. (1994) Bilateral motor improvement and alteration of L-dopa effect in two patients with Parkinson's disease following intrastriatal transplantation of foetal ventral mesencephalon. *Brain* **117** (**Pt 3**); 487-99.
- Petersen, B. E., Bowen, W. C., Patrene, K. D., Mars, W. M., Sullivan, A. K., Murase, N., Boggs, S. S., Greenberger, J. S., and Goff, J. P. (1999) Bone marrow as a potential source of hepatic oval cells. *Science* 284; 1168-70.
- Phinney, D. G., Hill, K., Michelson, C., DuTreil, M., Hughes, C., Humphries, S., Wilkinson, R., Baddoo, M., and Bayly, E. (2006) Biological activities encoded by the murine mesenchymal stem cell transcriptome provide a basis for their developmental potential and broad therapeutic efficacy. *Stem Cells* 24; 186-98.
- Phinney, D. G., and Prockop, D. J. (2007) Concise review: mesenchymal stem/multipotent stromal cells: the state of transdifferentiation and modes of tissue repair--current views. *Stem Cells* 25; 2896-902.
- Piccini, P., Brooks, D. J., Bjorklund, A., Gunn, R. N., Grasby, P. M., Rimoldi, O., Brundin, P., Hagell, P., Rehncrona, S., Widner, H., and Lindvall, O. (1999) Dopamine release from nigral transplants visualized in vivo in a Parkinson's patient. *Nature Neuroscience* 2; 1137-40.
- Pierret, C., Spears, K., Maruniak, J. A., and Kirk, M. D. (2006) Neural crest as the source of adult stem cells. *Stem Cells Dev* **15**; 286-91.
- Pittenger, M. F., Mackay, A. M., Beck, S. C., Jaiswal, R. K., Douglas, R., Mosca, J. D., Moorman, M. A., Simonetti, D. W., Craig, S., and Marshak, D. R. (1999) Multilineage potential of adult human mesenchymal stem cells. *Science* 284; 143-7.
- Pochampally, R. R., Neville, B. T., Schwarz, E. J., Li, M. M., and Prockop, D. J. (2004a) Rat adult stem cells (marrow stromal cells) engraft and differentiate in

chick embryos without evidence of cell fusion. *Proc Natl Acad Sci U S A* **101**; 9282-5.

- Pochampally, R. R., Smith, J. R., Ylostalo, J., and Prockop, D. J. (2004b) Serum deprivation of human marrow stromal cells (hMSCs) selects for a subpopulation of early progenitor cells with enhanced expression of OCT-4 and other embryonic genes. *Blood* 103; 1647-52.
- Prakash, N., and Wurst, W. (2006) Development of dopaminergic neurons in the mammalian brain. *Cell Mol Life Sci* 63; 187-206.
- Prockop, D. J. (1997) Marrow stromal cells as stem cells for nonhematopoietic tissues. *Science* 276; 71-4.
- Prockop, D. J. (2007) "Stemness" does not explain the repair of many tissues by mesenchymal stem/multipotent stromal cells (MSCs). *Clin Pharmacol Ther* 82; 241-3.
- Qu, T. Y., Dong, X. J., Sugaya, I., Vaghani, A., Pulido, J., and Sugaya, K. (2004) Bromodeoxyuridine increases multipotency of human bone marrow-derived stem cells. *Restor Neurol Neurosci* 22; 459-68.
- Quirici, N., Soligo, D., Bossolasco, P., Servida, F., Lumini, C., and Deliliers, G. L. (2002) Isolation of bone marrow mesenchymal stem cells by anti-nerve growth factor receptor antibodies. *Experimental Hematology*. **30**; 783-91.
- Ramasamy, R., Lam, E. W., Soeiro, I., Tisato, V., Bonnet, D., and Dazzi, F. (2007) Mesenchymal stem cells inhibit proliferation and apoptosis of tumor cells: impact on in vivo tumor growth. *Leukemia* 21; 304-10.
- Ramasamy, R., Tong, C. K., Seow, H. F., Vidyadaran, S., and Dazzi, F. (2008) The immunosuppressive effects of human bone marrow-derived mesenchymal stem cells target T cell proliferation but not its effector function. *Cell Immunol* 251; 131-6.
- Ramon-Cueto, A., and Nieto-Sampedro, M. (1992) Glial cells from adult rat olfactory bulb: immunocytochemical properties of pure cultures of ensheathing cells. *Neuroscience* 47; 213-20.
- Rasmusson, I., Ringden, O., Sundberg, B., and Le Blanc, K. (2003) Mesenchymal stem cells inhibit the formation of cytotoxic T lymphocytes, but not activated cytotoxic T lymphocytes or natural killer cells. *Transplantation* **76**; 1208-13.
- Rasmusson, I., Ringden, O., Sundberg, B., and Le Blanc, K. (2005) Mesenchymal stem cells inhibit lymphocyte proliferation by mitogens and alloantigens by different mechanisms. *Exp Cell Res* **305**; 33-41.

- Ray, J., Peterson, D. A., Schinstine, M., and Gage, F. H. (1993) Proliferation, differentiation, and long-term culture of primary hippocampal neurons. *Proc Natl Acad Sci U S A* 90; 3602-6.
- Redmond, D. E., Sladek, J. R., Jr., Roth, R. H., Collier, T. J., Elsworth, J. D., Deutch, A. Y., and Haber, S. (1986) Fetal neuronal grafts in monkeys given methylphenyltetrahydropyridine. *Lancet* 1; 1125-7.
- Reh, T. A. (2002) Neural stem cells: form and function. Nat Neurosci 5; 392-4.
- Remy, S., Canova, C., Daguin-Nerriere, V., Martin, C., Melchior, B., Neveu, I., Charreau, B., Soulillou, J. P., and Brachet, P. (2001) Different mechanisms mediate the rejection of porcine neurons and endothelial cells transplanted into the rat brain. *Xenotransplantation* 8; 136-48.
- Reyes, M., Lund, T., Lenvik, T., Aguiar, D., Koodie, L., and Verfaillie, C. M. (2001) Purification and ex vivo expansion of postnatal human marrow mesodermal progenitor cells. *Blood* 98; 2615-25.
- Reynolds, B. A., Tetzlaff, W., and Weiss, S. (1992) A multipotent EGF-responsive striatal embryonic progenitor cell produces neurons and astrocytes. *J Neurosci* 12; 4565-74.
- Richards, L. J., Kilpatrick, T. J., and Bartlett, P. F. (1992) De novo generation of neuronal cells from the adult mouse brain. *Proc Natl Acad Sci U S A* **89**; 8591-5.
- Rickard, D. J., Kassem, M., Hefferan, T. E., Sarkar, G., Spelsberg, T. C., and Riggs, B. L. (1996) Isolation and characterization of osteoblast precursor cells from human bone marrow. *J Bone Miner Res* 11; 312-24.
- Rickard, D. J., Sullivan, T. A., Shenker, B. J., Leboy, P. S., and Kazhdan, I. (1994) Induction of rapid osteoblast differentiation in rat bone marrow stromal cell cultures by dexamethasone and BMP-2. *Dev Biol* 161; 218-28.
- Rismanchi, N., Floyd, C. L., Berman, R. F., and Lyeth, B. G. (2003) Cell death and long-term maintenance of neuron-like state after differentiation of rat bone marrow stromal cells: a comparison of protocols. *Brain Res* **991**; 46-55.
- Roche, S., Richard, M. J., and Favrot, M. C. (2007) Oct-4, Rex-1, and Gata-4 expression in human MSC increase the differentiation efficiency but not hTERT expression. J Cell Biochem 101; 271-80.
- Rojas, M., Xu, J., Woods, C. R., Mora, A. L., Spears, W., Roman, J., and Brigham, K. L. (2005) Bone marrow-derived mesenchymal stem cells in repair of the injured lung. *Am J Respir Cell Mol Biol* 33; 145-52.
- Roy, N. S., Cleren, C., Singh, S. K., Yang, L., Beal, M. F., and Goldman, S. A. (2006) Functional engraftment of human ES cell-derived dopaminergic neurons enriched by coculture with telomerase-immortalized midbrain astrocytes. *Nat Med* 12; 1259-68.
- Ruitenberg, M. J., Vukovic, J., Sarich, J., Busfield, S. J., and Plant, G. W. (2006) Olfactory ensheathing cells: characteristics, genetic engineering, and therapeutic potential. *J Neurotrauma* 23; 468-78.
- Russo, F. P., Alison, M. R., Bigger, B. W., Amofah, E., Florou, A., Amin, F., Bou-Gharios, G., Jeffery, R., Iredale, J. P., and Forbes, S. J. (2006) The bone marrow functionally contributes to liver fibrosis. *Gastroenterology* **130**; 1807-21.
- Sacchetti, B., Funari, A., Michienzi, S., Di Cesare, S., Piersanti, S., Saggio, I., Tagliafico, E., Ferrari, S., Robey, P. G., Riminucci, M., and Bianco, P. (2007) Self-renewing osteoprogenitors in bone marrow sinusoids can organize a hematopoietic microenvironment. *Cell* 131; 324-36.
- Sachs, C., and Jonsson, G. (1975) Mechanisms of action of 6-hydroxydopamine. *Biochem Pharmacol* 24; 1-8.
- Sadan, O., Bahat-Stromza, M., Barhum, Y., Levy, Y. S., Pisnevsky, A., Peretz, H., Bar Ilan, A., Bulvik, S., Shemesh, N., Krepel, D., Cohen, Y., Melamed, E., and Offen, D. (2009) Protective effects of neurotrophic factors secreting cells in a 60HDA rat model of Parkinson disease. *Stem Cells Dev* 18; 1179-90.
- Sanchez-Pernaute, R., Studer, L., Bankiewicz, K. S., Major, E. O., and McKay, R. D. (2001) In vitro generation and transplantation of precursor-derived human dopamine neurons. *Journal of Neuroscience Research* 65; 284-8.
- Sanchez-Ramos, J., Song, S., Cardozo-Pelaez, F., Hazzi, C., Stedeford, T., Willing, A., Freeman, T. B., Saporta, S., Janssen, W., Patel, N., Cooper, D. R., and Sanberg, P. R. (2000) Adult bone marrow stromal cells differentiate into neural cells in vitro. *Exp Neurol* 164; 247-56.
- Sato, K., Ozaki, K., Oh, I., Meguro, A., Hatanaka, K., Nagai, T., Muroi, K., and Ozawa, K. (2007) Nitric oxide plays a critical role in suppression of T-cell proliferation by mesenchymal stem cells. *Blood* 109; 228-34.
- Sato, Y., Araki, H., Kato, J., Nakamura, K., Kawano, Y., Kobune, M., Sato, T., Miyanishi, K., Takayama, T., Takahashi, M., Takimoto, R., Iyama, S., Matsunaga, T., Ohtani, S., Matsuura, A., Hamada, H., and Niitsu, Y. (2005) Human mesenchymal stem cells xenografted directly to rat liver are differentiated into human hepatocytes without fusion. *Blood* 106; 756-63.
- Sayles, M., Jain, M., and Barker, R. A. (2004) The cellular repair of the brain in Parkinson's disease--past, present and future. *Transpl Immunol* **12**; 321-42.

- Schulz, J. B. (2008) Update on the pathogenesis of Parkinson's disease. J Neurol 255 Suppl 5; 3-7.
- Schulz, J. B., and Falkenburger, B. H. (2004) Neuronal pathology in Parkinson's disease. *Cell Tissue Res* 318; 135-47.
- Schutze, N., Noth, U., Schneidereit, J., Hendrich, C., and Jakob, F. (2005) Differential expression of CCN-family members in primary human bone marrow-derived mesenchymal stem cells during osteogenic, chondrogenic and adipogenic differentiation. *Cell Commun Signal* 3; 5.
- Schwarting, R. K., and Huston, J. P. (1996) The unilateral 6-hydroxydopamine lesion model in behavioral brain research. Analysis of functional deficits, recovery and treatments. *Prog Neurobiol* 50; 275-331.
- Schwartz, R. E., Reyes, M., Koodie, L., Jiang, Y., Blackstad, M., Lund, T., Lenvik, T., Johnson, S., Hu, W. S., and Verfaillie, C. M. (2002) Multipotent adult progenitor cells from bone marrow differentiate into functional hepatocyte-like cells. J Clin Invest 109; 1291-302.
- Schwarz, E. J., Reger, R. L., Alexander, G. M., Class, R., Azizi, S. A., and Prockop, D. J. (2001) Rat marrow stromal cells rapidly transduced with a self-inactivating retrovirus synthesize L-DOPA in vitro. *Gene Ther* 8; 1214-23.
- Sejersen, T., and Lendahl, U. (1993) Transient expression of the intermediate filament nestin during skeletal muscle development. *J Cell Sci* **106** (**Pt 4**); 1291-300.
- Seshi, B., Kumar, S., and King, D. (2003) Multilineage gene expression in human bone marrow stromal cells as evidenced by single-cell microarray analysis. *Blood Cells, Molecules, & Diseases* 31; 268-85.
- Shahdadfar, A., Fronsdal, K., Haug, T., Reinholt, F. P., and Brinchmann, J. E. (2005) In vitro expansion of human mesenchymal stem cells: choice of serum is a determinant of cell proliferation, differentiation, gene expression, and transcriptome stability. *Stem Cells* 23; 1357-66.
- Shen, B., Bhargav, D., Wei, A., Williams, L. A., Tao, H., Ma, D. D., and Diwan, A. D. (2009a) BMP-13 emerges as a potential inhibitor of bone formation. *Int J Biol Sci* 5; 192-200.
- Shen, B., Wei, A., Tao, H., Diwan, A. D., and Ma, D. D. (2009b) BMP-2 Enhances TGF-beta3-Mediated Chondrogenic Differentiation of Human Bone Marrow Multipotent Mesenchymal Stromal Cells in Alginate Bead Culture. *Tissue Eng Part A* 15, 1311-20.
- Shetty, P., Ravindran, G., Sarang, S., Thakur, A. M., Rao, H. S., and Viswanathan, C. (2009) Clinical grade mesenchymal stem cells transdifferentiated under xenofree

conditions alleviates motor deficiencies in a rat model of Parkinson's disease. *Cell Biol Int* **33**; 830-8.

- Shi, X. L., Mao, L., Xu, B. Y., Xie, T., Zhu, Z. H., Chen, J. H., Li, L., and Ding, Y. T. (2008) Optimization of an effective directed differentiation medium for differentiating mouse bone marrow mesenchymal stem cells into hepatocytes in vitro. *Cell Biol Int* 32; 959-65.
- Shi, X. L., Qiu, Y. D., Li, Q., Xie, T., Zhu, Z. H., Chen, L. L., Li, L., and Ding, Y. T. (2005) Hepatocyte-like cells from directed differentiation of mouse bone marrow cells in vitro. *Acta Pharmacol Sin* 26; 469-76.
- Shintani, A., Nakao, N., Kakishita, K., and Itakura, T. (2007) Protection of dopamine neurons by bone marrow stromal cells. *Brain Res* **1186**; 48-55.
- Shukla, S., Chaturvedi, R. K., Seth, K., Roy, N. S., and Agrawal, A. K. (2009) Enhanced survival and function of neural stem cells-derived dopaminergic neurons under influence of olfactory ensheathing cells in parkinsonian rats. J Neurochem 109; 436-51.
- Simmons, P. J., and Torok-Storb, B. (1991) Identification of stromal cell precursors in human bone marrow by a novel monoclonal antibody, STRO-1. *Blood* 78; 55-62.
- Singhal, S., Lawrence, J. M., Bhatia, B., Ellis, J. S., Kwan, A. S., Macneil, A., Luthert, P. J., Fawcett, J. W., Perez, M. T., Khaw, P. T., and Limb, G. A. (2008) Chondroitin sulfate proteoglycans and microglia prevent migration and integration of grafted Muller stem cells into degenerating retina. *Stem Cells* 26; 1074-82.
- Smidt, M. P., and Burbach, J. P. (2007) How to make a mesodiencephalic dopaminergic neuron. *Nat Rev Neurosci* 8; 21-32.
- Snykers, S., Vanhaecke, T., Papeleu, P., Luttun, A., Jiang, Y., Vander Heyden, Y., Verfaillie, C., and Rogiers, V. (2006) Sequential exposure to cytokines reflecting embryogenesis: the key for in vitro differentiation of adult bone marrow stem cells into functional hepatocyte-like cells. *Toxicol Sci* 94; 330-41; discussion 235-9.
- Song, H., Stevens, C. F., and Gage, F. H. (2002a) Astroglia induce neurogenesis from adult neural stem cells. *Nature* **417**; 39-44.
- Song, H. J., Stevens, C. F., and Gage, F. H. (2002b) Neural stem cells from adult hippocampus develop essential properties of functional CNS neurons. *Nature Neuroscience*. 5; 438-45.

- Song, S., Zhang, H., Cuevas, J., and Sanchez-Ramos, J. (2007) Comparison of neuronlike cells derived from bone marrow stem cells to those differentiated from adult brain neural stem cells. *Stem Cells Dev* 16; 747-56.
- Sorrentino, A., Ferracin, M., Castelli, G., Biffoni, M., Tomaselli, G., Baiocchi, M., Fatica, A., Negrini, M., Peschle, C., and Valtieri, M. (2008) Isolation and characterization of CD146+ multipotent mesenchymal stromal cells. *Exp Hematol* 36; 1035-46.
- Sotiropoulou, P. A., Perez, S. A., Gritzapis, A. D., Baxevanis, C. N., and Papamichail, M. (2006) Interactions between human mesenchymal stem cells and natural killer cells. *Stem Cells* 24; 74-85.
- Spaggiari, G. M., Capobianco, A., Abdelrazik, H., Becchetti, F., Mingari, M. C., and Moretta, L. (2008) Mesenchymal stem cells inhibit natural killer-cell proliferation, cytotoxicity, and cytokine production: role of indoleamine 2,3dioxygenase and prostaglandin E2. *Blood* 111; 1327-33.
- Spaggiari, G. M., Capobianco, A., Becchetti, S., Mingari, M. C., and Moretta, L. (2006) Mesenchymal stem cell-natural killer cell interactions: evidence that activated NK cells are capable of killing MSCs, whereas MSCs can inhibit IL-2-induced NK-cell proliferation. *Blood* 107; 1484-90.
- Spees, J. L., Gregory, C. A., Singh, H., Tucker, H. A., Peister, A., Lynch, P. J., Hsu, S. C., Smith, J., and Prockop, D. J. (2004) Internalized antigens must be removed to prepare hypoimmunogenic mesenchymal stem cells for cell and gene therapy. *Mol Ther* **9**; 747-56.
- Spees, J. L., Olson, S. D., Whitney, M. J., and Prockop, D. J. (2006) Mitochondrial transfer between cells can rescue aerobic respiration. *Proc Natl Acad Sci U S A* 103; 1283-8.
- Spees, J. L., Olson, S. D., Ylostalo, J., Lynch, P. J., Smith, J., Perry, A., Peister, A., Wang, M. Y., and Prockop, D. J. (2003) Differentiation, cell fusion, and nuclear fusion during ex vivo repair of epithelium by human adult stem cells from bone marrow stroma. *Proc Natl Acad Sci U S A* **100**; 2397-402.
- Stagg, J. (2007) Immune regulation by mesenchymal stem cells: two sides to the coin. *Tissue Antigens* **69**; 1-9.
- Streit, W. J., Graeber, M. B., and Kreutzberg, G. W. (1988) Functional plasticity of microglia: a review. *Glia* 1; 301-7.
- Stromberg, I., Bygdeman, M., Goldstein, M., Seiger, A., and Olson, L. (1986) Human fetal substantia nigra grafted to the dopamine-denervated striatum of immunosuppressed rats: evidence for functional reinnervation. *Neurosci Lett* 71; 271-6.

- Studer, L., Tabar, V., and McKay, R. D. (1998) Transplantation of expanded mesencephalic precursors leads to recovery in parkinsonian rats. *Nature Neuroscience* 1; 290-5.
- Sudres, M., Norol, F., Trenado, A., Gregoire, S., Charlotte, F., Levacher, B., Lataillade, J. J., Bourin, P., Holy, X., Vernant, J. P., Klatzmann, D., and Cohen, J. L. (2006) Bone marrow mesenchymal stem cells suppress lymphocyte proliferation in vitro but fail to prevent graft-versus-host disease in mice. *J Immunol* 176; 7761-7.
- Suon, S., Jin, H., Donaldson, A. E., Caterson, E. J., Tuan, R. S., Deschennes, G., Marshall, C., and Iacovitti, L. (2004) Transient differentiation of adult human bone marrow cells into neuron-like cells in culture: development of morphological and biochemical traits is mediated by different molecular mechanisms. *Stem Cells Dev* 13; 625-35.
- Suon, S., Yang, M., and Iacovitti, L. (2006) Adult human bone marrow stromal spheres express neuronal traits in vitro and in a rat model of Parkinson's disease. *Brain Res* 1106; 46-51.
- Suzuki, H., Taguchi, T., Tanaka, H., Kataoka, H., Li, Z., Muramatsu, K., Gondo, T., and Kawai, S. (2004) Neurospheres induced from bone marrow stromal cells are multipotent for differentiation into neuron, astrocyte, and oligodendrocyte phenotypes. *Biochem Biophys Res Commun* **322**; 918-22.
- Svendsen, C. (2008) Stem cells and Parkinson's disease: toward a treatment, not a cure. *Cell Stem Cell* **2**; 412-3.
- Svendsen, C. N., Bhattacharyya, A., and Tai, Y. T. (2001) Neurons from stem cells: preventing an identity crisis. *Nat Rev Neurosci* **2**; 831-4.
- Svendsen, C. N., Caldwell, M. A., Shen, J., ter Borg, M. G., Rosser, A. E., Tyers, P., Karmiol, S., and Dunnett, S. B. (1997) Long-term survival of human central nervous system progenitor cells transplanted into a rat model of Parkinson's disease. *Exp Neurol* 148; 135-46.
- Svendsen, C. N., Clarke, D. J., Rosser, A. E., and Dunnett, S. B. (1996) Survival and differentiation of rat and human epidermal growth factor-responsive precursor cells following grafting into the lesioned adult central nervous system. *Exp Neurol* 137; 376-88.
- Tabar, V., Tomishima, M., Panagiotakos, G., Wakayama, S., Menon, J., Chan, B., Mizutani, E., Al-Shamy, G., Ohta, H., Wakayama, T., and Studer, L. (2008) Therapeutic cloning in individual parkinsonian mice. *Nat Med* 14; 379-81.
- Takahashi, K., and Yamanaka, S. (2006) Induction of pluripotent stem cells from mouse embryonic and adult fibroblast cultures by defined factors. *Cell* **126**; 663-76.

- Takayama, H., Ray, J., Raymon, H. K., Baird, A., Hogg, J., Fisher, L. J., and Gage, F. H. (1995) Basic fibroblast growth factor increases dopaminergic graft survival and function in a rat model of Parkinson's disease. *Nat Med* 1; 53-8.
- Takeda, J., Seino, S., and Bell, G. I. (1992) Human Oct3 gene family: cDNA sequences, alternative splicing, gene organization, chromosomal location, and expression at low levels in adult tissues. *Nucleic Acids Res* 20; 4613-20.
- Tao, H., Rao, R., and Ma, D. D. (2005) Cytokine-induced stable neuronal differentiation of human bone marrow mesenchymal stem cells in a serum/feeder cell-free condition. *Dev Growth Differ* 47; 423-33.
- Tatard, V. M., D'Ippolito, G., Diabira, S., Valeyev, A., Hackman, J., McCarthy, M., Bouckenooghe, T., Menei, P., Montero-Menei, C. N., and Schiller, P. C. (2007) Neurotrophin-directed differentiation of human adult marrow stromal cells to dopaminergic-like neurons. *Bone* 40; 360-73.
- Tavassoli, M., and Crosby, W. H. (1968) Transplantation of marrow to extramedullary sites. *Science* 161; 54-6.
- Teismann, P., and Schulz, J. B. (2004) Cellular pathology of Parkinson's disease: astrocytes, microglia and inflammation. *Cell Tissue Res* **318**; 149-61.
- Teng, X., Nagata, I., Li, H. P., Kimura-Kuroda, J., Sango, K., Kawamura, K., Raisman, G., and Kawano, H. (2008) Regeneration of nigrostriatal dopaminergic axons after transplantation of olfactory ensheathing cells and fibroblasts prevents fibrotic scar formation at the lesion site. *J Neurosci Res* 86; 3140-50.
- Terada, N., Hamazaki, T., Oka, M., Hoki, M., Mastalerz, D. M., Nakano, Y., Meyer, E. M., Morel, L., Petersen, B. E., and Scott, E. W. (2002) Bone marrow cells adopt the phenotype of other cells by spontaneous cell fusion. *Nature* 416; 542-5.
- Theise, N. D., Badve, S., Saxena, R., Henegariu, O., Sell, S., Crawford, J. M., and Krause, D. S. (2000a) Derivation of hepatocytes from bone marrow cells in mice after radiation-induced myeloablation. *Hepatology* 31; 235-40.
- Theise, N. D., Nimmakayalu, M., Gardner, R., Illei, P. B., Morgan, G., Teperman, L., Henegariu, O., and Krause, D. S. (2000b) Liver from bone marrow in humans. *Hepatology* 32; 11-6.
- Till, J. E., and Mc, C. E. (1961) A direct measurement of the radiation sensitivity of normal mouse bone marrow cells. *Radiat Res* 14; 213-22.
- Tondreau, T., Dejeneffe, M., Meuleman, N., Stamatopoulos, B., Delforge, A., Martiat, P., Bron, D., and Lagneaux, L. (2008) Gene expression pattern of functional neuronal cells derived from human bone marrow mesenchymal stromal cells. *BMC Genomics* 9; 166.

- Tondreau, T., Lagneaux, L., Dejeneffe, M., Delforge, A., Massy, M., Mortier, C., and Bron, D. (2004a) Isolation of BM mesenchymal stem cells by plastic adhesion or negative selection: phenotype, proliferation kinetics and differentiation potential. *Cytotherapy* 6; 372-9.
- Tondreau, T., Lagneaux, L., Dejeneffe, M., Massy, M., Mortier, C., Delforge, A., and Bron, D. (2004b) Bone marrow-derived mesenchymal stem cells already express specific neural proteins before any differentiation. *Differentiation* 72; 319-26.
- Tondreau, T., Meuleman, N., Delforge, A., Dejeneffe, M., Leroy, R., Massy, M., Mortier, C., Bron, D., and Lagneaux, L. (2005) Mesenchymal stem cells derived from CD133-positive cells in mobilized peripheral blood and cord blood: proliferation, Oct4 expression, and plasticity. *Stem Cells* 23; 1105-12.
- Tran, S. D., Pillemer, S. R., Dutra, A., Barrett, A. J., Brownstein, M. J., Key, S., Pak, E., Leakan, R. A., Kingman, A., Yamada, K. M., Baum, B. J., and Mezey, E. (2003) Differentiation of human bone marrow-derived cells into buccal epithelial cells in vivo: a molecular analytical study. *Lancet* 361; 1084-8.
- Tremain, N., Korkko, J., Ibberson, D., Kopen, G. C., DiGirolamo, C., and Phinney, D. G. (2001) MicroSAGE analysis of 2,353 expressed genes in a single cell-derived colony of undifferentiated human mesenchymal stem cells reveals mRNAs of multiple cell lineages. *Stem Cells* 19; 408-18.
- Tropel, P., Platet, N., Platel, J. C., Noel, D., Albrieux, M., Benabid, A. L., and Berger, F. (2006) Functional neuronal differentiation of bone marrow-derived mesenchymal stem cells. *Stem Cells* 24; 2868-76.
- Trzaska, K. A., Kuzhikandathil, E. V., and Rameshwar, P. (2007) Specification of a dopaminergic phenotype from adult human mesenchymal stem cells. *Stem Cells* 25; 2797-808.
- Tse, W. T., Pendleton, J. D., Beyer, W. M., Egalka, M. C., and Guinan, E. C. (2003) Suppression of allogeneic T-cell proliferation by human marrow stromal cells: implications in transplantation. *Transplantation* **75**; 389-97.
- Tseng, P. Y., Chen, C. J., Sheu, C. C., Yu, C. W., and Huang, Y. S. (2007) Spontaneous differentiation of adult rat marrow stromal cells in a long-term culture. J Vet Med Sci 69; 95-102.
- Ungerstedt, U. (1968) 6-Hydroxy-dopamine induced degeneration of central monoamine neurons. *Eur J Pharmacol* **5**; 107-10.
- Ungerstedt, U., and Arbuthnott, G. W. (1970) Quantitative recording of rotational behavior in rats after 6-hydroxy-dopamine lesions of the nigrostriatal dopamine system. *Brain Res* 24; 485-93.

- Vaittinen, S., Lukka, R., Sahlgren, C., Hurme, T., Rantanen, J., Lendahl, U., Eriksson, J. E., and Kalimo, H. (2001) The expression of intermediate filament protein nestin as related to vimentin and desmin in regenerating skeletal muscle. *J Neuropathol Exp Neurol* 60; 588-97.
- Vassilopoulos, G., Wang, P. R., and Russell, D. W. (2003) Transplanted bone marrow regenerates liver by cell fusion. *Nature* **422**; 901-4.
- Veldman, B. A., Wijn, A. M., Knoers, N., Praamstra, P., and Horstink, M. W. (1998) Genetic and environmental risk factors in Parkinson's disease. *Clin Neurol Neurosurg* 100; 15-26.
- Vogel, W., Grunebach, F., Messam, C. A., Kanz, L., Brugger, W., and Buhring, H. J. (2003) Heterogeneity among human bone marrow-derived mesenchymal stem cells and neural progenitor cells. *Haematologica*. 88; 126-33.
- Wagers, A. J., and Weissman, I. L. (2004) Plasticity of adult stem cells. *Cell.* **116**; 639-48.
- Wagner, J. P., Black, I. B., and DiCicco-Bloom, E. (1999) Stimulation of neonatal and adult brain neurogenesis by subcutaneous injection of basic fibroblast growth factor. J Neurosci 19; 6006-16.
- Wagner, W., Feldmann, R. E., Jr., Seckinger, A., Maurer, M. H., Wein, F., Blake, J., Krause, U., Kalenka, A., Burgers, H. F., Saffrich, R., Wuchter, P., Kuschinsky, W., and Ho, A. D. (2006) The heterogeneity of human mesenchymal stem cell preparations--evidence from simultaneous analysis of proteomes and transcriptomes. *Exp Hematol* 34; 536-48.
- Wakitani, S., Saito, T., and Caplan, A. I. (1995) Myogenic cells derived from rat bone marrow mesenchymal stem cells exposed to 5-azacytidine. *Muscle Nerve* 18; 1417-26.
- Wallgren, A. C., Karlsson-Parra, A., and Korsgren, O. (1995) The main infiltrating cell in xenograft rejection is a CD4+ macrophage and not a T lymphocyte. *Transplantation* 60; 594-601.
- Wang, G., Bunnell, B. A., Painter, R. G., Quiniones, B. C., Tom, S., Lanson, N. A., Jr., Spees, J. L., Bertucci, D., Peister, A., Weiss, D. J., Valentine, V. G., Prockop, D. J., and Kolls, J. K. (2005) Adult stem cells from bone marrow stroma differentiate into airway epithelial cells: potential therapy for cystic fibrosis. *Proc Natl Acad Sci U S A* 102; 186-91.
- Wang, P. P., Wang, J. H., Yan, Z. P., Hu, M. Y., Lau, G. K., Fan, S. T., and Luk, J. M. (2004) Expression of hepatocyte-like phenotypes in bone marrow stromal cells after HGF induction. *Biochem Biophys Res Commun* 320; 712-6.

- Wang, X., Willenbring, H., Akkari, Y., Torimaru, Y., Foster, M., Al-Dhalimy, M., Lagasse, E., Finegold, M., Olson, S., and Grompe, M. (2003) Cell fusion is the principal source of bone-marrow-derived hepatocytes. *Nature* 422; 897-901.
- Weimann, J. M., Johansson, C. B., Trejo, A., and Blau, H. M. (2003) Stable reprogrammed heterokaryons form spontaneously in Purkinje neurons after bone marrow transplant. *Nat Cell Biol* 5; 959-66.
- Wennberg, L., Czech, K. A., Larsson, L. C., Mirza, B., Bennet, W., Song, Z., and Widner, H. (2001) Effects of immunosuppressive treatment on host responses against intracerebral porcine neural tissue xenografts in rats. *Transplantation* **71**; 1797-806.
- Wenning, G. K., Odin, P., Morrish, P., Rehncrona, S., Widner, H., Brundin, P., Rothwell, J. C., Brown, R., Gustavii, B., Hagell, P., Jahanshahi, M., Sawle, G., Bjorklund, A., Brooks, D. J., Marsden, C. D., Quinn, N. P., and Lindvall, O. (1997) Short- and long-term survival and function of unilateral intrastriatal dopaminergic grafts in Parkinson's disease. *Ann Neurol* 42; 95-107.
- Wernig, M., Zhao, J. P., Pruszak, J., Hedlund, E., Fu, D., Soldner, F., Broccoli, V., Constantine-Paton, M., Isacson, O., and Jaenisch, R. (2008) Neurons derived from reprogrammed fibroblasts functionally integrate into the fetal brain and improve symptoms of rats with Parkinson's disease. *Proc Natl Acad Sci U S A* **105**; 5856-61.
- Widner, H., and Brundin, P. (1988) Immunological aspects of grafting in the mammalian central nervous system. A review and speculative synthesis. *Brain Res* 472; 287-324.
- Wislet-Gendebien, S., Hans, G., Leprince, P., Rigo, J. M., Moonen, G., and Rogister, B. (2005) Plasticity of cultured mesenchymal stem cells: switch from nestinpositive to excitable neuron-like phenotype. *Stem Cells* 23; 392-402.
- Wislet-Gendebien, S., Leprince, P., Moonen, G., and Rogister, B. (2003) Regulation of neural markers nestin and GFAP expression by cultivated bone marrow stromal cells. J Cell Sci 116; 3295-302.
- Woodbury, D., Reynolds, K., and Black, I. B. (2002) Adult bone marrow stromal stem cells express germline, ectodermal, endodermal, and mesodermal genes prior to neurogenesis. *Journal of Neuroscience Research* **69**; 908-17.
- Woodbury, D., Schwarz, E. J., Prockop, D. J., and Black, I. B. (2000) Adult rat and human bone marrow stromal cells differentiate into neurons. *Journal of Neuroscience Research* 61; 364-70.
- Woodhall, E., West, A. K., and Chuah, M. I. (2001) Cultured olfactory ensheathing cells express nerve growth factor, brain-derived neurotrophic factor, glia cell

line-derived neurotrophic factor and their receptors. *Brain Res Mol Brain Res* **88**; 203-13.

- Wuerthele, S. M., Freed, W. J., Olson, L., Morihisa, J., Spoor, L., Wyatt, R. J., and Hoffer, B. J. (1981) Effect of dopamine agonists and antagonists on the electrical activity of substantia nigra neurons transplanted into the lateral ventricle of the rat. *Exp Brain Res* 44; 1-10.
- Wulf, G. G., Jackson, K. A., and Goodell, M. A. (2001) Somatic stem cell plasticity: current evidence and emerging concepts. *Exp Hematol* **29**; 1361-70.
- Yamanaka, S. (2007) Strategies and new developments in the generation of patientspecific pluripotent stem cells. *Cell Stem Cell* 1; 39-49.
- Yamazaki, K., and Allen, T. D. (1990) Ultrastructural morphometric study of efferent nerve terminals on murine bone marrow stromal cells, and the recognition of a novel anatomical unit: the "neuro-reticular complex". Am J Anat 187; 261-76.
- Yanez, R., Lamana, M. L., Garcia-Castro, J., Colmenero, I., Ramirez, M., and Bueren, J. A. (2006) Adipose tissue-derived mesenchymal stem cells have in vivo immunosuppressive properties applicable for the control of the graft-versus-host disease. *Stem Cells* 24; 2582-91.
- Ye, W., Shimamura, K., Rubenstein, J. L., Hynes, M. A., and Rosenthal, A. (1998) FGF and Shh signals control dopaminergic and serotonergic cell fate in the anterior neural plate. *Cell.* 93; 755-66.
- Yim, E. K., Pang, S. W., and Leong, K. W. (2007) Synthetic nanostructures inducing differentiation of human mesenchymal stem cells into neuronal lineage. *Exp Cell Res* 313; 1820-9.
- Ying, Q. L., Nichols, J., Evans, E. P., and Smith, A. G. (2002) Changing potency by spontaneous fusion. *Nature* **416**; 545-8.
- Yoo, J. U., Barthel, T. S., Nishimura, K., Solchaga, L., Caplan, A. I., Goldberg, V. M., and Johnstone, B. (1998) The chondrogenic potential of human bone-marrowderived mesenchymal progenitor cells. *J Bone Joint Surg Am* 80; 1745-57.
- Zangrossi, S., Marabese, M., Broggini, M., Giordano, R., D'Erasmo, M., Montelatici, E., Intini, D., Neri, A., Pesce, M., Rebulla, P., and Lazzari, L. (2007) Oct-4 expression in adult human differentiated cells challenges its role as a pure stem cell marker. *Stem Cells* 25; 1675-80.
- Zhang, W., Ge, W., Li, C., You, S., Liao, L., Han, Q., Deng, W., and Zhao, R. C. (2004) Effects of mesenchymal stem cells on differentiation, maturation, and function of human monocyte-derived dendritic cells. *Stem Cells Dev* 13; 263-71.

- Zhang, Z., Wang, X., and Wang, S. (2008) Isolation and characterization of mesenchymal stem cells derived from bone marrow of patients with Parkinson's disease. *In Vitro Cell Dev Biol Anim* 44; 169-77.
- Zhao, L. R., Duan, W. M., Reyes, M., Keene, C. D., Verfaillie, C. M., and Low, W. C. (2002) Human bone marrow stem cells exhibit neural phenotypes and ameliorate neurological deficits after grafting into the ischemic brain of rats. *Exp Neurol* 174; 11-20.
- Zhao, L. X., Zhang, J., Cao, F., Meng, L., Wang, D. M., Li, Y. H., Nan, X., Jiao, W. C., Zheng, M., Xu, X. H., and Pei, X. T. (2004) Modification of the brain-derived neurotrophic factor gene: a portal to transform mesenchymal stem cells into advantageous engineering cells for neuroregeneration and neuroprotection. *Exp Neurol* 190; 396-406.

CHAPTER 8

SUPPLEMENTARY FILES

The Supplementary CD is attached to the back cover of this thesis, and contains 'Supplementary File 1 - Part 1' and 'Supplementary File 1 - Part 2' video footage discussed in Section 4.2.1, and 'Supplementary File 2' video discussed in Section 5.2.1.